WEB TABLES

U.S. DEPARTMENT OF EDUCATION
AUGUST 2010 NCES 2010-162

Student Financing of Undergraduate Education: 2007–08

In 2007–08, approximately 21 million students were enrolled in undergraduate postsecondary education in the United States. These Web Tables provide a comprehensive source of information on financial aid that was awarded to undergraduate students during the 2007-08 academic year. Included are estimates of tuition, price of attendance, and financial aid, shown by the enrollment and demographic characteristics of students. The tables are similar to those produced for the 1995-96, 1999-2000, and 2003-04 National Postsecondary Student Aid Study (NPSAS) reports on student financing of undergraduate education, and are grouped into five separate sections:

Section 1 displays the percentile distribution among all undergraduates for family income, tuition, and price of attendance. These distributions were used to create the categories for these variables in the subsequent sections.

Section 2 shows undergraduates' average tuition and fees and average price of attendance. The price of attendance includes books and supplies, room and board (or housing and meal allowances for off-campus students), transportation, and other personal living expenses.

Section 3 looks at the various types of financial aid that undergraduates received—federal, state, and institutional aid and combined amounts—by type of institution attended.

Section 4 shows net tuition (tuition and fees minus all grants), net price of attendance (price minus all grants), the out-of-pocket net price (price minus all aid), and financial need.

Section 5 examines students' dependency and attendance status, residence, race/ethnicity, gender, citizenship, family income, and their distribution by the type of institution attended.

Figures showing sticker, net, and outof-pocket prices for college in 2007–08 can be viewed at

http://nces.ed.gov/pubsearch/pubsinfo.asp?pubid=2010162

RELATED NCES REPORTS

2007–08 National Postsecondary Student Aid Study (NPSAS:08): Student Financial Aid Estimates for 2007–08: First Look (NCES 2009-166).

Web Tables—Undergraduate Financial Aid Estimates by Type of Institution in 2007– 08 (NCES 2009-201).

DATA

The estimates presented in these tables were generated from the 2007–08 National Postsecondary Student Aid Study (NPSAS:08), a comprehensive, nationally representative survey of how students finance their postsecondary education conducted by the National Center for Education Statistics (NCES). NPSAS also includes a broad

These Web Tables were authored by Christina Chang Wei of MPR Associates, Inc. The NCES Project Officer was Tracy Hunt-White. For questions about content or to view this report online, go to: http://nces.ed.gov/pubsearch/pubsinfo.asp?pubid=2010162



array of demographic and enrollment characteristics.

NPSAS uses a two-stage sampling design. Institutions are selected for inclusion in the first stage, and from these institutions students are selected in the second stage. The NPSAS:08 target population consisted of all eligible undergraduate and graduate students enrolled any time between July 1, 2007, and June 30, 2008 at Title IV¹ eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

Approximately 114,000 undergraduates and 14,000 graduate students were study respondents for NPSAS:08.

For an overview of the survey methodology, see appendix B of the report 2007–08 National Postsecondary Student Aid Study (NPSAS:08): Student Financial Aid Estimates for 2007–08: First Look (http://nces.ed.gov/pubs2009/2009166.pdf).

DATA ANALYSIS SYSTEM (DAS)

These estimates were produced using the Data Analysis System (DAS), a web-based software application that enables users to generate tables for most of the postsecondary surveys conducted by NCES. The DAS produces the design-adjusted standard errors necessary for testing the statistical significance of differences between the estimates. The DAS also contains a

detailed description of how each variable was created and includes the wording of questions for variables coming directly from the interview.

With the DAS, users can replicate or expand upon the tables presented here. The output from the DAS includes the parameter estimates (e.g., percentages or means), their standard errors,² and weighted sample sizes. If the number of valid cases is too small (i.e., fewer than 30 cases) to produce a reliable estimate, the DAS prints the message "low-N" instead of the estimate.

In addition to generating tables, DAS users may conduct covariance analyses with either weighted least squares or logistic regression. Many options are available for output with the regression results. For example, a Winsor filter can be used to eliminate cases with extreme values by deleting some cases from the top and bottom of the range.

The DAS can be accessed electronically at http://nces.ed.gov/DAS. If users are new to the DAS, the DAS User Help Center (http://nces.ed.gov/dasol/help) provides online tutorials offering step-by-step instructions on how to use all the functions of the DAS. For a description of all the options available, users should access the DAS at http://nces.ed.gov/dasolv2.

For more information, contact

Aurora D'Amico
Postsecondary Studies Division
National Center for Education Statistics
1990 K Street NW
Washington, DC 20006-5652
(202) 502-7334

aurora.d'amico@ed.gov

ENDNOTES

¹ Title IV institutions are those eligible to participate in the federal financial aid programs included in Title IV of the Higher Education Act. These programs include Pell grants, federal student loans, work-study, and other federal aid.

² NPSAS samples are not simple random samples; therefore, simple random sample techniques for estimating sampling errors cannot be applied to these data. The DAS takes into account the complexity of the sampling procedures and calculates standard errors appropriate for such samples using a bootstrap technique. This technique approximates the estimator by replications of the sampled population.

Table 1. Family income, tuition and fees, and price of attendance among all undergraduates enrolled in postsecondary institutions, by selected percentiles and institutional and student characteristics: 2007–08

Institutional and student characteristics	10th percentile	25th percentile	50th percentile	75th percentile	90th percentile
Student Characteristics	Total percentile	25th percentile	Jour percentile	75tii percentile	John percentile
Income by dependency status					
Dependent family income	\$18,000	\$36,100	\$66,600	\$104,600	\$149,000
Independent student income	3,000	11,000	26,000	48,400	80,600
Unmarried, no dependents	700	6,000	15,500	27,800	40,500
Married, no dependents	12,600	26,900	47,500	73,700	104,500
Single parent	2,200	7,500	18,000	30,900	46,700
Married parent	19,000	32,700	52,800	80,000	108,200
Tuition and fees by institutional type ¹	300	900	2,800	6,900	16,600
Public					
Less-than-2-year	‡	‡	‡	5,000 !	7,100
2-year	200	300	800	1,600	2,700
4-year	1,500	2,800	4,800	7,000	9,800
Non-doctorate-granting	1,000	1,800	3,500	5,500	7,400
Doctorate-granting	1,900	3,400	5,400	7,700	10,600
Private nonprofit					
Less-than-4-year	1,200	2,900	5,000	9,300	14,800
4-year	3,100	6,800	18,700	26,600	34,300
Non-doctorate-granting	2,500	5,000	15,000	23,600	29,500
Doctorate-granting	3,900	10,500	23,000	30,500	35,300
For-profit	3,300	5,500	8,400	12,900	19,700
Price of attendance by institutional type ¹	3,500	5,900	11,000	18,600	30,100
Public					
Less-than-2-year	2,100	4,400	8,900	13,500	18,500
2-year	2,900	3,800	6,200	9,200	12,700
4-year	5,700	10,200	15,500	19,000	23,500
Non-doctorate-granting	4,600	8,000	12,600	16,800	20,100
Doctorate-granting	7,500	11,800	17,000	20,200	25,000
Private nonprofit	,,,,,,,,	,	,	,	
Less-than-4-year	4,700	8,600	14,300	19,700	25,100
4-year	8,800	16,100	29,300	39,000	46,800
Non-doctorate-granting	7,700	14,000	26,900	35,100	41,100
Doctorate-granting	10,600	20,300	34,100	43,000	48,800
For-profit	8,200	12,700	19,300	29,400	32,100

[!] Interpret data with caution (estimates are unstable).

NOTE: Dependent family income is the parents' income. Independent student income is the student's own income and their spouse's income, if married. The price of attendance includes tuition and fees, books and supplies, housing, meals, transportation, and personal or miscellaneous expenses. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[‡] Reporting standards not met.

¹ Undergraduates attending more than one institution are not included in the estimates for tuition and fees and price of attendance.

Table S1. Standard errors for table 1: Family income, tuition and fees, and price of attendance among all undergraduates enrolled in postsecondary institutions, by selected percentiles and institutional and student characteristics: 2007–08

Institutional and student characteristics	10th percentile	25th percentile	50th percentile	75th percentile	90th percentile
student characteristics	Total percentale	25th percentile	John percentile	75th percentile	John percentile
Income by dependency status					
Dependent family income	\$280	\$290	\$440	\$480	\$1,000
Independent student income	70	150	290	510	840
Unmarried, no dependents	60	110	300	370	880
Married, no dependents	740	830	930	1,110	1,300
Single parent	110	180	290	500	1,250
Married parent	570	410	330	970	1,660
Tuition and fees by institutional type Public	10	30	30	60	540
Less-than-2-year	†	†	†	1,620	2,530
2-year	10	20	30	30	40
4-year	40	80	120	100	130
Non-doctorate-granting	90	80	140	100	370
Doctorate-granting	50	50	100	110	170
Private nonprofit					
Less-than-4-year	80	410	840	1,550	1,260
4-year	150	390	470	310	570
Non-doctorate-granting	260	320	1,220	630	640
Doctorate-granting	170	970	650	820	90
For-profit	170	310	570	690	1,620
Price of attendance by institutional type Public	70	90	70	110	280
Less-than-2-year	620	1,280	1,250	1,600	3,090
2-year	60	40	120	80	120
4-year	140	130	130	80	190
Non-doctorate-granting	90	200	230	150	370
Doctorate-granting	200	160	120	100	160
Private nonprofit					
Less-than-4-year	640	420	1,150	1,250	1,290
4-year	170	340	430	450	310
Non-doctorate-granting	210	400	540	640	680
Doctorate-granting	370	1,140	520	780	340
For-profit	470	390	620	1,100	400

[†] Not applicable.

Table 1.1. Family income, tuition and fees, and price of attendance among full-time/full-year undergraduates enrolled in postsecondary institutions, by selected percentiles and institutional and student characteristics: 2007–08

Institutional and student characteristics	10th percentile	25th percentile	50th percentile	75th percentile	90th percentile
Income by dependency status					
Dependent family income	\$19,600	\$39,400	\$73,100	\$111,000	\$156,000
Independent student income	1,500	6,200	17,100	33,500	54,700
Unmarried, no dependents	#	3,000	8,800	18,700	30,000
Married, no dependents	7,200	17,100	31,400	49,000	71,100
Single parent	1,700	5,300	13,000	23,600	35,100
Married parent	13,800	25,600	40,000	59,700	86,200
Tuition and fees by institutional type ¹	2,200	3,800	6,500	14,300	26,000
Public					
Less-than-2-year	1,300 !	1,800 !	3,900 !	7,100	‡
2-year	600	1,500	2,400	3,100	3,800
4-year	3,400	4,600	6,200	8,200	11,300
Non-doctorate-granting	3,000	4,000	5,200	6,600	9,400
Doctorate-granting	3,700	5,000	6,700	8,900	12,700
Private nonprofit					
Less-than-4-year	3,100 !	4,700 !	9,300 !	14,100	19,300
4-year	9,800	18,100	24,400	30,100	35,300
Non-doctorate-granting	7,100	15,900	21,300	26,500	31,300
Doctorate-granting	13,800	20,700	26,500	33,700	35,500
For-profit	4,700	6,500	10,300	15,200	21,000
Price of attendance by institutional type ¹	11,100	14,900	19,000	29,200	38,200
Public					
Less-than-2-year	8,700	11,700	14,800	19,300	22,900
2-year	8,800	9,700	12,100	14,800	16,900
4-year	13,100	15,700	18,000	21,300	25,600
Non-doctorate-granting	11,100	13,900	16,400	18,500	22,400
Doctorate-granting	14,300	16,600	18,700	22,400	27,100
Private nonprofit					
Less-than-4-year	11,300	12,800	19,700	26,100	33,500
4-year	23,500	28,800	35,700	42,400	48,400
Non-doctorate-granting	20,300	27,300	32,400	38,600	44,200
Doctorate-granting	25,700	32,100	38,800	46,500	49,300
For-profit	19,500	23,900	30,800	31,600	34,500

[#] Rounds to zero.

NOTE: Dependent family income is the parents' income. Independent student income is the student's own income and their spouse's income, if married. The price of attendance includes tuition and fees, books and supplies, housing, meals, transportation, and personal or miscellaneous expenses. This table excludes students attending more than one institution. Full-time/full-year students represent 38.5 percent of all undergraduates attending only one institution. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[!] Interpret data with caution (estimates are unstable).

[‡] Reporting standards not met.

¹ Undergraduates attending more than one institution are not included in the estimates for tuition and fees and price of attendance.

Table S1.1. Standard errors for table 1.1: Family income, tuition and fees, and price of attendance among full-time/full-year undergraduates enrolled in postsecondary institutions, by selected percentiles and institutional and student characteristics: 2007–08

Institutional and student characteristics	10th percentile	25th percentile	50th percentile	75th percentile	90th percentile
			1	1	
Income by dependency status					
Dependent family income	\$260	\$350	\$520	\$690	\$1,190
Independent student income	120	200	430	680	1,470
Unmarried, no dependents	†	170	370	600	1,130
Married, no dependents	440	1,180	1,150	1,480	4,560
Single parent	180	250	500	710	1,170
Married parent	880	880	1,210	1,720	2,880
Tuition and fees by institutional type Public	60	60	40	440	290
Less-than-2-year	460	660	1,910	2,020	†
2-year	10	50	50	40	50
4-year	20	90	100	150	160
Non-doctorate-granting	230	180	140	150	970
Doctorate-granting	60	70	60	90	290
Private nonprofit					
Less-than-4-year	1,160	1,500	2,940	1,730	3,580
4-year	1,240	570	330	510	110
Non-doctorate-granting	1,390	1,560	480	630	910
Doctorate-granting	910	680	370	900	140
For-profit	290	30	630	1,020	1,140
Price of attendance by institutional type Public	110	80	90	280	390
Less-than-2-year	640	1,190	1,120	2,940	2,960
2-year	100	90	120	140	260
4-year	170	90	70	160	220
Non-doctorate-granting	150	150	140	180	420
Doctorate-granting	140	90	90	140	250
Private nonprofit	-				
Less-than-4-year	1,270	2,330	2,990	2,780	3,290
4-year	800	430	380	520	260
Non-doctorate-granting	730	330	610	710	920
Doctorate-granting	1,010	740	620	470	180
For-profit	550	620	270	210	860

[†] Not applicable.

Table 2.1-A. Average price of attendance for undergraduate students, by type of institution and selected student characteristics: 2007–08

		Public	4-year	Priv nonprof			
		Non-		Non-			
Student characteristics	Public 2-year	doctorate- granting	Doctorate- granting	doctorate- granting	Doctorate- granting	For-profit	All institutions ¹
Total	\$7,000	\$12,700	\$16,600	\$25,200	\$31,600	\$20,600	\$14,000
Attendance status							
Full-time/full-year	12,600	16,700	19,900	32,600	38,300	28,600	22,400
Full-time/part-year	6,800	10,100	12,500	17,900	22,500	16,700	12,200
Part-time/full-year	8,000	11,200	13,800	17,800	21,300	19,300	10,700
Part-time/part-year	3,900	5,700	7,300	8,400	10,400	11,800	5,200
Housing							
On campus	8,800	15,800	19,100	33,100	37,900	30,400	25,300
Off campus	7,000	12,300	16,100	18,600	25,000	20,800	12,600
Living with parents	7,000	10,900	13,800	22,100	27,000	19,600	10,400
Tuition and fees							
Less than \$1,000	5,200	4,400	4,800	4,900	‡	8,400	5,100
\$1,000–2,999	8,700	8,900	9,000	7,900	7,800	8,300	8,700
\$3,000–6,999	13,200	14,800	15,700	13,100	13,900	18,200	15,300
\$7,000 or more	19,800	20,900	22,800	31,700	36,300	23,800	27,800
Gender							
Male	6,900	12,600	16,800	24,800	32,200	21,800	14,000
Female	7,100	12,700	16,500	25,500	31,200	20,100	14,000
Race/ethnicity ²							
White	7,000	13,200	16,700	26,700	32,200	20,900	14,400
Black	7,000	12,800	15,900	21,400	26,800	20,100	13,200
Hispanic	7,000	10,500	15,500	19,500	27,800	19,800	12,400
Asian	7,300	13,300	18,300	29,200	37,000	22,200	15,100
American Indian	7,300	10,400	15,100	27,600	‡	25,000	12,100
Pacific Islander	6,700	13,000	16,900	18,700	40,500	23,100	12,900
Two or more races	7,500	13,000	17,300	28,200	34,500	23,200	15,100
Other	6,800	10,900	15,700	16,500	‡	24,100	12,800
Age as of 12/31/07							
18 years or younger	8,300	14,900	18,500	31,800	37,600	19,900	17,100
19–23 years	7,400	13,800	17,700	30,100	35,200	20,400	16,100
24–29 years	6,900	10,700	13,000	17,500	20,800	20,300	11,600
30–39 years	6,500	9,700	12,500	15,600	17,800	21,300	11,000
40 years or older	6,000	9,100	10,600	14,700	14,500	21,300	9,300
Dependency status							
Dependent	7,600	14,100	17,900	31,000	36,100	20,500	16,700
Independent	6,600	10,400	13,000	16,600	18,900	20,700	11,000
Unmarried, no dependents	6,500	10,500	13,000	17,900	21,900	20,200	11,100
Married, no dependents	6,300	10,400	13,300	16,000	16,800	22,600	10,500
Single parent	6,900	10,900	13,700	17,400	18,600	20,200	11,700
Married parent	6,500	9,600	11,900	14,800	16,200	21,400	10,400

Table 2.1-A. Average price of attendance for undergraduate students, by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year	Priv nonprof			
		Non-	4 your	Non-	it + your		
Student characteristics	Public 2-year	doctorate- granting	Doctorate- granting	doctorate- granting	Doctorate- granting	For-profit	All institutions ¹
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$7,400	\$13,000	\$17,600	\$26,900	\$31,700	\$18,200	\$13,900
\$20,000-39,999	7,300	13,600	17,300	29,200	33,600	19,700	14,500
\$40,000-59,999	7,500	13,500	16,900	29,200	35,300	21,000	14,600
\$60,000-79,999	7,700	14,000	17,600	30,800	34,800	21,100	15,800
\$80,000-99,999	8,000	14,800	18,100	31,800	37,700	22,900	17,900
\$100,000 or more	8,000	15,300	18,700	33,400	37,500	25,300	20,400
Independent							
Less than \$10,000	7,500	12,400	14,400	20,300	23,300	19,100	12,700
\$10,000-19,999	7,000	10,700	13,500	16,900	21,200	19,600	11,600
\$20,000-29,999	6,800	10,100	13,000	16,400	19,600	20,700	11,200
\$30,000-49,999	6,400	9,500	11,900	16,400	17,100	21,800	10,600
\$50,000 or more	5,800	8,700	11,100	14,200	15,400	24,300	9,200
Income group ³							
Lowest 25 percent	7,400	12,800	16,200	24,700	29,300	19,100	13,500
Middle 50 percent	7,100	12,600	16,300	24,900	31,800	20,800	13,700
Highest 25 percent	6,400	12,600	17,400	26,000	32,600	24,600	15,300
Aid status							
No aid	5,900	9,900	15,100	17,800	29,300	17,000	9,100
Received aid	8,300	13,800	17,200	26,300	32,100	20,800	16,600
Grant status							
No grant	6,300	11,500	15,700	19,600	29,000	21,200	11,100
Received grant	8,100	13,700	17,400	26,900	32,700	20,400	16,700
Loan status ⁴							
No loan	6,600	10,800	15,600	21,400	29,300	15,600	10,500
Received loan	9,700	15,100	17,700	27,600	33,500	21,100	19,700

[‡] Reporting standards not met.

NOTE: This table excludes students attending more than one institution. Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

¹ All institutions also include public less-than-2-year and private nonprofit less-than-4-year institutions in addition to those listed separately in this

² Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

³ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁴ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S2.1-A. Standard errors for table 2.1-A: Average price of attendance for undergraduate students, by type of institution and selected student characteristics: 2007–08

		Public	4-year	Priv nonprof			All institutions
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	
Total	\$70	\$170	\$90	\$350	\$370	\$470	\$70
Attendance status							
Full-time/full-year	110	150	90	410	420	360	110
Full-time/part-year	90	420	320	500	2,400	510	270
Part-time/full-year	60	190	150	440	530	600	80
Part-time/part-year	80	160	120	270	300	580	100
Housing							
On campus	390	250	130	480	410	1340	220
Off campus	70	210	100	370	490	500	90
Living with parents	80	260	160	640	1,110	550	100
Tuition and fees							
Less than \$1,000	70	180	210	420	†	1550	70
\$1,000-2,999	80	120	150	250	250	300	60
\$3,000-6,999	190	110	60	590	450	730	130
\$7,000 or more	630	320	110	380	320	480	170
Gender							
Male	80	190	120	500	500	730	100
Female	70	180	110	520	390	450	90
Race/ethnicity							
White	100	180	100	430	340	610	110
Black	90	380	180	770	770	710	200
Hispanic	80	270	230	870	1,960	620	250
Asian	160	370	350	1,680	1,020	1,230	300
American Indian	330	600	820	3,400	†	2,850	910
Pacific Islander	290	1,130	980	3,500	3,090	2,360	660
Two or more races	210	620	360	2,010	1,250	1,040	430
Other	410	1260	950	2,550	†	2,480	940
Age as of 12/31/07							
18 years or younger	120	300	210	670	700	720	230
19–23 years	80	190	100	430	430	520	90
24–29 years	90	210	160	660	630	550	140
30–39 years	110	220	270	630	900	700	160
40 years or older	80	270	450	1,070	570	760	200
Dependency status							
Dependent	90	190	100	440	420	590	90
Independent	70	140	120	460	410	510	110
Unmarried, no dependents	110	210	140	610	740	740	130
Married, no dependents	140	270	310	860	760	1,240	230
Single parent	90	320	300	740	890	560	190
Married parent	90	240	340	680	680	650	160

Table S2.1-A. Standard errors for table 2.1-A: Average price of attendance for undergraduate students, by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year	Priv nonprof			
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All institutions
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$140	\$260	\$210	\$1,060	\$1,570	\$650	\$230
\$20,000-39,999	110	260	180	870	1,160	850	190
\$40,000-59,999	150	320	200	690	620	1,030	180
\$60,000-79,999	170	300	180	810	750	1,510	270
\$80,000-99,999	180	230	190	870	610	870	310
\$100,000 or more	200	300	130	540	480	1,100	180
Independent							
Less than \$10,000	120	280	180	720	1,140	770	220
\$10,000-19,999	110	260	250	740	1,150	580	150
\$20,000-29,999	100	250	300	660	960	840	230
\$30,000-49,999	110	200	320	890	660	600	180
\$50,000 or more	120	270	380	780	590	700	190
Income group							
Lowest 25 percent	90	180	130	600	1,140	630	160
Middle 50 percent	90	190	120	460	380	490	90
Highest 25 percent	120	290	150	560	590	630	140
Aid status							
No aid	70	240	190	1150	850	630	80
Received aid	100	170	90	390	420	480	190
Grant status							
No grant	90	230	140	750	620	440	100
Received grant	110	170	100	410	470	540	200
Loan status							
No loan	80	180	150	510	510	600	60
Received loan	180	170	100	430	410	490	170

[†] Not applicable.

Table 2.1-B. Average price of attendance for full-time/full-year undergraduates, by type of institution and selected student characteristics: 2007–08

		Public	4-year		vate it 4-year		
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions ¹
Total	\$12,600	\$16,700	\$19,900	\$32,600	\$38,300	\$28,600	\$22,400
Housing							
On campus	11,200	17,600	20,500	35,600	39,900	34,500	27,700
Off campus	14,300	17,700	20,500	28,800	37,100	29,100	22,100
Living with parents	11,400	14,000	16,400	26,600	32,000	26,300	16,000
Tuition and fees							
Less than \$1,000	12,200	‡	‡	‡	‡	‡	12,100
\$1,000–2,999	12,000	13,200	15,400	‡	‡	14,000	12,300
\$3,000-6,999	13,800	15,800	17,300	16,700	17,900	29,300	17,300
\$7,000 or more	20,100	21,300	23,200	33,900	39,300	28,300	30,100
Gender							
Male	12,600	16,800	20,000	32,600	38,600	29,900	22,400
Female	12,700	16,600	19,700	32,600	38,100	28,000	22,300
Race/ethnicity ²							
White	12,600	16,900	20,000	33,500	38,700	29,100	22,700
Black	12,800	17,100	19,800	31,500	35,500	29,000	22,000
Hispanic	12,300	15,000	18,300	25,400	34,000	26,500	20,000
Asian	13,400	16,600	21,300	37,200	42,100	31,100	23,800
American Indian	12,100	15,500	18,500	‡	‡	29,100	19,100
Pacific Islander	12,900	‡	19,800	32,100	‡	31,900	22,900
Two or more races	12,300	16,900	19,800	34,000	39,100	30,300	22,800
Other	‡	‡	19,400	‡	‡	‡	22,500
Age as of 12/31/07							
18 years or younger	11,900	16,500	19,500	33,700	39,400	24,400	21,700
19–23 years	12,200	16,600	20,000	33,700	38,700	28,300	22,500
24–29 years	14,200	17,300	19,900	27,200	33,200	28,700	21,900
30–39 years	14,700	16,400	20,500	26,800	31,700	29,700	23,500
40 years or older	14,700	18,200	19,600	27,200	30,100	29,100	23,000
Dependency status							
Dependent	11,900	16,600	19,900	34,000	39,000	27,600	22,400
Independent	14,200	17,100	19,900	27,300	32,600	29,000	22,200
Unmarried, no dependents	13,900	17,200	19,800	28,100	35,600	29,100	21,700
Married, no dependents	14,600	16,500	20,100	27,000	30,600	30,900	22,500
Single parent	14,100	17,300	19,900	26,600	30,500	28,100	22,100
Married parent	14,800	17,000	20,000	27,300	29,800	29,400	23,000

Table 2.1-B. Average price of attendance for full-time/full-year undergraduates, by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year		vate it 4-year		
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions ¹
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$11,700	\$15,300	\$19,300	\$29,800	\$34,000	\$23,500	\$19,200
\$20,000–39,999	11,600	15,900	19,000	31,900	36,000	27,200	20,300
\$40,000–59,999	11,800	16,200	19,100	32,100	38,600	28,800	20,700
\$60,000–79,999	12,100	16,500	19,500	33,300	37,700	29,000	21,400
\$80,000–99,999	12,200	16,700	20,000	34,700	39,800	28,900	23,100
\$100,000 or more	12,300	17,700	20,800	36,700	40,700	32,100	25,500
Independent							
Less than \$10,000	13,900	17,200	19,600	27,100	33,800	27,000	20,900
\$10,000–19,999	14,000	16,800	19,800	26,400	32,300	28,500	21,400
\$20,000–29,999	14,500	16,700	20,600	26,400	29,700	28,700	22,200
\$30,000–49,999	14,500	16,400	19,700	28,000	31,800	29,800	23,600
\$50,000 or more	15,100	18,800	21,000	28,600	33,700	31,800	25,400
Income group ³							
Lowest 25 percent	12,400	16,100	19,200	29,600	34,800	26,600	20,100
Middle 50 percent	12,700	16,600	19,600	32,200	38,000	29,000	21,900
Highest 25 percent	13,000	17,700	20,900	35,700	40,500	31,900	25,800
Aid status							
No aid	12,300	16,300	20,500	35,500	41,200	28,900	19,600
Received aid	12,800	16,700	19,700	32,400	37,800	28,600	23,100
Grant status							
No grant	12,700	16,900	20,300	34,800	40,500	30,600	20,900
Received grant	12,600	16,500	19,600	32,200	37,600	27,900	23,100
Loan status ⁴							
No loan	12,400	15,800	19,800	32,400	38,600	23,600	19,600
Received loan	13,400	17,400	19,900	32,700	38,100	29,100	24,800

[‡] Reporting standards not met.

NOTE: This table excludes students attending more than one institution. Full-time/full-year students represent 38.5 percent of all undergraduates attending only one institution. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

¹ All institutions also include public less-than-2-year and private nonprofit less-than-4-year institutions in addition to those listed separately in this table.

² Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

³ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁴ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S2.1-B. Standard errors for table 2.1-B: Average price of attendance for full-time/full-year undergraduates, by type of institution and selected student characteristics: 2007–08

		Public	4-year	Priv nonprof	/ate it 4-year		
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions
Total	\$110	\$150	\$90	\$410	\$420	\$360	\$110
Housing							
On campus	310	250	130	450	380	950	230
Off campus	150	250	100	390	640	370	140
Living with parents	100	200	180	590	1,570	540	140
Tuition and fees							
Less than \$1,000	180	†	†	†	†	†	180
\$1,000-2,999	130	330	240	†	†	3,880	120
\$3,000–6,999	210	110	90	460	960	390	140
\$7,000 or more	660	360	120	440	330	470	180
Gender							
Male	120	180	120	530	540	570	160
Female	130	160	110	490	390	360	130
Race/ethnicity							
White	120	160	100	450	380	490	130
Black	190	450	170	930	910	370	270
Hispanic	170	250	190	1,130	2,640	920	370
Asian	420	350	270	960	610	650	420
American Indian	620	910	610	†	†	3,550	1,320
Pacific Islander	560	†	920	3,180	†	1,810	1,190
Two or more races	310	420	370	1,860	950	1,020	610
Other	†	†	930	†	†	†	1,490
Age as of 12/31/07							
18 years or younger	130	330	200	630	500	780	210
19–23 years	110	150	90	390	420	580	130
24-29 years	170	300	220	820	930	360	240
30-39 years	310	470	380	750	2,330	470	390
40 years or older	230	480	420	670	1,460	520	440
Dependency status							
Dependent	100	160	100	400	390	700	130
Independent	180	260	180	460	920	340	220
Unmarried, no dependents	190	270	220	930	1,050	520	240
Married, no dependents	400	480	320	620	1,470	440	410
Single parent	230	370	350	890	1,660	440	320
Married parent	300	460	470	640	1,410	400	380

Table S2.1-B. Standard errors for table 2.1-B: Average price of attendance for full-time/full-year undergraduates, by type of institution and selected student characteristics: 2007–08

—Continued

		Public 4-year		Priv nonprof	/ate it 4-year		
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$160	\$230	\$200	\$1,030	\$1,610	\$810	\$280
\$20,000–39,999	150	250	170	700	980	1,090	240
\$40,000–59,999	150	280	160	580	510	1,450	250
\$60,000-79,999	180	210	180	780	690	1,310	280
\$80,000–99,999	200	210	200	790	500	1,420	310
\$100,000 or more	190	270	140	460	500	1,100	220
Independent							
Less than \$10,000	240	280	210	860	1,580	600	260
\$10,000–19,999	210	290	410	760	1,220	520	270
\$20,000–29,999	280	440	430	1,010	1,720	650	400
\$30,000-49,999	290	520	330	700	1,390	340	410
\$50,000 or more	380	570	460	720	1,240	380	470
Income group							
Lowest 25 percent	130	170	130	560	1,190	550	150
Middle 50 percent	130	190	110	470	400	390	130
Highest 25 percent	200	260	150	570	480	370	200
Aid status							
No aid	170	230	180	1,370	910	730	210
Received aid	120	160	90	400	440	370	120
Grant status							
No grant	140	190	130	970	670	390	170
Received grant	130	160	100	390	480	400	130
Loan status							
No loan	110	160	120	660	590	970	130
Received loan	160	180	110	430	390	360	130

[†] Not applicable.

Table 2.1-C. Percentage distribution of undergraduates, by price of attendance and selected institutional and student characteristics: 2007–08

Institutional and student characteristics	Less than \$6,000	\$6,000– 10,999	\$11,000- 18,999	\$19,000 or more
Total	25.4	24.5	26.3	23.7
Institution type				
Public				
Less-than-2-year	33.6	28.9	28.2	9.4 !
2-year	48.2	35.9	14.4	1.4
4-year	10.7	18.2	45.9	25.1
Non-doctorate-granting	17.0	24.6	46.2	12.2
Doctorate-granting	7.3	14.7	45.8	32.2
Private nonprofit				
Less-than-4-year	13.2	22.4	36.1	28.3
4-year	4.6	9.8	15.3	70.3
Non-doctorate-granting	6.1	11.8	17.3	64.9
Doctorate-granting	2.8	7.6	13.2	76.4
For-profit	2.3	15.7	30.6	51.4
Attendance status				
Full-time/full-year	#	9.7	40.0	50.2
Full-time/part-year	18.6	40.8	24.9	15.7
Part-time/full-year	14.0	51.6	26.5	7.9
Part-time/part-year	77.2	14.8	6.3	1.7
Housing				
	2.5	6.6	30.9	60.0
On campus Off campus	27.9	26.1	26.1	19.9
	33.4	31.7	24.1	10.9
Living with parents	33.4	31.7	24.1	10.9
Tuition and fees				
Less than \$1,000	71.3	23.3	5.1	0.4
\$1,000–2,999	25.4	52.2	20.6	1.8
\$3,000–6,999	0.4	22.0	61.6	16.0
\$7,000 or more	#	0.4	20.4	79.1
Gender				
Male	26.4	24.1	25.9	23.6
Female	24.7	24.9	26.6	23.8
Race/ethnicity ¹				
White	25.0	23.3	26.5	25.2
Black	25.6	26.0	26.6	21.8
Hispanic	27.7	28.4	26.5	17.5
Asian	25.0	22.9	24.3	27.7
American Indian	28.6	29.0	24.3	18.1
Pacific Islander	29.1	29.8	18.8	22.4
Two or more races	20.6	25.0	27.9	26.5
Other	25.5	30.0	26.1	18.4
	20.0	30.0	20.1	10.4
Age as of 12/31/07				a
18 years or younger	15.3	21.8	31.3	31.6
19–23 years	18.9	21.5	29.9	29.7
24–29 years	31.0	28.4	24.6	16.1
30–39 years	35.4	28.7	20.0	15.9
40 years or older	43.9	29.8	15.4	10.8

Table 2.1-C. Percentage distribution of undergraduates, by price of attendance and selected institutional and student characteristics: 2007–08—Continued

Institutional and	Less than	\$6,000-	\$11,000-	\$19,000
student characteristics	\$6,000	10,999	18,999	or more
Dependency status				
Dependent	17.2	20.8	30.7	31.3
Independent	34.5	28.7	21.5	15.4
Unmarried, no dependents	33.6	28.2	23.4	14.8
Married, no dependents	39.2	28.1	18.5	14.2
Single parent	30.8	28.3	23.2	17.7
Married parent	37.4	30.0	18.6	14.1
Dependency and income in 2006 Dependent				
Less than \$20,000	21.2	25.4	32.4	20.9
\$20,000–39,999	20.8	24.8	30.9	23.5
\$40,000–59,999	21.1	24.5	30.2	24.2
\$60,000–79,999	19.0	19.6	33.2	28.2
\$80,000–99,999	14.5	18.3	31.2	35.9
\$100,000 or more	11.3	16.0	28.5	44.3
Independent				
Less than \$10,000	24.8	27.3	28.3	19.5
\$10,000–19,999	29.4	29.7	24.9	16.0
\$20,000-29,999	32.0	29.9	22.5	15.6
\$30,000-49,999	37.2	28.5	19.8	14.4
\$50,000 or more	46.6	28.5	13.3	11.6
Income group ²				
Lowest 25 percent	22.9	26.3	30.2	20.7
Middle 50 percent	25.6	24.9	26.9	22.6
Highest 25 percent	27.7	21.9	21.1	29.3
Aid status				
No aid	47.5	26.9	16.3	9.3
Received aid	13.7	23.3	31.6	31.4
	10.7	20.0	01.0	01.4
Grant status	07.4	05.0	04.0	45.0
No grant	37.4	25.6	21.8	15.2
Received grant	14.3	23.6	30.5	31.6
Loan status ³				
No loan	37.9	29.4	20.2	12.4
Received loan	5.0	16.5	36.2	42.2

[#] Rounds to zero.

NOTE: This table excludes students attending more than one institution. Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico. Detail may not sum to totals because of rounding.

[!] Interpret data with caution (estimates are unstable).

¹ Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

² Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

³ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S2.1-C. Standard errors for table 2.1-C: Percentage distribution of undergraduates, by price of attendance and selected institutional and student characteristics: 2007–08

Institutional and student characteristics	Less than \$6,000	\$6,000– 10,999	\$11,000- 18,999	\$19,000 or more
Total	0.43	0.47	0.31	0.32
Institution type				
Public				
Less-than-2-year	6.13	3.21	3.73	4.56
2-year	0.88	0.77	0.38	0.14
4-year	0.38	0.43	0.62	0.54
Non-doctorate-granting	0.85	0.65	0.99	0.76
Doctorate-granting	0.29	0.49	0.69	0.69
Private nonprofit				
Less-than-4-year	1.76	3.07	3.04	5.44
4-year	0.45	0.67	0.69	0.81
Non-doctorate-granting	0.82	1.04	1.04	1.25
Doctorate-granting	0.29	0.54	1.21	1.25
For-profit	0.30	1.00	1.78	2.33
Attendance status				
Full-time/full-year	†	0.34	0.55	0.54
Full-time/part-year	0.69	1.11	0.74	1.08
Part-time/full-year	0.60	0.61	0.54	0.35
Part-time/part-year	1.00	0.73	0.40	0.16
Housing				
On campus	0.22	0.28	0.75	0.84
Off campus	0.63	0.56	0.40	0.45
Living with parents	0.66	0.64	0.51	0.36
Tuition and fees				
Less than \$1,000	0.88	0.71	0.34	0.07
\$1,000–2,999	0.70	0.57	0.48	0.14
\$3,000–6,999	0.07	0.65	0.69	0.65
\$7,000 or more	†	0.10	0.72	0.72
Gender				
Male	0.52	0.60	0.36	0.38
Female	0.47	0.46	0.40	0.41
Race/ethnicity				
White	0.62	0.54	0.36	0.39
Black	1.01	0.56	0.85	0.79
Hispanic	0.78	0.73	0.70	0.90
Asian	1.22	0.93	0.92	1.01
American Indian	2.99	2.80	2.24	3.87
Pacific Islander	2.58	2.76	2.00	2.51
Two or more races	1.64	1.54	1.46	1.51
Other	3.55	3.94	3.23	3.82
Age as of 12/31/07				
18 years or younger	0.80	0.82	0.76	0.87
19–23 years	0.36	0.45	0.42	0.38
24–29 years	0.78	0.75	0.57	0.62
30–39 years	1.27	0.90	0.89	0.68
40 years or older	1.31	0.83	0.72	0.76

Table S2.1-C. Standard errors for table 2.1-C: Percentage distribution of undergraduates, by price of attendance and selected institutional and student characteristics: 2007–08—Continued

Institutional and	Less than	\$6,000-	\$11,000-	\$19,000
student characteristics	\$6,000	10,999	18,999	or more
Dependency status				
Dependent	0.35	0.40	0.45	0.38
Independent	0.82	0.64	0.48	0.51
Unmarried, no dependents	1.14	0.80	0.66	0.52
Married, no dependents	1.73	1.45	0.83	0.85
Single parent	0.93	0.71	0.65	0.83
Married parent	0.91	0.89	0.65	0.74
Dependency and income in 2006 Dependent				
Less than \$20,000	1.06	0.72	1.01	0.77
\$20,000–39,999	0.84	0.79	0.71	0.76
\$40,000–59,999	0.88	0.84	0.72	0.68
\$60,000–79,999	0.94	0.76	0.76	0.84
\$80,000–99,999	0.77	0.84	0.90	1.04
\$100,000 or more	0.48	0.48	0.65	0.70
Independent				
Less than \$10,000	0.66	1.02	0.87	1.04
\$10,000-19,999	1.11	1.10	0.74	0.65
\$20,000-29,999	1.04	0.79	0.82	0.99
\$30,000-49,999	1.17	0.97	0.88	0.77
\$50,000 or more	1.58	0.82	0.90	0.57
Income group				
Lowest 25 percent	0.51	0.61	0.64	0.64
Middle 50 percent	0.64	0.53	0.38	0.34
Highest 25 percent	0.74	0.51	0.43	0.45
Aid status				
No aid	0.63	0.61	0.38	0.26
Received aid	0.46	0.54	0.49	0.72
Grant status				
No grant	0.75	0.49	0.40	0.29
Received grant	0.73	0.59	0.53	0.29
•	0.51	0.55	0.55	0.73
Loan status	0.00	0.00	0.00	0.00
No loan Received loan	0.60 0.24	0.63 0.46	0.29 0.56	0.23 0.74
Neceived Iddii	0.24	0.40	0.00	0.74

[†] Not applicable.

Table 2.2-A. Average tuition and fees for undergraduates, by type of institution and selected student characteristics: 2007–08

		Public	: 4-year		vate iit 4-year		
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate-	For-profit	All insti- tutions ¹
Total	\$1,200	\$4,100	\$6,300	\$15,200	\$20,700	\$10,200	\$5,800
Attendance status							
Full-time/full-year	2,400	5,700	7,700	20,800	25,800	11,900	10,300
Full-time/part-year	1,300	3,100	4,900	9,500	13,400	9,800	5,400
Part-time/full-year	1,200	3,300	4,800	8,800	11,900	11,100	3,200
Part-time/part-year	500	1,500	2,400	3,600	5,100	6,100	1,400
Housing							
On campus	1,900	5,700	7,700	22,100	25,500	18,300	14,100
Off campus	1,000	3,600	5,700	9,400	15,200	9,900	4,500
Living with parents	1,300	3,700	5,500	13,100	18,100	10,900	3,800
Price of attendance							
Less than \$6,000	600	1,100	1,300	1,400	1,900	2,100	700
\$6,000-10,999	1,400	2,600	3,000	3,300	3,700	4,100	2,000
\$11,000-18,999	2,200	4,900	5,400	6,500	7,000	8,300	5,000
\$19,000 or more	3,300	8,300	10,200	21,000	25,400	13,500	16,100
Gender							
Male	1,200	4,100	6,500	15,000	21,100	11,000	5,800
Female	1,200	4,000	6,100	15,400	20,300	9,800	5,800
Race/ethnicity ²							
White	1,200	4,500	6,400	16,600	21,200	10,200	6,200
Black	1,200	3,800	5,600	11,700	16,100	9,700	5,100
Hispanic	900	2,600	5,100	10,000	17,400	10,100	4,500
Asian	1,200	4,700	7,700	19,600	25,200	10,200	6,700
American Indian	1,100	2,600	5,200	17,000	‡	14,900	4,400
Pacific Islander	1,000	3,200	6,000	10,000	27,100	11,900	5,100
Two or more races	1,100	4,000	6,400	17,500	22,900	10,800	6,200
Other	1,000	3,300	5,600	8,700	‡	11,200	5,000
Age as of 12/31/07							
18 years or younger	1,500	5,100	7,300	20,700	25,300	12,000	7,900
19–23 years	1,400	4,700	6,900	19,600	23,600	11,000	7,300
24–29 years	1,000	3,000	4,300	8,700	11,800	10,000	4,100
30–39 years	900	2,500	3,800	6,600	9,300	9,400	3,500
40 years or older	800	2,400	3,000	5,800	7,200	9,000	2,600
Dependency status							
Dependent	1,400	4,900	7,000	20,400	24,300	12,100	7,700
Independent	900	2,800		7,700	10,400	9,600	3,700
Unmarried, no dependents	1,000	2,900		9,400		10,100	4,000
Married, no dependents	900	2,800		7,300		9,500	3,200
Single parent	1,000	2,900		7,600		9,400	3,900
Married parent	900	2,400		6,100		9,300	3,200

Table 2.2-A. Average tuition and fees for undergraduates, by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year		/ate it 4-year		
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions ¹
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$1,400	\$4,000	\$6,500	\$16,400	\$20,500	\$10,800	\$5,700
\$20,000-39,999	1,300	4,600	6,400	18,900	22,300	11,900	6,200
\$40,000-59,999	1,400	4,600	6,400	19,000	23,900	12,800	6,400
\$60,000-79,999	1,500	4,900	6,800	20,500	23,100	12,600	7,100
\$80,000-99,999	1,500	5,300	7,200	21,100	25,500	12,900	8,500
\$100,000 or more	1,500	5,600	7,600	22,300	25,400	13,500	10,300
Independent							
Less than \$10,000	1,200	3,500	4,900	10,500	13,800	9,700	4,700
\$10,000-19,999	1,000	2,800	4,500	8,400	12,400	9,600	4,100
\$20,000–29,999	1,000	2,600	4,200	7,600	10,600	10,000	3,800
\$30,000-49,999	900	2,500	3,500	7,100	8,900	9,400	3,300
\$50,000 or more	800	2,300	3,400	5,800	7,700	8,900	2,600
Income group ³							
Lowest 25 percent	1,300	4,000	5,800	14,600	18,700	10,200	5,300
Middle 50 percent	1,200	4,100	6,100	15,100	20,700	10,300	5,500
Highest 25 percent	1,000	4,300	7,000	15,900	21,500	9,700	6,800
Aid status							
No aid	1,000	3,300	6,300	10,700	19,600	7,600	3,100
Received aid	1,400	4,400	6,300	15,900	20,900	10,300	7,200
Grant status							
No grant	1,000	3,800	6,300	11,200	19,000	10,100	4,100
Received grant	1,300	4,300	6,300	16,500	21,300	10,200	7,400
Loan status ⁴							
No loan	1,100	3,400	6,200	12,800	19,200	7,100	3,700
Received loan	1,800	5,000	6,300	16,800	21,800	10,500	9,200

[‡] Reporting standards not met.

NOTE: This table excludes students attending more than one institution. Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

¹ All institutions also include public less-than-2-year and private nonprofit less-than-4-year institutions in addition to those listed separately in this table

² Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

³ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁴ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S2.2-A. Standard errors for table 2.2-A: Average tuition and fees for undergraduates, by type of institution and selected student characteristics: 2007–08

		Public	: 4-year		/ate it 4-year	For-profit	
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non-	Doctorate- granting		All insti- tutions
Total	\$20	\$130	\$70	\$410	\$320	\$500	\$70
Attendance status							
Full-time/full-year	30	150	80	520	390	570	110
Full-time/part-year	40	190	250	370	1,700	590	240
Part-time/full-year	20	140	120	410	570	590	70
Part-time/part-year	20	90	70	240	230	510	40
Housing							
On campus	110	140	110	440	370	1,110	190
Off campus	20	140	80	350	420	530	90
Living with parents	30	200	90	590	1,140	530	60
Price of attendance							
Less than \$6,000	20	40	40	90	70	100	10
\$6,000-10,999	20	60	30	150	160	230	20
\$11,000–18,999	40	110	50	200	260	220	50
\$19,000 or more	150	260	110	540	270	700	200
Gender							
Male	20	140	90	460	440	560	80
Female	20	140	90	540	330	520	90
Race/ethnicity							
White	30	140	70	490	300	660	90
Black	30	280	100	690	680	670	180
Hispanic	30	150	110	780	1,920	340	170
Asian	50	200	250	1,750	880	840	230
American Indian	80	280	420	3,110	†	2,310	690
Pacific Islander	100	370	570	2,870	2,520	1,520	510
Two or more races	50	300	260	1,870	980	990	320
Other	120	510	620	1,620	†	2,510	590
Age as of 12/31/07							
18 years or younger	40	220	150	640	620	560	170
19–23 years	20	140	80	410	370	470	70
24-29 years	20	100	100	590	560	640	130
30–39 years	20	130		280	670	540	120
40 years or older	20	140		350	470	770	150
Dependency status							
Dependent	30	150	80	400	370	540	70
Independent	20	100		270	360	550	110
Unmarried, no dependents	20	120		570	550	640	100
Married, no dependents	30	110		460	570	940	140
Single parent	20	130		470	710	520	180
Married parent	20	120		260	570	680	120

Table S2.2-A. Standard errors for table 2.2-A: Average tuition and fees for undergraduates, by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year		/ate it 4-year		
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$40	\$180	\$130	\$890	\$1,440	\$550	\$160
\$20,000–39,999	30	190	150	710	1,060	750	150
\$40,000–59,999	40	210	130	640	540	850	130
\$60,000-79,999	50	190	120	710	690	1,330	210
\$80,000–99,999	50	160	130	790	510	940	240
\$100,000 or more	40	180	100	520	430	840	140
Independent							
Less than \$10,000	30	150	100	710	960	690	180
\$10,000–19,999	20	100	150	590	820	450	110
\$20,000–29,999	20	130	170	440	790	880	190
\$30,000-49,999	30	110	100	420	600	570	130
\$50,000 or more	20	130	120	300	470	500	100
Income group							
Lowest 25 percent	20	140	100	570	1,070	570	110
Middle 50 percent	30	150	90	470	340	520	80
Highest 25 percent	20	170	110	600	500	430	110
Aid status							
No aid	20	170	120	900	690	390	50
Received aid	20	130	70	440	370	510	120
Grant status							
No grant	20	170	100	660	520	480	60
Received grant	20	120	70	440	420	530	130
Loan status							
No loan	20	150	110	480	440	280	40
Received loan	40	130	60	490	360	530	150

[†] Not applicable.

Table 2.2-B. Average tuition and fees for full-time/full-year undergraduates, by type of institution and selected student characteristics: 2007–08

		Public	: 4-year		vate fit 4-year		
			7 your		-		AII
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	insti- tutions ¹
Total	\$2,400	\$5,700	\$7,700	\$20,800	\$25,800	\$11,900	\$10,300
Housing							
On campus	2,600	6,500	8,200	24,100	27,100	20,100	15,700
Off campus	2,500	5,700	7,600	15,900	24,300	11,400	8,700
Living with parents	2,400	4,800	6,600	16,400	22,300	13,300	6,300
Price of attendance							
Less than \$6,000	‡	‡	‡	‡	‡	‡	‡
\$6,000-10,999	2,100	3,400		‡		4,100	2,400
\$11,000–18,999	2,500	5,100	5,600	5,600	5,200	8,500	4,700
\$19,000 or more	3,800	8,600	10,100	21,900	26,700	12,300	16,400
Gender							
Male	2,500	5,800	7,900	20,800	26,100	13,200	10,500
Female	2,400	5,600	7,500	20,900	25,600	11,400	10,200
Race/ethnicity ²							
White	2,600	6,100	7,800	21,900	26,300	11,900	10,900
Black	2,300	5,600	7,200	18,500	22,600	11,600	9,100
Hispanic	1,700	4,000		13,800		11,800	8,100
Asian	2,700	5,900	9,100	26,000	29,200	12,700	11,700
American Indian	2,100	4,200	6,700	‡	‡	19,600	8,000
Pacific Islander	2,700	‡	7,500	19,400		13,200	10,400
Two or more races	2,000	5,500		22,200		12,200	10,400
Other	‡	‡	7,700	‡	‡	‡	9,900
Age as of 12/31/07							
18 years or younger	2,300	5,800	7,600	22,100	26,800	14,400	10,600
19–23 years	2,400	5,900	7,800	22,500	26,300	13,800	10,900
24–29 years	2,500	5,100	6,900	14,800	20,800	12,100	8,400
30–39 years	2,500	4,500	6,500	12,300	18,200	10,200	8,000
40 years or older	2,400	5,000	5,700	10,700	17,200	9,600	7,200
Dependency status							
Dependent	2,400	5,900	7,800	22,700	26,500	15,800	11,100
Independent	2,400	4,900	6,600	13,800	20,000	10,800	8,000
Unmarried, no dependents	2,500	5,100	6,900	16,000	22,700	12,100	8,600
Married, no dependents	2,700	4,600	6,600	14,000	17,800	10,400	8,000
Single parent	2,200	4,800	6,400	12,600		10,600	7,600
Married parent	2,400	4,800	5,900	11,900	17,100	10,100	7,600

Table 2.2-B. Average tuition and fees for full-time/full-year undergraduates, by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year		vate it 4-year		
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions ¹
Dependency and income in 2006	2 year	granting	granting	granting	granting	TOI PIOIN	tutions
Dependent Dependent							
Less than \$20,000	\$2,300	\$4,800	\$7,100	\$18,200	\$22,400	\$13,100	\$8,300
\$20,000–39,999	2,300	5,500	7,100	21,000		16,000	9,400
\$40,000–59,999	2,500	5,800	7,300	21,200	,	16,500	9,800
\$60,000–79,999	2,500	6,000	7,500	22,400		16,900	10,300
\$80,000–99,999	2,500	6,100	8,000	23,500		17,500	11,600
\$100,000 or more	2,400	6,600	8,500	24,900	27,800	17,300	13,600
Independent	_,	2,222	-,	_ ,,,,,,,	,	,	,
Less than \$10,000	2,400	5,100	6,700	15,000	21,200	11,400	7,900
\$10,000–19,999	2,300	4,700	6,900	14,100		11,400	7,900
\$20,000-29,999	2,400	4,600	6,800	13,400	16,600	11,400	8,000
\$30,000-49,999	2,400	4,800	5,900	13,000	19,400	10,000	7,900
\$50,000 or more	2,800	5,000	6,700	12,300	20,100	9,500	8,300
Income group ³							
Lowest 25 percent	2,300	5,100	6,900	18,200	22,800	12,400	8,500
Middle 50 percent	2,500	5,800	7,500	20,600	25,600	12,000	9,900
Highest 25 percent	2,500	6,400	8,600	23,400	27,600	10,700	13,200
Aid status							
No aid	2,400	5,800	8,700	23,600	28,400	10,200	8,400
Received aid	2,400	5,700	7,300	20,600	25,400	12,000	10,800
Grant status							
No grant	2,500	6,100	8,400	22,100	27,700	12,000	9,000
Received grant	2,300	5,500	7,200	20,600	25,300	11,900	11,100
Loan status ⁴							
No loan	2,300	5,300	8,000	20,800	26,200	8,700	8,400
Received loan	2,800	6,100	7,300	20,900	25,600	12,200	12,000

[‡] Reporting standards not met.

NOTE: This table excludes students attending more than one institution. Full-time/full-year students represent 38.5 percent of all undergraduates attending only one institution. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

¹ All institutions also include public less-than-2-year and private nonprofit less-than-4-year institutions in addition to those listed separately in this table.

² Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

³ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁴ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S2.2-B. Standard errors for table 2.2-B: Average tuition and fees for full-time/full-year undergraduates, by type of institution and selected student characteristics: 2007–08

		Public	4-year	Priv nonprof			
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions
Total	\$30	\$150	\$80	\$520	\$390	\$570	\$110
Housing							
On campus	120	160	120	410	370	650	210
Off campus	50	190	80	860	620	630	150
Living with parents	40	210	110	680	1,600	570	100
Price of attendance							
Less than \$6,000	†	†	†	†	†	†	†
\$6,000-10,999	40	180	80	†	†	210	40
\$11,000-18,999	50	100	50	240	350	250	40
\$19,000 or more	150	280	110	560	310	630	210
Gender							
Male	40	150	90	570	490	740	140
Female	40	150	100	610	370	560	130
Race/ethnicity							
White	40	140	80	610	350	730	130
Black	80	390	140	910	1,140	680	250
Hispanic	70	200	130	1,130	2,540	580	250
Asian	160	250	240	1,170	640	1,530	400
American Indian	320	320	510	t	†	4,170	1,270
Pacific Islander	360	†	760	2,810	†	3,250	1,120
Two or more races	150	310	300	1,910	960	1,880	600
Other	†	†	900	†	†	†	1,320
Age as of 12/31/07							
18 years or younger	50	270	160	570	480	610	190
19–23 years	40	130	90	390	380	650	120
24–29 years	70	170	170	950	910	760	230
30–39 years	110	350	230	1,260	1,760	700	310
40 years or older	100	260	290	1,390	1,390	860	370
Dependency status							
Dependent	40	150	90	380	370	630	110
Independent	50	160	130	970	810	630	200
Unmarried, no dependents	90	200	190	1,010	810	740	240
Married, no dependents	170	220	210	1,030	1,450	770	250
Single parent	70	190	200	1,200	1,670	710	300
Married parent	70	250	220	1,390	1,250	790	330

Table S2.2-B. Standard errors for table 2.2-B: Average tuition and fees for full-time/full-year undergraduates, by type of institution and selected student characteristics: 2007–08

—Continued

		Public	4-year	Priv nonprof			
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$60	\$180	\$150	\$910	\$1,560	\$640	\$250
\$20,000-39,999	60	190	160	630	930	1,050	210
\$40,000–59,999	70	210	140	700	490	1,450	210
\$60,000-79,999	70	200	160	710	650	1,180	270
\$80,000-99,999	90	170	150	760	440	1,070	290
\$100,000 or more	70	210	130	430	460	1,160	210
Independent							
Less than \$10,000	80	200	160	1,070	1,420	780	230
\$10,000–19,999	70	160	290	910	1,070	650	220
\$20,000–29,999	90	220	400	940	1,620	990	360
\$30,000–49,999	90	320	240	1,850	1,210	850	420
\$50,000 or more	140	240	350	1,860	990	480	280
Income group							
Lowest 25 percent	40	150	110	610	1,150	640	140
Middle 50 percent	40	170	90	560	370	650	130
Highest 25 percent	80	190	140	760	440	440	210
Aid status							
No aid	60	200	190	1,300	810	790	200
Received aid	30	140	70	520	420	570	120
Grant status							
No grant	50	190	120	990	580	560	150
Received grant	30	130	70	530	450	620	130
Loan status							
No loan	30	160	110	730	540	450	120
Received loan	50	150	80	560	370	590	160

[†] Not applicable.

Table 2.2-C. Average tuition and fees for part-time or part-year undergraduates, by type of institution and selected student characteristics: 2007–08

		Public	: 4-year	Priv nonprof			
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions ¹
Total	\$900	\$2,500	\$4,000	\$7,400	\$10,400	\$9,100	\$3,000
Attendance status							
Full-time/full-year	†	†	†	†	†	†	†
Full-time/part-year	1,300	3,100		9,500	13,400	9,800	5,400
Part-time/full-year	1,200	3,300	4,800	8,800	11,900	11,100	3,200
Part-time/part-year	500	1,500		3,600	5,100	6,100	1,400
Housing							
On campus	1,300	3,600	6,000	13,600	16,900	15,000	8,400
Off campus	800	2,300	3,600	5,600	8,000	9,000	2,800
Living with parents	1,000	2,400	3,900	7,900	10,500	9,600	2,300
Price of attendance							
Less than \$6,000	600	1,100	1,300	1,400	1,900	2,100	700
\$6,000-10,999	1,200	2,400	3,000	3,300	3,700	4,100	2,000
\$11,000-18,999	1,700	4,100		6,700	7,400	8,200	5,400
\$19,000 or more	1,600	6,600		16,400	18,800	15,800	14,700
Gender							
Male	900	2,500	4,200	7,200	10,300	9,700	2,800
Female	900	2,500	3,900	7,500	10,400	8,900	3,100
Race/ethnicity ²							
White	900	2,800	4,200	8,100	10,300	9,200	3,000
Black	1,000	2,200	3,600	5,700	8,500	8,800	3,200
Hispanic	700	1,700		5,600	9,900	9,000	2,700
Asian	700	3,400	5,200	9,400	14,800	8,700	3,000
American Indian	900	1,600	3,000	‡	‡	13,000	2,800
Pacific Islander	700	2,100	3,300	‡	‡	11,000	2,600
Two or more races	900	2,700	3,800	8,300	14,500	9,500	3,000
Other	800	‡	‡	‡	‡	12,400	3,000
Age as of 12/31/07							
18 years or younger	1,000	3,200	6,000	13,200	17,700	10,400	3,700
19–23 years	1,000	2,900	4,700	11,300	13,800	9,500	3,500
24–29 years	900	2,200	3,200	5,700	7,700	9,000	2,900
30–39 years	800	2,100	3,000	4,600	6,700	8,800	2,600
40 years or older	700	2,000	2,600	4,500	5,900	8,500	2,000
Dependency status							
Dependent	1,000	3,000	4,900	12,300	15,000	10,100	3,600
Independent	800	2,100		5,000	6,900	8,800	2,600
Unmarried, no dependents	800	2,100		6,200	8,200	9,100	2,800
Married, no dependents	800	2,200		4,500	5,400	8,800	2,200
Single parent	800	2,200		4,900	6,900	8,600	2,900
Married parent	800	2,000		4,300	6,000	8,800	2,300

Table 2.2-C. Average tuition and fees for part-time or part-year undergraduates, by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year	Private nonprofit 4-year			
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions ¹
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$1,000	\$2,600	\$4,700	\$10,700	\$12,400	\$9,600	\$3,100
\$20,000-39,999	1,000	3,000	4,300	10,200	14,600	9,700	3,100
\$40,000–59,999	1,000	2,600	4,300	11,900	14,100	10,800	3,000
\$60,000-79,999	1,000	3,000	5,100	13,300	13,000	10,500	3,300
\$80,000-99,999	1,100	3,600	5,000	11,800	17,400	10,600	3,900
\$100,000 or more	1,100	3,300	5,300	13,400	15,700	10,400	4,800
Independent							
Less than \$10,000	900	2,400	3,400	5,900	8,000	8,900	3,300
\$10,000–19,999	900	2,100	3,200	5,200	7,100	8,700	2,900
\$20,000–29,999	800	2,000	3,300	5,400	7,900	9,100	2,700
\$30,000–49,999	800	2,000	2,800	5,200	6,700	9,000	2,400
\$50,000 or more	700	2,100	2,900	4,400	6,000	8,400	1,900
Income group ³							
Lowest 25 percent	900	2,600	3,900	7,600	10,100	9,100	3,200
Middle 50 percent	900	2,400	3,900	7,400	10,600	9,200	2,900
Highest 25 percent	800	2,500	4,400	7,100	10,200	8,900	2,900
Aid status							
No aid	800	2,200	3,800	5,300		6,400	1,600
Received aid	1,000	2,700	4,200	7,900	10,700	9,200	4,100
Grant status							
No grant	800	2,400	3,900	5,500	9,100	9,100	2,200
Received grant	1,000	2,600	4,300	8,500	11,200	9,100	4,000
Loan status ⁴							
No loan	800	2,200	3,900	5,800	9,100	6,200	1,800
Received loan	1,400	3,200	4,200	8,900	11,900	9,400	6,000

[†] Not applicable.

NOTE: This table excludes students attending more than one institution. Includes full-time/part-year, part-time/full-year, and part-time/part-year students. Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Part-time or part-year students comprise 59 percent of all undergraduates attending only one institution. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[‡] Reporting standards not met.

¹ All institutions also include public less-than-2-year and private nonprofit less-than-4-year institutions in addition to those listed separately in this table.

² Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

³ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁴ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S2.2-C. Standard errors for table 2.2-C: Average tuition and fees for part-time or part-year undergraduates, by type of institution and selected student characteristics: 2007–08

		Public	: 4-year	Priv nonprof	/ate it 4-vear		
			4-yeai		it 4-yeai	,	
	Dublic	Non-	Dantaunta	Non-	Dantaunta		All
Student characteristics	Public 2-year	granting	Doctorate- granting	granting	Doctorate- granting	For-profit	insti- tutions
Total	\$20	\$110	\$110	\$230	\$770	\$540	\$70
Attendance status							
Full-time/full-year	†	†	†	†	†	†	†
Full-time/part-year	40	190	250	370	1,700	590	240
Part-time/full-year	20	140		410	570	590	70
Part-time/part-year	20	90	70	240	230	510	40
Housing							
On campus	130	190		640	1,590	2,830	400
Off campus	20	100		260	430	560	80
Living with parents	20	170	110	630	890	640	60
Price of attendance							
Less than \$6,000	20	40		90	70	100	10
\$6,000–10,999	20	50	40	150	160	240	20
\$11,000–18,999	50	180		230	190	250	80
\$19,000 or more	190	740	650	610	1,050	840	490
Gender							
Male	20	110	150	430	860	590	70
Female	20	120	110	280	820	560	90
Race/ethnicity							
White	20	130	130	340	840	730	90
Black	30	110	90	350	570	770	190
Hispanic	20	100	130	380	1,300	390	110
Asian	50	240	350	1,720	3,150	850	240
American Indian	70	250		†	†	2,270	510
Pacific Islander	80	160		†	†	1,730	410
Two or more races	40	380		1,390	1,760	610	190
Other	110	†	†	†	†	3,240	510
Age as of 12/31/07							
18 years or younger	30	180		1,010	3,240	730	290
19–23 years	20	160	170	470	1,070	520	80
24–29 years	20	80		480	480	660	110
30–39 years	30	120		210	410	670	110
40 years or older	20	140	160	330	330	810	130
Dependency status							
Dependent	20	180		420	1,330	600	110
Independent	20	90		210	260	600	100
Unmarried, no dependents	20	90		490	440	760	90
Married, no dependents	30	130		370	350	1,340	140
Single parent	20	120		280	590	530	170
Married parent	20	100	130	260	410	770	100

Table S2.2-C. Standard errors for table 2.2-C: Average tuition and fees for part-time or part-year undergraduates, by type of institution and selected student characteristics: 2007–08

—Continued

Student characteristics		Public	4-year	Private nonprofit 4-year			
	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$40	\$210	\$250	\$1,790	\$1,470	\$720	\$160
\$20,000-39,999	30	250	170	1,040	2,040	770	130
\$40,000-59,999	40	190	180	870	1,920	870	140
\$60,000-79,999	40	180	250	1,220	1,300	1,470	170
\$80,000-99,999	50	300	200	1,230	2,850	1,140	290
\$100,000 or more	40	180	330	850	1,390	1,110	200
Independent							
Less than \$10,000	20	110	110	480	610	780	180
\$10,000–19,999	20	100	110	410	740	510	100
\$20,000–29,999	30	130	140	400	760	990	160
\$30,000–49,999	30	90	90	370	380	570	100
\$50,000 or more	20	140	130	300	310	680	100
Income group							
Lowest 25 percent	20	140	90	570	1,010	630	110
Middle 50 percent	20	120	110	360	880	520	80
Highest 25 percent	20	130	230	480	880	610	90
Aid status							
No aid	20	130	150	410	670	350	30
Received aid	20	110	110	240	850	550	120
Grant status							
No grant	20	130	130	350	640	580	50
Received grant	20	110	120	270	940	560	130
Loan status							
No loan	20	120	140	260	680	240	30
Received loan	30	110	100	290	1,020	570	210

[†] Not applicable.

Table 2.2-D. Percentage distribution of all undergraduates, by tuition and fees and selected institutional and student characteristics: 2007–08

Institutional and	Less than	\$1,000-	\$3,000-	\$7,000
Total	\$1,000 26.7	2,999 24.7	6,999 24.2	or more 24.4
	20.7	24.7	24.2	24.4
Institution type				
Public	20.2	04.01	22.4	_
Less-than-2-year	39.3 56.6	24.0 ! 36.4	23.1 6.8	‡
2-year	50.0 5.1	21.6	48.2	0.1 25.1
4-year Non-doctorate-granting	9.5	32.1	46.7	11.7
Doctorate-granting	2.8	16.0	49.0	32.3
Private nonprofit	2.0	10.0	49.0	32.3
Less-than-4-year	3.4	21.9	38.5	36.2
4-year	1.8	7.8	15.9	74.5
Non-doctorate-granting	2.9	9.5	18.7	68.9
Doctorate-granting	0.5	5.9	12.7	80.8
For-profit	0.6 !	7.3	34.5	57.5
			00	0.10
Attendance status	2.2	40.5	20.5	45.0
Full-time/full-year	3.3	12.5	38.5	45.8
Full-time/part-year	15.4 29.7	35.8 41.7	24.1 17.4	24.8 11.2
Part-time/full-year Part-time/part-year	29.7 66.1	22.6	8.3	3.0
•	00.1	22.0	0.3	3.0
Housing				
On campus	1.4	6.6	30.5	61.5
Off campus	30.9	26.1	23.6	19.4
Living with parents	32.2	32.4	21.7	13.7
Gender				
Male	27.5	24.1	24.3	24.1
Female	26.1	25.2	24.1	24.6
Race/ethnicity ¹				
White	24.4	24.9	25.0	25.8
Black	25.6	28.2	23.4	22.8
Hispanic	35.0	24.1	22.3	18.6
Asian	31.3	17.3	23.3	28.1
American Indian	33.4	29.6	21.4	15.6
Pacific Islander	38.9	22.1	18.2	20.8
Two or more races	27.6	21.5	24.2	26.7
Other	33.6	21.3	24.7	20.4
Age as of 12/31/07				
18 years or younger	17.6	22.8	26.2	33.5
19–23 years	18.6	22.6	28.0	30.9
24–29 years	32.3	29.1	22.0	16.6
30–39 years	38.7	27.6	19.1	14.7
40 years or older	49.4	25.8	14.5	10.2

Table 2.2-D. Percentage distribution of all undergraduates, by tuition and fees and selected institutional and student characteristics: 2007–08—Continued

Institutional and	Less than	\$1,000-	\$3,000-	\$7,000
student characteristics	\$1,000	2,999	6,999	or more
Dependency status				
Dependent	17.1	21.8	28.3	32.8
Independent	37.4	28.0	19.6	15.1
Unmarried, no dependents	35.1	27.3	21.5	16.1
Married, no dependents	40.4	28.4	19.3	11.9
Single parent	35.4	27.0	20.0	17.6
Married parent	41.0	29.6	16.8	12.6
Dependency and income in 2006				
Dependent				
Less than \$20,000	22.5	25.6	28.1	23.9
\$20,000–39,999	21.5	24.8	27.7	26.0
\$40,000–59,999	21.1	25.3	27.3	26.2
\$60,000-79,999	17.9	23.1	28.8	30.1
\$80,000–99,999	14.2	20.0	29.1	36.7
\$100,000 or more	10.6	16.3	28.8	44.4
Independent	07.4	07.0	05.4	00.5
Less than \$10,000	27.4	27.0	25.1	20.5
\$10,000–19,999	31.5	29.1	22.7	16.7
\$20,000–29,999	35.4 40.8	30.1	18.6	15.9
\$30,000–49,999 \$50,000 or more	40.8 49.4	28.0 26.7	17.6 14.4	13.6 9.5
	49.4	20.7	14.4	9.5
Income group ²				
Lowest 25 percent	24.5	26.3	26.7	22.5
Middle 50 percent	26.9	25.7	24.0	23.4
Highest 25 percent	28.7	21.0	21.8	28.5
Aid status				
No aid	46.9	27.5	15.0	10.5
Received aid	16.0	23.2	29.0	31.8
Grant status				
No grant	37.1	26.7	20.2	15.9
Received grant	17.1	22.8	27.9	32.3
Loan status ³				
No loan	40.1	29.1	17.9	12.9
Received loan	4.8	17.5	34.4	43.2

[!] Interpret data with caution (estimates are unstable).

NOTE: This table excludes students attending more than one institution. Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico. Detail may not sum to totals because of rounding.

[‡] Reporting standards not met.

¹ Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

² Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

³ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S2.2-D. Standard errors for table 2.2-D: Percentage distribution of all undergraduates, by tuition and fees and selected institutional and student characteristics: 2007–08

Institutional and	Less than	\$1,000-	\$3,000-	\$7,000
student characteristics	\$1,000	2,999	6,999	or more
Total	0.48	0.61	0.33	0.33
Institution type				
Public				
Less-than-2-year	7.66	7.37	3.88	†
2-year	1.03	1.01	0.34	0.03
4-year	0.36	0.74	0.72	0.87
Non-doctorate-granting	1.03	1.12	1.34	1.28
Doctorate-granting	0.21	0.72	0.81	0.89
Private nonprofit	0.04	4.45	T 40	7.00
Less-than-4-year	0.84	4.45	5.40	7.22
4-year	0.41 0.74	0.55 0.97	0.84	0.90
Non-doctorate-granting	0.74	0.54	1.63	1.46
Doctorate-granting	0.12	1.00	1.20 1.63	1.38 1.93
For-profit	0.21	1.00	1.03	1.93
Attendance status				
Full-time/full-year	0.17	0.53	0.70	0.58
Full-time/part-year	1.08	1.02	0.90	1.33
Part-time/full-year	0.78	0.85	0.44	0.35
Part-time/part-year	1.04	0.87	0.43	0.28
Housing				
On campus	0.17	0.33	0.67	0.75
Off campus	0.69	0.65	0.38	0.38
Living with parents	0.75	0.91	0.55	0.42
Gender				
Male	0.59	0.68	0.44	0.44
Female	0.50	0.61	0.36	0.41
Race/ethnicity				
White	0.72	0.67	0.40	0.38
Black	1.02	0.88	0.70	0.99
Hispanic	1.09	1.02	0.76	0.98
Asian	1.44	1.39	1.02	1.08
American Indian	3.70	2.24	2.75	3.58
Pacific Islander	3.89	3.64	2.20	2.63
Two or more races	1.80	1.38	1.59	1.56
Other	4.37	3.14	3.79	3.34
Age as of 12/31/07				
_	0.91	1.09	0.87	1.00
18 years or younger 19–23 years	0.42	0.57	0.42	0.43
24–29 years	0.42	0.80	0.52	0.43
30–39 years	1.14	0.90	0.61	0.62
40 years or older	1.14	0.90	0.67	0.00

Table S2.2-D. Standard errors for table 2.2-D: Percentage distribution of all undergraduates, by tuition and fees and selected institutional and student characteristics: 2007–08—Continued

Institutional and	Less than	\$1,000-	\$3,000-	\$7,000
student characteristics	\$1,000	2,999	6,999	or more
Dependency status				
Dependent	0.42	0.62	0.44	0.47
Independent	0.79	0.67	0.40	0.44
Unmarried, no dependents	1.09	0.89	0.55	0.53
Married, no dependents	1.40	1.15	0.80	0.77
Single parent	1.18	0.79	0.63	0.78
Married parent	0.83	0.76	0.58	0.64
Dependency and income in 2006				
Dependent				
Less than \$20,000	0.99	0.96	1.07	0.78
\$20,000–39,999	0.77	0.91	0.66	0.87
\$40,000–59,999	0.85	0.91	0.71	0.76
\$60,000–79,999	1.03	0.94	0.77	0.84
\$80,000–99,999	0.96	0.90	0.76	0.96
\$100,000 or more	0.51	0.57	0.62	0.63
Independent				
Less than \$10,000	0.81	0.83	0.73	0.81
\$10,000–19,999	0.83	0.80	0.69	0.59
\$20,000–29,999	1.06	1.00	0.83	0.90
\$30,000–49,999	1.31	1.07	0.66	0.67
\$50,000 or more	1.52	1.02	0.67	0.61
Income group				
Lowest 25 percent	0.56	0.62	0.53	0.59
Middle 50 percent	0.67	0.70	0.38	0.37
Highest 25 percent	0.76	0.68	0.47	0.47
Aid status				
No aid	0.83	0.86	0.36	0.28
Received aid	0.57	0.63	0.52	0.73
Grant status				
No grant	0.89	0.79	0.39	0.30
Received grant	0.63	0.61	0.54	0.71
Loan status				
No loan	0.68	0.86	0.33	0.27
Received loan	0.29	0.47	0.57	0.69

[†] Not applicable.

Table 3.1-A. Percentage of undergraduates receiving aid, by type of aid received and selected institutional and student characteristics: 2007–08

Institutional and	Any	Any	Any	Any	Any other
student characteristics	aid¹	grants	work-study	loan²	type of aid
Total	65.6	51.7	7.4	38.5	6.8
Institution type					
Public					
Less-than-2-year	54.1	44.5	#	17.8	10.5
2-year	47.6	39.6	3.3	13.2	3.2
4-year	71.3	52.9	7.7	46.2	8.7
Non-doctorate-granting	70.2	52.5	7.3	43.4	7.2
Doctorate-granting	71.9	53.1	8.0	47.8	9.4
Private nonprofit	70.0	50.0	0.0	40.0	0.01
Less-than-4-year	79.6	59.8	2.8	43.9	9.2 !
4-year	84.7	73.6	23.2	58.9	11.4
Non-doctorate-granting	87.3	76.2	23.2	61.2	12.1
Doctorate-granting	81.7	70.7	23.2	56.5	10.7
For-profit	96.3	70.4	2.0	91.6	8.6
More than one institution	68.9	51.3	7.7	44.7	8.3
Attendance status					
Full-time/full-year	79.5	64.4	13.5	52.9	10.2
Full-time/part-year	71.0	53.3	5.2	46.0	7.9
Part-time/full-year	61.2	48.7	4.1	30.0	4.2
Part-time/part-year	44.3	33.3	2.0	18.9	3.1
Housing ³					
On campus	80.1	66.6	22.5	57.2	13.8
Off campus	65.5	51.3	4.5	38.3	5.9
Living with parents	56.1	43.9	4.9	36.3 25.7	4.3
	30.1	43.3	4.3	25.7	4.5
Price of attendance ³					
Less than \$6,000	35.3	29.1	0.8	7.5	1.9
\$6,000–10,999	62.0	49.7	3.4	25.6	4.1
\$11,000–18,999	78.5	60.0	7.3	52.2	8.0
\$19,000 or more	86.4	69.0	18.7	67.5	13.0
Gender					
Male	61.4	46.7	7.2	34.8	8.2
Female	68.8	55.5	7.6	41.3	5.8
Daga (athorists 4					
Race/ethnicity ⁴	20.5	40.0	7.0	00.0	- .
White	63.5	48.2	7.3	38.2	7.4
Black	76.2	63.5	7.2	49.5	6.3
Hispanic	69.0	58.1	7.2	34.3	5.5
Asian	53.4	43.1	8.7	25.5	4.3
American Indian	70.8	60.4	5.3	35.5	4.7
Pacific Islander	61.5	49.3	7.7	31.9	8.8
Two or more races	68.1	53.8	11.4	41.4	7.9
Other	64.1	51.1	6.1	35.1	6.7
Age as of 12/31/07					
18 years or younger	70.9	60.2	11.8	38.9	9.2
19–23 years	66.1	51.0	10.1	40.2	7.6
24–29 years	66.3	52.0	3.8	41.6	5.9
30–39 years	66.4	53.0	2.7	37.6	4.9
40 years or older	56.2	45.0	2.0	25.8	4.4

Table 3.1-A. Percentage of undergraduates receiving aid, by type of aid received and selected institutional and student characteristics: 2007–08—Continued

Institutional and student characteristics	Any aid¹	Any	Any	Any Ioan²	Any other
	alu.	grants	work-study	ioan	type of aid
Dependency status					
Dependent	66.5	51.4	11.0	40.2	8.3
Independent	64.6	52.0	3.3	36.6	5.1
Unmarried, no dependents	62.5	47.1	4.3	38.4	5.7
Married, no dependents	56.4	38.7	3.5	27.9	5.3
Single parent	72.2	66.3	3.2	42.0	3.5
Married parent	63.0	49.1	2.1	32.5	6.1
Dependency and income in 2006 Dependent					
Less than \$20,000	80.7	76.7	12.1	39.8	4.7
\$20,000–39,999	78.5	71.8	14.6	44.4	6.8
\$40,000–59,999	66.8	53.1	12.5	43.0	7.6
\$60,000-79,999	63.3	41.7	11.0	42.8	9.9
\$80,000–99,999	63.1	41.7	10.2	42.2	10.4
\$100,000 or more	56.4	37.4	8.0	33.7	9.3
Independent		-			
Less than \$10,000	74.0	67.4	6.8	45.0	4.9
\$10,000–19,999	71.5	62.3	3.7	45.6	5.1
\$20,000–29,999	68.6	51.5	3.0	40.6	5.6
\$30,000–49,999	62.9	49.3	1.8	34.9	5.1
\$50,000 or more	49.7	32.4	1.2	20.9	5.1
Income group ⁵					
Lowest 25 percent	77.3	71.5	10.2	43.6	5.3
Middle 50 percent	65.8	49.9	7.3	41.3	7.3
Highest 25 percent	53.1	34.9	4.8	27.5	7.3
· ·	33.1	34.3	4.0	21.5	7.5
Aid status					
No aid	†	†	†	_ †	†
Received aid	100.0	78.8	11.3	58.7	10.4
Grant status					
No grant	28.8	†	2.9	24.6	5.7
Received grant	100.0	100.0	11.6	51.4	7.9
Loan status ²					
No loan	44.1	40.8	4.2	†	3.5
Received loan	100.0	69.1	12.5	100.0	12.1

[†] Not applicable.

NOTE: Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[#] Rounds to zero.

[!] Interpret data with caution (estimates are unstable).

¹ Any aid excludes federal education tax benefits but includes federal Parent PLUS loans, veterans benefits, and WIA job training funds. For students attending more than one institution, includes aid received at the NPSAS institution, but may not include aid received at other institutions.

² Loans only include loans to students. Excludes federal Parent PLUS loans.

 $^{^{\}rm 3}$ Excludes students attending more than one institution.

⁴ Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

⁵ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

Table S3.1-A. Standard errors for table 3.1-A: Percentage of undergraduates receiving aid, by type of aid received and selected institutional and student characteristics: 2007–08

Institutional and	Any	Any	Any	Any	Any other
student characteristics	aid	grants	work-study	loan	type of aid
Total	1.20	0.96	0.20	0.23	0.18
Institution type					
Public					
Less-than-2-year	3.11	2.19	†	1.49	2.91
2-year	2.42	1.99	0.27	0.41	0.34
4-year	0.73	0.59	0.32	0.29	0.26
Non-doctorate-granting	0.76	0.87	0.50	0.78	0.34
Doctorate-granting	0.86	0.58	0.33	0.59	0.34
Private nonprofit					
Less-than-4-year	2.99	2.55	0.59	1.28	3.40
4-year	0.97	0.93	0.79	0.54	0.45
Non-doctorate-granting	1.06	1.13	1.28	1.04	0.66
Doctorate-granting	1.21	1.18	0.87	1.01	0.54
For-profit	0.25	0.91	0.29	0.26	0.63
More than one institution	2.05	1.66	0.44	2.73	0.40
Attendance status					
Full-time/full-year	0.37	0.38	0.29	0.42	0.20
Full-time/part-year	1.52	1.24	0.37	0.83	0.47
Part-time/full-year	1.24	1.15	0.26	0.40	0.21
Part-time/part-year	2.51	1.96	0.26	1.09	0.30
, ,	2.01	1.00	0.20	1.00	0.00
Housing					
On campus	1.21	1.14	0.85	0.83	0.46
Off campus	1.53	1.19	0.21	0.51	0.25
Living with parents	1.66	1.67	0.26	0.44	0.20
Price of attendance					
Less than \$6,000	2.40	2.08	0.15	0.50	0.27
\$6,000-10,999	1.65	1.48	0.29	0.53	0.31
\$11,000–18,999	1.03	0.82	0.28	0.52	0.22
\$19,000 or more	0.68	0.70	0.61	0.58	0.35
Gender					
Male	1.38	1.16	0.31	0.44	0.31
Female	1.04	0.78	0.18	0.25	0.15
	1.01	0.70	0.10	0.20	0.10
Race/ethnicity	4.00	2.25	0.05	0.50	0.07
White	1.32	0.95	0.25	0.52	0.27
Black	1.56	1.31	0.35	1.04	0.31
Hispanic	0.76	0.99	0.33	1.10	0.32
Asian	1.96	1.54	0.56	0.88	0.35
American Indian	2.92	3.12	1.12	3.34	1.06
Pacific Islander	3.28	3.10	1.46	2.99	1.77
Two or more races	1.51	1.43	1.02	1.54	0.70
Other	4.03	3.96	1.60	4.02	1.85
Age as of 12/31/07					
18 years or younger	0.99	1.15	0.44	0.79	0.45
19–23 years	1.08	0.97	0.29	0.31	0.19
24–29 years	1.60	1.10	0.23	0.61	0.35
30-39 years	1.82	1.48	0.28	0.78	0.36
40 years or older	1.48	1.09	0.17	0.83	0.49

Table S3.1-A. Standard errors for table 3.1-A: Percentage of undergraduates receiving aid, by type of aid received and selected institutional and student characteristics: 2007–08

—Continued

Institutional and	Any	Any	Any	Any	Any other
student characteristics	aid	grants	work-study	loan	type of aid
Dependency status					
Dependent	1.06	0.94	0.29	0.32	0.19
Independent	1.42	1.06	0.16	0.35	0.29
Unmarried, no dependents	1.63	1.16	0.22	0.73	0.41
Married, no dependents	1.30	0.89	0.33	1.31	0.55
Single parent	2.16	1.93	0.25	0.85	0.24
Married parent	1.09	1.04	0.23	0.67	0.44
Dependency and income in 2006					
Dependent					
Less than \$20,000	1.13	1.25	0.59	1.04	0.32
\$20,000–39,999	1.07	1.10	0.51	0.82	0.37
\$40,000–59,999	1.69	1.41	0.53	0.80	0.29
\$60,000-79,999	1.27	1.17	0.71	0.89	0.51
\$80,000-99,999	0.98	1.02	0.60	0.77	0.48
\$100,000 or more	1.06	0.81	0.31	0.69	0.37
Independent					
Less than \$10,000	1.25	1.08	0.33	0.80	0.34
\$10,000-19,999	1.74	1.50	0.30	0.78	0.40
\$20,000-29,999	2.17	1.89	0.37	0.89	0.49
\$30,000-49,999	1.74	1.27	0.22	0.89	0.51
\$50,000 or more	1.07	0.79	0.18	0.75	0.43
Income group					
Lowest 25 percent	1.07	1.02	0.28	0.59	0.22
Middle 50 percent	1.45	1.15	0.28	0.45	0.22
Highest 25 percent	0.90	0.61	0.19	0.56	0.31
Aid status					
No aid	†	†	†	†	†
Received aid	†	0.23	0.23	0.76	0.20
Grant status					
No grant	1.13	†	0.23	0.82	0.33
Received grant	†	†	0.26	0.82	0.19
Loan status					
No loan	1.77	1.58	0.25	†	0.25
Received loan	†	0.36	0.28	†	0.26

[†] Not applicable.

Table 3.1-B. Average amount of aid received by undergraduates receiving a specific type of aid, by type of aid and selected institutional and student characteristics: 2007–08

		Any	Any	Any	Any
Institutional and	Any aid	grant	work-study	loan	other
student characteristics	amount ¹	amount	amount	amount ²	type of aid
Total	\$9,100	\$4,900	\$2,400	\$7,100	\$8,100
Institution type					
Public					
Less-than-2-year	4,700	2,700	‡	5,700	3,000
2-year	3,400	2,200	3,000	4,100	3,600
4-year	9,400	5,200	2,500	6,600	8,200
Non-doctorate-granting	8,000	4,300	2,500	6,300	6,500
Doctorate-granting	10,100	5,600	2,500	6,800	8,900
Private nonprofit					
Less-than-4-year	7,800	4,000	2,000	7,000	7,400
4-year	17,400	10,200	2,100	9,100	12,200
Non-doctorate-granting	16,000	9,300	1,900	8,400	10,500
Doctorate-granting	19,000	11,100	2,200	9,900	14,200
For-profit	10,800	3,200	3,500	8,100	7,900
More than one institution	9,000	4,400	2,200	6,800	8,400
Attendance status					
Full-time/full-year	12,700	7,100	2,300	8,000	9,900
Full-time/part-year	7,500	3,400	2,200	6,300	6,300
Part-time/full-year	6,200	2,900	2,800	6,700	6,100
Part-time/part-year	3,800	1,700	3,300	4,800	3,900
Housing ³					
On campus	16,200	9,600	2,100	7,800	11,500
Off campus	7,900	3,700	2,600	7,200	6,700
Living with parents	6,500	3,800	2,900	6,100	6,100
Price of attendance ³					
Less than \$6,000	1,700	1,200	1,600	2,700	1,600
\$6,000–10,999	4,200	2,500	2,600	4,400	3,200
\$11,000–18,999	8,500	4,300	2,500	6,500	6,400
\$19,000 or more	16,700	8,900	2,400	9,200	11,900
Gender					
Male	9,300	5,000	2,500	7,200	8,100
Female	9,000	4,800	2,300	7,000	8,200
Dago/othnicity. ⁴					
Race/ethnicity ⁴ White	9,400	5,000	2,400	7,200	8,500
Black	9,000	4,400	2,400	7,000	7,000
Hispanic	7,900	4,300	2,600	6,900	7,600
Asian	9,600	6,400	2,500	6,800	8,800
American Indian	7,200	4,300	2,100	5,800	6,400
Pacific Islander	8,600	4,300 4,100	2,500	7,400	8,000
Two or more races	10,300	5,800	2,600	7,400 7,200	7,600
Other	7,800	4,200	1,600	6,700	7,000 ‡

Table 3.1-B. Average amount of aid received by undergraduates receiving a specific type of aid, by type of aid and selected institutional and student characteristics: 2007–08—Continued

		Any	Any	Any	Any	
Institutional and	Any aid	grant	work-study	loan	other	
student characteristics	amount ¹	amount	amount	amount ²	type of aid	
Age as of 12/31/07						
18 years or younger	\$10,600	\$6,600	\$2,100	\$6,100	\$10,400	
19–23 years	10,400	6,000	2,400	7,200	9,400	
24–29 years	7,800	3,200	2,700	7,400	5,700	
30–39 years	6,900	2,900	2,600	7,300	4,400	
40 years or older	5,800	2,700	3,000	7,100	3,800	
Dependency status						
Dependent	10,800	6,400	2,300	7,000	9,900	
Independent	7,100	3,100	2,700	7,200	4,900	
Unmarried, no dependents	7,900	3,400	2,600	7,600	5,300	
Married, no dependents	6,600	3,000	3,000	7,700	5,500	
Single parent	7,100	3,200	2,600	6,700	4,100	
Married parent	6,400	2,800	2,800	7,200	4,600	
Dependency and income in 2006						
Dependent						
Less than \$20,000	9,700	6,400	2,200	5,900	6,200	
\$20,000-39,999	10,600	6,500	2,300	6,400	7,200	
\$40,000-59,999	10,700	6,000	2,300	7,000	8,500	
\$60,000-79,999	10,700	6,100	2,400	7,100	8,900	
\$80,000-99,999	11,600	6,800	2,400	7,500	10,300	
\$100,000 or more	11,500	6,700	2,400	7,700	13,000	
Independent	·		•	•	•	
Less than \$10,000	8,400	4,000	2,400	7,000	5,000	
\$10,000-19,999	7,600	3,100	2,700	6,900	5,200	
\$20,000-29,999	7,200	3,100	2,900	7,200	5,000	
\$30,000-49,999	6,500	2,500	3,100	7,300	5,000	
\$50,000 or more	5,400	2,400	3,600	7,900	4,400	
Income group ⁵						
Lowest 25 percent	9,400	5,400	2,300	6,600	5,900	
Middle 50 percent	9,000	4,500	2,500	7,100	7,900	
Highest 25 percent	8,900	4,900	2,500	7,900	10,200	
	,	•	•	•	•	
Aid status No aid	1	_	_	_	_	
	†	1 000	†	7 100	†	
Received aid	9,100	4,900	2,400	7,100	8,100	
Grant status	0.400	1	2 200	7 200	7.000	
	*	7 4 000	,		7,900 8,300	
No grant Received grant	8,100 9,400	† 4,900	3,200 2,200	7,300 7,000		

Table 3.1-B. Average amount of aid received by undergraduates receiving a specific type of aid, by type of aid and selected institutional and student characteristics: 2007–08—Continued

Institutional and student characteristics	Any aid amount¹	Any grant amount	Any work-study amount	Any Ioan amount²	Any other type of aid
Loan status ²					
No loan	\$4,300	\$3,900	\$2,800	†	\$5,200
Received loan	12,500	5,700	2,200	7,100	9,500

[†] Not applicable.

NOTE: Average aid amounts are calculated only for students receiving a particular type of aid. Those not receiving a specific type of aid (i.e., zero values) are not included in the average for that aid. Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[‡] Reporting standards not met.

¹ Total aid excludes the federal education tax benefits but includes federal Parent PLUS loans, veterans benefits, and WIA job training funds. For students attending more than one institution, includes aid received at the NPSAS institution, but may not include aid received at other institutions.

 $^{^{\}rm 2}$ Loans only include loans to students. Excludes federal Parent PLUS loans.

³ Excludes students attending more than one institution.

⁴ Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

⁵ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

Table S3.1-B. Standard errors for table 3.1-B: Average amount of aid received by undergraduates receiving a specific type of aid, by type of aid and selected institutional and student characteristics: 2007–08

		Any	Any	Any	Any
Institutional and	Any aid	grant	work-study	loan	other
student characteristics	amount	amount	amount	amount	type of aid
Total	\$150	\$80	\$30	\$50	\$160
Institution type					
Public					
Less-than-2-year	400	190	†	370	520
2-year	100	60	120	120	140
4-year	60	50	40	60	170
Non-doctorate-granting	120	90	70	80	230
Doctorate-granting	90	60	50	80	220
Private nonprofit					
Less-than-4-year	970	910	240	740	880
4-year	260	170	40	130	350
Non-doctorate-granting	370	220	40	160	500
Doctorate-granting	370	260	50	220	540
For-profit	170	50	290	130	350
More than one institution	370	150	80	140	310
Attendance status					
Full-time/full-year	120	70	30	60	170
Full-time/part-year	220	120	100	140	360
Part-time/full-year	110	50	110	70	200
Part-time/part-year	70	40	160	80	240
Housing					
On campus	160	140	40	110	270
Off campus	120	60	50	60	180
Living with parents	130	70	90	80	200
Price of attendance					
Less than \$6,000	30	20	190	50	110
\$6,000–10,999	70	40	140	50	130
\$11,000–18,999	80	50	60	60	110
\$19,000 or more	150	130	40	90	210
Gender					
Male	160	90	50	90	230
Female	150	90	30	50	160
Race/ethnicity					
White	130	70	30	70	180
Black	180	120	70	90	260
Hispanic	270	140	100	100	320
Asian	350	250	100	150	580
American Indian	670	470	280	430	1,140
Pacific Islander	530	330	270	490	950
Two or more races	300	260	120	250	640
Other	680	530	310	420	†

Table S3.1-B. Standard errors for table 3.1-B: Average amount of aid received by undergraduates receiving a specific type of aid, by type of aid and selected institutional and student characteristics: 2007–08—Continued

		Any	Any	Any	Any
Institutional and student characteristics	Any aid amount	grant amount	work-study amount	loan amount	other type of aid
Age as of 12/31/07					
18 years or younger	\$250	\$150	\$60	\$120	\$400
19–23 years	140	100	30	60	160
24–29 years	190	70	100	110	230
30–39 years	150	60	140	140	210
40 years or older	130	70	190	130	240
Dependency status					
Dependent	150	110	30	60	160
Independent	140	60	70	70	150
Unmarried, no dependents	170	100	90	90	210
Married, no dependents	200	100	180	270	360
Single parent	180	70	120	90	250
Married parent	170	60	170	130	250
Dependency and income in 2006 Dependent					
Less than \$20,000	250	150	70	110	400
\$20,000–39,999	280	160	70	120	370
\$40,000-59.999	260	200	60	160	370
\$60,000-79,999	190	170	70	120	270
\$80,000–79,999	230	240	80	160	270
\$100,000 or more	160	130	70	120	290
Independent	100	130	70	120	290
Less than \$10,000	240	100	80	110	300
\$10,000–19,999	210	90	150	100	250
\$20,000–29,999	150	100	170	120	240
\$30,000-49,999	160	60	180	140	240
\$50,000 - 49,333 \$50,000 or more	110	70	340	210	270
Income group					
Lowest 25 percent	220	110	40	70	190
Middle 50 percent	150	90	40	50	180
Highest 25 percent	130	90	70	110	320
Aid status					
No aid	†	†	†	†	†
Received aid	150	80	30	50	160
Grant status					
No grant	120	†	80	100	190
Received grant	160	80	30	50	200
Loan status					
No loan	80	80	60	_†	170
Received loan	110	70	30	50	150

[†] Not applicable

Table 3.1-C. Percentage of undergraduates receiving any aid, by type of institution and selected student characteristics: 2007–08

		Public	: 4-year		vate fit 4-year		
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions ¹
Total	47.6	70.2	71.9	87.3	81.7	96.3	65.6
Attendance status							
Full-time/full-year	65.7	81.5	76.9	93.6	85.2	96.7	79.5
Full-time/part-year	53.9	68.9	68.6	83.6	73.8	95.1	71.0
Part-time/full-year	52.1	65.9	68.9	85.1	82.7	98.3	61.2
Part-time/part-year	34.4	47.6	53.7	66.5	69.1	96.5	44.3
Housing ²							
On campus	65.4	78.9	75.1	91.5	81.1	93.5	80.1
Off campus	49.8	69.5	70.7	83.1	80.7	96.7	65.5
Living with parents	43.5	64.4	70.5	88.8	88.0	95.1	56.1
Tuition and fees ²							
Less than \$1,000	38.8	39.8	43.4	48.4	‡	89.0	39.1
\$1,000-2,999	58.3	64.9	62.5	70.9	66.3	92.1	61.4
\$3,000-6,999	63.8	78.2	76.5	84.1	77.9	95.1	78.4
\$7,000 or more	44.9	77.7	72.2	92.1	83.5	97.7	85.0
Price of attendance ²							
Less than \$6,000	33.9	38.6	38.3	47.6	53.7	76.2	35.3
\$6,000-10,999	56.5	67.1	63.5	75.8	69.5	94.3	62.0
\$11,000–18,999	68.7	79.7	77.1	87.8	78.8	97.4	78.5
\$19,000 or more	77.6	84.5	76.2	93.0	84.5	97.3	86.4
Gender							
Male	42.6	66.8	69.4	85.8	78.1	96.5	61.4
Female	51.6	73.0	74.2	88.5	84.8	96.3	68.8
Race/ethnicity ³							
White	44.9	69.9	69.6	86.9	79.9	95.8	63.5
Black	59.2	81.4	85.0	87.2	91.6	98.1	76.2
Hispanic	51.9	65.9	78.3	92.9	90.8	95.4	69.0
Asian	35.0	55.1	63.2	78.3	71.2	94.9	53.4
American Indian	59.8	72.2	83.1	85.6	‡	98.3	70.8
Pacific Islander	47.3	72.4	69.0	79.3	75.7	98.5	61.5
Two or more races	48.2	74.3	80.2	92.2	84.4	96.4	68.1
Other	48.0	72.5	70.4	65.0	‡	96.1	64.1

Table 3.1-C. Percentage of undergraduates receiving any aid, by type of institution and selected student characteristics: 2007–08—Continued

		Public	: 4-year		vate iit 4-year		
			4-yeai		it 4-year		All
	Public	Non- doctorate-	Doctorate-	Non- doctorate-	Doctorate-		insti-
Student characteristics	2-year	granting	granting	granting	granting	For-profit	tutions ¹
Age as of 12/31/07							
18 years or younger	50.2	81.2	80.4	91.7	87.0	95.3	70.9
19–23 years	46.3	70.2	71.5	90.6	80.3	95.8	66.1
24–29 years	50.4	65.1	69.9	84.6	79.5	96.9	66.3
30-39 years	51.0	69.0	72.4	83.4	86.0	96.8	66.4
40 years or older	43.0	66.9	61.2	73.6	81.4	96.1	56.2
Dependency status							
Dependent	45.3	72.3	72.9	91.0	81.4	94.8	66.5
Independent	49.4	66.9	69.4	82.0	82.5	96.8	64.6
Unmarried, no dependents	45.9	64.6	68.7	79.9	82.1	96.5	62.5
Married, no dependents	40.1	65.1	65.1	80.0	82.0	94.9	56.4
Single parent	58.0	71.8	74.6	90.5	81.5	97.5	72.2
Married parent	48.6	66.8	69.4	78.3	84.4	96.7	63.0
Dependency and income in 2006							
Dependent							
Less than \$20,000	67.2	88.2	85.8	94.8	95.2	98.2	80.7
\$20,000-39,999	60.4	87.9	89.9	98.3	93.5	96.6	78.5
\$40,000–59,999	43.7	72.8	80.6	97.0	93.5	96.2	66.8
\$60,000-79,999	35.9	70.5	77.8	93.1	87.8	92.7	63.3
\$80,000-99,999	34.3	68.3	71.5	93.5	84.3	93.8	63.1
\$100,000 or more	27.6	57.7	59.3	82.7	70.1	82.7	56.4
Independent							
Less than \$10,000	60.4	73.3	73.4	87.7	82.8	96.8	74.0
\$10,000–19,999	57.2	70.9	74.2	86.9	81.4	98.1	71.5
\$20,000-29,999	54.9	68.8	72.6	83.0	81.9	97.4	68.6
\$30,000-49,999	47.8	64.4	67.8	84.4	87.8	97.1	62.9
\$50,000 or more	35.3	56.8	57.6	73.4	79.8	93.7	49.7
Income group ⁴							
Lowest 25 percent	62.6	81.9	83.2	93.0	90.2	97.0	77.3
Middle 50 percent	46.5	70.1	74.9	90.5	86.7	96.9	65.8
Highest 25 percent	33.0	56.2	58.3	78.5	72.1	91.9	53.1
Aid status							
No aid	†	100.0	100.0	100.0	100.0	100.0	100.0
Received aid	100.0	100.0	100.0	100.0	100.0	100.0	100.0

Table 3.1-C. Percentage of undergraduates receiving any aid, by type of institution and selected student characteristics: 2007–08—Continued

Student characteristics		Public 4-year		Private nonprofit 4-year			
	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions ¹
Grant status							
No grant	13.3	37.2	40.2	46.7	37.6	87.6	28.8
Received grant	100.0	100.0	100.0	100.0	100.0	100.0	100.0
Loan status ⁵							
No loan	39.7	47.3	46.3	67.4	58.0	56.8	44.1
Received loan	100.0	100.0	100.0	100.0	100.0	100.0	100.0

[†] Not applicable.

NOTE: Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Excludes federal education tax benefits. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[‡] Reporting standards not met.

¹ All institutions also include those attending public less-than-2-year, private not-for-profit less-than-4-year, and multiple institutions, in addition to the institutions listed separately in this table.

 $^{^{\}rm 2}\,{\rm Excludes}$ those attending more than one institution.

³ Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

⁴ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁵ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S3.1-C. Standard errors for table 3.1-C: Percentage of undergraduates receiving any aid, by type of institution and selected student characteristics: 2007–08

		Public 4-year			vate fit 4-year		
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions
Total	2.42	0.76	0.86	1.06	1.21	0.25	1.20
Attendance status							
Full-time/full-year	1.32	0.84	0.95	0.79	1.14	0.33	0.37
Full-time/part-year	3.62	1.58	1.26	2.60	2.71	0.54	1.52
Part-time/full-year	1.94	1.66	1.66	1.86	1.67	0.45	1.24
Part-time/part-year	2.93	1.59	1.39	3.34	3.30	0.63	2.51
Housing							
On campus	5.26	1.57	1.12	1.18	1.90	2.77	1.21
Off campus	2.41	0.86	0.97	1.58	1.29	0.33	1.53
Living with parents	2.49	1.26	1.32	1.80	1.73	0.60	1.66
Tuition and fees							
Less than \$1,000	2.82	2.57	3.40	7.01	†	3.87	2.62
\$1,000-2,999	1.72	1.40	1.34	3.31	4.91	1.59	1.48
\$3,000-6,999	2.05	1.26	1.11	2.14	3.09	0.41	1.03
\$7,000 or more	8.54	1.74	0.99	0.84	1.20	0.27	0.69
Price of attendance							
Less than \$6,000	2.79	1.92	2.06	4.42	4.63	3.75	2.40
\$6,000-10,999	2.05	1.77	1.57	2.93	3.86	0.61	1.65
\$11,000–18,999	1.65	0.73	1.18	2.33	2.80	0.34	1.03
\$19,000 or more	3.28	1.64	1.05	0.79	1.15	0.30	0.68
Gender							
Male	2.60	1.12	1.19	1.54	1.67	0.38	1.38
Female	2.26	0.81	0.70	1.00	1.06	0.36	1.04
Race/ethnicity							
White	2.46	0.82	0.92	1.34	1.54	0.44	1.32
Black	2.88	2.05	1.16	1.87	1.29	0.32	1.56
Hispanic	2.24	1.63	1.55	1.27	1.60	0.49	0.76
Asian	2.94	3.97	2.44	3.64	4.35	1.15	1.96
American Indian	5.47	4.71	5.06	7.93	†	1.57	2.92
Pacific Islander	5.37	8.87	6.09	5.11	13.75	1.11	3.28
Two or more races	3.03	3.74	2.52	2.30	4.19	1.32	1.51
Other	7.06	10.68	8.28	12.67	†	3.70	4.03

Table S3.1-C. Standard errors for table 3.1-C: Percentage of undergraduates receiving any aid, by type of institution and selected student characteristics: 2007–08—Continued

		Public	: 4-year		vate it 4-year		
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions
Age as of 12/31/07							
18 years or younger	2.28	1.77	1.25	2.70	1.74	0.86	0.99
19–23 years	2.60	0.86	0.94	0.96	1.47	0.34	1.08
24–29 years	2.82	1.97	1.39	1.86	3.36	0.37	1.60
30–39 years	3.03	2.23	1.84	2.80	2.11	0.37	1.82
40 years or older	1.93	1.95	2.99	3.42	2.29	0.65	1.48
Dependency status							
Dependent	2.58	0.87	0.95	1.04	1.40	0.45	1.06
Independent	2.37	1.09	1.02	1.82	1.31	0.26	1.42
Unmarried, no dependents	2.82	1.71	1.54	2.15	1.79	0.38	1.63
Married, no dependents	1.74	2.85	2.76	2.87	2.47	1.02	1.30
Single parent	3.49	2.38	1.88	1.29	2.35	0.33	2.16
Married parent	1.91	2.18	1.80	3.73	2.58	0.51	1.09
Dependency and income in 2006							
Dependent							
Less than \$20,000	2.24	1.88	1.43	1.57	1.36	0.44	1.13
\$20,000-39,999	2.45	1.57	0.97	0.52	1.39	0.71	1.07
\$40,000–59,999	3.35	1.90	1.77	1.19	0.97	0.99	1.69
\$60,000-79,999	3.15	1.67	1.15	1.29	2.72	1.62	1.27
\$80,000-99,999	2.30	1.97	1.18	1.84	2.74	2.26	0.98
\$100,000 or more	2.26	1.60	1.31	2.51	2.42	3.18	1.06
Independent							
Less than \$10,000	2.77	1.75	1.37	2.09	2.87	0.39	1.25
\$10,000-19,999	3.38	1.84	2.42	2.03	4.14	0.32	1.74
\$20,000-29,999	3.37	2.19	2.68	2.56	3.59	0.46	2.17
\$30,000-49,999	2.69	2.89	2.51	2.95	1.85	0.54	1.74
\$50,000 or more	1.26	2.45	2.41	3.06	2.27	0.98	1.07
Income group							
Lowest 25 percent	2.34	1.42	0.74	0.87	1.41	0.29	1.07
Middle 50 percent	2.91	0.91	1.00	0.80	1.20	0.31	1.45
Highest 25 percent	1.29	1.43	1.21	2.26	2.18	1.11	0.90
Aid status							
No aid	†	†	†	†	†	†	†
Received aid	†	†	†	†	†	†	†

Table S3.1-C. Standard errors for table 3.1-C: Percentage of undergraduates receiving any aid, by type of institution and selected student characteristics: 2007–08—Continued

		Public 4-year		Private nonprofit 4-year			
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions
Grant status							
No grant	1.27	0.91	1.33	3.14	2.46	0.73	1.13
Received grant	†	†	†	†	†	†	†
Loan status							
No loan	2.54	1.39	1.18	2.43	2.50	2.34	1.77
Received loan	†	†	†	†	†	†	†

[†] Not applicable.

Table 3.1-D. Average amount of total aid received by undergraduates who received any aid, by type of institution and selected student characteristics: 2007–08

		Public	4-year	Priv nonprof			
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions ¹
Total	\$3,400	\$8,000	\$10,100	\$16,000	\$19,000	\$10,800	\$9,100
Attendance status							
Full-time/full-year	5,400	9,900	11,500	20,000	22,400	13,100	12,700
Full-time/part-year	3,100	6,100	7,300	10,500	13,700	9,900	7,500
Part-time/full-year	3,400	6,500	8,800	10,900	12,400	11,400	6,200
Part-time/part-year	1,800	4,100	5,100	5,900	6,800	7,200	3,800
Housing ²							
On campus	5,500	10,500	11,800	21,300	22,800	22,500	16,200
Off campus	3,400	7,700	9,900	11,300	14,700	10,600	7,900
Living with parents	3,300	6,200	7,500	13,300	16,800	11,100	6,500
Tuition and fees ²							
Less than \$1,000	2,200	2,300	3,400	4,100	‡	7,300	2,200
\$1,000-2,999	4,200	5,300	6,100	4,800	5,100	5,800	4,700
\$3,000-6,999	5,900	9,000	9,700	7,200	8,400	8,500	8,700
\$7,000 or more	6,200	12,900	12,800	19,600	21,400	12,800	16,200
Price of attendance ²							
Less than \$6,000	1,500	2,200	2,700	2,400	2,900	3,300	1,700
\$6,000-10,999	3,500	4,800	5,300	5,100	5,300	5,800	4,200
\$11,000-18,999	5,900	9,000	9,400	8,500	8,400	9,400	8,500
\$19,000 or more	7,300	13,500	13,800	20,200	22,200	13,400	16,700
Gender							
Male	3,300	8,000	10,100	15,700	19,200	11,800	9,300
Female	3,400	8,100	10,100	16,200	18,800	10,400	9,000
Race/ethnicity ³							
White	3,500	8,200	9,600	16,900	19,000	10,900	9,400
Black	3,500	9,600	11,900	14,200	18,400	10,600	9,000
Hispanic	2,900	6,200	10,300	11,900	18,300	10,600	7,900
Asian	3,200	7,500	10,800	17,600	19,500	11,700	9,600
American Indian	3,500	6,900	9,400	17,700	‡	9,800	7,200
Pacific Islander	2,900	6,700	12,800	16,500	15,900	11,900	8,600
Two or more races	3,900	8,400	11,000	19,400	22,500	11,400	10,300
Other	2,900	‡	11,800	‡	‡	9,200	7,800
Age as of 12/31/07							
18 years or younger	3,600	8,400	10,500	19,800	21,500	12,100	10,600
19–23 years	3,500	8,600	10,400	19,100	21,200	11,100	10,400
24–29 years	3,700	7,700	9,600	10,800	13,800	10,900	7,800
30–39 years	3,200	6,600	8,900	9,200	10,900	10,400	6,900
40 years or older	2,800	5,500	7,000	8,500	8,400	10,300	5,800

Table 3.1-D. Average amount of total aid received by undergraduates who received any aid, by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year	Priv nonprof			
		Non-		Non-			All
	Public	doctorate-	Doctorate-	doctorate-	Doctorate-		insti-
Student characteristics	2-year	granting	granting	granting	granting	For-profit	tutions ¹
Dependency status							
Dependent	\$3,500	\$8,700	\$10,400	\$19,500	\$21,500	\$12,100	\$10,800
Independent	3,300	7,000	9,200	10,200	11,800	10,400	7,100
Unmarried, no dependents	3,400	7,700	9,800	12,200	14,900	10,800	7,900
Married, no dependents	2,900	5,600	8,200	9,500	9,900	10,800	6,600
Single parent	3,600	7,300	9,700	10,300	11,600	10,000	7,100
Married parent	3,100	6,300	8,000	8,500	9,000	10,600	6,400
Dependency and income in 2006							
Dependent							
Less than \$20,000	3,900	9,300	13,300	17,900	21,600	10,200	9,700
\$20,000–39,999	3,800	9,200	12,200	22,200	24,400	12,200	10,600
\$40,000–59,999	3,200	8,700	10,900	21,000	24,000	12,200	10,700
\$60,000-79,999	3,200	8,300	9,700	19,800	22,000	12,400	10,700
\$80,000–99,999	3,300	8,400	9,500	19,500	23,100	13,700	11,600
\$100,000 or more	3,100	8,100	9,000	18,000	18,800	15,700	11,500
Independent							
Less than \$10,000	4,000	8,800	11,400	13,900	15,600	10,200	8,400
\$10,000–19,999	3,800	8,000	9,700	11,000	14,600	10,200	7,600
\$20,000-29,999	3,600	6,400	8,900	10,300	13,200	10,600	7,200
\$30,000-49,999	2,800	6,400	7,600	9,300	10,300	10,400	6,500
\$50,000 or more	2,300	4,400	6,400	7,500	8,100	11,000	5,400
Income group ⁴							
Lowest 25 percent	3,900	9,100	12,200	17,500	20,800	10,500	9,400
Middle 50 percent	3,300	7,900	9,700	16,400	20,400	10,800	9,000
Highest 25 percent	2,500	6,700	8,600	14,000	15,900	11,800	8,900
Aid status							
No aid	†	†	†	†	†	†	†
Received aid	3,400	8,000	10,100	16,000	19,000	10,800	9,100
Grant status							
No grant	4,100	7,500	9,100	9,600	13,000	10,500	8,100
Received grant	3,300	8,200	10,400	16,900	19,900	10,900	9,400

Table 3.1-D. Average amount of total aid received by undergraduates who received any aid, by type of institution and selected student characteristics: 2007–08—Continued

		Public 4-year		Private nonprofit 4-year			
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions ¹
Loan status ⁵							
No loan	\$2,400	\$4,500	\$6,300	\$8,400	\$10,500	\$4,000	\$4,300
Received loan	5,900	10,200	12,000	19,300	22,800	11,200	12,500

[†] Not applicable.

NOTE: Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Excludes federal education tax benefits. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[‡] Reporting standards not met.

¹ All institutions also include those attending public less-than-2-year, private not-for-profit less-than-4-year, and multiple institutions, in addition to the institutions listed separately in this table.

² Excludes those attending more than one institution.

³ Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

⁴ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁵ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S3.1-D. Standard errors for table 3.1-D: Average amount of total aid received by undergraduates who received any aid, by type of institution and selected student characteristics: 2007–08

		Public	Public 4-year		rate it 4-year		
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions
Total	\$100	\$120	\$90	\$370	\$370	\$170	\$150
Attendance status							
Full-time/full-year	180	140	110	450	470	250	120
Full-time/part-year	140	260	180	400	1,420	350	220
Part-time/full-year	130	170	180	490	480	210	110
Part-time/part-year	50	220	170	320	350	250	70
Housing							
On campus	330	250	130	440	440	1,520	160
Off campus	110	160	120	280	410	180	120
Living with parents	90	130	150	450	1,200	290	130
Tuition and fees							
Less than \$1,000	60	180	380	480	†	1,710	60
\$1,000-2,999	140	90	170	220	420	320	90
\$3,000-6,999	180	130	110	300	470	270	70
\$7,000 or more	870	410	180	380	360	230	140
Price of attendance							
Less than \$6,000	30	120	120	150	220	180	30
\$6,000–10,999	80	110	90	230	290	130	70
\$11,000–18,999	230	100	80	180	290	220	80
\$19,000 or more	300	310	200	410	350	280	150
Gender							
Male	80	150	110	430	530	330	160
Female	120	130	130	450	420	190	150
Race/ethnicity							
White	80	120	110	450	400	240	130
Black	170	360	200	610	600	260	180
Hispanic	130	240	270	710	1,990	260	270
Asian	140	350	330	1,650	740	890	350
American Indian	290	790	720	3,850	†	2,110	670
Pacific Islander	290	1,120	1,370	3,540	1,150	2,020	530
Two or more races	260	650	430	990	1,320	750	300
Other	500	†	1,020	†	†	960	680
Age as of 12/31/07							
18 years or younger	130	300	230	620	590	580	250
19–23 years	100	140	100	420	460	280	140
24–29 years	140	180	180	490	520	290	190
30-39 years	130	200	310	360	620	300	150
40 years or older	100	260	450	330	570	340	130

Table S3.1-D. Standard errors for table 3.1-D: Average amount of total aid received by undergraduates who received any aid, by type of institution and selected student characteristics: 2007–08

—Continued

		Public	4-year	Priv nonprof	/ate it 4-year		
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions
Dependency status							
Dependent	\$90	\$150	\$90	\$410	\$440	\$370	\$150
Independent	110	140	170	270	330	170	140
Unmarried, no dependents	120	190	170	600	600	250	170
Married, no dependents	110	260	270	480	580	770	200
Single parent	170	290	330	400	750	220	180
Married parent	130	250	420	360	420	300	170
Dependency and income in 2006 Dependent							
Less than \$20,000	180	310	290	630	1680	390	250
\$20,000-39,999	150	280	180	890	1,030	470	280
\$40,000-59,999	100	270	210	700	870	670	260
\$60,000-79,999	110	280	220	750	730	1,120	190
\$80,000-99,999	160	240	240	800	820	910	230
\$100,000 or more	140	270	160	490	550	1,030	160
Independent							
Less than \$10,000	230	290	240	800	990	270	240
\$10,000–19,999	170	240	270	650	790	270	210
\$20,000–29,999	170	340	310	470	840	300	150
\$30,000-49,999	110	250	320	360	400	270	160
\$50,000 or more	80	260	370	370	540	390	110
Income group							
Lowest 25 percent	180	200	140	610	1,140	250	220
Middle 50 percent	100	130	120	390	440	170	150
Highest 25 percent	80	230	190	510	490	380	130
Aid status							
No aid	†	†	†	†	†	†	†
Received aid	100	120	90	370	370	170	150
Grant status							
No grant	90	150	130	440	460	280	120
Received grant	110	130	110	390	410	190	160
Loan status							
No loan	40	120	120	270	410	180	80
Received loan	210	140	100	450	410	180	110

[†] Not applicable.

Table 3.2-A. Percentage of undergraduates receiving federal aid, by type of aid and selected institutional and student characteristics: 2007–08

Institutional and	Any federal	Federal	Federal	Federal	Federal Parent
student characteristics	aid ¹	grants	work-study	loans ²	PLUS loans
Total	47.0	27.6	5.6	34.7	3.8
Institution type					
Public					
Less-than-2-year	38.5	33.9	#	14.6	‡
2-year	27.4	21.1	2.6	10.2	0.2
4-year	51.2	25.6	5.7	41.5	5.9
Non-doctorate-granting	52.2	29.8	5.4	39.1	3.9
Doctorate-granting	50.6	23.4	5.9	42.9	6.9
Private nonprofit					
Less-than-4-year	72.0	49.6	2.4	40.2	4.8 !
4-year	62.5	26.2	17.8	54.3	8.6
Non-doctorate-granting	65.8	29.9	17.3	57.4	8.4
Doctorate-granting	58.8	22.1	18.3	51.0	8.7
For-profit	93.0	62.8	1.7	87.9	5.1
More than one institution	51.9	28.0	5.9	40.5	5.0
Attendance status					
Full-time/full-year	62.2	33.0	10.3	49.1	7.2
Full-time/part-year	53.4	33.7	4.0	41.1	3.5
Part-time/full-year	42.1	28.3	3.1	26.8	1.4
Part-time/part-year	23.3	14.7	1.5	15.1	0.5
Housing ³					
On campus	60.2	23.5	17.1	52.9	12.0
Off campus	46.7	29.9	3.4	34.6	2.0
Living with parents	38.1	24.5	3.8	22.0	2.6
Price of attendance ³					
Less than \$6,000	13.2	11.2	0.5	3.7	0.1 !
\$6,000–10,999	40.1	29.8	2.4	21.7	0.6
\$11,000–18,999	62.8	37.6	5.6	48.6	4.4
\$19,000 or more	70.9	31.5	14.5	63.7	10.0
Gender					
Male	41.1	21.7	5.5	30.8	3.9
Female	51.4	32.0	5.8	37.6	3.7
Race/ethnicity ⁴					
White	43.1	20.7	5.4	34.3	4.3
Black	62.2	46.4	5.8	46.0	2.8
Hispanic	52.4	39.5	5.7	30.2	2.9
Asian	36.1	22.7	6.5	22.3	2.7
American Indian	53.5	39.3	3.9	31.9	1.5
Pacific Islander	40.8	26.6	4.6	27.7	5.0
Two or more races	51.2	31.2	9.0	38.4	3.5
Other	46.8	33.2	3.9!	32.4	2.6 !

Table 3.2-A. Percentage of undergraduates receiving federal aid, by type of aid and selected institutional and student characteristics: 2007–08—Continued

In attitution of an a	Any federal	Fadanal	Fodovol	Federal	Federal
Institutional and student characteristics	aid ¹	Federal grants	Federal work-study	loans ²	Parent PLUS loans
Age as of 12/31/07					
18 years or younger	51.7	27.4	9.4	36.2	8.2
19–23 years	48.1	24.6	7.6	35.8	6.0
24–29 years	50.0	36.5	2.9	37.4	†
30–39 years	46.8	32.6	2.0	34.6	†
40 years or older	32.3	21.1	1.5	23.3	†
Dependency status					
Dependent	47.8	22.1	8.4	36.0	7.1
Independent	46.0	33.7	2.6	33.2	†
Unmarried, no dependents	45.0	29.8	3.2	34.3	†
Married, no dependents	31.3	12.7	2.6	24.8	†
Single parent	58.6	54.0	2.7	38.5	†
Married parent	40.4	26.6	1.5	29.9	†
Dependency and income in 2006					
Dependent					
Less than \$20,000	71.1	68.2	10.1	36.5	3.2
\$20,000–39,999	68.3	59.3	12.2	41.4	5.3
\$40,000–59,999	48.9	23.7	10.1	38.7	6.3
\$60,000-79,999	41.1	1.7	8.3	38.1	8.4
\$80,000–99,999	39.8	0.5	7.3	36.8	9.5
\$100,000 or more	32.6	0.6	4.9	29.4	8.6
Independent					
Less than \$10,000	62.1	58.3	5.3	40.6	†
\$10,000–19,999	60.9	52.3	3.0	42.0	†
\$20,000–29,999	52.1	34.9	2.3	36.9	†
\$30,000-49,999	42.2	28.4	1.4	31.8	†
\$50,000 or more	19.0	0.3 !	8.0	18.4	†
Income group ⁵					
Lowest 25 percent	66.6	61.8	8.3	40.0	2.2
Middle 50 percent	47.1	23.3	5.6	37.1	4.2
Highest 25 percent	26.2	0.6	2.9	24.1	4.6
Aid status					
No aid	†	†	†	†	†
Received aid	71.6	42.0	8.6	52.8	5.8
Grant status					
No grant	21.3	†	2.1	19.7	2.9
Received grant	70.9	53.3	9.0	48.7	4.6

Table 3.2-A. Percentage of undergraduates receiving federal aid, by type of aid and selected institutional and student characteristics: 2007–08—Continued

Institutional and student characteristics	Any federal aid ¹	Federal grants	Federal work-study	Federal loans ²	Federal Parent PLUS Ioans
Loan status ⁶					
No loan	18.9	16.8	2.9	†	0.4
Received loan	91.9	44.8	10.0	90.1	9.3

[†] Not applicable.

NOTE: Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Federal aid excludes veterans benefits and education tax benefits. For students attending more than one institution, includes federal aid received at any institution. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[#] Rounds to zero.

[!] Interpret data with caution (estimates are unstable).

[‡] Reporting standards not met.

¹ Includes all federal grants, loans, college work-study awards, and federal Parent PLUS loans. Excludes veterans benefits and education tax credit and tax deduction benefits.

² Federal loans includes Stafford and Perkins loans and other loans to students. Excludes federal Parent PLUS loans.

³ Excludes students attending more than one institution.

⁴ Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

⁵ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁶ Loan status refers to loans that were made only to students, regardless of whether federal Parent PLUS loans were also received.

Table S3.2-A. Standard errors for table 3.2-A: Percentage of undergraduates receiving federal aid, by type of aid and selected institutional and student characteristics: 2007–08

Institutional and student characteristics	Any federal aid	Federal grants	Federal work-study	Federal Ioans	Federal Parent PLUS Ioans
Total	0.78	0.47	0.16	0.16	0.09
Institution type					
Public					
Less-than-2-year	1.47	1.35	†	1.36	†
2-year	1.64	1.18	0.18	0.34	0.03
4-year	0.75	0.42	0.27	0.28	0.22
Non-doctorate-granting	0.75	0.81	0.43	0.74	0.27
Doctorate-granting	0.93	0.35	0.29	0.66	0.31
Private nonprofit					
Less-than-4-year	3.04	2.62	0.47	0.78	2.12
4-year	0.52	0.31	0.66	0.32	0.42
Non-doctorate-granting	0.98	1.06	1.02	0.95	0.63
Doctorate-granting	1.07	1.08	0.79	1.08	0.51
For-profit	0.38	0.74	0.27	0.18	0.39
More than one institution	3.12	2.18	0.41	2.79	0.33
Attendance status					
Full-time/full-year	0.33	0.38	0.24	0.44	0.18
Full-time/part-year	1.23	0.99	0.30	0.74	0.30
Part-time/full-year	0.80	0.60	0.16	0.39	0.12
Part-time/part-year	2.02	1.36	0.22	1.00	0.07
Housing					
On campus	1.00	0.57	0.76	0.71	0.43
Off campus	1.22	0.77	0.16	0.47	0.08
Living with parents	1.10	0.95	0.20	0.40	0.14
Price of attendance					
Less than \$6,000	1.61	1.46	0.09	0.37	0.03
\$6,000–10,999	1.20	0.88	0.22	0.48	0.06
\$11,000–18,999	0.88	0.59	0.24	0.51	0.17
\$19,000 or more	0.69	0.53	0.50	0.56	0.31
Gender					
Male	0.87	0.59	0.24	0.36	0.14
Female	0.70	0.37	0.15	0.24	0.11
Race/ethnicity					
White	0.92	0.39	0.20	0.50	0.15
Black	1.58	1.19	0.29	1.01	0.13
Hispanic	0.74	0.63	0.27	1.16	0.24
Asian	1.16	0.96	0.46	0.90	0.24
American Indian	3.19	3.51	0.98	3.49	0.43
Pacific Islander	3.31	3.02	1.08	2.82	1.23
Two or more races	1.46	1.31	0.96	1.59	0.40
Other	3.77	3.10	1.19	3.88	0.96

Table S3.2-A. Standard errors for table 3.2-A: Percentage of undergraduates receiving federal aid, by type of aid and selected institutional and student characteristics: 2007–08—Continued

	Any				Federal
Institutional and	federal	Federal	Federal	Federal	Parent
student characteristics	aid	grants	work-study	loans	PLUS loans
Age as of 12/31/07					
18 years or younger	0.74	0.85	0.40	0.83	0.47
19–23 years	0.73	0.37	0.22	0.29	0.16
24–29 years	0.97	0.72	0.19	0.49	†
30–39 years	1.41	0.96	0.24	0.68	†
40 years or older	1.37	0.81	0.16	0.86	†
Dependency status					
Dependent	0.70	0.37	0.23	0.29	0.17
Independent	0.95	0.63	0.13	0.28	†
Unmarried, no dependents	0.98	0.65	0.19	0.61	†
Married, no dependents	1.35	0.62	0.28	1.18	†
Single parent	1.80	1.52	0.21	0.82	†
Married parent	0.81	0.67	0.19	0.67	†
Dependency and income in 2006					
Dependent					
Less than \$20,000	0.81	0.84	0.53	1.00	0.23
\$20,000-39,999	0.78	0.76	0.45	0.91	0.33
\$40,000-59,999	1.28	0.94	0.47	0.70	0.26
\$60,000–79,999	1.00	0.18	0.63	0.90	0.45
\$80,000-99,999	0.78	0.14	0.49	0.74	0.46
\$100,000 or more	0.79	0.08	0.21	0.73	0.35
Independent					
Less than \$10,000	0.68	0.59	0.28	0.93	†
\$10,000–19,999	1.11	1.09	0.26	0.77	†
\$20,000–29,999	1.78	1.31	0.27	0.93	†
\$30,000–49,999	1.44	1.08	0.19	0.80	†
\$50,000 or more	0.81	0.09	0.17	0.75	†
Income group					
Lowest 25 percent	0.51	0.45	0.23	0.68	0.11
Middle 50 percent	1.03	0.65	0.23	0.38	0.13
Highest 25 percent	0.63	0.07	0.15	0.58	0.21
Aid status					
No aid	†	†	†	†	†
Received aid	0.25	0.20	0.17	0.73	0.17
Grant status					
No grant	0.88	†	0.19	0.73	0.12
Received grant	0.34	0.27	0.21	0.89	0.17

Table S3.2-A. Standard errors for table 3.2-A: Percentage of undergraduates receiving federal aid, by type of aid and selected institutional and student characteristics: 2007–08—Continued

Institutional and student characteristics	Any federal aid	Federal grants	Federal work-study	Federal Ioans	Federal Parent PLUS Ioans
Loan status					
No loan	1.03	0.90	0.18	†	0.03
Received loan	0.18	0.46	0.24	0.21	0.24

[†] Not applicable.

Table 3.2-B. Average amount of federal aid received by undergraduates who received a specific type of federal aid, by type of aid and selected institutional and student characteristics: 2007–08

	Any				Federal
Institutional and	federal	Federal	Federal	Federal	Parent
student characteristics	aid ¹	grants	work-study	loans ²	PLUS loans
Total	\$6,600	\$2,800	\$2,300	\$5,100	\$10,800
Institution type					
Public					
Less-than-2-year	4,200	2,600	‡	4,900	‡
2-year	3,500	2,300	3,200	3,700	4,800
4-year	7,100	3,200	2,400	5,100	9,600
Non-doctorate-granting	6,400	3,100	2,300	5,100	8,000
Doctorate-granting	7,500	3,300	2,400	5,100	10,000
Private nonprofit					
Less-than-4-year	5,400	3,100	2,100	4,600	8,200
4-year	8,700	3,500	1,900	5,500	14,100
Non-doctorate-granting	8,300	3,300	1,800	5,300	12,700
Doctorate-granting	9,200	3,700	2,100	5,600	15,600
For-profit	7,500	2,600	3,500	5,500	9,000
More than one institution	6,600	2,700	2,100	5,000	9,900
Attendance status					
Full-time/full-year	7,900	3,700	2,200	5,400	11,400
Full-time/part-year	5,600	2,300	2,200	4,400	8,500
Part-time/full-year	5,500	2,300	2,900	5,400	9,400
Part-time/part-year	3,700	1,300	3,500	3,900	7,600
Housing ³					
On campus	8,600	3,800	1,900	4,800	12,100
Off campus	6,300	2,600	2,700	5,400	10,800
Living with parents	5,200	2,800	3,000	4,400	7,500
Price of attendance ³					
Less than \$6,000	1,900	1,300	1,700	2,400	‡
\$6,000–10,999	3,900	2,200	2,600	3,800	3,500
\$11,000–18,999	6,500	3,100	2,600	5,100	7,300
\$19,000 or more	9,100	3,600	2,300	5,800	13,100
Gender					
Male	6,700	2,900	2,500	5,000	11,200
Female	6,500	2,800	2,200	5,100	10,400
Race/ethnicity ⁴					
White	6,700	2,700	2,300	5,000	11,000
Black	6,700	2,800	2,300	5,400	9,900
Hispanic	6,000	2,900	2,600	5,100	10,200
Asian	6,400	3,300	2,500	5,000	11,000
American Indian	5,600	3,200	2,000	4,700	‡
Pacific Islander	6,800	2,600	2,600	5,200	10,500
Two or more races	7,000	3,000	2,600	5,400	10,500
Other	5,900	2,700	‡	5,000	‡

Table 3.2-B. Average amount of federal aid received by undergraduates who received a specific type of federal aid, by type of aid and selected institutional and student characteristics: 2007–08

—Continued

	Any				Federal
Institutional and	federal	Federal	Federal	Federal	Parent
student characteristics	aid ¹	grants	work-study	loans ²	PLUS Ioans
Age as of 12/31/07					
18 years or younger	\$6,700	\$3,300	\$2,200	\$4,000	\$11,100
19–23 years	6,800	3,000	2,300	4,800	10,700
24–29 years	6,400	2,500	2,700	5,900	†
30-39 years	6,100	2,400	2,700	5,800	†
40 years or older	5,800	2,400	3,000	5,800	†
Dependency status					
Dependent	6,900	3,200	2,200	4,500	10,800
Independent	6,200	2,500	2,700	5,800	†
Unmarried, no dependents	6,600	2,600	2,600	6,100	†
Married, no dependents	6,100	2,500	3,200	6,100	†
Single parent	6,100	2,600	2,500	5,500	†
Married parent	5,900	2,300	2,800	5,800	†
Dependency and income in 2006					
Dependent					
Less than \$20,000	6,600	3,800	2,200	4,600	7,400
\$20,000–39,999	6,700	3,200	2,200	4,700	8,100
\$40,000–59,999	6,300	2,100	2,100	4,600	9,200
\$60,000-79,999	6,600	1,400	2,300	4,400	9,700
\$80,000-99,999	7,100	‡	2,400	4,400	10,700
\$100,000 or more	7,900	2,700	2,300	4,300	13,500
Independent					
Less than \$10,000	6,800	3,000	2,300	5,700	†
\$10,000–19,999	6,000	2,300	2,700	5,700	†
\$20,000-29,999	6,000	2,600	3,100	5,800	†
\$30,000-49,999	5,800	1,800	3,100	6,000	†
\$50,000 or more	6,000	‡	4,000	6,100	†
Income group ⁵					
Lowest 25 percent	6,700	3,300	2,200	5,200	7,600
Middle 50 percent	6,300	2,200	2,400	5,100	10,100
Highest 25 percent	7,300	1,800	2,600	5,000	13,600
Aid status					
No aid	†	†	†	†	†
Received aid	6,600	2,800	2,300	5,100	10,800

Table 3.2-B. Average amount of federal aid received by undergraduates who received a specific type of federal aid, by type of aid and selected institutional and student characteristics: 2007–08

—Continued

Institutional and student characteristics	Any federal aid ¹	Federal grants	Federal work-study	Federal Ioans ²	Federal Parent PLUS Ioans	
Grant status						
No grant	\$6,400	†	\$3,400	\$4,900	\$11,400	
Received grant	6,600	2,800	2,100	5,200	10,400	
Loan status ⁶						
No loan	3,000	2,600	3,000	†	13,200	
Received loan	7,700	2,900	2,100	5,100	10,600	

[†] Not applicable.

NOTE: Average aid amounts are calculated only for students receiving a particular type of aid. Those not receiving a specific type of aid (i.e., zero values) are not included in the average for that aid. Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Federal aid excludes veterans benefits and education tax benefits. For students attending more than one institution, includes federal aid received at any institution. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[‡] Reporting standards not met.

¹ Includes all federal grants, loans, college work-study awards, and federal Parent PLUS loans. Excludes veterans benefits and education tax credit and tax deduction benefits.

² Total federal loans includes Stafford and Perkins loans and other federal loans to students. Excludes federal Parent PLUS loans.

³ Excludes students attending more than one institution.

⁴ Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

⁵ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁶ Loan status refers to loans that were made only to students, regardless of whether federal Parent PLUS loans were also received.

Table S3.2-B. Standard errors for table 3.2-B: Average amount of federal aid received by undergraduates who received a specific type of federal aid, by type of aid and selected institutional and student characteristics: 2007–08

Institutional and student characteristics	Any federal aid	Federal	Federal	Federal Ioans	Federal Parent
Total	\$110	grants \$50	work-study \$30	\$20	PLUS Ioans \$160
Institution type	•	, , ,	,	•	,
Public					
Less-than-2-year	250	90	†	230	t
2-year	160	80	100	150	420
4-year	60	30	40	60	200
Non-doctorate-granting	90	40	90	70	280
Doctorate-granting	70	40	50	60	260
Private nonprofit	10		00	00	200
Less-than-4-year	500	540	320	260	1,680
4-year	140	50	40	50	380
Non-doctorate-granting	210	60	50	70	530
Doctorate-granting	200	60	60	80	620
For-profit	110	30	260	70	440
More than one institution	230	90	80	170	320
Attendance status					
Full-time/full-year	70	30	30	20	190
Full-time/part-year	130	40	110	70	390
Part-time/full-year	110	30	100	70	350
Part-time/part-year	80	20	180	50	920
Housing					
On campus	100	40	40	40	290
Off campus	110	40	60	30	260
Living with parents	100	50	90	50	260
Price of attendance					
Less than \$6,000	40	30	160	50	†
\$6,000–10,999	70	30	130	40	280
\$11,000–18,999	60	40	70	30	150
\$19,000 or more	80	30	40	30	240
Gender					
Male	120	60	50	40	250
Female	100	50	40	30	180
Race/ethnicity					
White	100	50	40	30	200
Black	140	60	70	50	380
Hispanic	150	60	100	50	380
Asian	190	70	120	90	690
American Indian	460	300	330	330	†
Pacific Islander	390	170	310	280	1,340
Two or more races	190	110	140	140	830
Other	360	150	†	290	

Table S3.2-B. Standard errors for table 3.2-B: Average amount of federal aid received by undergraduates who received a specific type of federal aid, by type of aid and selected institutional and student characteristics: 2007–08—Continued

Institutional and student characteristics	Any federal aid	Federal grants	Federal work-study	Federal Ioans	Federal Parent PLUS Ioans
	alu	grants	work-study	Ioans	1 LOS IOAIIS
Age as of 12/31/07	# 400	#70	#70	# 40	#070
18 years or younger	\$160	\$70	\$70	\$40	\$370
19–23 years	80	50	40	20	180
24–29 years	140	50	130	80	†
30–39 years	140	50	150	80	†
40 years or older	220	70	220	140	†
Dependency status					
Dependent	90	50	30	20	160
Independent	140	50	70	60	†
Unmarried, no dependents	130	60	110	70	†
Married, no dependents	190	90	210	160	†
Single parent	180	50	120	90	†
Married parent	150	60	190	80	†
Dependency and income in 2006					
Dependent					
Less than \$20,000	140	60	80	60	500
\$20,000-39,999	130	50	80	40	420
\$40,000–59,999	120	50	60	40	390
\$60,000-79,999	130	190	90	50	280
\$80,000-99,999	140	†	90	50	280
\$100,000 or more	160	550	90	30	280
Independent					
Less than \$10,000	160	50	80	60	†
\$10,000-19,999	140	50	170	80	†
\$20,000-29,999	160	70	200	100	†
\$30,000-49,999	180	50	200	100	†
\$50,000 or more	120	†	410	110	†
Income group					
Lowest 25 percent	120	50	50	40	330
Middle 50 percent	110	40	40	30	180
Highest 25 percent	120	340	100	40	300
Aid status					
No aid	†	†	†	†	†
Received aid	110	50	30	20	160

Table S3.2-B. Standard errors for table 3.2-B: Average amount of federal aid received by undergraduates who received a specific type of federal aid, by type of aid and selected institutional and student characteristics: 2007–08—Continued

Institutional and student characteristics	Any federal aid	Federal grants	Federal work-study	Federal Ioans	Federal Parent PLUS Ioans	
Grant status						
No grant	\$80	†	\$90	\$40	\$260	
Received grant	120	50	30	30	210	
Loan status						
No loan	60	60	60	†	720	
Received loan	80	40	30	20	160	

[†] Not applicable.

Table 3.2-C. Percentage of undergraduates receiving federal aid, by type of institution and selected student characteristics: 2007–08

		Public 4-year		Priv nonprof			
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions ¹
Total	27.4	52.2	50.6	65.8	58.8	93.0	47.0
Attendance status							
Full-time/full-year	47.7	65.1	56.7	74.0	65.2	93.1	62.2
Full-time/part-year	33.5	52.9	42.6	61.5	46.1	91.5	53.4
Part-time/full-year	31.2	48.2	50.7	65.7	57.3	96.5	42.1
Part-time/part-year	13.6	24.4	27.6	35.1	35.5	93.1	23.3
Housing ²							
On campus	48.2	62.9	54.3	70.5	60.2	91.6	60.2
Off campus	28.4	51.0	50.0	60.3	53.7	93.6	46.7
Living with parents	25.1	45.9	45.7	70.1	69.6	90.7	38.1
Price of attendance ²							
Less than \$6,000	12.8	15.0	11.5	8.1	11.7	56.4	13.2
\$6,000–10,999	34.9	45.1	35.4	46.6	32.2	90.1	40.1
\$11,000–18,999	53.2	64.7	57.4	71.8	52.8	95.1	62.8
\$19,000 or more	69.3	71.3	56.8	73.1	64.2	94.3	70.9
Gender							
Male	21.2	47.4	47.5	60.0	54.3	92.8	41.1
Female	32.2	56.2	53.4	70.0	62.5	93.1	51.4
Race/ethnicity ³							
White	24.4	50.2	45.9	62.8	54.5	91.8	43.1
Black	40.2	67.7	72.3	72.2	74.6	95.9	62.2
Hispanic	29.8	50.9	61.0	81.5	75.4	92.1	52.4
Asian	18.4	40.0	45.3	45.8	49.0	92.1	36.1
American Indian	38.3	55.9	61.3	60.9	‡	97.0	53.5
Pacific Islander	21.1	51.4	51.5	48.2 !		95.8	40.8
Two or more races	30.1	52.6	59.2	76.6	66.7	93.4	51.2
Other	25.3	48.8	63.5	50.0	‡	92.9	46.8
Age as of 12/31/07							
18 years or younger	32.1	62.2	56.1	72.9	64.7	91.8	51.7
19–23 years	28.4	53.9	50.3	68.9	59.2	92.4	48.1
24–29 years	30.0	50.0	53.0	64.9	62.6	94.1	50.0
30–39 years	27.7	46.7	49.8	62.6	55.0	93.7	46.8
40 years or older	18.0	36.4	32.5	45.6	39.9	91.2	32.3
Dependency status							
Dependent	27.4	55.7	51.0	69.5	60.0	90.7	47.8
Independent	27.3	46.9	49.7	60.2	55.2	93.7	46.0
Unmarried, no dependents	24.2	47.6	52.2	62.9	59.9	93.3	45.0
Married, no dependents	13.6	35.7	39.7	49.2	44.7	88.7	31.3
Single parent	40.3	58.2	56.9	76.6	59.9	95.6	58.6
Married parent	23.6	41.0	44.1	49.6	50.5	92.5	40.4

Table 3.2-C. Percentage of undergraduates receiving federal aid, by type of institution and selected student characteristics: 2007–08—Continued

		Public	Public 4-year		Private nonprofit 4-year		
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions ¹
Dependency and income in 2006							
Dependent							
Less than \$20,000	53.1	81.7	79.0	88.8	85.1	96.5	71.1
\$20,000-39,999	46.4	80.4	80.7	90.9	87.5	94.0	68.3
\$40,000-59,999	23.8	56.5	62.6	76.4	75.3	91.3	48.9
\$60,000-79,999	14.8	47.8	51.2	70.7	64.1	87.5	41.1
\$80,000-99,999	12.2	48.6	43.5	64.8	62.9	88.0	39.8
\$100,000 or more	10.0	35.6	32.1	54.4	42.9	72.9	32.6
Independent							
Less than \$10,000	43.8	61.8	62.0	78.7	65.3	94.5	62.1
\$10,000-19,999	42.7	61.3	63.2	76.6	75.4	96.4	60.9
\$20,000-29,999	33.8	51.0	52.0	70.3	67.7	94.9	52.1
\$30,000-49,999	24.0	43.9	40.8	61.5	56.9	93.7	42.2
\$50,000 or more	5.9	16.8	23.7	33.6	33.1	85.7	19.0
Income group ⁴							
Lowest 25 percent	47.5	73.2	74.0	85.6	79.2	94.8	66.6
Middle 50 percent	26.1	51.5	52.0	69.8	66.9	93.9	47.1
Highest 25 percent	7.2	28.1	29.8	46.0	39.9	83.8	26.2
Aid status							
No aid	†	†	†	†	†	†	†
Received aid	57.4	74.4	70.3	75.3	71.9	96.5	71.6
Grant status							
No grant	7.3	28.2	31.0	35.0	27.4	80.5	21.3
Received grant	58.0	74.0	67.9	75.4	71.8	98.2	70.9
Loan status ⁵							
No loan	19.2	22.1	13.8	19.8	15.4	41.7	18.9
Received loan	81.5	91.5	90.9	95.0	92.2	97.7	91.9

[†] Not applicable.

NOTE: Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Federal aid includes Parent PLUS loans and excludes veterans benefits and the education tax credit and tax deduction benefits. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[!] Interpret data with caution (estimates are unstable).

[‡] Reporting standards not met.

¹ All institutions also include those attending public less-than-2-year, private not-for-profit less-than-4-year, and multiple institutions, in addition to the institutions listed separately in this table.

² Excludes those attending more than one institution.

³ Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

⁴ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁵ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S3.2-C. Standard errors for table 3.2-C: Percentage of undergraduates receiving federal aid, by type of institution and selected student characteristics: 2007–08

		Public	: 4-year		/ate it 4-year		
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions
Total	1.64	0.75	0.93	0.98	1.07	0.38	0.78
Attendance status							
Full-time/full-year	0.82	1.05	1.01	1.17	1.25	0.68	0.33
Full-time/part-year	3.12	1.49	1.48	2.97	3.53	0.94	1.23
Part-time/full-year	1.19	1.34	1.80	2.40	2.55	0.83	0.80
Part-time/part-year	2.20	1.56	1.34	2.46	2.49	1.14	2.02
Housing							
On campus	4.12	1.58	1.22	1.60	1.52	2.87	1.00
Off campus	1.73	0.81	1.07	1.62	1.71	0.48	1.22
Living with parents	1.60	1.41	1.43	2.13	2.89	0.87	1.10
Price of attendance							
Less than \$6,000	1.84	1.57	1.15	1.87	3.20	5.41	1.61
\$6,000–10,999	1.50	2.03	1.50	4.05	3.12	0.93	1.20
\$11,000–18,999	1.16	0.86	1.13	2.65	4.70	0.53	0.88
\$19,000 or more	3.80	2.21	1.37	1.13	1.10	0.53	0.69
Gender							
Male	1.65	0.88	1.11	1.59	1.58	0.72	0.87
Female	1.59	0.93	0.93	1.12	1.30	0.58	0.70
Race/ethnicity							
White	1.67	0.69	0.96	1.12	1.21	0.78	0.92
Black	2.51	2.68	1.61	2.96	2.41	0.53	1.58
Hispanic	1.38	2.16	1.77	1.99	3.71	0.92	0.74
Asian	1.98	2.77	2.61	4.52	3.45	1.62	1.16
American Indian	5.24	4.72	5.63	14.47	†	2.88	3.19
Pacific Islander	4.12	9.75	6.51	15.01	14.76	1.99	3.31
Two or more races	1.91	3.26	2.60	4.93	5.28	2.14	1.46
Other	4.54	11.54	7.39	11.28	†	3.11	3.77
Age as of 12/31/07							
18 years or younger	1.58	1.96	1.41	2.93	2.23	1.05	0.74
19–23 years	1.78	0.86	0.95	1.22	1.30	0.52	0.73
24–29 years	1.82	1.78	1.44	2.78	2.94	0.46	0.97
30–39 years	2.21	2.25	2.45	2.98	3.57	0.57	1.41
40 years or older	1.40	2.85	3.18	3.99	3.22	1.20	1.37
Dependency status							
Dependent	1.70	0.93	0.97	1.27	1.22	0.66	0.70
Independent	1.66	1.03	1.33	1.95	2.19	0.40	0.95
Unmarried, no dependents	1.63	1.58	1.60	2.85	2.50	0.54	0.98
Married, no dependents	1.09	2.66	2.59	3.38	2.92	1.55	1.35
Single parent	3.01	2.04	2.02	2.17	4.46	0.43	1.80
Married parent	1.38	2.18	2.49	3.65	3.95	0.88	0.81

Table S3.2-C. Standard errors for table 3.2-C: Percentage of undergraduates receiving federal aid, by type of institution and selected student characteristics: 2007–08—Continued

		Public 4-year		Private nonprofit 4-year			
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions
Dependency and income in 2006							
Dependent							
Less than \$20,000	1.68	1.68	1.40	2.14	2.40	0.64	0.81
\$20,000-39,999	1.94	1.39	1.05	2.30	1.70	0.96	0.78
\$40,000-59,999	2.26	2.17	2.11	3.13	2.51	1.47	1.28
\$60,000-79,999	1.83	1.69	1.46	2.81	3.03	2.45	1.00
\$80,000-99,999	1.36	1.91	1.22	2.98	3.39	3.29	0.78
\$100,000 or more	1.24	1.44	1.27	2.31	1.68	4.26	0.79
Independent							
Less than \$10,000	1.76	1.72	1.41	3.30	3.51	0.67	0.68
\$10,000-19,999	2.28	2.01	1.97	2.50	4.76	0.44	1.11
\$20,000-29,999	2.43	2.28	3.27	2.87	3.97	0.71	1.78
\$30,000-49,999	1.94	2.03	2.72	4.57	3.42	0.77	1.44
\$50,000 or more	0.77	1.65	2.39	3.12	3.21	1.80	0.81
Income group							
Lowest 25 percent	1.49	1.23	0.72	1.44	1.89	0.50	0.51
Middle 50 percent	1.92	0.88	1.24	1.30	1.33	0.44	1.03
Highest 25 percent	0.75	1.17	1.19	1.58	1.39	1.95	0.63
Aid status							
No aid	†	†	†	†	†	†	†
Received aid	0.72	0.69	0.66	0.83	1.19	0.23	0.25
Grant status							
No grant	0.87	0.75	1.36	2.89	2.10	0.95	0.88
Received grant	0.73	0.84	0.62	0.86	1.32	0.19	0.34
Loan status							
No loan	1.55	1.44	0.74	1.75	1.52	3.21	1.03
Received loan	0.93	0.58	0.47	0.74	0.74	0.19	0.18

[†] Not applicable.

Table 3.2-D. Average amount of federal aid received by undergraduates who received federal aid, by type of institution and selected student characteristics: 2007–08

		Public	4-year		vate it 4-year		
			. , , , ,		ii i youi		All
	Public	Non-	Doctorate-	Non-	Doctorate-		insti-
Student characteristics	2-year	granting	granting	granting	granting	For-profit	tutions ¹
Total	\$3,500	\$6,400	\$7,500	\$8,300	\$9,200	\$7,500	\$6,600
Attendance status							
Full-time/full-year	4,800	7,200	8,100	9,200	10,000	8,900	7,900
Full-time/part-year	2,800	4,800	5,700	6,200	6,700	7,000	5,600
Part-time/full-year	3,500	5,700	7,300	7,500	8,400	7,900	5,500
Part-time/part-year	2,100	4,400	4,600	4,700	5,400	4,900	3,700
Housing ²							
On campus	4,700	7,500	8,100	9,300	9,900	11,500	8,600
Off campus	3,700	6,500	7,600	7,500	8,800	7,400	6,300
Living with parents	3,200	5,000	5,800	6,800	7,900	7,900	5,200
Price of attendance ²							
Less than \$6,000	1,700	2,100	2,600	2,000	‡	2,600	1,900
\$6,000–10,999	3,400	4,300	4,700	4,200	4,600	4,600	3,900
\$11,000–18,999	4,900	7,000	7,200	6,100	6,500	6,900	6,500
\$19,000 or more	5,300	8,200	9,100	9,400	9,900	8,800	9,100
Gender							
Male	3,500	6,300	7,500	8,400	9,400	7,800	6,700
Female	3,500	6,500	7,600	8,200	9,100	7,400	6,500
Race/ethnicity ³							
White	3,700	6,500	7,400	8,400	9,200	7,500	6,700
Black	3,500	7,600	8,500	8,700	10,200	7,300	6,700
Hispanic	3,200	5,100	7,100	7,000	8,600	7,600	6,000
Asian	3,400	5,200	7,200	9,300	8,500	7,500	6,400
American Indian	3,600	5,500	7,200	‡	‡	6,700	5,600
Pacific Islander	3,200	‡	8,900	9,200	‡	8,800	6,800
Two or more races	3,700	7,100	7,500	9,500	10,900	8,000	7,000
Other	3,000	‡	7,400	‡	‡	7,000	5,900
Age as of 12/31/07							
18 years or younger	3,300	6,200	7,300	9,000	9,400	8,900	6,700
19–23 years	3,400	6,300	7,400	8,700	9,500	7,900	6,800
24–29 years	3,800	6,700	8,000	7,400	8,600	7,400	6,400
30–39 years	3,600	6,800	8,200	7,000	8,800	7,100	6,100
40 years or older	3,700	6,600	7,400	7,100	7,700	6,900	5,800
Dependency status							
Dependent	3,400	6,300	7,400	8,800	9,500	8,400	6,900
Independent	3,600	6,600	8,000	7,400	8,500	7,200	6,200
Unmarried, no dependents	3,600	6,800	8,100	8,000	9,400	7,200	6,600
Married, no dependents	3,900	5,900	6,900	7,500	6,700	7,100	6,100
Single parent	3,600	6,700	8,700	7,400	8,800	7,400	6,100
Married parent	3,600	6,600	7,500	6,700	7,600	7,000	5,900

Table 3.2-D. Average amount of federal aid received by undergraduates who received federal aid, by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year	Priv nonprof	vate it 4-year		
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions ¹
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$3,600	\$7,000	\$8,400	\$8,900	\$10,500	\$7,700	\$6,600
\$20,000-39,999	3,400	6,300	7,500	10,100	10,500	8,700	6,700
\$40,000-59,999	2,900	5,900	6,500	8,600	9,200	8,100	6,300
\$60,000-79,999	3,400	5,900	6,700	7,700	8,900	7,900	6,600
\$80,000-99,999	3,700	6,000	7,300	8,400	8,300	9,400	7,100
\$100,000 or more	3,400	6,400	7,900	8,600	9,700	9,800	7,900
Independent							
Less than \$10,000	3,900	7,500	8,800	8,800	9,900	7,600	6,800
\$10,000-19,999	3,400	6,400	7,800	7,300	8,800	7,300	6,000
\$20,000-29,999	3,700	6,000	7,300	7,100	8,100	7,400	6,000
\$30,000-49,999	3,200	6,300	7,300	7,000	7,800	7,100	5,800
\$50,000 or more	4,300	6,000	6,900	6,300	7,000	6,200	6,000
Income group ⁴							
Lowest 25 percent	3,700	6,900	8,200	9,200	10,300	7,800	6,700
Middle 50 percent	3,300	6,000	7,000	8,000	8,700	7,500	6,300
Highest 25 percent	4,000	6,400	7,900	7,800	9,100	6,700	7,300
Aid status							
No aid	†	†	†	†	†	†	†
Received aid	3,500	6,400	7,500	8,300	9,200	7,500	6,600
Grant status							
No grant	3,800	6,000	7,200	6,600	9,400	6,900	6,400
Received grant	3,500	6,500	7,700	8,500	9,200	7,700	6,600
Loan status ⁵							
No loan	2,600	3,300	4,200	3,700	4,900	2,900	3,000
Received loan	5,000	7,400	8,100	8,900	9,800	7,700	7,700

[†] Not applicable.

NOTE: Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Federal aid includes Parent PLUS loans and excludes veterans benefits and the education tax credit and tax deduction benefits. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[‡] Reporting standards not met.

¹ All institutions also include those attending public less-than-2-year, private not-for-profit less-than-4-year, and multiple institutions, in addition to the institutions listed separately in this table.

² Excludes those attending more than one institution.

³ Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

⁴ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁵ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S3.2-D. Standard errors for table 3.2-D: Average amount of federal aid received by undergraduates who received federal aid, by type of institution and selected student characteristics: 2007–08

		Public	4-year		vate it 4-year		
		Non-		Non-			All
	Public	doctorate-	Doctorate-	doctorate-	Doctorate-		insti-
Student characteristics	2-year	granting	granting	granting	granting	For-profit	tutions
Total	\$160	\$90	\$70	\$210	\$200	\$110	\$110
Attendance status							
Full-time/full-year	190	120	90	230	220	150	70
Full-time/part-year	100	200	210	380	180	270	130
Part-time/full-year	190	110	120	350	380	170	110
Part-time/part-year	60	220	160	240	320	200	80
Housing							
On campus	280	200	120	310	270	2,320	100
Off campus	200	100	80	200	270	100	110
Living with parents	100	110	110	210	460	240	100
Price of attendance							
Less than \$6,000	40	180	190	260	†	200	40
\$6,000–10,999	110	100	90	190	270	120	70
\$11,000-18,999	250	100	60	160	310	160	60
\$19,000 or more	240	280	160	230	190	170	80
Gender							
Male	150	120	110	270	320	230	120
Female	170	100	90	210	200	100	100
Race/ethnicity							
White	170	90	100	240	230	180	100
Black	200	280	150	290	390	140	140
Hispanic	140	150	180	300	740	200	150
Asian	170	310	320	730	300	690	190
American Indian	320	590	620	†	†	1,690	460
Pacific Islander	350	†	1,150	1,090	†	930	390
Two or more races	210	570	340	740	670	330	190
Other	370	†	560	†	†	770	360
Age as of 12/31/07							
18 years or younger	130	200	210	530	480	450	160
, , ,							
19–23 years	110	110	80	240	250	180	80
24–29 years	210	150	120	350	330	190	140
30–39 years	230	240	230	210	570	140	140
40 years or older	270	310	330	320	300	280	220
Dependency status							
Dependent	100	130	90	250	230	250	90
Independent	210	140	90	210	260	100	140
Unmarried, no dependents	190	160	130	350	410	200	130
Married, no dependents	190	260	250	340	320	400	190
Single parent	230	280	170	300	390	200	180
Married parent	250	280	250	250	450	190	150

Table S3.2-D. Standard errors for table 3.2-D: Average amount of federal aid received by undergraduates who received federal aid, by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year	Private ear nonprofit 4-year			
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$140	\$210	\$170	\$350	\$660	\$290	\$140
\$20,000-39,999	130	200	120	430	440	430	130
\$40,000–59,999	110	170	170	360	620	390	120
\$60,000-79,999	130	220	200	480	350	570	130
\$80,000–99,999	160	240	280	520	360	940	140
\$100,000 or more	190	260	250	450	500	820	160
Independent							
Less than \$10,000	250	240	160	390	580	160	160
\$10,000-19,999	180	230	180	310	490	130	140
\$20,000-29,999	220	310	190	280	480	250	160
\$30,000-49,999	240	230	180	250	330	190	180
\$50,000 or more	250	280	280	280	270	240	120
Income group							
Lowest 25 percent	180	140	90	250	430	170	120
Middle 50 percent	150	100	100	240	200	150	110
Highest 25 percent	180	210	250	350	430	240	120
Aid status							
No aid	†	†	†	†	†	†	†
Received aid	160	90	70	210	200	110	110
Grant status							
No grant	120	130	140	350	560	200	80
Received grant	170	100	80	230	200	100	120
Loan status							
No loan	70	70	100	180	440	150	60
Received loan	260	120	80	210	190	110	80

[†] Not applicable.

Table 3.2-E. Percentage of undergraduates receiving federal Pell grants and Stafford loans, by type of aid and selected institutional and student characteristics: 2007–08

Institutional and	Federal	Any Stafford Loop	Stafford	Stafford
student characteristics	Pell grant	Stafford Ioan	subsidized	unsubsidized
Total	27.3	34.5	29.7	22.0
Institution type				
Public				
Less-than-2-year	33.8	14.6	13.6	11.3
2-year	21.0	10.2	8.3	5.8
4-year	25.3	41.3	33.3	23.4
Non-doctorate-granting	29.6	38.9	31.7	23.3
Doctorate-granting	23.0	42.5	34.1	23.4
Private nonprofit				
Less-than-4-year	48.0	40.2	37.8	29.5
4-year	25.6	53.8	45.8	27.9
Non-doctorate-granting	29.7	56.8	48.1	31.4
Doctorate-granting	21.1	50.4	43.2	24.1
For-profit	62.7	87.9	86.4	78.1
More than one institution	27.8	40.3	34.1	25.0
Attendance status				
Full-time/full-year	32.6	48.7	40.8	27.9
Full-time/part-year	33.5	41.0	36.9	29.9
Part-time/full-year	28.2	26.7	23.7	17.9
Part-time/part-year	14.6	15.0	13.3	11.5
Housing ¹				
On campus	22.8	52.3	41.3	24.2
Off campus	29.8	34.5	31.2	24.8
Living with parents	24.4	21.8	18.0	13.1
Price of attendance ¹				
Less than \$6,000	11.1	3.6	2.3	2.4
\$6,000-10,999	29.8	21.6	18.3	14.3
\$11,000–18,999	37.4	48.4	41.9	31.5
\$19,000 or more	30.9	63.3	55.7	39.1
Gender				
Male	21.5	30.7	25.9	18.7
Female	31.8	37.3	32.6	24.4
Race/ethnicity ²				
White	20.5	34.1	28.1	21.4
Black	46.3	45.9	42.7	33.1
Hispanic	39.4	30.0	27.7	18.0
Asian	22.4	21.8	19.6	10.6
American Indian	36.1	31.9	29.5	22.6
Pacific Islander	26.2	27.7	23.5	18.0
Two or more races	30.8	37.8	33.0	24.4
Other	33.2	32.4	27.5	21.7

Table 3.2-E. Percentage of undergraduates receiving federal Pell grants and Stafford loans, by type of aid and selected institutional and student characteristics: 2007–08—Continued

Institutional and	Federal	Any	Stafford	Stafford
student characteristics	Pell grant	Stafford loan	subsidized	unsubsidized
Age as of 12/31/07				
18 years or younger	27.1	35.9	28.1	16.9
19–23 years	24.2	35.6	28.9	18.7
24–29 years	36.4	37.3	36.0	30.0
30–39 years	32.6	34.6	32.7	29.5
40 years or older	21.0	23.1	21.2	19.7
Dependency status				
Dependent	21.8	35.7	28.0	17.3
Independent	33.6	33.1	31.6	27.1
Unmarried, no dependents	29.7	34.2	33.0	27.6
Married, no dependents	12.5	24.7	21.4	20.4
Single parent	54.0	38.3	37.7	31.7
Married parent	26.5	29.8	27.9	24.8
Dependency and income in 2006				
Dependent				
Less than \$20,000	68.1	35.9	35.2	11.8
\$20,000-39,999	59.2	41.1	39.9	14.0
\$40,000-59,999	23.5	38.4	36.2	12.9
\$60,000-79,999	1.2	37.9	31.7	17.8
\$80,000-99,999	#	36.7	24.6	22.1
\$100,000 or more	#	29.3	12.6	21.9
Independent				
Less than \$10,000	58.3	40.4	39.7	31.8
\$10,000-19,999	52.3	41.9	41.2	33.5
\$20,000-29,999	34.8	36.8	36.0	30.5
\$30,000-49,999	28.2	31.7	30.6	26.2
\$50,000 or more	0.2 !	18.4	14.8	16.6
Income group ³				
Lowest 25 percent	61.7	39.6	38.8	22.1
Middle 50 percent	23.1	37.0	33.1	23.0
Highest 25 percent	0.2	24.0	13.4	19.6
Aid status				
No aid	†	†	†	†
Received aid	41.7	52.5	45.3	33.5
Grant status				
No grant	†	19.6	14.5	14.2
Received grant	52.8	48.3	43.9	29.2

Table 3.2-E. Percentage of undergraduates receiving federal Pell grants and Stafford loans, by type of aid and selected institutional and student characteristics: 2007–08—Continued

Institutional and student characteristics	Federal Pell grant	Any Stafford Ioan	Stafford subsidized	Stafford unsubsidized
Loan status ⁴				
No loan	16.6	†	†	†
Received loan	44.5	89.6	77.2	57.0

[†] Not applicable.

NOTE: Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. For students attending more than one institution, includes federal aid received at any institution. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[#] Rounds to zero.

[!] Interpret data with caution (estimates are unstable).

¹ Excludes students attending more than one institution.

² Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

³ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁴ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S3.2-E. Standard errors for table 3.2-E: Percentage of undergraduates receiving federal Pell grants and Stafford loans, by type of aid and selected institutional and student characteristics: 2007–08

Institutional and student characteristics	Federal Pell grant	Any Stafford Ioan	Stafford subsidized	Stafford unsubsidized
Total	0.47	0.15	† 1	† ¹
Institution type				
Public				
Less-than-2-year	1.32	1.36	0.81	0.83
2-year	1.17	0.34	0.19	0.13
4-year	0.40	0.29	0.10	0.10
Non-doctorate-granting	0.77	0.74	0.71	0.63
Doctorate-granting	0.35	0.67	0.36	0.35
Private nonprofit				
Less-than-4-year	2.62	0.81	0.49	0.42
4-year	0.33	0.31	0.20	0.16
Non-doctorate-granting	1.06	0.95	0.99	0.82
Doctorate-granting	1.06	1.10	1.06	0.90
For-profit	0.74	0.18	0.11	0.14
More than one institution	2.19	2.77	2.37	1.90
Attendance status				
Full-time/full-year	0.37	0.45	0.58	0.45
Full-time/part-year	1.00	0.74	0.70	0.65
Part-time/full-year	0.60	0.39	0.38	0.36
Part-time/part-year	1.34	1.00	0.86	0.74
Housing				
On campus	0.58	0.69	0.51	0.40
Off campus	0.77	0.47	0.32	0.28
Living with parents	0.94	0.40	0.38	0.35
Price of attendance				
Less than \$6,000	1.44	0.37	0.26	0.25
\$6,000–10,999	0.87	0.48	0.41	0.40
\$11,000–18,999	0.58	0.51	0.47	0.43
\$19,000 or more	0.54	0.56	0.49	0.56
Gender				
Male	0.60	0.36	0.30	0.32
Female	0.37	0.24	0.22	0.22
Race/ethnicity				
White	0.39	0.50	0.32	0.28
Black	1.20	1.01	0.95	0.92
Hispanic	0.63	1.15	1.18	1.02
Asian	0.99	0.90	0.93	0.64
American Indian	3.43	3.49	3.41	3.42
Pacific Islander	3.02	2.82	2.68	2.64
Two or more races	1.33	1.70	1.59	1.52
Other	3.10	3.88	3.75	3.71

Table S3.2-E. Standard errors for table 3.2-E: Percentage of undergraduates receiving federal Pell grants and Stafford loans, by type of aid and selected institutional and student characteristics: 2007–08—Continued

Institutional and	Federal	Any	Stafford	Stafford
student characteristics	Pell grant	Stafford loan	subsidized	unsubsidized
Age as of 12/31/07				
18 years or younger	0.86	0.83	0.72	0.61
19–23 years	0.37	0.28	0.22	0.21
24–29 years	0.73	0.50	0.49	0.48
30–39 years	0.96	0.68	0.64	0.57
40 years or older	0.79	0.85	0.81	0.84
Dependency status				
Dependent	0.37	0.28	0.23	0.21
Independent	0.62	0.28	0.26	0.24
Unmarried, no dependents	0.64	0.61	0.60	0.55
Married, no dependents	0.63	1.18	1.12	1.12
Single parent	1.51	0.82	0.81	0.85
Married parent	0.67	0.67	0.69	0.78
Dependency and income in 2006				
Dependent				
Less than \$20,000	0.85	1.03	1.03	0.56
\$20,000-39,999	0.76	0.90	0.89	0.62
\$40,000-59,999	0.90	0.70	0.68	0.49
\$60,000-79,999	0.14	0.88	0.71	0.52
\$80,000-99,999	†	0.73	0.64	0.64
\$100,000 or more	†	0.73	0.35	0.61
Independent				
Less than \$10,000	0.58	0.92	0.94	0.96
\$10,000-19,999	1.09	0.77	0.77	0.77
\$20,000-29,999	1.31	0.92	0.89	0.86
\$30,000-49,999	1.07	0.80	0.77	0.67
\$50,000 or more	0.09	0.74	0.65	0.65
Income group				
Lowest 25 percent	0.44	0.69	0.70	0.52
Middle 50 percent	0.66	0.38	0.26	0.19
Highest 25 percent	0.05	0.57	0.33	0.50
Aid status				
No aid	†	†	†	†
Received aid	0.20	0.74	0.81	0.60
Grant status				
No grant	†	0.73	0.44	0.44
Received grant	0.27	0.90	0.90	0.65

Table S3.2-E. Standard errors for table 3.2-E: Percentage of undergraduates receiving federal Pell grants and Stafford loans, by type of aid and selected institutional and student characteristics: 2007–08—Continued

Institutional and student characteristics	Federal Pell grant	Any Stafford Ioan	Stafford subsidized	Stafford unsubsidized
Loan status				
No loan	0.89	†	†	†
Received loan	0.44	0.23	0.46	0.34

[†] Not applicable.

¹ The percentage of students who received Stafford loans is computed from known population totals, obtained from the U.S. Department of Education's Office of Postsecondary Education and from Integrated Postsecondary Education Data System, and therefore is not subject to sampling variance.

Table 3.2-F. Average amount of federal aid in Pell grants and Stafford loans received by undergraduates who received Pell grants and Stafford loans, by type of aid and selected institutional and student characteristics: 2007–08

Institutional and	Federal	Any	Stafford	Stafford
student characteristics	Pell grant	Stafford Ioan	subsidized	unsubsidized
Total	\$2,600	\$5,000	\$3,400	\$3,200
Institution type				
Public				
Less-than-2-year	2,500	4,900	2,600	3,100
2-year	2,300	3,700	2,700	2,700
4-year	2,800	5,000	3,700	3,500
Non-doctorate-granting	2,800	4,900	3,600	3,400
Doctorate-granting	2,900	5,000	3,700	3,600
Private nonprofit				
Less-than-4-year	2,600	4,600	2,800	2,700
4-year	2,900	5,200	3,800	3,600
Non-doctorate-granting	2,800	5,100	3,800	3,500
Doctorate-granting	2,900	5,200	4,000	3,700
For-profit	2,500	5,500	2,900	2,900
More than one institution	2,600	4,900	3,400	3,300
Attendance status				
Full-time/full-year	3,300	5,200	3,800	3,600
Full-time/part-year	2,100	4,400	2,600	2,700
Part-time/full-year	2,200	5,400	3,500	3,400
Part-time/part-year	1,300	3,900	2,300	2,400
Housing ¹				
On campus	3,000	4,400	3,600	3,400
Off campus	2,400	5,300	3,300	3,200
Living with parents	2,700	4,400	3,100	3,000
Price of attendance ¹				
Less than \$6,000	1,300	2,400	1,700	1,900
\$6,000–10,999	2,100	3,700	2,500	2,500
\$11,000–18,999	2,900	5,000	3,300	3,300
\$19,000 or more	3,000	5,500	3,800	3,600
Gender				
Male	2,600	4,900	3,400	3,300
Female	2,500	5,000	3,300	3,200
Race/ethnicity ²				
White	2,400	4,900	3,400	3,300
Black	2,600	5,300	3,200	3,100
Hispanic	2,700	5,000	3,300	3,200
Asian	2,900	4,800	3,500	3,300
American Indian	2,500	4,600	2,800	2,800
Pacific Islander	2,400	5,000	3,400	3,400
Two or more races	2,600	5,200	3,500	3,400
Other	2,500	4,900	3,400	3,100

Table 3.2-F. Average amount of federal aid in Pell grants and Stafford loans received by undergraduates who received Pell grants and Stafford loans, by type of aid and selected institutional and student characteristics: 2007–08—Continued

Institutional and	Federal	Any	Stafford	Stafford
student characteristics	Pell grant	Stafford Ioan	subsidized	unsubsidized
Age as of 12/31/07				
18 years or younger	\$2,800	\$3,700	\$3,000	\$2,900
19–23 years	2,700	4,600	3,500	3,300
24–29 years	2,400	5,800	3,300	3,300
30–39 years	2,300	5,800	3,200	3,200
40 years or older	2,200	5,800	3,200	3,300
Dependency status				
Dependent	2,800	4,300	3,500	3,300
Independent	2,400	5,700	3,200	3,200
Unmarried, no dependents	2,400	6,000	3,400	3,300
Married, no dependents	2,300	6,100	3,400	3,800
Single parent	2,500	5,400	3,100	2,900
Married parent	2,200	5,700	3,200	3,300
Dependency and income in 2006				
Dependent				
Less than \$20,000	3,400	4,400	3,400	3,000
\$20,000–39,999	2,800	4,400	3,600	2,800
\$40,000–59,999	1,600	4,300	3,600	2,800
\$60,000–79,999	600	4,200	3,400	2,900
\$80,000–99,999	‡	4,300	3,300	3,400
\$100,000 or more	‡	4,300	3,300	3,800
Independent				
Less than \$10,000	2,800	5,600	3,300	3,000
\$10,000–19,999	2,200	5,600	3,200	3,100
\$20,000–29,999	2,500	5,700	3,200	3,100
\$30,000–49,999	1,700	5,900	3,300	3,300
\$50,000 or more	‡	6,000	3,100	3,900
Income group ³				
Lowest 25 percent	3,000	5,000	3,400	3,000
Middle 50 percent	2,000	5,000	3,400	3,100
Highest 25 percent	600	4,900	3,200	3,900
Aid status				
No aid	†	†	†	†
Received aid	2,600	5,000	3,400	3,200
Grant status				
No grant	†	4,900	3,200	3,500
Received grant	2,600	5,000	3,400	3,100

Table 3.2-F. Average amount of federal aid in Pell grants and Stafford loans received by undergraduates who received Pell grants and Stafford loans, by type of aid and selected institutional and student characteristics: 2007–08—Continued

Institutional and student characteristics	Federal Pell grant	Any Stafford Ioan	Stafford subsidized	Stafford unsubsidized
Loan status ⁴				
No loan	\$2,500	†	†	†
Received loan	2,600	5,000	3,400	3,200

[†] Not applicable.

NOTE: Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. For students attending more than one institution, includes federal aid received at any institution. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[‡] Reporting standards not met.

¹ Excludes students attending more than one institution.

² Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

³ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁴ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S3.2-F. Standard errors for table 3.2-F: Average amount of federal aid in Pell grants and Stafford loans received by undergraduates who received Pell grants and Stafford loans, by type of aid and selected institutional and student characteristics: 2007–08

Institutional and	Federal	Any	Stafford	Stafford
student characteristics	Pell grant	Stafford Ioan	subsidized	unsubsidized
Total	\$40	\$20	\$10	\$10
Institution type				
Public				
Less-than-2-year	100	230	160	180
2-year	70	150	80	100
4-year	20	60	60	70
Non-doctorate-granting	30	70	50	80
Doctorate-granting	30	60	60	80
Private nonprofit				
Less-than-4-year	140	260	180	220
4-year	30	40	30	50
Non-doctorate-granting	40	50	40	70
Doctorate-granting	60	80	40	70
For-profit	30	70	30	50
More than one institution	90	160	90	120
Attendance status				
Full-time/full-year	30	20	20	20
Full-time/part-year	30	70	40	70
Part-time/full-year	30	70	30	40
Part-time/part-year	20	60	40	50
Housing				
On campus	30	30	40	70
Off campus	40	30	10	20
Living with parents	50	50	30	50
Price of attendance				
Less than \$6,000	30	50	40	70
\$6,000–10,999	30	40	30	40
\$11,000–18,999	40	30	30	40
\$19,000 or more	30	30	30	40
Gender				
Male	50	40	20	20
Female	40	20	10	20
Race/ethnicity				
White	40	30	10	20
Black	50	50	40	40
Hispanic	50	50	40	60
Asian	60	90	90	90
American Indian	120	330	160	240
Pacific Islander	150	280	190	240
Two or more races	80	120	80	90
Other	130	290	190	270

Table S3.2-F. Standard errors for table 3.2-F: Average amount of federal aid in Pell grants and Stafford loans received by undergraduates who received Pell grants and Stafford loans, by type of aid and selected institutional and student characteristics: 2007–08—Continued

Institutional and	Federal	Any	Stafford	Stafford
student characteristics	Pell grant	Stafford Ioan	subsidized	unsubsidized
Age as of 12/31/07				
18 years or younger	\$60	\$40	\$30	\$40
19–23 years	40	20	20	40
24–29 years	50	80	30	50
30–39 years	50	80	40	50
40 years or older	60	140	80	100
Dependency status				
Dependent	40	20	20	40
Independent	50	50	20	30
Unmarried, no dependents	50	70	40	40
Married, no dependents	80	160	90	110
Single parent	50	90	50	60
Married parent	60	80	40	60
Dependency and income in 2006				
Dependent				
Less than \$20,000	50	50	40	70
\$20,000-39,999	40	40	30	50
\$40,000–59,999	30	40	30	70
\$60,000–79,999	70	40	50	80
\$80,000–99,999	†	50	50	80
\$100,000 or more	†	30	50	40
Independent				
Less than \$10,000	50	60	30	40
\$10,000-19,999	40	80	40	40
\$20,000-29,999	60	100	50	70
\$30,000-49,999	50	100	50	50
\$50,000 or more	†	110	60	90
Income group				
Lowest 25 percent	40	30	20	30
Middle 50 percent	40	30	10	20
Highest 25 percent	90	40	40	30
Aid status				
No aid	†	†	†	†
Received aid	40	20	10	10
Grant status				
No grant	†	40	30	30
Received grant	40	20	10	20

Table S3.2-F. Standard errors for table 3.2-F: Average amount of federal aid in Pell grants and Stafford loans received by undergraduates who received Pell grants and Stafford loans, by type of aid and selected institutional and student characteristics: 2007–08—Continued

Institutional and student characteristics	Federal Pell grant	Any Stafford Ioan	Stafford subsidized	Stafford unsubsidized
Loan status				
No loan	\$60	†	†	†
Received loan	30	20	10	10

[†] Not applicable.

Table 3.2-G. Percentage of undergraduates receiving federal campus-based aid, by type of aid and selected institutional and student characteristics: 2007–08

Institutional and student characteristics	Any federal campus-based aid	Supplemental Educational Opportunity Grant (SEOG)	Perkins Ioan	Federal work-study
Total	11.7	6.1	2.8	5.6
Institution type				
Public				
Less-than-2-year	‡	‡	#	#
2-year	5.1	2.8	0.1	2.6
4-year	11.6	5.1	4.0	5.7
Non-doctorate-granting	10.8	5.1	3.0	5.4
Doctorate-granting	12.0	5.1	4.5	5.9
Private nonprofit				
Less-than-4-year	17.4	14.8	‡	2.4
4-year	26.3	10.3	9.4	17.8
Non-doctorate-granting	25.4	11.2	7.5	17.3
Doctorate-granting	27.2	9.2	11.5	18.3
For-profit	21.0	19.1	1.2	1.7
More than one institution	11.0	4.3	3.1	5.9
Attendance status				
Full-time/full-year	18.7	8.7	5.6	10.3
Full-time/part-year	11.8	7.9	1.5	4.0
Part-time/full-year	7.9	4.7	1.2	3.1
Part-time/part-year	3.8	2.1	0.4	1.5
Housing ¹				
On campus	24.5	8.6	9.7	17.1
Off campus	9.9	6.3	1.8	3.4
Living with parents	8.5	4.7	0.9	3.8
Price of attendance ¹				
Less than \$6,000	1.4	0.9	#!	0.5
\$6,000–10,999	6.7	4.4	0.5	2.4
\$11,000–18,999	14.4	8.6	2.8	5.6
\$19,000 or more	25.2	11.2	8.0	14.5
Gender				
Male	10.0	4.5	2.7	5.5
Female	12.9	7.3	2.8	5.8
Race/ethnicity ²				
White	10.2	4.5	2.8	5.4
Black	15.5	10.2	3.0	5.8
Hispanic	13.7	8.5	2.0	5.7
Asian	12.3	5.5	3.4	6.5
American Indian	11.0	7.0 !	1.7	3.9
Pacific Islander	11.4	6.3	2.6 !	4.6
Two or more races	16.0	7.6	4.3	9.0
Other	9.2	6.6	2.1 !	3.9!

Table 3.2-G. Percentage of undergraduates receiving federal campus-based aid, by type of aid and selected institutional and student characteristics: 2007–08—Continued

		Supplemental Educational		
Institutional and	Any federal	Opportunity	Perkins	Federal
student characteristics	campus-based aid	Grant (SEOG)	loan	work-study
Age as of 12/31/07				
18 years or younger	16.3	7.1	5.1	9.4
19–23 years	13.2	5.7	3.5	7.6
24–29 years	10.7	7.5	1.7	2.9
30–39 years	8.7	6.6	1.2	2.0
40 years or older	5.1	3.7	0.7	1.5
Dependency status				
Dependent	13.6	5.2	4.1	8.4
Independent	9.5	7.0	1.3	2.6
Unmarried, no dependents	9.3	6.1	1.9	3.2
Married, no dependents	5.1	2.6	0.8	2.6
Single parent	14.6	12.1	1.3	2.7
Married parent	6.4	4.7	0.8	1.5
Dependency and income in 2006				
Dependent				
Less than \$20,000	25.9	19.1	6.0	10.1
\$20,000-39,999	23.0	13.4	6.6	12.2
\$40,000–59,999	15.5	4.5	6.1	10.1
\$60,000–79,999	10.2	0.4	3.7	8.3
\$80,000-99,999	8.8	‡	2.5	7.3
\$100,000 or more	5.8	‡	1.5	4.9
Independent				
Less than \$10,000	18.9	14.8	2.8	5.3
\$10,000–19,999	14.1	11.0	1.7	3.0
\$20,000-29,999	9.1	6.5	1.0	2.3
\$30,000-49,999	5.4	3.4	0.9	1.4
\$50,000 or more	1.0	‡	0.2	0.8
ncome group ³				
Lowest 25 percent	21.8	15.5	4.7	8.3
Middle 50 percent	10.6	4.2	2.8	5.6
Highest 25 percent	3.4	#!	0.8	2.9
Aid status				
No aid	†	†	†	†
Received aid	17.8	9.2	4.2	8.6
Grant status				
No grant	2.4	<u>†</u>	0.4	2.1
Received grant	20.4	11.7	5.0	9.0

Table 3.2-G. Percentage of undergraduates receiving federal campus-based aid, by type of aid and selected institutional and student characteristics: 2007–08—Continued

Institutional and student characteristics	Any federal campus-based aid	Supplemental Educational Opportunity Grant (SEOG)	Perkins Ioan	Federal work-study
Loan status ⁴				
No loan	5.1	2.5	†	2.9
Received loan	22.3	11.8	7.2	10.0

[†] Not applicable.

NOTE: Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. For students attending more than one institution, includes aid received at the NPSAS institution, but may not include aid received at other institutions. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[#] Rounds to zero.

[!] Interpret data with caution (estimates are unstable).

[‡] Reporting standards not met.

¹ Excludes students attending more than one institution.

² Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

³ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁴ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S3.2-G. Standard errors for table 3.2-G: Percentage of undergraduates receiving federal campus-based aid, by type of aid and selected institutional and student characteristics: 2007–08

		Supplemental		
		Educational		
Institutional and	Any federal	Opportunity	Perkins	Federal
student characteristics	campus-based aid	Grant (SEOG)	loan	work-study
Total	0.22	0.17	0.07	0.16
Institution type				
Public				
Less-than-2-year	†	†	†	†
2-year	0.26	0.16	0.02	0.18
4-year	0.29	0.12	0.14	0.27
Non-doctorate-granting	0.43	0.24	0.25	0.43
Doctorate-granting	0.32	0.14	0.20	0.29
Private nonprofit				
Less-than-4-year	3.20	3.13	†	0.47
4-year	0.69	0.33	0.44	0.66
Non-doctorate-granting	1.06	0.58	0.53	1.02
Doctorate-granting	0.78	0.42	0.74	0.79
For-profit	1.49	1.47	0.19	0.27
More than one institution	0.73	0.35	0.32	0.41
Attendance status				
Full-time/full-year	0.30	0.27	0.16	0.24
Full-time/part-year	0.60	0.52	0.16	0.30
Part-time/full-year	0.33	0.24	0.09	0.16
Part-time/part-year	0.35	0.24	0.07	0.22
Housing				
On campus	0.79	0.27	0.42	0.76
Off campus	0.32	0.26	0.07	0.16
Living with parents	0.31	0.22	0.07	0.20
Price of attendance				
Less than \$6,000	0.19	0.15	†	0.09
\$6,000–10,999	0.35	0.27	0.07	0.22
\$11,000–18,999	0.46	0.40	0.12	0.24
\$19,000 or more	0.60	0.32	0.30	0.50
Gender				
Male	0.26	0.14	0.10	0.24
Female	0.30	0.26	0.09	0.15
Race/ethnicity				
White	0.24	0.15	0.11	0.20
Black	0.58	0.51	0.18	0.29
Hispanic	0.48	0.39	0.16	0.27
Asian	0.52	0.36	0.31	0.46
American Indian	2.39	2.45	0.44	0.98
Pacific Islander	1.77	1.27	1.02	1.08
Two or more races	1.07	0.92	0.73	0.96
Other	1.86	1.59	0.89	1.19

Table S3.2-G. Standard errors for table 3.2-G: Percentage of undergraduates receiving federal campus-based aid, by type of aid and selected institutional and student characteristics: 2007–08—Continued

		Supplemental Educational		
Institutional and student characteristics	Any federal campus-based aid	Opportunity Grant (SEOG)	Perkins Ioan	Federal work-study
Age as of 12/31/07	- Campac Gacca and	Crain (CECC)		
18 years or younger	0.46	0.29	0.25	0.40
19–23 years	0.40	0.29	0.23	0.40
24–29 years	0.48	0.45	0.10	0.19
30–39 years	0.49	0.40	0.10	0.19
40 years or older	0.49	0.29	0.08	0.16
Dependency status				
Dependent	0.25	0.13	0.12	0.23
Independent	0.32	0.30	0.06	0.13
Unmarried, no dependents	0.36	0.31	0.13	0.19
Married, no dependents	0.38	0.28	0.12	0.28
Single parent	0.70	0.68	0.09	0.21
Married parent	0.38	0.38	0.08	0.19
Dependency and income in 2006				
Dependent				
Less than \$20,000	0.91	0.83	0.41	0.53
\$20,000–39,999	0.55	0.44	0.32	0.45
\$40,000–59,999	0.56	0.26	0.37	0.47
\$60,000–79,999	0.68	0.08	0.33	0.63
\$80,000–99,999	0.54	†	0.28	0.49
\$100,000 or more	0.25	†	0.14	0.21
Independent				
Less than \$10,000	0.70	0.71	0.19	0.28
\$10,000–19,999	0.65	0.58	0.16	0.26
\$20,000–29,999	0.58	0.51	0.11	0.27
\$30,000–49,999	0.49	0.46	0.10	0.19
\$50,000 or more	0.18	†	0.04	0.17
Income group				
Lowest 25 percent	0.47	0.47	0.17	0.23
Middle 50 percent	0.32	0.21	0.10	0.23
Highest 25 percent	0.17	†	0.07	0.15
Aid status				
No aid	†	†	†	†
Received aid	0.34	0.32	0.14	0.17
Grant status				
No grant	0.20	†	0.05	0.19
Received grant	0.46	0.40	0.17	0.21

Table S3.2-G. Standard errors for table 3.2-G: Percentage of undergraduates receiving federal campus-based aid, by type of aid and selected institutional and student characteristics: 2007–08—Continued

Institutional and student characteristics	Any federal campus-based aid	Supplemental Educational Opportunity Grant (SEOG)	Perkins Ioan	Federal work-study
Loan status				
No loan	0.23	0.11	†	0.18
Received loan	0.46	0.40	0.19	0.24

[†] Not applicable.

Table 3.2-H. Average amount of federal campus-based aid received by undergraduates who received campus-based aid, by type of aid and selected institutional and student characteristics: 2007–08

Institutional and student characteristics	Any federal campus- based aid	Supplemental Educational Opportunity Grant (SEOG)	Perkins Ioan	Federal work-study
Total	\$2,000	\$700	\$2,000	\$2,300
Institution type Public				
Less-than-2-year	200 !	200 !	‡	‡
2-year	1,900	400	1,300	3,200
4-year	2,200	800	2,000	2,400
Non-doctorate-granting	2,100	700	2,100	2,300
Doctorate-granting	2,300	900	2,000	2,400
Private nonprofit				
Less-than-4-year	600	300	‡	2,100
4-year	2,500	1,200	2,100	1,900
Non-doctorate-granting	2,200	1,000	2,000	1,800
Doctorate-granting	2,800	1,500	2,100	2,100
For-profit	700	400	1,800	3,500
More than one institution	1,900	700	1,900	2,100
Attendance status				
Full-time/full-year	2,200	900	2,100	2,200
Full-time/part-year	1,200	500	1,600	2,200
Part-time/full-year	1,700	500	1,900	2,900
Part-time/part-year	1,800	300	1,500	3,500
Housing ¹				
On campus	2,600	1,200	2,100	1,900
Off campus	1,600	600	2,000	2,700
Living with parents	1,800	500	1,900	3,000
Price of attendance ¹				
Less than \$6,000	800	300	‡	1,700
\$6,000–10,999	1,300	400	1,300	2,600
\$11,000–18,999	1,700	600	1,900	2,600
\$19,000 or more	2,400	1,000	2,100	2,300
Gender				
Male	2,300	800	2,000	2,500
Female	1,800	600	2,000	2,200
Race/ethnicity ²				
White	2,100	700	1,900	2,300
Black	1,700	700	2,100	2,300
Hispanic	1,800	600	2,000	2,600
Asian	2,300	900	2,300	2,500
American Indian	1,300	500	‡	2,000
Pacific Islander	1,800	700	‡	2,600
Two or more races	2,400	800	2,200	2,600
Other	1,500	700	‡	‡

Table 3.2-H. Average amount of federal campus-based aid received by undergraduates who received campus-based aid, by type of aid and selected institutional and student characteristics: 2007–08—Continued

Institutional and student characteristics	Any federal campus- based aid	Supplemental Educational Opportunity Grant (SEOG)	Perkins Ioan	Federal work-study
Age as of 12/31/07		, ,		
18 years or younger	\$2,300	\$900	\$2,000	\$2,200
19–23 years	2,200	800	2,000	2,300
24–29 years	1,400	600	1,900	2,700
30–39 years	1,300	500	2,100	2,700
40 years or older	1,400	500	1,900	3,000
Dependency status				
Dependent	2,300	900	2,000	2,200
Independent	1,400	500	2,000	2,700
Unmarried, no dependents	1,800	700	2,000	2,600
Married, no dependents	2,300	600	2,100	3,200
Single parent	1,000	500	1,900	2,500
Married parent	1,300	500	2,000	2,800
Dependency and income in 2006 Dependent				
Less than \$20,000	1,900	800	2,000	2,200
\$20,000-39,999	2,300	900	2,200	2,200
\$40,000-59,999	2,400	900	2,000	2,100
\$60,000-79,999	2,600	900	1,900	2,300
\$80,000-99,999	2,600	‡	1,900	2,400
\$100,000 or more	2,500	‡	2,000	2,300
Independent				
Less than \$10,000	1,400	600	2,100	2,300
\$10,000–19,999	1,200	500	1,800	2,700
\$20,000-29,999	1,300	500	2,000	3,100
\$30,000-49,999	1,400	500	1,800	3,100
\$50,000 or more	3,600	‡	2,000	4,000
Income group ³				
Lowest 25 percent	1,800	800	2,100	2,200
Middle 50 percent	2,000	600	1,900	2,400
Highest 25 percent	2,700	‡	2,000	2,600
Aid status				
No aid	†	†	†	†
Received aid	2,000	700	2,000	2,300
Grant status				
No grant	3,200	†	2,000	3,400
Received grant	1,800	700	2,000	2,100

Table 3.2-H. Average amount of federal campus-based aid received by undergraduates who received campus-based aid, by type of aid and selected institutional and student characteristics: 2007–08—Continued

Institutional and student characteristics	Any federal campus- based aid	Supplemental Educational Opportunity Grant (SEOG)	Perkins Ioan	Federal work-study
Loan status ⁴				
No loan	\$2,000	\$600	†	\$3,000
Received loan	2,000	700	2,000	2,100

[†] Not applicable.

NOTE: Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. For students attending more than one institution, includes aid received at the NPSAS institution, but may not include aid received at other institutions. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[!] Interpret data with caution (estimates are unstable).

[‡] Reporting standards not met.

¹ Excludes students attending more than one institution.

² Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

³ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁴ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S3.2-H. Standard errors for table 3.2-H: Average amount of federal campus-based aid received by undergraduates who received campus-based aid, by type of aid and selected institutional and student characteristics: 2007–08

Institutional and	Any federal campus-	Supplemental Educational Opportunity	Perkins	Federal
student characteristics	based aid	Grant (SEOG)	loan	work-study
Total	\$30	\$20	\$30	\$30
Institution type				
Public				
Less-than-2-year	90	90	†	†
2-year	80	10	150	100
4-year	30	20	30	40
Non-doctorate-granting	70	20	80	90
Doctorate-granting	40	30	40	50
Private nonprofit				
Less-than-4-year	140	80	†	320
4-year	60	40	60	40
Non-doctorate-granting	80	50	60	50
Doctorate-granting	90	90	80	60
For-profit	60	20	240	260
More than one institution	70	30	70	80
Attendance status				
Full-time/full-year	30	40	40	30
Full-time/part-year	80	80	80	110
Part-time/full-year	60	70	70	100
Part-time/part-year	120	120	120	180
Housing				
On campus	50	40	40	40
Off campus	50	20	50	60
Living with parents	60	20	80	90
Price of attendance				
Less than \$6,000	100	20	†	160
\$6,000-10,999	80	20	90	130
\$11,000–18,999	50	20	50	70
\$19,000 or more	40	30	40	40
Gender				
Male	40	30	50	50
Female	40	20	30	40
Race/ethnicity				
White	40	20	40	40
Black	70	40	70	70
Hispanic	70	20	60	100
Asian	100	60	100	120
American Indian	330	130	†	330
Pacific Islander	240	120	†	310
Two or more races	190	80	110	140
Other	300	90	†	†

Table S3.2-H. Standard errors for table 3.2-H: Average amount of federal campus-based aid received by undergraduates who received campus-based aid, by type of aid and selected institutional and student characteristics: 2007–08—Continued

	Any federal	Supplemental Educational		
Institutional and	campus-	Opportunity	Perkins	Federal
student characteristics	based aid	Grant (SEOG)	loan	work-study
Age as of 12/31/07				
18 years or younger	\$70	\$40	\$60	\$70
19–23 years	30	20	40	40
24–29 years	60	30	60	130
30–39 years	90	30	130	150
40 years or older	100	30	110	220
Dependency status				
Dependent	30	20	40	30
Independent	50	20	50	70
Unmarried, no dependents	70	30	70	110
Married, no dependents	160	60	140	210
Single parent	50	20	70	120
Married parent	90	30	110	190
Dependency and income in 2006				
Dependent				
Less than \$20,000	70	30	60	80
\$20,000-39,999	60	30	50	80
\$40,000-59,999	60	40	60	60
\$60,000-79,999	70	180	80	90
\$80,000-99,999	90	†	110	90
\$100,000 or more	80	†	120	90
Independent				
Less than \$10,000	50	20	80	80
\$10,000-19,999	60	20	70	170
\$20,000-29,999	110	20	120	200
\$30,000-49,999	130	50	130	200
\$50,000 or more	350	†	240	410
Income group				
Lowest 25 percent	40	20	40	50
Middle 50 percent	40	20	40	40
Highest 25 percent	80	†	120	100
Aid status				
No aid	†	†	†	†
Received aid	30	20	30	30
Grant status				
No grant	90	†	130	90
Received grant	30	20	30	30

Table S3.2-H. Standard errors for table 3.2-H: Average amount of federal campus-based aid received by undergraduates who received campus-based aid, by type of aid and selected institutional and student characteristics: 2007–08—Continued

Institutional and student characteristics	Any federal campus- based aid	Supplemental Educational Opportunity Grant (SEOG)	Perkins Ioan	Federal work-study	
Loan status					
No loan	\$60	\$20	†	\$60	
Received loan	40	20	30	30	

[†] Not applicable.

Table 3.3-A. Percentage of undergraduates receiving any state aid, by type of aid and selected institutional and student characteristics: 2007–08

Institutional and	Any state	Any state	State	State	State merit-only
student characteristics	aid ¹	grants	loans	work-study	grants
Total	17.5	16.4	0.4	0.4	4.3
Institution type					
Public					
Less-than-2-year	12.0	2.8 !	‡	#	‡
2-year	12.8	11.6	0.1	0.4!	3.4
4-year	24.1	23.1	0.8	0.5	7.6
Non-doctorate-granting	23.0	22.2	0.5	0.3	5.7
Doctorate-granting	24.8	23.7	0.9	0.6	8.7
Private nonprofit					
Less-than-4-year	24.5	20.4	‡	‡	#
4-year	25.3	24.1	0.6	0.7	2.8
Non-doctorate-granting	28.2	26.7	0.7	1.0	2.8
Doctorate-granting	22.0	21.1	0.6	0.4	2.7
For-profit	8.1	7.1	0.1 !	#	‡
More than one institution	15.3	14.5	0.4	0.4	3.6
Attendance status					
Full-time/full-year	27.7	26.4	0.7	0.6	6.6
Full-time/part-year	15.0	13.6	0.2 !	0.3	4.0
Part-time/full-year	13.7	12.7	0.2	0.4 !	2.8
Part-time/part-year	6.1	5.5	0.1	0.1 !	2.0
Housing ²					
On campus	27.5	26.2	0.9	0.7	7.0
Off campus	15.0	13.8	0.3	0.3	3.3
Living with parents	18.2	17.3	0.1	0.4 !	5.1
Price of attendance ²					
Less than \$6,000	5.7	5.0	#	‡	2.1
\$6,000–10,999	15.9	14.8	0.1	0.4!	4.7
\$11,000–18,999	25.7	24.3	0.4	0.4	6.4
\$19,000 or more	23.7	22.3	1.0	0.7	4.1
Gender					
Male	15.6	14.6	0.4	0.4	3.8
Female	19.0	17.8	0.4	0.4	4.6

Table 3.3-A. Percentage of undergraduates receiving any state aid, by type of aid and selected institutional and student characteristics: 2007–08—Continued

	Any				State
Institutional and	state	Any state	State	State	merit-only
student characteristics	aid ¹	grants	loans	work-study	grants
Race/ethnicity ³					
White	16.9	15.8	0.4	0.4	5.0
Black	19.8	18.9	0.2	0.2	4.2
Hispanic	18.9	17.5	0.4	0.7	2.5
Asian	15.7	14.9	0.4!	0.4	2.4
American Indian	16.8	15.6	‡	0.6 !	2.0 !
Pacific Islander	12.9	11.7	‡	‡	1.4 !
Two or more races	18.4	17.4	‡	0.4	3.9
Other	16.2	15.0	#	‡	3.5 !
Age as of 12/31/07					
18 years or younger	27.1	26.3	0.4	0.6	8.7
19–23 years	20.3	19.2	0.5	0.5	5.7
24–29 years	12.4	11.4	0.2	0.3	1.3
30–39 years	12.1	11.0	0.2	0.3 !	1.5
40 years or older	10.2	8.4	0.1 !	0.1	1.4
Dependency status					
Dependent	22.0	20.9	0.6	0.5	6.6
Independent	12.5	11.4	0.2	0.3	1.7
Unmarried, no dependents	12.5	11.4	0.2	0.3	1.3
Married, no dependents	8.4	7.5	0.1 !	0.3 !	2.0
Single parent	16.3	15.0	0.1	0.3	2.2
Married parent	10.4	9.1	0.2	0.2 !	1.5
Dependency and income in 2006					
Dependent					
Less than \$20,000	32.3	31.4	0.3	0.8	4.9
\$20,000–39,999	34.6	33.2	0.5	1.0	6.0
\$40,000–59,999	26.5	25.5	0.6	0.7	6.7
\$60,000–79,999	18.9	17.5	0.6	0.6	6.9
\$80,000–99,999	15.7	14.4	0.9	0.3	7.2
\$100,000 or more	12.0	11.2	0.6	0.1	7.1
Independent					
Less than \$10,000	18.4	17.0	0.3	0.5	2.0
\$10,000–19,999	16.9	15.5	0.2	0.3 !	1.9
\$20,000-29,999	13.7	12.6	0.2 !	0.3 !	1.8
\$30,000–49,999	10.8	9.7	0.1 !	0.2 !	1.7
\$50,000 or more	4.3	3.5	0.1 !	0.1 !	1.2
Income group ⁴					
Lowest 25 percent	26.7	25.4	0.3	0.7	3.9
Middle 50 percent	17.4	16.3	0.4	0.4	4.5
Highest 25 percent	8.2	7.4	0.3	0.1	4.2

Table 3.3-A. Percentage of undergraduates receiving any state aid, by type of aid and selected institutional and student characteristics: 2007–08—Continued

Institutional and student characteristics	Any state aid ¹	Any state grants	State Ioans	State work-study	State merit-only grants
Aid status					
No aid	†	†	†	†	†
Received aid	26.7	25.0	0.6	0.6	6.5
Grant status					
No grant	0.9	†	0.2	0.1!	†
Received grant	33.1	31.8	0.5	0.7	8.2
Loan status ⁵					
No loan	13.0	12.1	†	0.3 !	4.6
Received loan	24.8	23.3	1.0	0.6	3.7

[†] Not applicable.

NOTE: Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. For students attending more than one institution, includes aid received at the NPSAS institution, but may not include aid received at other institutions. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[#] Rounds to zero.

[!] Interpret data with caution (estimates are unstable).

[‡] Reporting standards not met.

¹ Includes all state need- and merit-based grants, scholarships, loans, and work-study assistance.

² Excludes students attending more than one institution.

³ Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

⁴ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁵ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S3.3-A. Standard errors for table 3.3-A: Percentage of undergraduates receiving any state aid, by type of aid and selected institutional and student characteristics: 2007–08

Institutional and student characteristics	Any state aid	Any state grants	State Ioans	State work-study	State merit-only grants
	uiu	granto	104.10	Work olday	granto
Total	0.35	0.33	0.03	0.07	0.17
Institution type					
Public					
Less-than-2-year	3.42	1.01	†	†	†
2-year	0.68	0.61	0.02	0.15	0.36
4-year	0.53	0.53	0.08	0.07	0.35
Non-doctorate-granting	0.72	0.70	0.11	0.06	0.56
Doctorate-granting	0.63	0.64	0.10	0.11	0.41
Private nonprofit					
Less-than-4-year	5.32	3.62	†	†	†
4-year	0.78	0.78	0.09	0.15	0.34
Non-doctorate-granting	1.25	1.23	0.14	0.28	0.50
Doctorate-granting	0.99	0.97	0.10	0.05	0.37
For-profit	0.64	0.59	0.04	†	†
More than one institution	0.60	0.59	0.07	0.08	0.21
Attendance status					
Full-time/full-year	0.43	0.43	0.05	0.08	0.26
Full-time/part-year	0.59	0.48	0.08	0.07	0.29
Part-time/full-year	0.39	0.37	0.03	0.16	0.19
Part-time/part-year	0.49	0.46	0.02	0.05	0.21
Housing					
On campus	0.87	0.84	0.10	0.14	0.36
Off campus	0.42	0.38	0.05	0.07	0.18
Living with parents	0.72	0.71	0.03	0.13	0.34
Price of attendance					
Less than \$6,000	0.65	0.56	†	†	0.24
\$6,000–10,999	0.61	0.56	0.03	0.16	0.34
\$11,000–18,999	0.48	0.49	0.06	0.05	0.31
\$19,000 or more	0.61	0.61	0.09	0.11	0.25
Gender					
Male	0.52	0.47	0.05	0.08	0.17
Female	0.29	0.29	0.03	0.08	0.19

Table S3.3-A. Standard errors for table 3.3-A: Percentage of undergraduates receiving any state aid, by type of aid and selected institutional and student characteristics: 2007–08—Continued

	Any				State
Institutional and	state	Any state	State	State	merit-only
student characteristics	aid	grants	loans	0.06 0.06 0.19 0.10 0.25	grants
Race/ethnicity					
White	0.48	0.45	0.04	0.06	0.22
Black	0.49	0.46	0.04		0.25
Hispanic	0.51	0.46	0.10		0.22
Asian	0.70	0.62	0.19		0.24
American Indian	2.22	2.29	†		0.71
Pacific Islander	1.74	1.73	†		0.47
Two or more races	0.87	0.89	†	•	0.41
Other	2.50	2.35	†		1.44
Age as of 12/31/07					
18 years or younger	0.77	0.75	0.09	0.16	0.47
19–23 years	0.43	0.45	0.05		0.25
24–29 years	0.43	0.39	0.04	0.05	0.16
30–39 years	0.51	0.45	0.04	0.13	0.16
40 years or older	0.59	0.42	0.04	0.04	0.12
Dependency status					
Dependent	0.47	0.48	0.05	0.09	0.26
Independent	0.36	0.29	0.02	0.07	0.13
Unmarried, no dependents	0.42	0.33	0.04	0.07	0.15
Married, no dependents	0.52	0.45	0.04	0.11	0.28
Single parent	0.57	0.56	0.03	0.07	0.20
Married parent	0.49	0.43	0.04	0.09	0.14
Dependency and income in 2006					
Dependent					
Less than \$20,000	0.82	0.80	0.08	0.14	0.42
\$20,000–39,999	0.64	0.64	0.12	0.29	0.35
\$40,000–59,999	0.75	0.77	0.09		0.45
\$60,000–79,999	0.83	0.82	0.11	0.14	0.44
\$80,000-99,999	0.68	0.73	0.18	0.08	0.45
\$100,000 or more	0.52	0.50	0.09	0.04	0.31
Independent					
Less than \$10,000	0.59	0.55	0.05	0.11	0.19
\$10,000–19,999	0.65	0.57	0.06	0.11	0.21
\$20,000–29,999	0.62	0.56	0.07	0.08	0.21
\$30,000-49,999	0.61	0.56	0.03	0.06	0.17
\$50,000 or more	0.39	0.34	0.04	0.06	0.15
Income group					
Lowest 25 percent	0.41	0.38	0.06	0.13	0.21
Middle 50 percent	0.49	0.49	0.04	0.07	0.22
Highest 25 percent	0.35	0.32	0.05	0.03	0.18

Table S3.3-A. Standard errors for table 3.3-A: Percentage of undergraduates receiving any state aid, by type of aid and selected institutional and student characteristics: 2007–08—Continued

Institutional and student characteristics	Any state aid	Any state grants	State Ioans	State work-study	State merit-only grants
Aid status					
No aid	†	†	†	†	†
Received aid	0.32	0.31	0.05	0.11	0.25
Grant status					
No grant	0.11	†	0.03	0.04	†
Received grant	0.40	0.39	0.05	0.10	0.31
Loan status					
No loan	0.55	0.51	†	0.10	0.22
Received loan	0.36	0.35	0.07	0.07	0.20

[†] Not applicable.

Table 3.3-B. Average amount of state aid received by undergraduates who received a specific type of state aid, by type of aid and selected institutional and student characteristics: 2007–08

Institutional and student characteristics	Any state aid¹	Any state grants	State loans	State work- study	State merit-only grants
Total	\$2,600	\$2,500	\$6,100	\$1,900	\$2,100
Institution type					
Public					
Less-than-2-year	2,300	‡	‡	‡	‡
2-year	1,300	1,300	3,700 !	1,100	1,100
4-year	3,100	3,000	5,900	2,500	2,800
Non-doctorate-granting	2,800	2,600	5,900	2,600	2,700
Doctorate-granting	3,300	3,200	5,900	2,400	2,800
Private nonprofit					
Less-than-4-year	2,900 !	1,900 !	5,500	‡	‡
4-year	3,500	3,200	7,200	2,400	2,100
Non-doctorate-granting	3,200	3,000	7,500	2,400	1,900
Doctorate-granting	3,800	3,600	6,700	2,400	2,300
For-profit	2,600	2,400	8,000	‡	1,800
More than one institution	2,700	2,500	7,400	1,900	2,100
Attendance status					
Full-time/full-year	3,200	3,000	6,800	2,300	2,600
Full-time/part-year	1,900	1,800	3,200	1,300	1,700
Part-time/full-year	1,700	1,600	4,900	1,200	1,500
Part-time/part-year	1,200	1,100	‡	‡	1,000
Housing ²					
On campus	3,600	3,400	7,400	2,300	2,700
Off campus	2,400	2,300	4,900	1,800	2,100
Living with parents	2,100	2,100	6,600	1,700	1,800
Price of attendance ²					
Less than \$6,000	900	800	‡	‡	900
\$6,000–10,999	1,600	1,500	2,500	1,100	1,700
\$11,000–18,999	2,500	2,400	4,700	1,900	2,500
\$19,000 or more	3,900	3,600	7,200	2,500	2,800
Gender					
Male	2,700	2,500	6,300	2,100	2,200
Female	2,600	2,500	6,000	1,700	2,100

Table 3.3-B. Average amount of state aid received by undergraduates who received a specific type of state aid, by type of aid and selected institutional and student characteristics: 2007–08

—Continued

	Any	Any		State	State
Institutional and	state	state	State	work-	merit-only
student characteristics	aid¹	grants	loans	study	grants
Race/ethnicity ³					
White	\$2,600	\$2,400	\$6,700	\$2,100	\$2,200
Black	2,400	2,300	4,000	1,600	1,800
Hispanic	2,500	2,400	5,000	1,600	2,100
Asian	3,500	3,400	‡	1,800	2,400
American Indian	2,200	2,000	‡	‡	‡
Pacific Islander	2,700	2,800	‡	‡	‡
Two or more races	2,900	2,800	‡	‡	2,300
Other	2,400	2,500	‡	‡	‡
Age as of 12/31/07					
18 years or younger	2,800	2,600	8,100	2,000	2,100
19–23 years	2,900	2,800	6,500	1,900	2,300
24–29 years	2,000	1,800	4,000	1,900	1,300
30-39 years	1,800	1,700	2,800	1,300 !	1,300
40 years or older	1,800	1,600	‡	‡	1,100
Dependency status					
Dependent	3,000	2,800	6,800	2,000	2,300
Independent	1,900	1,800	3,600	1,600	1,300
Unmarried, no dependents	2,200	2,000	4,000	1,700	1,400
Married, no dependents	2,000	1,800	‡	‡	1,300
Single parent	1,800	1,700	2,400	1,800	1,200
Married parent	1,800	1,600	3,400	1,500 !	1,300
Dependency and income in 2006					
Dependent					
Less than \$20,000	2,900	2,800	5,300	1,900	2,000
\$20,000–39,999	3,000	3,000	5,300	1,700	2,100
\$40,000–59,999	3,000	2,800	5,600	2,200	2,100
\$60,000–79,999	2,900	2,600	7,000	2,200	2,400
\$80,000–99,999	2,700	2,400	6,900	1,900	2,400
\$100,000 or more	3,000	2,800	8,500	2,300	2,600
Independent					
Less than \$10,000	2,100	2,000	2,800	1,700	1,400
\$10,000—19,999	1,900	1,800	3,700	1,700	1,100
\$20,000–29,999	1,900	1,700	‡	‡	1,300
\$30,000–49,999	1,800	1,600	‡	‡	1,300
\$50,000 or more	1,900	1,600	‡	‡	1,200
Income group ⁴					
Lowest 25 percent	2,700	2,600	4,700	1,800	1,900
Middle 50 percent	2,500	2,400	6,000	2,000	2,100
Highest 25 percent	2,800	2,500	8,000	1,800	2,400

Table 3.3-B. Average amount of state aid received by undergraduates who received a specific type of state aid, by type of aid and selected institutional and student characteristics: 2007–08

—Continued

Institutional and student characteristics	Any state aid¹	Any state grants	State loans	State work- study	State merit-only grants
Aid status					
No aid	†	†	†	†	†
Received aid	\$2,600	\$2,500	\$6,100	\$1,900	\$2,100
Grant status					
No grant	3,600	†	7,100	1,600	†
Received grant	2,600	2,500	5,800	1,900	2,100
Loan status ⁵					
No loan	2,200	2,200	†	1,400	2,100
Received loan	3,000	2,700	6,100	2,300	2,300

[†] Not applicable.

NOTE: Average aid amounts are calculated only for students receiving a particular type of aid. Those not receiving a specific type of aid (i.e., zero values) are not included in the average for that aid. Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. For students attending more than one institution, includes aid received at the NPSAS institution, but may not include aid received at other institutions. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[!] Interpret data with caution (estimates are unstable).

[‡] Reporting standards not met.

¹ Includes all state need- and merit-based grants, scholarships, loans, and work-study assistance.

² Excludes students attending more than one institution.

³ Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

⁴ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁵ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S3.3-B. Standard errors for table 3.3-B: Average amount of state aid received by undergraduates who received a specific type of state aid, by type of aid and selected institutional and student characteristics: 2007–08

Institutional and student characteristics	Any state aid	Any state grants	State Ioans	State work- study	State merit-only grants
		<u> </u>			
Total	\$40	\$40	\$340	\$190	\$70
Institution type					
Public					
Less-than-2-year	540	†	†	†	†
2-year	30	30	1,370	220	30
4-year	50	50	380	120	70
Non-doctorate-granting	90	90	1,560	270	180
Doctorate-granting	50	40	250	140	60
Private nonprofit					
Less-than-4-year	910	790	970	†	†
4-year	100	100	570	110	120
Non-doctorate-granting	120	120	940	150	160
Doctorate-granting	150	140	760	140	200
For-profit	170	170	1,890	†	400
More than one institution	80	60	780	240	100
Attendance status					
Full-time/full-year	40	40	360	100	80
Full-time/part-year	60	50	620	170	120
Part-time/full-year	30	30	850	310	60
Part-time/part-year	50	50	†	†	60
Housing					
On campus	70	70	540	120	100
Off campus	50	40	280	220	90
Living with parents	40	40	1,030	320	60
Price of attendance					
Less than \$6,000	40	40	†	†	40
\$6,000–10,999	40	40	460	310	70
\$11,000–18,999	40	30	440	130	70
\$19,000 or more	60	60	370	80	120
Gender					
Male	40	40	610	230	70
Female	40	50	290	190	80

Table S3.3-B. Standard errors for table 3.3-B: Average amount of state aid received by undergraduates who received a specific type of state aid, by type of aid and selected institutional and student characteristics: 2007–08—Continued

	Any	Any		State	State
Institutional and	state	state	State	work-	merit-only
student characteristics	aid	grants	loans	study	grants
Race/ethnicity					
White	\$40	\$40	\$420	\$150	\$80
Black	60	50	640	300	80
Hispanic	70	70	840	290	90
Asian	120	110	†	340	180
American Indian	360	410	†	†	t
Pacific Islander	290	300	†	†	†
Two or more races	130	120	†	†	180
Other	440	460	†	†	†
Age as of 12/31/07					
18 years or younger	70	60	1,160	320	90
19–23 years	40	50	300	180	80
24–29 years	50	50	840	210	70
30–39 years	70	60	810	490	90
40 years or older	100	80	†	†	50
Dependency status					
Dependent	40	50	350	160	70
Independent	40	40	490	300	50
Unmarried, no dependents	70	50	870	250	70
Married, no dependents	100	90	†	†	80
Single parent	60	50	570	360	70
Married parent	70	70	760	600	100
Dependency and income in 2006					
Dependent					
Less than \$20,000	60	60	590	260	130
\$20,000–39,999	50	50	590	310	90
\$40,000–59,999	80	80	420	130	100
\$60,000–79,999	80	70	790	290	90
\$80,000–99,999	130	110	910	150	110
\$100,000 or more	100	90	750	210	100
Independent					
Less than \$10,000	60	60	640	240	100
\$10,000–19,999	60	60	870	430	70
\$20,000–29,999	70	60	†	†	90
\$30,000–49,999	60	70	†	†	90
\$50,000 or more	140	120	†	†	60
Income group					
Lowest 25 percent	40	40	410	210	80
Middle 50 percent	50	40	330	180	80
Highest 25 percent	80	80	770	280	90

Table S3.3-B. Standard errors for table 3.3-B: Average amount of state aid received by undergraduates who received a specific type of state aid, by type of aid and selected institutional and student characteristics: 2007–08—Continued

Institutional and student characteristics	Any state aid	Any state grants	State Ioans	State work- study	State merit-only grants
Aid status					
No aid	†	†	†	†	†
Received aid	\$40	\$40	\$340	\$190	\$70
Grant status					
No grant	300	†	660	410	†
Received grant	40	40	300	170	70
Loan status					
No loan	50	60	†	270	70
Received loan	40	30	340	90	80

[†] Not applicable.

Table 3.3-C. Percentage of undergraduates receiving any state aid, by type of institution and selected student characteristics: 2007–08

		Public	4-year	Priv nonprof			
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions ¹
Total	12.8	23.0	24.8	28.2	22.0	8.1	17.5
Attendance status							
Full-time/full-year	27.0	34.7	29.9	35.7	27.0	9.5	27.7
Full-time/part-year	15.8	16.3	21.6	21.9	11.8	6.7	15.0
Part-time/full-year	13.0	15.1	17.1	22.9	19.0	10.0	13.7
Part-time/part-year	5.4	5.8	10.2	8.1	6.4	6.1	6.1
Housing ²							
On campus	19.1	31.5	28.5	30.3	21.6	12.1	27.5
Off campus	11.9	19.1	22.0	23.3	18.1	7.0	15.0
Living with parents	14.0	24.4	27.7	40.1	37.6	12.0	18.2
Price of attendance ²							
Less than \$6,000	5.9	6.1	5.3	‡	‡	3.8 !	5.7
\$6,000–10,999	16.3	16.9	20.1	10.7	5.7	8.1	15.9
\$11,000–18,999	25.8	30.0	28.9	26.9	15.8	10.0	25.7
\$19,000 or more	23.4	32.5	25.4	34.1	25.5	7.1	23.7
Gender							
Male	10.2	20.8	21.9	24.5	20.2	8.0	15.6
Female	14.7	24.9	27.3	30.9	23.6	8.1	19.0
Race/ethnicity ³							
White	12.2	22.7	22.2	27.6	19.7	7.0	16.9
Black	18.1	25.9	29.2	30.3	27.9	7.3	19.8
Hispanic	12.1	22.3	35.1	31.2	33.3	11.9	18.9
Asian	8.8	21.3	27.9	20.7	18.2	8.7	15.7
American Indian	13.5	21.7	21.6	23.7 !	‡	‡	16.8
Pacific Islander	7.1 !	26.4	28.7	‡	‡	‡	12.9
Two or more races	11.5	25.7	28.6	34.4	27.7	3.7	18.4
Other	10.5 !	20.0	30.0 !	‡	‡	‡	16.2
Age as of 12/31/07							
18 years or younger	19.6	37.6	35.5	33.5	26.3	16.2	27.1
19–23 years	14.2	25.4	26.1	31.1	22.6	8.9	20.3
24–29 years	10.5	14.6	17.0	20.8	21.2	7.5	12.4
30–39 years	11.6	13.6	13.8	23.8	18.5	6.2	12.1
40 years or older	8.6	14.5	11.3	22.6	11.9	7.2	10.2
Dependency status							
Dependent	15.0	28.2	27.5	31.6	23.4	10.7	22.0
Independent	11.1	15.1	17.1	23.1	18.3	7.3	12.5
Unmarried, no dependents	9.6	14.3	17.8	23.3	23.1	7.3	12.5
Married, no dependents	6.5	13.0	13.3	15.8	10.6	2.7	8.4
Single parent Married parent	15.8 10.1	19.5 12.8	23.4 12.3	34.8 17.2	21.6 13.2	9.2 5.3	16.3 10.4

Table 3.3-C. Percentage of undergraduates receiving any state aid, by type of institution and selected student characteristics: 2007–08—Continued

		Public	Public 4-year		Private nonprofit 4-year		
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions ¹
Dependency and income in 2006							
Dependent							
Less than \$20,000	23.4	44.1	47.8	39.4	47.0	14.5	32.3
\$20,000-39,999	23.8	44.8	50.3	54.2	45.2	12.4	34.6
\$40,000-59,999	14.3	34.0	40.5	39.9	38.6	14.1	26.5
\$60,000-79,999	9.9	22.6	24.7	33.1	28.9	4.4 !	18.9
\$80,000-99,999	9.0	18.7	16.8	29.4	20.3	‡	15.7
\$100,000 or more	6.9	14.7	15.0	17.4	9.5	2.1 !	12.0
Independent							
Less than \$10,000	16.8	22.2	26.0	29.2	27.7	10.3	18.4
\$10,000-19,999	16.2	20.4	21.7	34.9	24.3	8.3	16.9
\$20,000–29,999	13.4	15.3	16.4	26.0	26.4	6.3	13.7
\$30,000-49,999	10.0	11.7	12.0	24.9	19.4	5.3	10.8
\$50,000 or more	3.8	4.8	3.8	10.5	4.8	2.8 !	4.3
Income group ⁴							
Lowest 25 percent	20.7	35.6	40.6	41.1	39.8	11.2	26.7
Middle 50 percent	12.3	22.0	24.6	31.5	26.9	7.1	17.4
Highest 25 percent	4.7	10.0	12.5	14.4	8.1	2.8	8.2
Aid status							
No aid	†	†	†	†	†	†	†
Received aid	26.8	32.8	34.4	32.3	27.0	8.4	26.7
Grant status							
No grant	0.9	0.8	0.8	1.4 !	0.3	0.8	0.9
Received grant	30.9	43.1	45.9	36.5	31.1	11.1	33.1
Loan status ⁵							
No loan	11.1	17.0	18.6	13.5	10.3	13.0	13.0
Received loan	23.7	30.8	31.5	37.6	31.1	7.6	24.8

[†] Not applicable.

NOTE: Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[!] Interpret data with caution (estimates are unstable).

[‡] Reporting standards not met.

¹ All institutions also include those attending public less-than-2-year, private nonprofit less-than-4-year, and multiple institutions, in addition to the institutions listed separately in this table.

² Excludes those attending more than one institution.

³ Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

⁴ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁵ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S3.3-C. Standard errors for table 3.3-C: Percentage of undergraduates receiving any state aid, by type of institution and selected student characteristics: 2007–08

		Public	: 4-year		/ate it 4-year		
		Non-		Non-			All
	Public		Doctorate-	doctorate-	Doctorate-		insti-
Student characteristics	2-year	granting	granting	granting	granting	For-profit	tutions
Total	0.68	0.72	0.63	1.25	0.99	0.64	0.35
Attendance status							
Full-time/full-year	0.88	1.12	0.81	1.96	1.12	1.16	0.43
Full-time/part-year	1.26	1.64	1.21	2.10	1.45	0.80	0.59
Part-time/full-year	0.65	1.02	1.08	2.15	1.88	1.10	0.39
Part-time/part-year	0.69	0.64	0.83	1.39	1.06	1.11	0.49
Housing							
On campus	2.45	1.95	0.94	1.94	1.63	2.89	0.87
Off campus	0.67	0.66	0.63	1.69	0.99	0.68	0.42
Living with parents	0.80	1.18	1.46	2.82	2.09	1.25	0.72
Price of attendance							
Less than \$6,000	0.74	0.93	0.76	†	†	1.58	0.65
\$6,000-10,999	0.78	1.35	1.32	1.66	1.43	1.23	0.61
\$11,000–18,999	0.79	1.01	0.98	1.85	2.98	0.94	0.48
\$19,000 or more	3.24	2.41	0.67	1.77	1.22	0.96	0.61
Gender							
Male	0.83	0.94	0.76	1.60	1.33	0.84	0.52
Female	0.60	0.77	0.68	1.37	1.05	0.76	0.29
Race/ethnicity							
White	0.87	0.89	0.64	1.54	1.10	0.66	0.48
Black	1.11	1.39	1.40	3.11	2.61	0.92	0.49
Hispanic	0.70	1.21	1.41	2.25	2.34	1.68	0.51
Asian	1.02	1.86	1.80	4.52	2.27	2.15	0.70
American Indian	2.77	6.12	4.40	10.84	†	†	2.22
Pacific Islander	2.26	7.57	7.04	†	†	t	1.74
Two or more races	1.17	3.00	2.56	5.73	4.29	1.00	0.87
Other	3.95	8.93	6.45	†	†	†	2.50
Age as of 12/31/07							
18 years or younger	1.14	2.09	1.90	2.31	1.98	2.35	0.77
19–23 years	0.81	1.03	0.73	1.68	1.26	0.72	0.43
24-29 years	0.78	1.11	1.06	1.47	1.97	1.07	0.43
30–39 years	0.91	1.15		3.40	2.40	0.80	0.51
40 years or older	0.79	1.86	1.28	3.98	1.74	1.23	0.59
Dependency status	0.00		2.22	4.00	4.00	4.40	o
Dependent	0.82	1.10		1.68	1.23	1.19	0.47
Independent	0.68	0.68		1.92	1.19	0.71	0.36
Unmarried, no dependents	0.86	1.03		1.81	1.68	0.78	0.42
Married, no dependents Single parent	0.88 0.94	1.58 1.52		2.10 3.69	2.42 1.94	0.60 1.17	0.52 0.57
Married parent	0.69	1.13		2.83	2.14	0.87	0.57

Table S3.3-C. Standard errors for table 3.3-C: Percentage of undergraduates receiving any state aid, by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year		vate it 4-year		
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions
Dependency and income in 2006							
Dependent							
Less than \$20,000	1.37	2.15	2.07	4.29	3.02	2.04	0.82
\$20,000-39,999	1.06	2.06	1.40	2.63	2.91	1.78	0.64
\$40,000-59,999	1.24	1.73	1.55	3.48	2.37	2.20	0.75
\$60,000-79,999	1.25	1.76	1.28	3.11	2.49	1.87	0.83
\$80,000-99,999	1.09	1.71	1.07	2.80	2.42	t	0.68
\$100,000 or more	0.85	1.48	0.80	2.57	1.43	0.71	0.52
Independent							
Less than \$10,000	0.88	1.81	1.22	2.67	2.86	1.48	0.59
\$10,000-19,999	1.11	1.66	1.50	2.90	2.75	1.15	0.65
\$20,000-29,999	1.12	1.74	1.61	2.85	3.39	0.81	0.62
\$30,000-49,999	0.94	1.60	1.37	3.75	2.29	0.83	0.61
\$50,000 or more	0.49	0.93	0.74	2.68	1.19	0.87	0.39
Income group							
Lowest 25 percent	0.76	1.26	0.93	1.98	1.97	1.10	0.41
Middle 50 percent	0.87	0.87	0.77	1.53	1.07	0.61	0.49
Highest 25 percent	0.47	1.01	0.73	1.83	1.17	0.72	0.35
Aid status							
No aid	†	†	†	†	†	†	†
Received aid	0.63	0.85	0.73	1.40	1.02	0.66	0.32
Grant status							
No grant	0.19	0.21	0.16	0.50	0.06	0.20	0.11
Received grant	0.69	0.97	0.94	1.60	1.17	0.90	0.40
Loan status							
No loan	0.82	0.91	0.86	1.48	1.06	1.79	0.55
Received loan	1.23	0.90	0.65	1.60	1.14	0.65	0.36

[†] Not applicable.

Table 3.3-D. Average amount of state aid received by undergraduates who received any state aid, by type of institution and selected student characteristics: 2007–08

		Public	: 4-year	Priv nonprof			
Student characteristics	Public		Doctorate-	Non- doctorate- granting	Doctorate-	Ear profit	All insti- tutions ¹
	2-year	granting	granting		granting	For-profit	
Total	\$1,300	\$2,800	\$3,300	\$3,200	\$3,800	\$2,600	\$2,600
Attendance status							
Full-time/full-year	1,700	3,100	3,700	3,600	4,100	3,000	3,200
Full-time/part-year	1,200	1,900	2,500	2,100	2,400	2,700	1,900
Part-time/full-year	1,200	1,800	2,500	2,300	2,500	2,400	1,700
Part-time/part-year	700	1,500	2,200	1,600	1,400	1,600	1,200
Housing ²							
On campus	1,100	3,200	3,600	3,800	4,100	3,200	3,600
Off campus	1,300	2,600	3,300	2,900	3,300	2,600	2,400
Living with parents	1,400	2,600	3,000	2,500	3,600	2,700	2,100
Price of attendance ²							
Less than \$6,000	800	900	1,500	‡	‡	‡	900
\$6,000-10,999	1,300	2,100	2,400	1,100	1,000 !	1,300	1,600
\$11,000–18,999	1,700	2,800	3,100	1,600	1,400	2,300	2,500
\$19,000 or more	1,900	3,800	4,200	3,700	4,100	3,400	3,900
Gender							
Male	1,400	2,800	3,300	3,100	3,600	2,600	2,700
Female	1,300	2,700	3,400	3,300	3,900	2,600	2,600
Race/ethnicity ³							
White	1,300	2,700	3,200	3,300	3,700	3,100	2,600
Black	1,200	3,100	3,100	3,200	3,700	2,300	2,400
Hispanic	1,300	2,300	3,700	2,400	3,600	2,200	2,500
Asian	1,500	3,700	4,300	4,100	4,700	3,700 !	3,500
American Indian	1,300	2,900	‡	‡	‡	‡	2,200
Pacific Islander	‡	‡	‡	‡	‡	‡	2,700
Two or more races	1,400	2,800	3,200	3,600	4,700	2,800	2,900
Other	‡	‡	‡	‡	‡	‡	2,400
Age as of 12/31/07							
18 years or younger	1,500	2,900	3,400	3,400	4,200	2,100	2,800
19–23 years	1,300	2,900	3,500	3,600	4,000	2,600	2,900
24-29 years	1,200	2,200	2,400	2,600	3,200	2,700	2,000
30–39 years	1,200	2,200	2,600	2,400	2,300	2,600	1,800
40 years or older	1,200	1,900	2,300	2,400	2,400	3,000	1,800
Dependency status							
Dependent	1,400	3,000	3,500	3,500	4,000	2,700	3,000
Independent	1,200	2,200	2,600	2,600	2,900	2,600	1,900
Unmarried, no dependents	1,300	2,400	2,700	2,800	3,500	2,700	2,200
Married, no dependents	1,200	2,100	2,600	2,600	2,900	3,600	2,000
Single parent	1,200	2,000	2,500	2,500	2,400	2,400	1,800
Married parent	1,200	2,100	2,300	2,500	2,300	2,800	1,800

Table 3.3-D. Average amount of state aid received by undergraduates who received any state aid, by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year	Priv nonprofi			
		Non-		Non-			All
	Public		Doctorate-	doctorate-			insti-
Student characteristics	2-year	granting	granting	granting	granting	For-profit	tutions ¹
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$1,500	\$2,900	\$3,900	\$4,000	\$4,200	\$2,300	\$2,900
\$20,000–39,999	1,400	3,200	3,600	4,200	4,600	2,900	3,000
\$40,000–59,999	1,400	2,900	3,500	3,600	4,800	2,500	3,000
\$60,000–79,999	1,300	2,800	3,400	3,100	3,500	2,600	2,900
\$80,000–99,999	1,500	2,300	3,200	2,900	3,200	‡	2,700
\$100,000 or more	1,300	3,300	3,300	3,000	3,300	2,800	3,000
Independent							
Less than \$10,000	1,300	2,400	2,800	3,000	3,700	2,400	2,100
\$10,000–19,999	1,200	2,100	2,400	2,600	2,800	2,300	1,900
\$20,000–29,999	1,200	1,900	2,600	2,600	2,600	2,800	1,900
\$30,000–49,999	1,100	2,000	2,400	2,200	2,400	3,100	1,800
\$50,000 or more	1,200	2,300	2,300	2,600	2,200	3,700 !	1,900
Income group ⁴							
Lowest 25 percent	1,400	2,900	3,500	3,800	4,300	2,400	2,700
Middle 50 percent	1,300	2,500	3,200	3,000	3,600	2,800	2,500
Highest 25 percent	1,200	3,200	3,200	3,000	3,000	3,600 !	2,800
Aid status							
No aid	†	†	†	†	†	†	†
Received aid	1,300	2,800	3,300	3,200	3,800	2,600	2,600
Grant status							
No grant	1,800	5,200	5,500	‡	‡	7,400	3,600
Received grant	1,300	2,700	3,300	3,200	3,800	2,500	2,600
Loan status ⁵							
No loan	1,300	2,700	3,400	2,600	3,400	2,400	2,200
Received loan	1,300	2,800	3,300	3,400	3,900	2,700	3,000

[†] Not applicable.

NOTE: Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[!] Interpret data with caution (estimates are unstable).

[‡] Reporting standards not met.

¹ All institutions also include those attending public less-than-2-year, private nonprofit less-than-4-year, and multiple institutions, in addition to the institutions listed separately in this table.

 $^{^{\}rm 2}\,{\rm Excludes}$ those attending more than one institution.

³ Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

⁴ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁵ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S3.3-D. Standard errors for table 3.3-D: Average amount of state aid received by undergraduates who received any state aid, by type of institution and selected student characteristics: 2007–08

		Public	4-year	Priv nonprof			
		Non-		Non-			All
Student characteristics	Public 2-year		Doctorate- granting	doctorate- granting	Doctorate- granting	For-profit	insti- tutions
Total	\$30	\$90	\$50	\$120	\$150	\$170	\$40
	φου	φθΟ	φου	φ120	φ130	\$170	φ40
Attendance status							
Full-time/full-year	40	90	50	140	150	230	40
Full-time/part-year	60	110	110	150	260	300	60
Part-time/full-year	30	110	110	220	260	210	30
Part-time/part-year	30	240	170	210	280	270	50
Housing							
On campus	100	140	90	140	170	700	70
Off campus	40	130	60	190	160	210	50
Living with parents	30	80	80	180	380	210	40
Price of attendance							
Less than \$6,000	40	80	140	†	†	†	40
\$6,000–10,999	30	100	110	140	450	220	40
\$11,000-18,999	50	100	40	120	370	200	40
\$19,000 or more	130	210	90	120	120	240	60
Gender							
Male	40	100	70	140	240	270	40
Female	30	120	60	140	120	180	40
Race/ethnicity							
White	30	80	60	120	130	190	40
Black	50	160	100	270	230	240	60
Hispanic	50	160	150	220	630	300	70
Asian	100	330	160	560	320	1,120	120
American Indian	140	440	†	†	†	†	360
Pacific Islander	†	†	†	†	†	†	290
Two or more races	100	310	230	380	420	490	130
Other	†	†	†	†	†	†	440
Age as of 12/31/07							
18 years or younger	60	150	110	220	270	250	70
19–23 years	30	90	60	150	150	210	40
24–29 years	50	140	110	180	410	310	50
30–39 years	60	170	170	290	210	360	70
40 years or older	60	160	220	510	330	270	100
Dependency status							
Dependent	30	90	50	140	150	220	40
Independent	40	100	90	210	220	210	40
Unmarried, no dependents	50	160	130	210	360	290	70
Married, no dependents	100	210	190	420	330	830	100
Single parent	50	130	150	340	250	270	60
Married parent	60	170	180	260	320	250	70

Table S3.3-D. Standard errors for table 3.3-D: Average amount of state aid received by undergraduates who received any state aid, by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year	Private nonprofit 4-year			
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$60	\$140	\$120	\$290	\$370	\$270	\$60
\$20,000-39,999	40	110	90	240	330	370	50
\$40,000–59,999	60	130	90	210	230	270	80
\$60,000–79,999	70	220	150	230	230	460	80
\$80,000–99,999	210	160	160	330	380	†	130
\$100,000 or more	70	360	120	240	290	630	100
Independent							
Less than \$10,000	60	150	110	280	480	300	60
\$10,000–19,999	50	170	190	270	310	260	60
\$20,000–29,999	60	200	210	270	320	290	70
\$30,000-49,999	60	210	190	280	190	270	60
\$50,000 or more	100	390	240	460	280	1,440	140
Income group							
Lowest 25 percent	40	90	60	200	340	210	40
Middle 50 percent	30	100	70	110	120	180	50
Highest 25 percent	70	350	120	220	260	1,140	80
Aid status							
No aid	†	†	†	†	†	†	†
Received aid	30	90	50	120	150	170	40
Grant status							
No grant	200	1,130	510	†	†	1,040	300
Received grant	30	80	50	120	150	170	40
Loan status							
No loan	30	130	70	210	290	380	50
Received loan	50	100	60	120	130	200	40

[†] Not applicable.

Table 3.4-A. Percentage of undergraduates receiving institutional aid, by type of aid and selected institutional and student characteristics: 2007–08

Institutional and student characteristics	Any institutional aid ¹	Any institutional grants	Institutional Ioans	Institutional work-study	Institutional merit-only grants
Total	20.8	19.9	0.5	1.5	8.8
Institution type					
Public					
Less-than-2-year	3.9!	3.9 !	#	#	‡
2-year	11.1	10.7	0.1 !	0.4	1.3
4-year	23.6	22.3	0.6	1.6	11.1
Non-doctorate-granting	16.6	15.4	0.1	1.6	8.4
Doctorate-granting	27.4	26.0	0.9	1.6	12.6
Private nonprofit					
Less-than-4-year	10.2 !	9.9 !	0.2 !	0.2 !	‡
4-year	53.1	52.2	1.2	5.2	31.3
Non-doctorate-granting	52.4	51.9	1.1	5.4	32.0
Doctorate-granting	53.8	52.5	1.3	4.9	30.4
For-profit	8.6	7.1	‡	0.3 !	3.2 !
More than one institution	21.5	20.5	0.3	1.5	7.8
Attendance status					
Full-time/full-year	34.0	32.7	0.8	2.8	17.5
Full-time/part-year	16.7	15.6	0.7 !	1.1	7.0
Part-time/full-year	14.6	13.9	0.3	0.6	2.8
Part-time/part-year	7.6	7.2	0.1 !	0.3	1.2
Housing ²					
On campus	49.6	48.2	1.1	5.2	29.4
Off campus	15.3	14.5	0.5	0.8	5.1
Living with parents	15.7	15.0	0.3	0.8	5.2
Price of attendance ²					
Less than \$6,000	7.2	6.9	0.1 !	0.2 !	0.5
\$6,000–10,999	12.2	11.5	0.2	0.7	2.6
\$11,000–18,999	21.4	20.1	0.5	1.4	8.9
\$19,000 or more	43.2	41.9	1.3	3.8	24.1
Gender					
Male	20.2	19.4	0.5	1.5	8.8
Female	21.2	20.3	0.5	1.5	8.7

Table 3.4-A. Percentage of undergraduates receiving institutional aid, by type of aid and selected institutional and student characteristics: 2007–08—Continued

	Any	Any			Institutional
Institutional and	institutional	institutional	Institutional	Institutional	merit-only
student characteristics	aid ¹	grants	loans	work-study	grants
Race/ethnicity ³					
White	20.8	20.0	0.4	1.6	10.7
Black	17.3	16.5	0.4	1.2	5.9
Hispanic	22.4	21.3	0.8	1.0	4.8
Asian	23.8	22.4	0.9 !	1.9	5.8
American Indian	15.9	15.3	‡	0.9 !	7.0
Pacific Islander	23.8	23.3	‡	2.2 !	4.5
Two or more races	25.1	23.7	0.9 !	2.2	8.7
Other	17.4	17.1	‡	‡	3.2
Age as of 12/31/07					
18 years or younger	32.9	32.0	0.8	1.9	17.3
19–23 years	25.9	24.9	0.6	2.2	12.4
24–29 years	13.3	12.4	0.5 !	0.6	2.6
30–39 years	10.5	9.9	0.4!	0.4	1.9
40 years or older	9.4	8.9	0.2 !	0.4	1.7
Dependency status					
Dependent	28.5	27.4	0.6	2.3	14.3
Independent	12.1	11.4	0.4 !	0.5	2.5
Unmarried, no dependents	14.4	13.6	0.5	0.7	2.7
Married, no dependents	11.6	10.9	0.4 !	0.7	3.4
Single parent	11.8	11.3	0.4 !	0.3	2.1
Married parent	9.7	9.1	0.3 !	0.4	2.3
Dependency and income in 2006					
Dependent					
Less than \$20,000	28.9	27.8	0.6	1.5	9.0
\$20,000-39,999	30.7	29.8	0.7	1.7	10.7
\$40,000-59,999	29.6	28.6	0.7	2.0	13.0
\$60,000-79,999	27.9	26.7	0.6	2.2	15.0
\$80,000–99,999	28.8	27.8	0.5	2.7	17.7
\$100,000 or more	26.4	25.2	0.6	3.0	17.5
Independent					
Less than \$10,000	17.9	16.9	0.5	1.1	3.9
\$10,000–19,999	14.1	13.4	0.5 !	0.4	2.6
\$20,000–29,999	12.8	12.1	0.4 !	0.5 !	2.4
\$30,000–49,999	10.8	10.2	‡	0.3	2.4
\$50,000 or more	5.7	5.3	0.2 !	0.3	1.3
Income group ⁴					
Lowest 25 percent	24.2	23.2	0.6	1.3	7.0
Middle 50 percent	21.1	20.2	0.5	1.4	9.1
Highest 25 percent	16.6	15.7	0.4	1.8	9.8

Table 3.4-A. Percentage of undergraduates receiving institutional aid, by type of aid and selected institutional and student characteristics: 2007–08—Continued

Institutional and student characteristics	Any institutional aid ¹	Any institutional grants	Institutional loans	Institutional work-study	Institutional merit-only grants
Aid status					
No aid	†	†	†	†	†
Received aid	31.6	30.3	0.8	2.2	13.3
Grant status					
No grant	1.0	†	0.2	0.7	†
Received grant	39.3	38.4	0.8	2.2	16.9
Loan status ⁵					
No loan	15.9	15.3	†	1.1	6.1
Received loan	28.6	27.2	1.3	2.1	13.0

[†] Not applicable.

NOTE: Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. For students attending more than one institution, includes only institutional aid awarded by the institution in the NPSAS sample. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[#] Rounds to zero.

[!] Interpret data with caution (estimates are unstable).

[‡] Reporting standards not met.

¹ Includes all institutional need- and merit-based grants, scholarships, tuition waivers, loans, and work-study assistance.

² Excludes students attending more than one institution.

³ Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

⁴ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁵ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S3.4-A. Standard errors for table 3.4-A: Percentage of undergraduates receiving institutional aid, by type of aid and selected institutional and student characteristics: 2007–08

Institutional and student characteristics	Any institutional aid	Any institutional grants	Institutional Ioans	Institutional work-study	Institutional merit-only grants
Total	0.60	0.57	0.09	0.07	0.23
Institution type					
Public					
Less-than-2-year	1.83	1.83	†	†	†
2-year	0.90	0.87	0.04	0.05	0.15
4-year	0.41	0.38	0.12	0.13	0.28
Non-doctorate-granting	0.64	0.60	0.02	0.26	0.52
Doctorate-granting	0.61	0.57	0.18	0.14	0.41
Private nonprofit					
Less-than-4-year	4.39	4.41	0.10	0.10	†
4-year	1.24	1.30	0.18	0.37	1.11
Non-doctorate-granting	1.79	1.90	0.26	0.54	1.51
Doctorate-granting	1.67	1.70	0.26	0.50	1.50
For-profit	1.70	1.54	†	0.13	1.48
More than one institution	0.58	0.58	0.05	0.13	0.41
Attendance status					
Full-time/full-year	0.52	0.52	0.11	0.14	0.43
Full-time/part-year	0.93	0.91	0.31	0.15	0.60
Part-time/full-year	0.81	0.77	0.06	0.07	0.16
Part-time/part-year	0.92	0.88	0.04	0.07	0.15
Housing					
On campus	1.00	1.02	0.16	0.31	1.00
Off campus	0.64	0.60	0.14	0.08	0.30
Living with parents	0.84	0.79	0.06	0.08	0.23
Price of attendance					
Less than \$6,000	0.90	0.85	0.06	0.06	0.09
\$6,000–10,999	0.75	0.72	0.05	0.07	0.23
\$11,000–18,999	0.46	0.42	0.10	0.10	0.28
\$19,000 or more	1.00	1.01	0.28	0.25	0.89
Gender					
Male	0.71	0.68	0.09	0.10	0.25
Female	0.71	0.54	0.10	0.09	0.27

Table S3.4-A. Standard errors for table 3.4-A: Percentage of undergraduates receiving institutional aid, by type of aid and selected institutional and student characteristics: 2007–08

—Continued

Institutional and	Any institutional	Any institutional	Institutional	Institutional	Institutional merit-only
student characteristics	aid	grants	loans	work-study	grants
Race/ethnicity					
White	0.57	0.53	0.07	0.10	0.31
Black	0.87	0.86	0.08	0.13	0.48
Hispanic	0.84	0.83	0.20	0.10	0.25
Asian	1.39	1.39	0.26	0.25	0.48
American Indian	2.11	2.09	†	0.43	1.32
Pacific Islander	2.51	2.47	†	1.02	0.93
Two or more races	1.31	1.24	0.43	0.40	0.93
Other	3.20	3.17	†	†	1.10
Age as of 12/31/07					
18 years or younger	0.75	0.74	0.13	0.23	0.60
19–23 years	0.59	0.56	0.07	0.12	0.26
24–29 years	0.79	0.76	0.16	0.09	0.37
30-39 years	0.92	0.91	0.13	0.07	0.29
40 years or older	0.82	0.83	0.09	0.08	0.49
Dependency status					
Dependent	0.59	0.56	0.07	0.12	0.31
Independent	0.73	0.71	0.13	0.05	0.32
Unmarried, no dependents	0.87	0.86	0.14	0.09	0.22
Married, no dependents	0.81	0.79	0.12	0.14	0.28
Single parent	0.85	0.82	0.15	0.08	0.35
Married parent	0.89	0.89	0.12	0.09	0.65
Dependency and income in 2006					
Dependent					
Less than \$20,000	0.97	0.92	0.15	0.21	0.48
\$20,000–39,999	0.77	0.75	0.11	0.15	0.42
\$40,000–59,999	0.90	0.85	0.11	0.20	0.45
\$60,000–79,999	1.12	1.06	0.16	0.23	0.67
\$80,000–99,999	0.85	0.86	0.11	0.30	0.71
\$100,000 or more	0.70	0.69	0.10	0.22	0.59
Independent					
Less than \$10,000	0.91	0.92	0.16	0.14	0.41
\$10,000–19,999	0.87	0.84	0.17	0.08	0.30
\$20,000–29,999	1.06	0.97	0.14	0.16	0.34
\$30,000–49,999	0.94	0.92	†	0.07	0.59
\$50,000 or more	0.49	0.48	0.07	0.08	0.18
Income group					
Lowest 25 percent	0.68	0.66	0.12	0.10	0.25
Middle 50 percent	0.67	0.64	0.10	0.10	0.30
Highest 25 percent	0.54	0.52	0.07	0.13	0.36

Table S3.4-A. Standard errors for table 3.4-A: Percentage of undergraduates receiving institutional aid, by type of aid and selected institutional and student characteristics: 2007–08

—Continued

Institutional and student characteristics	Any institutional aid	Any institutional grants	Institutional Ioans	Institutional work-study	Institutional merit-only grants
Aid status					
No aid	†	†	†	†	†
Received aid	0.48	0.46	0.13	0.09	0.33
Grant status					
No grant	0.11	†	0.07	0.07	†
Received grant	0.54	0.55	0.12	0.10	0.40
Loan status					
No loan	0.82	0.80	†	0.07	0.17
Received loan	0.55	0.52	0.23	0.13	0.48

[†] Not applicable.

Table 3.4-B. Average amount of institutional aid received by undergraduates who received a specific type of institutional aid, by type of aid and selected institutional and student characteristics: 2007–08

Institutional and	Any institutional	Any institutional	Institutional Ioan	Institutional work-study	Institutional merit-only
student characteristics	aid ¹	grant amount	amount	amount	grants
Total	\$5,000	\$5,000	\$3,200	\$2,600	\$5,300
Institution type					
Public					
Less-than-2-year	700	700	‡	‡	‡
2-year	800	700	700	2,700	1,500
4-year	3,700	3,700	3,300 !	2,800	3,900
Non-doctorate-granting	3,200	3,100	‡	2,900	3,200
Doctorate-granting	3,900	3,900	3,400 !	2,700	4,100
Private nonprofit					
Less-than-4-year	2,100	2,100	‡	‡	‡
4-year	9,600	9,500	3,000	2,400	7,400
Non-doctorate-granting	8,900	8,700	3,000	2,200	7,000
Doctorate-granting	10,400	10,400	3,100	2,600	7,800
For-profit	2,100	1,600	4,300	3,500	1,600
More than one institution	3,900	3,800	2,600	2,700	4,700
Attendance status					
Full-time/full-year	6,500	6,400	3,300	2,500	5,900
Full-time/part-year	3,700	3,600	3,600	2,600	3,900
Part-time/full-year	2,100	2,000	2,400	3,300	3,100
Part-time/part-year	1,300	1,200	2,300 !	2,900	2,400
Housing ²					
On campus	8,500	8,400	2,800	2,500	6,700
Off campus	3,300	3,200	3,700	2,700	4,100
Living with parents	2,900	2,800	2,200	2,900	3,900
Price of attendance ²					
Less than \$6,000	500	500	‡	2,100	800
\$6,000–10,999	1,200	1,100	1,200	3,400	1,700
\$11,000–18,999	2,500	2,400	2,200	2,500	2,800
\$19,000 or more	8,500	8,400	4,100	2,500	7,000
Gender					
Male	5,200	5,200	2,800	2,700	5,400
Female	4,900	4,800	3,400	2,500	5,300

Table 3.4-B. Average amount of institutional aid received by undergraduates who received a specific type of institutional aid, by type of aid and selected institutional and student characteristics: 2007–08—Continued

Institutional and	Any institutional	Any institutional	Institutional Ioan	Institutional work-study	Institutional merit-only
student characteristics	aid ¹	grant amount	amount	amount	grants
Race/ethnicity ³					
White	\$5,600	\$5,500	\$3,100	\$2,600	\$5,400
Black	4,400	4,400	2,900	2,900	5,200
Hispanic	3,300	3,200	3,100	2,400	4,800
Asian	5,300	5,300	3,000	2,700	6,500
American Indian	3,400	3,400	‡	‡	3,900
Pacific Islander	3,300	2,900	‡	‡	4,600
Two or more races	5,200	5,000	‡	2,700	5,100
Other	3,500	3,300	‡	‡	‡
Age as of 12/31/07					
18 years or younger	6,000	6,000	2,500	1,900	5,700
19–23 years	5,900	5,800	3,100	2,600	5,700
24–29 years	2,200	2,100	3,800	2,800	2,500
30–39 years	1,700	1,600	3,100	3,400	1,900
40 years or older	1,600	1,500	3,400 !	3,400	1,800
Dependency status					
Dependent	6,100	6,100	2,800	2,500	5,700
Independent	2,200	2,000	3,700	2,900	2,800
Unmarried, no dependents	2,500	2,400	3,200	2,900	3,200
Married, no dependents	2,700	2,600	3,800 !	2,500	3,600
Single parent	1,800	1,600	4,100 !	3,000	2,400
Married parent	1,800	1,700	4,000 !	3,300	1,900
Dependency and income in 2006					
Dependent					
Less than \$20,000	4,300	4,200	3,800 !	2,500	5,000
\$20,000–39,999	5,100	5,000	2,400	2,700	5,300
\$40,000–59,999	5,800	5,700	2,900	2,900	5,300
\$60,000–79,999	6,400	6,400	2,600	2,700	5,400
\$80,000-99,999	7,400	7,300	2,900	2,400	6,200
\$100,000 or more	7,100	7,000	2,800	2,400	6,200
Independent					
Less than \$10,000	2,600	2,500	3,200	2,900	3,500
\$10,000–19,999	2,100	2,000	4,200 !	3,000	2,800
\$20,000-29,999	1,900	1,800	3,000	2,400	2,300
\$30,000–49,999	1,800	1,600	4,300 !	3,600	2,100
\$50,000 or more	2,100	2,000	‡	‡	2,200
Income group ⁴					
Lowest 25 percent	4,000	3,900	3,100	2,600	4,700
Middle 50 percent	5,100	5,100	3,300	2,700	5,200
Highest 25 percent	6,300	6,300	2,800	2,400	6,100

Table 3.4-B. Average amount of institutional aid received by undergraduates who received a specific type of institutional aid, by type of aid and selected institutional and student characteristics: 2007–08—Continued

Institutional and student characteristics	Any institutional aid ¹	Any institutional grant amount	Institutional Ioan amount	Institutional work-study amount	Institutional merit-only grants
Aid status					
No aid	†	†	†	†	†
Received aid	\$5,000	\$5,000	\$3,200	\$2,600	\$5,300
Grant status					
No grant	3,000	†	3,100	3,000	†
Received grant	5,100	5,000	3,200	2,500	5,300
Loan status ⁵					
No loan	3,800	3,800	†	2,700	5,600
Received loan	6,100	6,100	3,200	2,500	5,200

[†] Not applicable.

NOTE: Average aid amounts are calculated only for students receiving a particular type of aid. Those not receiving a specific type of aid (i.e., zero values) are not included in the average for that aid. Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. For students attending more than one institution, includes only institutional aid awarded by the institution in the NPSAS sample. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[!] Interpret data with caution (estimates are unstable).

[‡] Reporting standards not met.

¹ Includes all institutional need- and merit-based grants, scholarships, tuition waivers, loans, and work-study assistance.

 $^{^{\}rm 2}$ Excludes students attending more than one institution.

³ Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

⁴ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁵ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S3.4-B. Standard errors for table 3.4-B: Average amount of institutional aid received by undergraduates who received a specific type of institutional aid, by type of aid and selected institutional and student characteristics: 2007–08

Institutional and student characteristics	Any institutional aid	Any institutional grant amount	Institutional Ioan amount	Institutional work-study amount	Institutional merit-only grants
Total	\$130	\$130	\$470	\$60	\$120
Institution type					
Public					
Less-than-2-year	120	120	†	†	†
2-year	30	30	80	240	100
4-year	80	70	1,050	100	100
Non-doctorate-granting	140	150	†	210	160
Doctorate-granting	100	80	1,100	110	130
Private nonprofit					
Less-than-4-year	600	600	†	†	†
4-year	140	140	230	80	130
Non-doctorate-granting	220	210	230	110	190
Doctorate-granting	180	180	440	110	190
For-profit	390	240	1,190	880	420
More than one institution	260	260	350	190	220
Attendance status					
Full-time/full-year	100	100	480	60	110
Full-time/part-year	300	320	490	260	390
Part-time/full-year	80	80	620	190	160
Part-time/part-year	80	70	930	440	290
Housing					
On campus	150	140	210	90	130
Off campus	110	110	810	120	200
Living with parents	110	110	430	200	160
Price of attendance					
Less than \$6,000	20	20	†	410	100
\$6,000–10,999	40	40	290	260	80
\$11,000–18,999	50	40	310	140	70
\$19,000 or more	140	130	630	80	150
Gender					
Male	140	140	340	90	120
Female	140	140	600	80	140

Table S3.4-B. Standard errors for table 3.4-B: Average amount of institutional aid received by undergraduates who received a specific type of institutional aid, by type of aid and selected institutional and student characteristics: 2007–08—Continued

Institutional and	Any institutional	Any institutional	Institutional Ioan	Institutional work-study	Institutional merit-only
student characteristics	aid	grant amount	amount	amount	grants
Race/ethnicity					
White	\$110	\$110	\$430	\$70	\$110
Black	250	260	640	230	420
Hispanic	200	190	840	180	240
Asian	370	380	570	240	650
American Indian	510	520	†	†	830
Pacific Islander	450	410	†	†	680
Two or more races	400	390	†	310	350
Other	940	950	†	†	†
Age as of 12/31/07					
18 years or younger	160	170	380	140	190
19–23 years	120	120	400	70	100
24–29 years	130	120	850	240	280
30–39 years	120	110	710	490	190
40 years or older	130	120	1,220	650	320
Dependency status					
Dependent	110	110	270	70	100
Independent	120	110	880	200	260
Unmarried, no dependents	150	150	670	230	230
Married, no dependents	190	190	1,160	350	340
Single parent	140	120	1,530	530	330
Married parent	150	120	1,250	530	320
Dependency and income in 2006					
Dependent					
Less than \$20,000	180	180	1,520	230	200
\$20,000–39,999	180	180	520	190	160
\$40,000–59,999	230	240	350	200	200
\$60,000–79,999	200	210	310	190	170
\$80,000-99,999	250	250	550	190	250
\$100,000 or more	140	150	410	90	150
Independent					
Less than \$10,000	150	140	570	260	300
\$10,000–19,999	160	140	1,380	460	290
\$20,000–29,999	160	160	830	440	310
\$30,000–49,999 \$50,000 or more	130 170	120 160	1,470 †	490 †	340 410
Income group					
Lowest 25 percent	130	130	620	130	140
Middle 50 percent	150	160	510	110	150
Highest 25 percent	150	150	390	90	150

Table S3.4-B. Standard errors for table 3.4-B: Average amount of institutional aid received by undergraduates who received a specific type of institutional aid, by type of aid and selected institutional and student characteristics: 2007–08—Continued

Institutional and student characteristics	Any institutional aid	Any institutional grant amount	Institutional Ioan amount	Institutional work-study amount	Institutional merit-only grants
Aid status					
No aid	†	†	†	†	†
Received aid	\$130	\$130	\$470	\$60	\$120
Grant status					
No grant	200	†	750	130	†
Received grant	130	130	440	70	120
Loan status					
No loan	140	140	†	90	140
Received loan	120	120	470	80	150

[†] Not applicable.

Table 3.4-C. Percentage of undergraduates receiving institutional aid, by type of institution and selected student characteristics: 2007–08

		Public	: 4-year	Priv nonprofi			
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions ¹
Total	11.1	16.6	27.4	52.4	53.8	8.6	20.8
Attendance status							
Full-time/full-year	18.1	25.5	33.9	70.0	65.1	8.2	34.0
Full-time/part-year	9.6	11.8	22.7	42.9	43.3	9.5	16.7
Part-time/full-year	13.4	9.8	17.1	24.2	31.9	10.3	14.6
Part-time/part-year	6.7	3.9	10.4	14.0	15.9	6.2	7.6
Housing ²							
On campus	22.7	27.0	36.5	79.3	65.3	21.8 !	49.6
Off campus	10.3	13.8	24.1	27.4	38.1	8.6	15.3
Living with parents	12.0	13.8	21.8	52.5	57.3	8.0	15.7
Price of attendance ²							
Less than \$6,000	7.9	2.7	5.6	5.0 !	4.6 !	5.3 !	7.2
\$6,000-10,999	13.5	8.5	13.4	11.3	15.1	‡	12.2
\$11,000–18,999	16.3	21.8	27.5	29.7	22.6	9.3	21.4
\$19,000 or more	8.1	32.5	38.6	70.4	64.8	9.7	43.2
Gender							
Male	10.0	16.3	26.0	51.6	51.4	8.7	20.2
Female	12.0	16.9	28.6	53.0	55.8	8.5	21.2
Race/ethnicity ³							
White	8.2	17.0	26.1	58.7	55.2	8.1	20.8
Black	10.5	17.8	27.5	34.9	45.9	9.4	17.3
Hispanic	21.1	13.7	32.6	35.9	51.4	8.2	22.4
Asian	13.9	15.4	31.2	50.1	54.6	9.8 !	23.8
American Indian	11.0	14.9	23.5	55.9	‡	13.1 !	15.9
Pacific Islander	20.6	28.3	22.7	59.2	53.7 !	‡	23.8
Two or more races	13.8	21.6	34.6	60.4	61.3	11.8 !	25.1
Other	16.7 !	‡	20.4	27.4 !	‡	‡	17.4
Age as of 12/31/07							
18 years or younger	15.3	26.6	39.1	78.5	69.8	11.1	32.9
19–23 years	11.9	19.1	28.7	70.7	61.7	8.4	25.9
24–29 years	11.0	10.0		24.3	29.5	9.0	13.3
30-39 years	9.2	8.0		15.4	17.4	7.8	10.5
40 years or older	8.3	8.5	13.8	12.0	16.0	8.4 !	9.4
Dependency status							
Dependent	12.6	21.1	30.6	73.8	63.9	9.8	28.5
Independent	10.0	9.7	18.4	20.9	25.1	8.2	12.1
Unmarried, no dependents	11.6	10.8	19.1	27.8	28.7	8.8	14.4
Married, no dependents	8.0	11.4	17.0	21.4	27.8	8.9 !	11.6
Single parent	11.1	9.5		19.7	25.6	6.7	11.8
Married parent	7.9	7.0	15.2	14.6	18.0	9.8 !	9.7

Table 3.4-C. Percentage of undergraduates receiving institutional aid, by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year		vate it 4-year		
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions ¹
Dependency and income in 2006							
Dependent							
Less than \$20,000	20.0	22.4	39.7	63.9	64.1	8.4	28.9
\$20,000-39,999	17.8	26.0	40.4	76.4	71.1	9.9	30.7
\$40,000-59,999	13.7	22.5	38.1	80.3	74.0	12.8 !	29.6
\$60,000-79,999	9.5	21.6	32.7	77.2	75.4	8.6 !	27.9
\$80,000-99,999	7.7	20.3	27.2	82.0	71.4	11.0 !	28.8
\$100,000 or more	5.1	16.4	22.5	68.0	53.5	8.1	26.4
Independent							
Less than \$10,000	16.0	14.3	23.9	36.6	41.6	7.7	17.9
\$10,000-19,999	12.7	12.3	21.4	25.1	32.8	6.9	14.1
\$20,000-29,999	11.2	8.7	19.4	24.4	25.1	8.4	12.8
\$30,000-49,999	9.2	6.8	15.3	16.7	18.8	11.1 !	10.8
\$50,000 or more	4.2	5.1	9.2	9.5	13.3	6.6 !	5.7
Income group ⁴							
Lowest 25 percent	17.6	19.8	34.4	55.7	58.8	7.8	24.2
Middle 50 percent	10.8	17.3	28.6	56.3	60.6	9.6	21.1
Highest 25 percent	4.3	10.9	19.7	44.2	43.8	6.8 !	16.6
Aid status							
No aid	†	†	†	†	†	†	†
Received aid	23.3	23.6	38.0	60.0	65.8	8.9	31.6
Grant status							
No grant	0.4	1.2	1.7	0.7 !	3.1	‡	1.0
Received grant	27.4	30.6	50.1	68.5	74.8	11.6	39.3
Loan status ⁵							
No loan	11.2	13.6	22.2	37.5	38.5	7.3 !	15.9
Received loan	10.5	20.5	33.0	61.9	65.6	8.7	28.6

[†] Not applicable.

NOTE: Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[!] Interpret data with caution (estimates are unstable).

[‡] Reporting standards not met.

¹ All institutions also include those attending public less-than-2-year, private nonprofit less-than-4-year, and multiple institutions, in addition to the institutions listed separately in this table.

² Excludes those attending more than one institution.

³ Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

⁴ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁵ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S3.4-C. Standard errors for table 3.4-C: Percentage of undergraduates receiving institutional aid, by type of institution and selected student characteristics: 2007–08

		Public	: 4-year		vate fit 4-year		
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions
Total	0.90	0.64	0.61	1.79	1.67	1.70	0.60
Attendance status							
Full-time/full-year	0.96	1.04	0.77	2.31	2.21	2.01	0.52
Full-time/part-year	1.67	1.10	0.92	2.69	4.06	2.76	0.93
Part-time/full-year	1.09	0.76	1.03	1.96	2.07	1.93	0.81
Part-time/part-year	0.96	0.56	0.91	1.98	1.67	1.45	0.92
Housing							
On campus	3.94	1.57	1.00	2.05	2.18	7.28	1.00
Off campus	0.89	0.74	0.64	1.97	1.55	1.93	0.64
Living with parents	1.07	0.79	1.19	2.65	4.38	1.34	0.84
Price of attendance							
Less than \$6,000	1.06	0.56	0.98	1.79	1.72	2.44	0.90
\$6,000-10,999	0.83	0.82	1.00	2.06	2.45	†	0.75
\$11,000-18,999	1.03	0.84	0.68	2.00	3.09	1.50	0.46
\$19,000 or more	1.67	2.43	0.98	2.47	1.82	2.26	1.00
Gender							
Male	0.98	0.71	0.80	2.45	2.18	1.94	0.71
Female	0.88	0.81	0.64	1.86	1.70	1.73	0.56
Race/ethnicity							
White	0.74	0.80	0.62	2.38	1.79	1.83	0.57
Black	1.22	1.46	1.25	3.02	2.92	2.40	0.87
Hispanic	1.64	1.22	1.70	3.01	6.75	1.67	0.84
Asian	1.93	1.92	2.06	5.46	3.96	3.78	1.39
American Indian	2.42	4.30	5.09	13.81	†	5.60	2.11
Pacific Islander	3.73	8.37	5.15	9.70	17.14	†	2.51
Two or more races	1.86	3.69	2.91	7.07	5.57	5.88	1.31
Other	6.21	†	5.96	8.74	†	†	3.20
Age as of 12/31/07							
18 years or younger	1.29	1.51	1.56	3.56	2.85	1.93	0.75
19–23 years	0.94	0.79	0.72	2.12	2.12	1.39	0.59
24–29 years	1.17	0.77	1.02	2.14	2.44	2.41	0.79
30-39 years	1.14	1.37	1.26	2.21	2.43	2.08	0.92
40 years or older	0.88	1.41	1.75	1.87	2.61	2.95	0.82
Dependency status							
Dependent	1.01	0.83	0.72	2.16	2.08	1.78	0.59
Independent	0.92	0.68	0.84	1.26	1.51	1.89	0.73
Unmarried, no dependents	1.29	0.83	1.10	2.16	2.66	1.64	0.87
Married, no dependents	1.16	1.45	1.75	2.95	3.11	3.09	0.81
Single parent	1.30	1.42	1.56	2.27	3.04	1.41	0.85
Married parent	0.76	1.08	1.33	1.91	2.39	4.61	0.89

Table S3.4-C. Standard errors for table 3.4-C: Percentage of undergraduates receiving institutional aid, by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year		vate fit 4-year		
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions
Dependency and income in 2006							
Dependent							
Less than \$20,000	1.39	1.87	1.79	4.03	5.52	1.64	0.97
\$20,000-39,999	1.34	1.41	1.48	2.81	4.56	2.73	0.77
\$40,000-59,999	1.67	1.59	1.30	4.48	3.27	3.86	0.90
\$60,000-79,999	1.28	1.83	1.53	3.37	3.11	3.61	1.12
\$80,000-99,999	1.10	1.60	1.29	3.66	2.88	5.02	0.85
\$100,000 or more	0.82	1.39	0.91	3.02	2.63	1.97	0.70
Independent							
Less than \$10,000	1.48	1.28	1.21	3.19	4.02	1.60	0.91
\$10,000-19,999	1.19	1.25	1.76	2.95	4.10	1.68	0.87
\$20,000-29,999	1.49	1.51	1.70	2.40	3.75	1.81	1.06
\$30,000-49,999	1.12	1.03	1.91	2.42	2.31	3.91	0.94
\$50,000 or more	0.52	1.15	1.28	1.71	1.98	2.35	0.49
Income group							
Lowest 25 percent	1.06	0.94	0.93	2.63	4.29	1.29	0.68
Middle 50 percent	1.05	0.86	0.72	2.10	2.04	2.19	0.67
Highest 25 percent	0.47	0.99	0.87	2.69	1.88	2.10	0.54
Aid status							
No aid	†	†	†	†	†	†	†
Received aid	0.88	0.92	0.65	2.09	1.93	1.76	0.48
Grant status							
No grant	0.09	0.18	0.25	0.26	0.51	†	0.11
Received grant	1.04	1.20	0.87	2.03	1.97	2.15	0.54
Loan status							
No loan	1.01	0.68	0.92	1.96	2.08	2.57	0.82
Received loan	0.69	0.90	0.71	2.34	2.11	1.86	0.55

[†] Not applicable.

Table 3.4-D. Average amount of institutional aid received by undergraduates who received institutional aid, by type of institution and selected student characteristics: 2007–08

		Public	: 4-year	Priv nonprof	/ate it 4-year		
Student characteristics	Public 2-year	Non-	Doctorate- granting	Non-	Doctorate- granting	For-profit	All insti- tutions ¹
Total	\$800	\$3,200	\$3,900	\$8,900	\$10,400	\$2,100	\$5,000
Attendance status							
Full-time/full-year	1,300	3,400	4,200	10,000	11,200	2,200	6,500
Full-time/part-year	800	2,100	3,300	5,200	8,300	2,500	3,700
Part-time/full-year	700	2,600	3,200	5,900	6,500	1,800	2,100
Part-time/part-year	500	1,600	2,500	3,100	4,200	1,100	1,300
Housing ²							
On campus	2,300	3,700	4,600	10,400	11,800	2,000 !	8,500
Off campus	700	3,000	3,700	6,000	8,400	2,100	3,300
Living with parents	900	2,700	2,800	6,500	8,300	2,000	2,900
Price of attendance ²							
Less than \$6,000	400	1,000	1,400	‡	‡	‡	500
\$6,000-10,999	900	1,800	1,900	1,800	3,100	900 !	1,200
\$11,000–18,999	1,200	2,800	2,800	2,900	3,600	1,900	2,500
\$19,000 or more	1,300	5,000	5,500	9,800	11,000	2,400	8,500
Gender							
Male	800	3,300	4,200	8,900	10,400	2,700	5,200
Female	800	3,000	3,700	8,900	10,400	1,900	4,900
Race/ethnicity ³							
White	1,000	3,000	3,800	9,100	10,400	1,600	5,600
Black	900	4,200	5,000	7,700	10,000	1,800	4,400
Hispanic	600	2,600	3,400	6,900	10,100	3,200	3,300
Asian	700	3,200	4,000	12,300	11,800	2,700	5,300
American Indian	‡	‡	‡	‡	‡	‡	3,400
Pacific Islander	900	‡	‡	7,900	‡	‡	3,300
Two or more races	700	3,200	4,100	10,300	11,100	‡	5,200
Other	‡	‡	‡	‡	‡	‡	3,500
Age as of 12/31/07							
18 years or younger	1,100	3,000	4,200	9,400	11,800	1,300	6,000
19–23 years	1,000	3,400	4,100	9,700	10,700	2,600	5,900
24–29 years	600	2,600	2,800	4,600	6,000	2,000	2,200
30–39 years	700	1,900	2,500	3,300	4,600	1,900	1,700
40 years or older	600	2,000	3,200	3,100	4,800	1,800	1,600
Dependency status	4.000	0.455	4.406	0.700	44.000	0.000	0.400
Dependent	1,000	3,400	4,100	9,700	11,000	2,300	6,100
Independent	600	2,400	3,000	4,600	6,200	2,000	2,200
Unmarried, no dependents	600	2,800	3,000	5,500	7,200	2,300	2,500
Married, no dependents	700	2,400	3,400	5,000	6,900 5,700	2,300	2,700
Single parent Married parent	600 700	2,000 2,200	3,000 2,800	3,600 3,500	4,300	1,800 2,000 !	1,800 1,800
maineu paient	700	2,200	2,000	3,500	4,500	۷,000 !	1,000

Table 3.4-D. Average amount of institutional aid received by undergraduates who received institutional aid, by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year	Priv nonprof			
		Non-		Non-			All
	Public	doctorate-	Doctorate-	doctorate-	Doctorate-		insti-
Student characteristics	2-year	granting	granting	granting	granting	For-profit	tutions ¹
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$900	\$2,600	\$4,200	\$9,000	\$10,600	\$1,500	\$4,300
\$20,000-39,999	800	3,000	4,400	9,700	11,500	1,800	5,100
\$40,000-59,999	1,000	3,400	4,000	10,200	11,800	2,900	5,800
\$60,000-79,999	1,500	3,400	3,800	10,500	11,200	2,000 !	6,400
\$80,000-99,999	1,400	4,200	4,400	9,800	11,800	3,000 !	7,400
\$100,000 or more	1,300	3,700	4,100	9,400	10,200	4,000 !	7,100
Independent							
Less than \$10,000	600	2,800	3,200	5,800	7,500	1,600	2,600
\$10,000-19,999	600	2,600	3,200	4,700	5,800	2,500	2,100
\$20,000-29,999	600	2,000	2,400	3,800	6,300	2,000	1,900
\$30,000-49,999	600	1,900	3,100	3,200	4,600	2,200	1,800
\$50,000 or more	900	2,000	2,500	3,800	5,000	2,200 !	2,100
Income group ⁴							
Lowest 25 percent	700	2,900	4,000	8,300	10,200	1,600	4,000
Middle 50 percent	900	3,200	3,900	9,200	10,900	2,400	5,100
Highest 25 percent	1,000	3,400	4,000	8,900	9,800	2,100	6,300
Aid status							
No aid	†	†	†	†	†	†	†
Received aid	800	3,200	3,900	8,900	10,400	2,100	5,000
Grant status							
No grant	2,500	3,000	3,000	‡	2,700	5,300	3,000
Received grant	800	3,200	4,000	8,900	10,600	1,900	5,100
Loan status ⁵							
No loan	800	3,400	4,600	8,700	9,800	‡	3,800
Received loan	900	2,900	3,400	9,000	10,700	2,100	6,100

[†] Not applicable.

NOTE: Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[!] Interpret data with caution (estimates are unstable).

[‡] Reporting standards not met.

¹ All institutions also include those attending public less-than-2-year, private nonprofit less-than-4-year, and multiple institutions, in addition to the institutions listed separately in this table.

² Excludes those attending more than one institution.

³ Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

⁴ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁵ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S3.4-D. Standard errors for table 3.4-D: Average amount of institutional aid received by undergraduates who received institutional aid, by type of institution and selected student characteristics: 2007–08

		Public	4-year	Priv nonprof	/ate it 4-year		
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions
Total	\$30	\$140	\$100	\$220	\$180	\$390	\$130
Attendance status							
Full-time/full-year	60	140		240	200	630	100
Full-time/part-year	120	160	240	350	800	560	300
Part-time/full-year	30	290	200	480	440	370	80
Part-time/part-year	40	240	300	480	530	190	80
Housing							
On campus	350	260	160	280	210	750	150
Off campus	30	150	120	320	390	480	110
Living with parents	50	140	160	330	430	320	110
Price of attendance							
Less than \$6,000	20	200	240	†	†	†	20
\$6,000–10,999	40	110	110	280	400	300	40
\$11,000–18,999	60	140	70	180	240	220	50
\$19,000 or more	250	310	170	230	200	600	140
Gender							
Male	40	220	110	270	230	470	140
Female	30	110	140	280	220	380	140
Race/ethnicity							
White	40	130	110	220	220	260	110
Black	90	370	290	760	440	380	250
Hispanic	40	210	200	540	410	620	200
Asian	60	320	260	1,430	620	640	370
American Indian	†	†	†	†	†	†	510
Pacific Islander	230	†	†	850	†	†	450
Two or more races	100	360	330	1,210	750	†	400
Other	†	†	†	†	†	†	940
Age as of 12/31/07							
18 years or younger	70	280	190	420	360	320	160
19–23 years	50	150	110	230	210	480	120
24–29 years	30	170	180	360	410	530	130
30-39 years	50	230	310	440	570	410	120
40 years or older	60	300	420	460	750	520	130
Dependency status							
Dependent	50	170	100	230	190	390	110
Independent	30	110	220	310	310	420	120
Unmarried, no dependents	30	210	150	470	440	450	150
Married, no dependents	60	250	360	670	830	650	190
Single parent	30	270		420	720	450	140
Married parent	80	280	470	400	550	630	150

Table S3.4-D. Standard errors for table 3.4-D: Average amount of institutional aid received by undergraduates who received institutional aid, by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year	Priv nonprof			
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$90	\$210	\$300	\$600	\$470	\$320	\$180
\$20,000-39,999	60	170	200	430	400	450	180
\$40,000-59,999	70	260	200	450	430	610	230
\$60,000-79,999	190	320	230	580	510	810	200
\$80,000-99,999	190	430	260	410	410	920	250
\$100,000 or more	120	290	170	270	310	1,530	140
Independent							
Less than \$10,000	40	220	210	520	430	330	150
\$10,000-19,999	30	290	400	730	520	590	160
\$20,000-29,999	50	340	240	490	1,160	460	160
\$30,000-49,999	40	280	420	400	930	650	130
\$50,000 or more	130	310	380	700	610	680	170
Income group							
Lowest 25 percent	40	120	150	310	280	310	130
Middle 50 percent	40	190	130	310	260	490	150
Highest 25 percent	100	280	190	260	320	500	150
Aid status							
No aid	†	†	†	†	†	†	†
Received aid	30	140	100	220	180	390	130
Grant status							
No grant	360	340	250	†	310	1,320	200
Received grant	30	140	100	220	190	310	130
Loan status							
No loan	30	180	130	320	300	†	140
Received loan	50	170	120	270	210	400	120

[†] Not applicable.

Table 3.5-A. Percentage distribution of undergraduates who received financial aid, by combination of financial aid received and selected institutional and student characteristics: 2007–08

Institutional and	Grant	Grant and work-	Grant, loan, and work-	Grant and	Loan	Loan and work-	Other combi-	Total
student characteristics	only	study	study ¹	loan ¹	only ¹	study ¹	nations ²	aid
Total	34.3	2.5	5.5	30.4	15.1	0.7	11.6	100.0
Institution type								
Public								
Less-than-2-year	52.3	#	#	19.6	8.7	#	19.3	100.0
2-year	61.7	3.1	1.1	14.2	10.8	0.3	8.9	100.0
4-year	28.8	2.5	5.2	30.9	18.6	0.9	13.2	100.0
Non-doctorate-granting	31.5	2.6	4.7	30.4	18.4	1.0	11.3	100.0
Doctorate-granting	27.3	2.4	5.5	31.1	18.6	8.0	14.2	100.0
Private nonprofit								
Less-than-4-year	39.9	0.8 !	1.6 !	27.6	17.7	0.4 !	11.8 !	100.0
4-year	23.4	3.8	17.1	31.5	8.8	8.0	14.5	100.0
Non-doctorate-granting	22.7	3.9	16.8	32.6	8.7	0.6	14.8	100.0
Doctorate-granting	24.3	3.8	17.6	30.2	8.8	1.1	14.2	100.0
For-profit	4.3	#	1.3	62.4	22.6	0.4 !	8.9	100.0
More than one institution	28.1	2.3	5.2	32.3	17.6	1.1	13.4	100.0
Attendance status								
Full-time/full-year	26.6	3.5	9.1	33.2	12.9	0.9	13.9	100.0
Full-time/part-year	28.3	1.6	3.6	35.7	18.1	0.6	12.2	100.0
Part-time/full-year	44.2	2.1	2.3	27.3	15.4	0.4	8.3	100.0
Part-time/part-year	49.4	1.2	8.0	21.2	18.2	0.4	8.8	100.0
Housing ³								
On campus	21.6	4.2	16.9	27.6	10.3	1.2	18.3	100.0
Off campus	34.7	1.8	3.0	33.9	16.1	0.5	9.9	100.0
Living with parents	46.4	3.1	2.3	22.5	15.3	0.5	9.9	100.0
Price of attendance ³								
Less than \$6,000	71.7	1.0	0.2 !	8.2	12.4	0.1 !	6.4	100.0
\$6,000–10,999	50.0	2.7	1.0	23.6	14.7	0.3	7.9	100.0
\$11,000–18,999	27.0	2.5	4.3	37.0	17.4	0.8	11.2	100.0
\$19,000 or more	15.5	3.2	12.3	37.8	13.5	0.9	16.7	100.0
Gender								
Male	34.3	2.6	5.3	26.5	15.7	0.8	14.8	100.0
Female	34.3	2.5	5.6	33.0	14.7	0.5	9.5	100.0

Table 3.5-A. Percentage distribution of undergraduates who received financial aid, by combination of financial aid received and selected institutional and student characteristics: 2007–08

—Continued

Institutional and student characteristics	Grant only	Grant and work- study	Grant, Ioan, and work- study ¹	Grant and loan ¹	Loan only ¹	Loan and work- study ¹	Other combinations ²	Total aid
Race/ethnicity ⁴								
White	32.7	2.2	5.5	28.5	17.3	0.8	13.0	100.0
Black	29.7	1.9	5.8	41.0	12.4	0.3	8.8	100.0
Hispanic	42.9	3.3	4.4	29.1	10.5	0.3	9.5	100.0
Asian	42.2	5.4	6.0	22.6	12.3	1.0 !	10.5	100.0
American Indian	44.5	2.1 !	4.4	29.5	12.3	‡	7.0	100.0
Pacific Islander	39.9	2.8 !	5.0	25.1	11.5	‡	15.5	100.0
Two or more races	29.5	3.9	8.2	30.5	13.6	1.2 !	13.2	100.0
Other	38.1	‡	5.1 !	28.4	14.5	‡	10.7	100.0
Age as of 12/31/07								
18 years or younger	39.1	3.3	8.3	25.0	9.1	0.9	14.3	100.0
19–23 years	31.9	3.4	7.5	27.3	15.8	0.9	13.3	100.0
24–29 years	29.8	1.3	2.8	39.6	16.6	0.3	9.7	100.0
30–39 years	37.0	1.3	1.7	36.1	15.7	0.2	8.0	100.0
40 years or older	46.5	1.1	1.2	28.0	14.5	0.3 !	8.4	100.0
Dependency status								
Dependent	32.7	3.5	8.1	25.1	15.4	1.0	14.2	100.0
Independent	36.1	1.4	2.4	36.4	14.7	0.3	8.6	100.0
Unmarried, no dependents	30.6	1.4	3.4	34.9	19.1	0.5	10.0	100.0
Married, no dependents	40.6	2.0	2.1	20.7	23.5	0.7 !	10.4	100.0
Single parent	36.9	1.5	2.2	47.9	6.1	0.1 !	5.3	100.0
Married parent	40.4	0.9	1.3	30.8	16.1	0.2 !	10.3	100.0
Dependency and income in 2006								
Dependent								
Less than \$20,000	43.9	4.4	8.6	32.9	3.1	‡	6.9	100.0
\$20,000-39,999	36.1	4.9	10.4	32.9	5.8	0.2	9.7	100.0
\$40,000–59,999	29.3	3.5	10.2	28.4	14.7	1.2	12.7	100.0
\$60,000-79,999	25.9	2.5	7.6	20.7	23.7	1.9	17.7	100.0
\$80,000–99,999	27.1	2.5	7.5	19.4	23.7	1.5	18.3	100.0
\$100,000 or more	32.5	2.9	5.2	17.2	21.8	1.1	19.3	100.0
Independent								
Less than \$10,000	32.0	2.3	5.1	46.8	6.0	0.3	7.6	100.0
\$10,000-19,999	30.4	1.4	2.6	48.0	9.8	0.2	7.6	100.0
\$20,000-29,999	33.6	1.2	1.6	35.1	18.9	0.5	9.0	100.0
\$30,000-49,999	37.8	0.8	1.0	34.6	17.0	0.2!	8.6	100.0
\$50,000 or more	48.5	8.0	0.4	12.6	26.4	0.3 !	11.0	100.0
Income group ⁵								
Lowest 25 percent	36.6	3.6	7.4	39.3	5.1	0.2	7.9	100.0
Middle 50 percent	30.8	2.1	5.3	31.0	17.6	0.9	12.2	100.0
Highest 25 percent	39.5	2.0	3.0	15.1	23.8	0.7	15.9	100.0

Table 3.5-A. Percentage distribution of undergraduates who received financial aid, by combination of financial aid received and selected institutional and student characteristics: 2007–08

—Continued

Institutional and student characteristics	Grant only	Grant and work- study	Grant, Ioan, and work- study ¹	Grant and loan ¹	Loan only ¹	Loan and work- study ¹	Other combi- nations ²	Total aid
Grant status								
No grant	†	†	†	†	71.2	3.1	25.7	100.0
Received grant	43.5	3.2	6.9	38.5	†	†	7.9	100.0
Loan status ¹								
No loan	82.9	6.1	†	†	†	†	11.0	100.0
Received loan	†	†	9.3	51.8	25.7	1.1	12.1	100.0

[†] Not applicable.

NOTE: Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. For students attending more than one institution, combinations of aid may include only aid awarded at the institution in the NPSAS sample and may not include aid awarded at other institutions. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico. Detail may not sum to totals because of rounding.

[#] Rounds to zero.

[!] Interpret data with caution (estimates are unstable).

[‡] Reporting standards not met.

¹ Loans only include loans to students. Excludes federal Parent PLUS loans.

² Includes federal Parent PLUS loans.

³ Excludes students attending more than one institution.

⁴ Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

⁵ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

Table S3.5-A. Standard errors for table 3.5-A: Percentage distribution of undergraduates who received financial aid, by combination of financial aid received and selected institutional and student characteristics: 2007–08

Institutional and student characteristics	Grant only	Grant and work- study	Grant, Ioan, and work- study	Grant and Ioan	Loan only	Loan and work- study	Other combi- nations	Total aid
Total	0.58	0.11	0.16	0.48	0.21	0.04	0.21	†
Institution type Public								
	3.94	+	+	2.00	1.78	+	4.58	+
Less-than-2-year	3.94 0.77	† 0.28	† 0.12	0.53	0.36	1	4.56 0.52	†
2-year	-		-			0.05		†
4-year	0.37	0.18	0.22	0.41	0.30	0.08	0.35	†
Non-doctorate-granting	0.94	0.28	0.36	0.74	0.77	0.12	0.49	†
Doctorate-granting	0.45	0.19	0.30	0.46	0.42	0.11	0.48	†
Private nonprofit	0.45	0.04	0.50	0.00	0.40	0.47	0.00	
Less-than-4-year	3.15	0.34	0.58	3.23	2.40	0.17	3.89	†
4-year	0.64	0.27	0.64	0.61	0.47	0.14	0.56	†
Non-doctorate-granting	0.87	0.41	1.03	1.03	0.75	0.16	0.80	†
Doctorate-granting	0.98	0.40	0.80	1.05	0.59	0.24	0.67	†
For-profit	0.24	†	0.22	0.85	0.66	0.14	0.66	†
More than one institution	1.93	0.22	0.38	1.62	0.50	0.13	0.50	†
Attendance status								
Full-time/full-year	0.51	0.16	0.24	0.60	0.26	0.07	0.25	†
Full-time/part-year	1.14	0.17	0.36	1.02	0.62	0.09	0.65	†
Part-time/full-year	0.94	0.24	0.13	0.69	0.42	0.06	0.30	†
Part-time/part-year	0.96	0.17	0.14	0.75	0.63	0.08	0.51	†
Housing								
On campus	0.49	0.28	0.58	0.57	0.39	0.15	0.59	†
Off campus	0.58	0.13	0.13	0.54	0.25	0.04	0.30	†
Living with parents	1.17	0.22	0.18	0.61	0.64	0.07	0.36	†
Price of attendance								
Less than \$6,000	0.85	0.22	0.07	0.50	0.63	0.04	0.58	†
\$6,000–10,999	0.80	0.23	0.12	0.65	0.49	0.05	0.43	†
\$11,000–18,999	0.51	0.17	0.18	0.50	0.31	0.07	0.29	†
\$19,000 or more	0.39	0.21	0.43	0.62	0.37	0.11	0.41	†
Gender								
Male	0.61	0.17	0.20	0.49	0.41	0.08	0.36	†
Female	0.69	0.11	0.20	0.61	0.25	0.04	0.22	÷

Table S3.5-A. Standard errors for table 3.5-A: Percentage distribution of undergraduates who received financial aid, by combination of financial aid received and selected institutional and student characteristics: 2007–08—Continued

		Grant	Grant,			Loan		
		and	loan, and	Grant		and	Other	
Institutional and	Grant	work-	work-	and	Loan	work-	combi-	Total
student characteristics	only	study	study	loan	only	study	nations	aid
			-		-			
Race/ethnicity								
White	0.46	0.11	0.19	0.50	0.27	0.06	0.31	†
Black	0.96	0.17	0.37	0.88	0.51	0.07	0.41	†
Hispanic	1.69	0.27	0.30	0.89	0.54	0.08	0.52	†
Asian	1.42	0.65	0.55	1.20	0.85	0.31	0.72	†
American Indian	3.83	0.73	1.21	3.92	1.96	†	1.39	†
Pacific Islander	3.77	1.06	1.44	3.62	2.19	†	2.62	†
Two or more races	1.61	0.63	1.15	1.68	1.14	0.53	1.10	†
Other	5.32	†	1.56	4.88	3.30	†	2.82	†
Age as of 12/31/07								
18 years or younger	1.27	0.26	0.47	0.88	0.46	0.16	0.70	†
19–23 years	0.65	0.16	0.22	0.44	0.35	0.07	0.28	†
24-29 years	0.84	0.15	0.22	0.97	0.47	0.05	0.46	†
30-39 years	0.99	0.22	0.19	1.03	0.57	0.07	0.51	†
40 years or older	1.15	0.16	0.20	0.95	0.74	0.09	0.83	†
Dependency status								
Dependent	0.63	0.15	0.23	0.43	0.30	0.07	0.30	†
Independent	0.65	0.13	0.14	0.71	0.26	0.04	0.37	†
Unmarried, no dependents	0.89	0.16	0.28	0.90	0.53	0.09	0.53	†
Married, no dependents	1.79	0.37	0.31	0.87	1.16	0.22	0.90	†
Single parent	1.29	0.17	0.18	1.28	0.47	0.02	0.36	†
Married parent	1.06	0.16	0.17	1.05	0.73	0.07	0.69	†
Dependency and income in 2006								
Dependent								
Less than \$20,000	1.41	0.37	0.53	1.14	0.41	†	0.46	†
\$20,000-39,999	1.32	0.34	0.49	0.96	0.37	0.05	0.49	†
\$40,000-59,999	1.04	0.34	0.45	0.91	0.59	0.20	0.51	†
\$60,000-79,999	0.70	0.28	0.58	0.64	0.82	0.23	0.75	†
\$80,000-99,999	0.91	0.31	0.61	0.79	0.80	0.31	0.82	†
\$100,000 or more	0.71	0.23	0.37	0.49	0.56	0.13	0.64	†
Independent								
Less than \$10,000	1.38	0.23	0.41	1.34	0.37	0.07	0.49	†
\$10,000–19,999	1.37	0.24	0.27	1.28	0.50	0.07	0.51	†
\$20,000–29,999	1.28	0.21	0.22	1.17	0.88	0.14	0.68	†
\$30,000–49,999	0.97	0.20	0.13	1.23	0.76	0.06	0.65	†
\$50,000 or more	1.33	0.20	0.11	0.77	0.85	0.10	0.80	†
Income group								
Lowest 25 percent	1.19	0.20	0.29	1.01	0.23	0.04	0.30	†
Middle 50 percent	0.60	0.13	0.19	0.52	0.32	0.07	0.27	†
Highest 25 percent	0.65	0.16	0.21	0.42	0.50	0.08	0.49	†

Table S3.5-A. Standard errors for table 3.5-A: Percentage distribution of undergraduates who received financial aid, by combination of financial aid received and selected institutional and student characteristics: 2007–08—Continued

Institutional and student characteristics	Grant only	Grant and work- study	Grant, Ioan, and work- study	Grant and loan	Loan only	Loan and work- study	Other combi- nations	Total aid
Grant status								
No grant	†	†	†	†	0.77	0.20	0.73	†
Received grant	0.72	0.14	0.20	0.60	†	†	0.19	†
Loan status								
No loan	0.47	0.24	†	†	†	†	0.39	†
Received loan	†	†	0.24	0.34	0.33	0.08	0.26	†

[†] Not applicable.

Table 3.5-B. Average amount of financial aid received by undergraduates who received various combinations of financial aid, by selected institutional and student characteristics: 2007–08

Institutional and student characteristics	Grant only	Grant and work- study	Grant, loan, and work- study ¹	Grant and Ioan ¹	Loan only ¹	Loan and work- study ¹	Other combi- nations ²	Total aid
Total	\$3,700	\$9,800	\$21,000	\$11,700	\$7,400	\$11,600	\$14,600	\$9,100
Institution type Public								
Less-than-2-year	2,500	‡	‡	8,900	5,800	‡	5,800	4,700
2-year	2,000	5,600	9,600	6,700	4,000	7,100	5,200	3,400
4-year	5,000	9,900	15,500	11,500	7,500	11,200	14,000	9,400
Non-doctorate-granting	4,000	8,800	14,300	10,500	6,900	9,800	11,600	8,000
Doctorate-granting	5,600	10,600	16,000	12,000	7,900	12,100	15,000	10,100
Private nonprofit								
Less-than-4-year	4,900 !	‡	‡	9,800	8,100	‡	12,000	7,800
4-year	8,300	15,800	26,200	17,900	10,300	14,000	25,300	17,400
Non-doctorate-granting	7,300	15,400	24,900	16,000	9,400	12,000	23,600	16,000
Doctorate-granting	9,400	16,400	27,700	20,300	11,300	15,100	27,500	19,000
For-profit	3,500	‡	14,400	10,900	9,600	17,100	15,900	10,800
More than one institution	3,500	8,800	18,500	11,000	7,100	10,000	14,200	9,000
Attendance status								
Full-time/full-year	6,200	11,800	22,400	14,200	9,200	12,400	19,000	12,700
Full-time/part-year	3,100	6,400	16,600	9,400	7,000	10,600	10,400	7,500
Part-time/full-year	2,400	6,200	16,000	10,000	7,100	10,900	9,700	6,200
Part-time/part-year	1,500	4,900	11,500	6,900	4,800	8,500	5,600	3,800
Housing ³								
On campus	8,500	14,900	23,900	16,200	9,400	13,300	22,200	16,200
Off campus	3,000	8,000	17,700	11,000	7,500	11,000	11,800	7,900
Living with parents	3,300	7,100	16,600	10,500	6,200	11,100	10,000	6,500
Price of attendance ³								
Less than \$6,000	1,200	2,800	‡	3,900	3,000	‡	2,000	1,700
\$6,000–10,999	2,700	5,500	7,200	6,400	4,900	6,700	5,000	4,200
\$11,000–18,999	4,600	8,400	13,100	10,400	7,600	9,700	11,100	8,500
\$19,000 or more	9,600	14,700	25,000	16,400	11,100	15,000	22,500	16,700
Gender								
Male	3,800	10,000	21,300	12,200	7,500	11,200	14,100	9,300
Female	3,600	9,600	20,800	11,400	7,400	12,000	15,100	9,000

Table 3.5-B. Average amount of financial aid received by undergraduates who received various combinations of financial aid, by selected institutional and student characteristics: 2007–08—Continued

Institutional and	Grant	Grant and work-	Grant, loan, and work-	Grant and	Loan	Loan and work-	Other combi-	Total
student characteristics	only	study	study ¹	loan ¹	only ¹	study ¹	nations ²	aid
Description 4								
Race/ethnicity ⁴	Φο οοο	40.700	004 400	# 44.000	#7.500	0.4.4.000	0.45 000	00.400
White	\$3,800	\$9,700	\$21,400	\$11,900	\$7,500	\$11,900	\$15,200	\$9,400
Black	3,200	10,300	19,000	11,200	7,500	8,600	13,300	9,000
Hispanic	3,200	8,300	20,400	11,400	7,200	15,700	12,900	7,900
Asian	4,700	13,300	24,300	13,000	7,900	9,100	13,500	9,600
American Indian	3,900	‡	16,000	9,700	5,700	‡	14,600	7,200
Pacific Islander	3,200	‡ 2.722	20,400	11,800	7,400	‡	14,200	8,600
Two or more races	4,400	9,700	22,100	12,400	7,500	‡	14,200	10,300
Other	2,600	‡	‡	11,100	6,600	‡	‡	7,800
Age as of 12/31/07								
18 years or younger	5,200	10,400	21,300	12,300	6,800	11,000	18,600	10,600
19–23 years	4,600	10,600	21,900	12,700	7,500	11,900	16,600	10,400
24–29 years	2,300	7,400	17,200	10,900	7,500	11,600	10,000	7,800
30–39 years	2,300	5,400	15,800	10,400	7,800	‡	8,600	6,900
40 years or older	2,100	6,000	15,200	10,600	6,900	‡	6,600	5,800
Dependency status								
Dependent	4,900	10,900	21,900	13,100	7,400	11,800	17,600	10,800
Independent	2,400	6,600	17,100	10,700	7,500	11,000	8,800	7,100
Unmarried, no dependents	2,400	7,400	19,000	11,300	7,600	12,200	9,700	7,900
Married, no dependents	2,400	6,000	16,600	11,200	8,000	10,800	8,000	6,600
Single parent	2,400	6,300	14,900	10,200	6,500	, ‡	8,700	7,100
Married parent	2,300	6,000	15,700	10,400	7,300	‡	8,300	6,400
Dependency and income in 2006								
Dependent								
Less than \$20,000	4,700	11,400	21,300	12,500	6,400	‡	14,800	9,700
\$20,000–39,999	4,600	10,800	21,100	12,700	7,300	‡	16,800	10,600
\$40,000–59,999	4,200	10,700	22,300	12,300	6,800	10,700	17,200	10,700
\$60,000-79,999	4,500	9,700	22,300	13,000	7,200	10,400	16,600	10,700
\$80,000-99,999	5,600	11,800	23,900	14,400	7,100	12,400	18,000	11,600
\$100,000 or more	5,900	11,200	21,800	14,600	8,200	14,600	19,000	11,500
Independent								
Less than \$10,000	3,100	7,200	18,700	11,000	6,700	9,200	10,200	8,400
\$10,000–19,999	2,500	6,800	14,200	10,300	7,100	‡	10,000	7,600
\$20,000–29,999	2,300	6,200	15,500	10,800	7,300	14,600	9,100	7,200
\$30,000–49,999	2,000	5,600	16,000	10,200	7,400	‡	8,400	6,500
\$50,000 or more	2,100	5,200	17,000	11,400	8,000	7,500	6,900	5,400
Income group ⁵								
Lowest 25 percent	4,000	10,000	20,300	11,700	6,900	8,500	13,500	9,400
Middle 50 percent	3,300	9,500	21,400	11,400	7,200	11,300	14,700	9,000
Highest 25 percent	3,900	10,100	21,600	13,600	8,200	13,900	15,200	8,900

Table 3.5-B. Average amount of financial aid received by undergraduates who received various combinations of financial aid, by selected institutional and student characteristics: 2007–08—Continued

Institutional and student characteristics	Grant only	Grant and work- study	Grant, loan, and work- study ¹	Grant and Ioan ¹	Loan only ¹	Loan and work- study ¹	Other combi- nations ²	Total aid
Aid status								
No aid	†	t	†	†	t	†	†	†
Received aid	\$3,700	\$9,800	\$21,000	\$11,700	\$7,400	\$11,600	\$14,600	\$9,100
Grant status								
No grant	†	†	†	†	7,400	11,600	9,500	8,100
Received grant	3,700	9,800	21,000	11,700	†	†	19,000	9,400
Loan status ¹								
No loan	3,700	9,800	†	t	†	†	6,100	4,300
Received loan	†	†	21,000	11,700	7,400	11,600	20,000	12,500

[†] Not applicable.

NOTE: Averages are for those who received the specified combinations only. Those not receiving a specific combination of aid are not included in the calculation of the average. Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. For students attending more than one institution, combinations of aid may include only aid awarded at the institution in the NPSAS sample and may not include aid awarded at other institutions. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[!] Interpret data with caution (estimates are unstable).

[‡] Reporting standards not met.

¹ Loans only include loans to students. Excludes federal Parent PLUS loans.

² Includes federal Parent PLUS loans.

³ Excludes students attending more than one institution.

⁴ Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

⁵ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

Table S3.5-B. Standard errors for table 3.5-B: Average amount of financial aid received by undergraduates who received various combinations of financial aid, by selected institutional and student characteristics: 2007–08

Institutional and	Grant	Grant and work-	Grant, loan, and work-	Grant and	Loan	Loan and work-	Other combi-	Total
student characteristics	only	study	study	loan	only	study	nations	aid
Total	\$80	\$310	\$280	\$110	\$110	\$470	\$330	\$150
Institution type								
Public								
Less-than-2-year	210	†	†	900	610	†	750	400
2-year	40	300	370	290	90	380	170	100
4-year	80	270	170	90	100	500	190	60
Non-doctorate-granting	110	350	350	150	150	770	290	120
Doctorate-granting	130	360	240	110	140	670	260	90
Private nonprofit								
Less-than-4-year	1,700	†	†	1,030	880	†	1,200	970
4-year	230	620	390	270	330	1,400	650	260
Non-doctorate-granting	290	870	580	390	380	2,130	930	370
Doctorate-granting	400	820	530	460	480	1,940	840	370
For-profit	150	†	1,100	170	300	1,850	550	170
More than one institution	250	920	440	180	170	630	590	370
Attendance status								
Full-time/full-year	100	340	300	110	140	560	280	120
Full-time/part-year	150	510	1,330	200	240	1,750	570	220
Part-time/full-year	60	490	490	100	140	1,060	320	110
Part-time/part-year	50	440	850	140	130	970	260	70
Housing								
On campus	230	430	380	220	300	1,040	420	160
Off campus	60	390	370	110	140	550	370	120
Living with parents	90	290	620	170	160	1,160	280	130
Price of attendance								
Less than \$6,000	30	420	†	70	70	†	140	30
\$6,000–10,999	50	240	250	80	80	410	140	70
\$11,000–18,999	60	250	140	60	130	420	190	80
\$19,000 or more	240	440	350	180	220	870	320	150
Gender								
Male	100	470	370	170	160	630	410	160
Female	80	340	340	120	100	610	350	150

Table S3.5-B. Standard errors for table 3.5-B: Average amount of financial aid received by undergraduates who received various combinations of financial aid, by selected institutional and student characteristics: 2007–08—Continued

Institutional and	Grant	Grant and work-	Grant, loan, and work-	Grant and	Loan	Loan and work-	Other combi-	Total
student characteristics	only	study	study	loan	only	study	nations	aid
Race/ethnicity								
White	\$70	\$350	\$340	\$130	\$130	\$520	\$360	\$130
Black	130	780	470	180	210	840	450	180
Hispanic	110	470	830	220	210	2,250	540	270
Asian	320	1,120	940	360	390	840	940	350
American Indian	710	†	2,770	690	800	†	2,020	670
Pacific Islander	410	†	2,200	1,120	730	†	1,750	530
Two or more races	290	1,200	900	460	500	†	1,110	300
Other	370	†	†	1,140	1,010	†	†	680
Age as of 12/31/07								
18 years or younger	200	600	550	230	330	1,620	500	250
19–23 years	110	320	330	130	110	520	290	140
24–29 years	70	550	730	180	170	1,000	480	190
30-39 years	60	390	1,120	200	240	†	440	150
40 years or older	70	410	1,300	220	290	†	520	130
Dependency status								
Dependent	120	330	310	130	140	550	280	150
Independent	50	330	450	120	140	770	380	140
Unmarried, no dependents	100	630	650	200	150	1,070	440	170
Married, no dependents	100	630	1,150	390	390	1,300	510	200
Single parent	60	310	670	150	360	†	580	180
Married parent	60	570	980	190	210	†	540	170
Dependency and income in 2006								
Dependent								
Less than \$20,000	160	610	540	220	650	†	680	250
\$20,000–39,999	210	600	420	220	440	†	660	280
\$40,000–59,999	240	910	650	300	230	1,370	550	260
\$60,000–79,999	170	860	700	350	230	790	570	190
\$80,000–99,999	340	890	1,020	490	260	1,370	500	230
\$100,000 or more	160	610	570	300	230	1,040	420	160
Independent								
Less than \$10,000	130	420	590	160	300	730	650	240
\$10,000–19,999	90	740	750	210	300	†	520	210
\$20,000–29,999	80	490	1,460	180	220	1,570	580	150
\$30,000–49,999	70	650	950	210	240	†	480	160
\$50,000 or more	70	840	1,810	410	310	840	440	110
Income group								
Lowest 25 percent	140	340	340	140	260	550	430	220
Middle 50 percent	100	480	400	120	90	560	400	150
Highest 25 percent	110	620	570	280	210	1,050	450	130

Table S3.5-B. Standard errors for table 3.5-B: Average amount of financial aid received by undergraduates who received various combinations of financial aid, by selected institutional and student characteristics: 2007–08—Continued

		Grant and	Grant, loan, and	Grant		Loan and	Other	
Institutional and	Grant	work-	work-	and	Loan	work-	combi-	Total
student characteristics	only	study	study	loan	only	study	nations	aid
Aid status								
No aid	†	†	†	†	†	†	†	†
Received aid	\$80	\$310	\$280	\$110	\$110	\$470	\$330	\$150
Grant status								
No grant	†	†	†	†	110	470	250	120
Received grant	80	310	280	110	†	†	380	160
Loan status								
No loan	80	310	†	†	†	†	160	80
Received Ioan	†	†	280	110	110	470	250	110

[†] Not applicable.

Table 3.6-A. Percentage of undergraduates in public 4-year institutions receiving financial aid, by type and source of aid and selected student characteristics: 2007–08

			Any		Any	Any	Any
	Any	Any	work-	Any	federal	state	institution
Student characteristics	aid	grants	study	loans ¹	aid ²	aid	aid
Total	71.3	52.9	7.7	46.2	51.2	24.1	23.6
Attendance status							
Full-time/full-year	78.3	60.4	10.5	52.7	59.2	31.4	31.4
Full-time/part-year	68.7	47.9	5.2	43.0	46.1	19.9	19.1
Part-time/full-year	67.6	47.6	4.9	44.4	49.6	16.3	14.0
Part-time/part-year	50.8	33.9	2.3	26.1	26.0	8.1	7.3
Housing							
On campus	76.2	58.4	12.0	53.9	56.7	29.4	33.9
Off campus	70.3	51.1	6.4	46.1	50.4	21.0	20.6
Living with parents	67.6	50.5	5.7	35.9	45.8	26.1	18.0
Price of attendance							
Less than \$6,000	38.5	26.2	0.9!	13.6	13.4	5.7	4.0
\$6,000-10,999	65.2	48.7	3.0	33.2	40.0	18.6	11.1
\$11,000–18,999	78.0	58.0	8.1	54.2	60.0	29.3	25.5
\$19,000 or more	77.6	58.0	13.4	54.9	59.3	26.6	37.5
Gender							
Male	68.5	49.6	7.2	43.6	47.5	21.5	22.7
Female	73.8	55.8	8.2	48.5	54.4	26.4	24.3
Race/ethnicity ³							
White	69.7	49.1	6.8	45.9	47.3	22.3	23.1
Black	83.5	66.8	10.8	63.8	70.5	27.9	23.6
Hispanic	72.4	59.5	8.9	38.3	56.2	29.0	23.6
Asian	61.0	49.3	8.3	32.5	43.8	26.1	26.8
American Indian	77.3	65.8	8.4	44.6	58.4	21.6	18.9
Pacific Islander	70.1	56.9	11.4	40.7	51.5	28.0	24.5
Two or more races	78.3	61.5	11.8	48.5	57.1	27.7	30.5
Other	71.2	60.0	6.4 !	40.9	58.2	26.4	15.3
Age as of 12/31/07							
18 years or younger	80.7	65.7	9.6	48.6	58.2	36.2	34.7
19–23 years	71.1	51.6	8.7	46.8	51.4	25.9	25.8
24–29 years	67.9	48.5	5.3	47.9	51.8	16.0	14.6
30–39 years	70.7	52.8	4.0	44.2	48.3	13.7	13.1
40 years or older	64.1	52.8	3.9	30.4	34.5	12.9	11.1
Dependency status							
Dependent	72.7	53.5	8.9	47.5	52.4	27.7	27.6
Independent	68.3	51.7	5.1	43.4	48.4	16.2	14.5
Unmarried, no dependents	67.1	48.5	5.8	47.0	50.4	16.4	15.9
Married, no dependents	65.1	44.4	4.5	34.0	38.0	13.2	14.6
Single parent	73.1	64.3	5.7	47.2	57.6	21.4	15.0
Married parent	68.1	50.8	3.7	38.6	42.6	12.5	11.1

Table 3.6-A. Percentage of undergraduates in public 4-year institutions receiving financial aid, by type and source of aid and selected student characteristics: 2007–08—Continued

			Any	_	Any	Any	Any
	Any	Any	work-	Any	federal	state	institution
Student characteristics	aid	grants	study	loans ¹	aid ²	aid	aid
Dependency and income in 2006							
Dependent							
Less than \$20,000	86.8	83.5	13.0	54.2	80.0	46.3	32.8
\$20,000–39,999	89.2	82.4	15.8	58.4	80.6	48.4	35.3
\$40,000–59,999	77.9	62.0	12.3	55.6	60.5	38.2	32.7
\$60,000-79,999	75.2	47.9	8.5	53.0	50.0	23.9	28.8
\$80,000-99,999	70.6	42.7	5.6	47.8	45.0	17.3	25.2
\$100,000 or more	58.9	35.6	4.8	34.4	32.9	14.9	21.1
Independent							
Less than \$10,000	73.4	64.1	9.4	50.6	61.9	24.5	20.1
\$10,000-19,999	72.8	59.1	6.0	53.7	62.4	21.1	17.5
\$20,000-29,999	70.7	46.8	3.4	45.8	51.5	15.9	14.1
\$30,000-49,999	66.2	45.5	3.0	42.0	42.2	11.8	11.3
\$50,000 or more	57.3	36.7	1.6 !	23.8	20.5	4.3	7.3
Income group ⁴							
Lowest 25 percent	82.7	76.2	12.6	54.3	73.7	38.7	28.7
Middle 50 percent	73.1	51.0	7.4	50.3	51.9	23.6	24.5
Highest 25 percent	57.7	35.0	3.9	31.2	29.3	11.8	17.2
Undergraduate class level							
First year	71.8	56.8	7.1	43.3	52.4	27.5	23.5
Second year	73.0	54.3	7.9	47.3	52.5	26.3	24.4
Third year	76.1	54.2	8.3	54.8	58.7	25.8	24.5
Fourth year or more	68.1	49.5	7.9	43.2	46.3	20.3	23.2
Aid status							
No aid	†	†	†	†	†	†	†
Received aid	100.0	74.2	10.9	64.8	71.7	33.8	33.0
Grant status							
No grant	39.1	†	3.5	35.2	30.0	0.8	1.5
Received grant	100.0	100.0	11.5	56.0	70.0	44.9	43.2
Loan status ¹							
No loan	46.7	43.3	5.0	†	16.9	18.0	19.0
Received loan	100.0	64.1	10.9	100.0	91.1	31.3	28.8

[†] Not applicable.

NOTE: Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. This table excludes students attending more than one institution. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto

[!] Interpret data with caution (estimates are unstable).

¹ Loans only include loans to students. Excludes federal Parent PLUS loans.

² Includes all federal grants, loans, college work-study awards, and federal Parent PLUS loans. Excludes veterans benefits and education tax credit and tax deduction benefits.

³ Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

⁴ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

Table S3.6-A. Standard errors for table 3.6-A: Percentage of undergraduates in public 4-year institutions receiving financial aid, by type and source of aid and selected student characteristics: 2007–08

			Any		Any	Any	Any
	Any	Any	work-	Any	federal	state	institution
Student characteristics	aid	grants	study	loans	aid	aid	aid
Total	0.73	0.59	0.32	0.29	0.75	0.53	0.41
Attendance status							
Full-time/full-year	0.85	0.89	0.39	0.51	0.93	0.73	0.57
Full-time/part-year	1.05	0.99	0.69	1.15	1.06	0.95	0.74
Part-time/full-year	1.32	1.32	0.34	0.90	1.26	0.81	0.71
Part-time/part-year	1.02	0.90	0.42	0.97	1.02	0.54	0.53
Housing							
On campus	1.16	1.16	0.74	0.99	1.16	0.99	0.86
Off campus	0.75	0.51	0.32	0.48	0.83	0.49	0.42
Living with parents	0.97	1.09	0.41	0.95	1.06	0.96	0.70
Price of attendance							
Less than \$6,000	1.39	1.16	0.35	0.99	1.09	0.63	0.54
\$6,000-10,999	1.36	1.15	0.37	0.91	1.36	1.04	0.72
\$11,000–18,999	0.90	0.80	0.36	0.48	0.89	0.78	0.50
\$19,000 or more	1.03	0.97	0.71	1.03	1.29	0.72	0.94
Gender							
Male	1.04	0.94	0.39	0.56	0.89	0.67	0.56
Female	0.56	0.50	0.36	0.36	0.76	0.52	0.48
Race/ethnicity							
White	0.76	0.65	0.31	0.45	0.70	0.57	0.47
Black	1.20	0.99	0.81	1.20	1.64	1.09	1.04
Hispanic	1.13	1.04	0.61	1.32	1.42	1.02	1.03
Asian	2.50	2.01	1.05	1.64	2.29	1.41	1.71
American Indian	3.68	3.70	2.23	4.66	3.83	3.66	3.21
Pacific Islander	4.48	4.55	3.14	4.61	5.49	4.99	4.55
Two or more races	2.17	2.27	1.34	2.24	2.05	1.86	2.45
Other	6.36	6.17	2.95	5.39	5.48	5.39	4.31
Age as of 12/31/07							
18 years or younger	1.17	1.75	0.72	1.09	1.22	1.61	1.12
19–23 years	0.77	0.74	0.36	0.42	0.77	0.65	0.53
24–29 years	1.36	0.97	0.40	1.04	1.25	0.81	0.63
30-39 years	1.44	1.34	0.77	1.32	1.66	0.84	0.97
40 years or older	1.78	1.85	0.55	1.51	2.24	1.23	1.06
Dependency status							
Dependent	0.81	0.84	0.39	0.39	0.81	0.72	0.54
Independent	0.87	0.53	0.35	0.62	0.96	0.54	0.51
Unmarried, no dependents	1.21	0.95	0.42	1.14	1.18	0.72	0.73
Married, no dependents	2.05	1.58	0.88	1.82	2.17	1.04	1.06
Single parent	1.57	1.45	0.72	1.44	1.44	1.18	1.06
Married parent	1.45	1.32	0.63	1.35	1.72	0.84	0.86

Table S3.6-A. Standard errors for table 3.6-A: Percentage of undergraduates in public 4-year institutions receiving financial aid, by type and source of aid and selected student characteristics: 2007–08—Continued

	Any	Any	Any work-	Any	Any federal	Any state	Any institution
Student characteristics	aid	grants	study	loans	aid	aid	aid
Dependency and income in 2006							
Dependent							
Less than \$20,000	1.35	1.30	0.80	1.62	1.05	1.62	1.12
\$20,000-39,999	0.89	1.04	0.96	1.04	0.86	1.16	1.05
\$40,000-59,999	1.34	1.47	0.88	1.04	1.53	1.16	1.00
\$60,000-79,999	0.89	1.07	0.74	0.99	0.99	1.16	1.22
\$80,000-99,999	1.06	1.20	0.64	1.05	1.04	0.90	1.02
\$100,000 or more	1.06	0.98	0.30	0.94	1.02	0.74	0.80
Independent							
Less than \$10,000	1.08	1.08	0.67	1.15	0.97	1.07	0.88
\$10,000-19,999	1.67	1.50	0.73	1.53	1.60	1.20	1.17
\$20,000-29,999	1.83	1.58	0.73	1.72	2.21	1.25	1.02
\$30,000-49,999	2.00	1.73	0.55	1.76	1.73	1.10	1.10
\$50,000 or more	1.81	1.54	0.52	1.34	1.49	0.60	0.78
Income group							
Lowest 25 percent	0.82	0.76	0.57	0.86	0.62	0.80	0.62
Middle 50 percent	0.72	0.63	0.40	0.46	0.86	0.64	0.53
Highest 25 percent	0.98	0.86	0.27	0.85	0.87	0.61	0.70
Undergraduate class level							
First year	1.05	1.28	0.44	0.86	0.99	1.11	0.75
Second year	1.15	1.14	0.63	0.83	1.27	1.08	0.80
Third year	1.17	1.05	0.45	1.34	1.47	0.80	0.65
Fourth year or more	0.80	0.58	0.40	0.55	0.82	0.54	0.64
Aid status							
No aid	†	†	†	†	†	†	†
Received aid	†	0.33	0.39	0.47	0.52	0.56	0.51
Grant status							
No grant	0.92	†	0.27	0.74	0.90	0.13	0.17
Received grant	†	†	0.43	0.53	0.53	0.71	0.62
Loan status							
No loan	1.17	1.04	0.36	†	0.97	0.74	0.54
Received loan	†	0.45	0.44	†	0.35	0.52	0.57

[†] Not applicable.

Table 3.6-B. Average amount of financial aid received by undergraduates who received a specific type of aid in public 4-year institutions, by type and source of aid and selected student characteristics: 2007–08

			Any		Any	Any	Any
		Any	work-	Any	federal	state	institution
Student characteristics	Any aid	grant	study	loan amount ¹	aid amount²	aid amount	aid
	amount	amount	amount				amount
Total	\$9,400	\$5,200	\$2,500	\$6,600	\$7,100	\$3,100	\$3,700
Attendance status							
Full-time/full-year	11,000	6,100	2,400	7,100	7,800	3,500	4,000
Full-time/part-year	6,900	3,800	2,300	5,200	5,400	2,300	3,000
Part-time/full-year	7,800	3,700	2,800	6,900	6,700	2,200	3,000
Part-time/part-year	4,600	2,500	3,400	4,800	4,500	1,900	2,300
Housing							
On campus	11,400	6,400	2,200	6,500	7,900	3,500	4,400
Off campus	9,100	4,800	2,600	6,900	7,200	3,100	3,500
Living with parents	6,900	4,500	3,000	5,700	5,400	2,800	2,800
Price of attendance							
Less than \$6,000	2,400	1,600	2,500	3,200	2,300	1,200	1,300
\$6,000-10,999	5,000	3,100	2,500	4,700	4,500	2,300	1,900
\$11,000-18,999	9,300	4,900	2,300	6,600	7,100	3,000	2,800
\$19,000 or more	13,800	7,600	2,700	7,900	8,900	4,100	5,400
Gender							
Male	9,400	5,200	2,600	6,600	7,100	3,100	4,000
Female	9,400	5,100	2,400	6,600	7,200	3,200	3,600
Race/ethnicity ³							
White	9,100	4,800	2,500	6,700	7,100	3,000	3,600
Black	11,000	6,000	2,200	6,700	8,100	3,100	4,800
Hispanic	8,500	5,300	2,500	6,200	6,200	3,200	3,200
Asian	9,900	6,800	2,600	6,100	6,700	4,200	3,900
American Indian	8,100	4,900	‡	5,700	6,300	2,700	3,500
Pacific Islander	10,800	5,200	‡	7,300	7,500	3,000	4,300
Two or more races	10,200	6,300	2,700	6,200	7,400	3,100	3,900
Other	9,000	4,600	‡	7,100	6,400	2,900	‡
Age as of 12/31/07							
18 years or younger	9,700	5,900	2,000	5,500	6,900	3,200	3,900
19–23 years	9,800	5,600	2,500	6,600	7,100	3,300	3,900
24–29 years	8,800	4,100	2,800	7,300	7,500	2,300	2,800
30–39 years	7,800	3,600	2,400	7,200	7,500	2,400	2,300
40 years or older	6,200	3,100	2,900	6,700	7,000	2,100	2,800
Dependency status							
Dependent	9,900	5,700	2,400	6,400	7,000	3,300	4,000
Independent	8,200	4,000	2,600	7,100	7,400	2,400	2,800
Unmarried, no dependents	9,000	4,300	2,700	7,200	7,600	2,600	2,900
Married, no dependents	7,000	3,900	2,400	7,100	6,500	2,400	3,000
Single parent	8,500	4,100	2,400	6,900	7,600	2,300	2,700
Married parent	7,200	3,400	2,900	7,000	7,100	2,200	2,600

Table 3.6-B. Average amount of financial aid received by undergraduates who received a specific type of aid in public 4-year institutions, by type and source of aid and selected student characteristics: 2007–08—Continued

Student characteristics	Any aid amount	Any grant amount	Any work- study amount	Any Ioan amount ¹	Any federal aid amount ²	Any state aid amount	Any institution aid amount
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$11,700	\$7,700	\$2,200	\$5,600	\$7,800	\$3,500	\$3,800
\$20,000-39,999	11,100	7,000	2,300	5,800	7,100	3,500	4,000
\$40,000-59,999	10,100	5,400	2,400	6,300	6,300	3,300	3,800
\$60,000-79,999	9,300	4,400	2,600	6,800	6,400	3,200	3,700
\$80,000-99,999	9,200	4,500	2,700	6,800	6,900	2,900	4,400
\$100,000 or more	8,800	4,600	2,600	6,800	7,500	3,300	4,000
Independent							
Less than \$10,000	10,300	5,300	2,500	7,100	8,300	2,600	3,100
\$10,000-19,999	9,000	3,800	2,800	7,100	7,200	2,300	3,000
\$20,000–29,999	7,700	3,700	2,600	7,100	6,700	2,300	2,300
\$30,000-49,999	7,000	3,000	3,100	6,800	6,800	2,200	2,700
\$50,000 or more	5,400	2,600	3,100	7,500	6,500	2,300	2,400
Income group ⁴							
Lowest 25 percent	11,000	6,700	2,400	6,200	7,700	3,300	3,700
Middle 50 percent	9,000	4,500	2,500	6,700	6,600	3,000	3,700
Highest 25 percent	8,100	4,100	2,700	7,000	7,500	3,200	3,900
Undergraduate class level							
First year	8,800	5,200	2,100	5,500	6,500	3,100	3,700
Second year	9,400	5,300	2,400	6,300	6,900	3,200	3,800
Third year	10,400	5,300	2,500	7,300	7,700	3,300	3,700
Fourth year or more	9,200	5,000	2,800	7,000	7,300	3,100	3,800
Aid status							
No aid	†	†	†	†	†	†	†
Received aid	9,400	5,200	2,500	6,600	7,100	3,100	3,700
Grant status							
No grant	8,600	†	3,000	7,200	6,800	5,400	3,000
Received grant	9,700	5,200	2,400	6,300	7,300	3,100	3,800

Table 3.6-B. Average amount of financial aid received by undergraduates who received a specific type of aid in public 4-year institutions, by type and source of aid and selected student characteristics: 2007–08—Continued

Student characteristics	Any aid amount	Any grant amount	Any work- study amount	Any Ioan amount ¹	Any federal aid amount ²	Any state aid amount	Any institution aid amount
Loan status ¹							
No loan	\$5,600	\$5,200	\$2,800	†	\$3,800	\$3,200	\$4,300
Received loan	11,400	5,200	2,300	6,600	7,800	3,100	3,300

[†] Not applicable.

NOTE: Average aid amounts are calculated only for students receiving a specific type of aid. Those not receiving a specific type of aid (i.e., zero values) are not included in the average for that aid. Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. This table excludes students attending more than one institution. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[‡] Reporting standards not met.

¹ Loans only include loans to students. Excludes federal Parent PLUS loans.

² Includes all federal grants, loans, college work-study awards, and federal Parent PLUS loans. Excludes veterans benefits and education tax credit and tax deduction benefits.

³ Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

⁴ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

Table S3.6-B. Standard errors for table 3.6-B: Average amount of financial aid received by undergraduates who received a specific type of aid in public 4-year institutions, by type and source of aid and selected student characteristics: 2007–08

Student characteristics	Any aid amount	Any grant amount	Any work- study amount	Any Ioan amount	Any federal aid amount	Any state aid amount	Any institution aid amount
Total	\$60	\$50	\$40	\$60	\$60	\$50	\$80
Attendance status							
Full-time/full-year	90	50	40	80	70	50	80
Full-time/part-year	160	120	160	110	140	90	190
Part-time/full-year	120	80	130	100	80	90	170
Part-time/part-year	130	100	210	110	120	170	230
Housing							
On campus	130	90	60	90	100	80	130
Off campus	80	50	60	70	70	70	100
Living with parents	100	70	120	120	80	50	120
Price of attendance							
Less than \$6,000	90	80	430	120	120	80	170
\$6,000–10,999	70	50	160	70	70	90	80
\$11,000–18,999	60	50	60	70	50	50	60
\$19,000 or more	180	110	60	120	140	80	150
Gender							
Male	90	70	60	80	80	50	90
Female	90	60	50	80	70	70	110
Race/ethnicity							
White	80	50	50	70	80	50	90
Black	180	120	100	100	130	80	220
Hispanic	220	160	110	130	130	140	170
Asian	280	230	180	220	240	150	230
American Indian	620	530	†	540	470	270	800
Pacific Islander	1,060	580	†	770	900	520	1,030
Two or more races	360	290	180	320	290	190	270
Other	1,050	730	†	910	570	500	†
Age as of 12/31/07	100	120	00	120	150	00	160
18 years or younger 19–23 years	190 70	130 60	90 40	130 70	150 70	90	160
	120	80		100	100	50 90	80 140
24–29 years 30–39 years	170	100	160 200	160	170	130	240
40 years or older	240	130	320	170	230	130	290
Dependency status							
Dependent	70	50	40	70	70	50	80
Independent	100	60	110	80	90	80	160
Unmarried, no dependents	120	90	150	110	110	120	120
Married, no dependents	190	170	240	210	180	140	270
Single parent	210	110	210	180	170	90	420
Married parent	210	110	240	180	190	130	320

Table S3.6-B. Standard errors for table 3.6-B: Average amount of financial aid received by undergraduates who received a specific type of aid in public 4-year institutions, by type and source of aid and selected student characteristics: 2007–08—Continued

	Any aid	Any grant	Any work- study	Any Ioan	Any federal aid	Any state aid	Any institution aid
Student characteristics	amount	amount	amount	amount	amount	amount	amount
Dependency and income in 2006 Dependent							
Less than \$20,000	\$270	\$150	\$110	\$140	\$150	\$90	\$240
\$20,000-39,999	150	130	80	110	100	60	160
\$40,000-59,999	160	130	80	120	130	80	170
\$60,000-79,999	180	130	120	150	170	130	190
\$80,000-99,999	210	150	180	180	230	130	220
\$100,000 or more	150	130	110	130	190	130	140
Independent							
Less than \$10,000	190	120	110	120	150	100	160
\$10,000–19,999	180	110	250	130	150	140	290
\$20,000–29,999	210	120	310	200	190	150	200
\$30,000-49,999	210	130	360	180	150	140	320
\$50,000 or more	220	150	460	240	200	250	290
Income group							
Lowest 25 percent	130	80	60	80	80	50	120
Middle 50 percent	80	60	60	80	80	70	100
Highest 25 percent	160	120	120	120	190	120	150
Undergraduate class level							
First year	140	80	80	110	100	80	140
Second year	130	110	90	120	140	100	140
Third year	130	100	70	90	100	70	160
Fourth year or more	100	80	60	70	100	60	100
Aid status							
No aid	†	†	†	†	†	†	†
Received aid	60	50	40	60	60	50	80
Grant status							
No grant	100	†	90	90	110	520	200
Received grant	70	50	40	80	60	50	80
Loan status							
No loan	80	80	70	†	70	80	110
Received loan	90	60	40	60	70	50	100

[†] Not applicable.

Table 3.7-A. Percentage of undergraduates in public 2-year institutions receiving financial aid, by type and source of aid and selected student characteristics: 2007–08

			Any		Any	Any	Any
	Any	Any	work-	Any	federal	state	institution
Student characteristics	aid	grants	study	loans ¹	aid ²	aid	aid
Total	47.6	39.6	3.3	13.2	27.4	12.8	11.1
Attendance status							
Full-time/full-year	65.7	55.7	6.9	22.5	47.7	27.0	18.1
Full-time/part-year	53.9	42.7	3.3	17.1	33.5	15.8	9.6
Part-time/full-year	52.1	44.5	3.4	13.8	31.2	13.0	13.4
Part-time/part-year	34.4	27.8	1.7	7.3	13.6	5.4	6.7
Housing							
On campus	65.4	58.0	14.9	33.1	48.2	19.1	22.7
Off campus	49.8	41.8	2.7	14.3	28.4	11.9	10.3
Living with parents	43.5	35.4	4.0	10.7	25.1	14.0	12.0
Price of attendance							
Less than \$6,000	33.9	29.0	0.8	5.8	12.8	5.9	7.9
\$6,000–10,999	56.5	47.4	3.8	15.9	34.9	16.3	13.5
\$11,000–18,999	68.7	55.0	8.0	28.5	53.2	25.8	16.3
\$19,000 or more	77.6	45.1	28.7	35.5	69.3	23.4	8.1
Gender							
Male	42.6	33.7	3.2	11.2	21.2	10.2	10.0
Female	51.6	44.1	3.4	14.6	32.2	14.7	12.0
Race/ethnicity ³							
White	44.9	35.9	2.8	14.3	24.4	12.2	8.2
Black	59.2	52.3	3.5	16.3	40.2	18.1	10.5
Hispanic	51.9	45.6	4.5	8.5	29.8	12.1	21.1
Asian	35.0	29.8	3.7	5.2	18.4	8.8	13.9
American Indian	59.8	50.8	2.9 !	16.9	38.3	13.5	11.0
Pacific Islander	47.3	39.2	3.5 !	9.4	21.1	7.1 !	20.6
Two or more races	48.2	38.8	5.3	13.7	30.1	11.5	13.8
Other	48.0	36.6	‡	12.6 !	25.3	10.5 !	16.7
Age as of 12/31/07							
18 years or younger	50.2	43.7	5.0	12.2	32.1	19.6	15.3
19–23 years	46.3	36.9	4.3	13.6	28.4	14.2	11.9
24–29 years	50.4	42.3	2.6	15.8	30.0	10.5	11.0
30-39 years	51.0	44.3	2.3	13.0	27.7	11.6	9.2
40 years or older	43.0	36.1	1.5	9.4	18.0	8.6	8.3
Dependency status							
Dependent	45.3	36.0	4.6	13.6	27.4	15.0	12.6
Independent	49.4	42.3	2.3	12.8	27.3	11.1	10.0
Unmarried, no dependents	45.9	37.2	2.4	13.5	24.2	9.6	11.6
Married, no dependents	40.1	31.1	2.4	9.2	13.6	6.5	8.0
Single parent	58.0	54.1	2.9	14.6	40.3	15.8	11.1
Married parent	48.6	41.0	1.6	11.8	23.6	10.1	7.9

Table 3.7-A. Percentage of undergraduates in public 2-year institutions receiving financial aid, by type and source of aid and selected student characteristics: 2007–08—Continued

			Any	_	Any	Any	Any
	Any	Any	work-	Any	federal	state	institution
Student characteristics	aid	grants	study	loans ¹	aid ²	aid	aid
Dependency and income in 2006							
Dependent							
Less than \$20,000	67.2	62.9	6.1	10.9	53.1	23.4	20.0
\$20,000-39,999	60.4	55.3	7.3	13.8	46.4	23.8	17.8
\$40,000-59,999	43.7	34.2	4.2	15.3	23.8	14.3	13.7
\$60,000-79,999	35.9	22.1	4.0	15.0	14.8	9.9	9.5
\$80,000-99,999	34.3	19.8	2.8	15.4	12.2	9.0	7.7
\$100,000 or more	27.6	16.4	2.7	11.2	10.0	6.9	5.1
Independent							
Less than \$10,000	60.4	55.1	4.3	16.5	43.8	16.8	16.0
\$10,000-19,999	57.2	50.9	2.8	18.4	42.7	16.2	12.7
\$20,000-29,999	54.9	45.4	2.8	15.7	33.8	13.4	11.2
\$30,000-49,999	47.8	41.3	1.4	11.2	24.0	10.0	9.2
\$50,000 or more	35.3	27.4	1.2	6.7	5.9	3.8	4.2
Income group ⁴							
Lowest 25 percent	62.6	57.8	5.4	14.3	47.5	20.7	17.6
Middle 50 percent	46.5	37.2	2.9	14.9	26.1	12.3	10.8
Highest 25 percent	33.0	24.3	1.7	7.8	7.2	4.7	4.3
Undergraduate class level							
First year	48.0	40.4	3.3	12.2	28.0	12.9	10.4
Second year	52.0	42.7	3.8	16.4	31.1	14.9	13.1
Third year	52.8	43.1	4.6	16.4	31.2	12.1	15.2
Aid status							
No aid	†	†	†	†	†	†	†
Received aid	100.0	83.1	6.9	27.6	57.4	26.8	23.3
Grant status							
No grant	13.3	†	2.0	9.1	7.3	0.9	0.4
Received grant	100.0	100.0	5.3	19.3	58.0	30.9	27.4
Loan status ¹							
No loan	39.7	36.8	3.0	†	19.2	11.1	11.2
Received loan	100.0	58.1	5.2	100.0	81.5	23.7	10.5

[†] Not applicable.

NOTE: Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. This table excludes students attending more than one institution. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto

[!] Interpret data with caution (estimates are unstable).

[‡] Reporting standards not met.

¹ Loans only include loans to students. Excludes federal Parent PLUS loans.

² Includes all federal grants, loans, college work-study awards, and federal Parent PLUS loans. Excludes veterans benefits and education tax credit and tax deduction benefits.

³ Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

⁴ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

Table S3.7-A. Standard errors for table 3.7-A: Percentage of undergraduates in public 2-year institutions receiving financial aid, by type and source of aid and selected student characteristics: 2007–08

			Any		Any	Any	Any
	Any	Any	work-	Any	federal	state	institution
Student characteristics	aid	grants	study	loans	aid	aid	aid
Total	2.42	1.99	0.27	0.41	1.64	0.68	0.90
Attendance status							
Full-time/full-year	1.32	0.95	0.41	0.81	0.82	0.88	0.96
Full-time/part-year	3.62	3.13	0.53	1.54	3.12	1.26	1.67
Part-time/full-year	1.94	1.77	0.40	0.40	1.19	0.65	1.09
Part-time/part-year	2.93	2.42	0.22	0.96	2.20	0.69	0.96
Housing							
On campus	5.26	4.64	2.28	3.71	4.12	2.45	3.94
Off campus	2.41	1.93	0.28	0.42	1.73	0.67	0.89
Living with parents	2.49	2.19	0.34	0.56	1.60	0.80	1.07
Price of attendance							
Less than \$6,000	2.79	2.50	0.16	0.57	1.84	0.74	1.06
\$6,000-10,999	2.05	1.67	0.40	0.57	1.50	0.78	0.83
\$11,000-18,999	1.65	1.15	0.54	1.00	1.16	0.79	1.03
\$19,000 or more	3.28	3.94	4.27	2.57	3.80	3.24	1.67
Gender							
Male	2.60	2.08	0.33	0.58	1.65	0.83	0.98
Female	2.26	1.90	0.27	0.37	1.59	0.60	0.88
Race/ethnicity							
White	2.46	1.99	0.24	0.60	1.67	0.87	0.74
Black	2.88	2.56	0.38	0.78	2.51	1.11	1.22
Hispanic	2.24	2.07	0.57	0.56	1.38	0.70	1.64
Asian	2.94	2.26	0.74	0.86	1.98	1.02	1.93
American Indian	5.47	4.58	1.40	2.64	5.24	2.77	2.42
Pacific Islander	5.37	5.17	1.61	2.19	4.12	2.26	3.73
Two or more races	3.03	2.69	0.87	1.49	1.91	1.17	1.86
Other	7.06	7.20	†	3.82	4.54	3.95	6.21
Age as of 12/31/07							
18 years or younger	2.28	2.04	0.51	0.75	1.58	1.14	1.29
19–23 years	2.60	2.29	0.40	0.60	1.78	0.81	0.94
24–29 years	2.82	2.15	0.32	0.66	1.82	0.78	1.17
30–39 years	3.03	2.59	0.42	0.73	2.21	0.91	1.14
40 years or older	1.93	1.52	0.22	0.47	1.40	0.79	0.88
Dependency status							
Dependent	2.58	2.19	0.37	0.67	1.70	0.82	1.01
Independent	2.37	1.92	0.24	0.35	1.66	0.68	0.92
Unmarried, no dependents	2.82	2.21	0.33	0.85	1.63	0.86	1.29
Married, no dependents	1.74	1.43	0.42	0.89	1.09	0.88	1.16
Single parent	3.49	3.42	0.41	0.56	3.01	0.94	1.30
Married parent	1.91	1.49	0.26	0.61	1.38	0.69	0.76

Table S3.7-A. Standard errors for table 3.7-A: Percentage of undergraduates in public 2-year institutions receiving financial aid, by type and source of aid and selected student characteristics: 2007–08—Continued

			Any		Any	Any	Any
	Any	Any	work-	Any	federal	state	institution
Student characteristics	aid	grants	study	loans	aid	aid	aid
Dependency and income in 2006							
Dependent							
Less than \$20,000	2.24	2.44	0.72	0.78	1.68	1.37	1.39
\$20,000-39,999	2.45	2.33	0.87	0.83	1.94	1.06	1.34
\$40,000-59,999	3.35	2.69	0.61	1.27	2.26	1.24	1.67
\$60,000-79,999	3.15	2.22	0.77	1.63	1.83	1.25	1.28
\$80,000-99,999	2.30	2.09	0.65	1.16	1.36	1.09	1.10
\$100,000 or more	2.26	1.50	0.53	1.20	1.24	0.85	0.82
Independent							
Less than \$10,000	2.77	2.63	0.46	0.76	1.76	0.88	1.48
\$10,000-19,999	3.38	2.93	0.50	0.93	2.28	1.11	1.19
\$20,000-29,999	3.37	2.99	0.43	0.79	2.43	1.12	1.49
\$30,000-49,999	2.69	2.28	0.25	0.60	1.94	0.94	1.12
\$50,000 or more	1.26	0.89	0.24	0.70	0.77	0.49	0.52
Income group							
Lowest 25 percent	2.34	2.30	0.46	0.47	1.49	0.76	1.06
Middle 50 percent	2.91	2.35	0.31	0.64	1.92	0.87	1.05
Highest 25 percent	1.29	0.81	0.25	0.64	0.75	0.47	0.47
Undergraduate class level							
First year	2.55	2.13	0.33	0.63	1.95	0.78	0.83
Second year	2.08	1.68	0.27	0.38	1.21	0.56	1.12
Third year	2.34	2.68	0.79	1.66	2.11	1.55	2.58
Aid status							
No aid	†	†	†	†	†	†	†
Received aid	†	0.54	0.49	0.69	0.72	0.63	0.88
Grant status							
No grant	1.27	†	0.26	0.81	0.87	0.19	0.09
Received grant	†	†	0.42	0.73	0.73	0.69	1.04
Loan status							
No loan	2.54	2.33	0.29	†	1.55	0.82	1.01
Received loan	†	1.31	0.47	†	0.93	1.23	0.69

[†] Not applicable.

Table 3.7-B. Average amount of financial aid received by undergraduates who received a specific type of aid in public 2-year institutions, by type and source of aid and selected student characteristics: 2007–08

Student characteristics	Any aid amount	Any grant amount	Any work- study amount	Any Ioan amount ¹	Any federal aid amount ²	Any state aid amount	Any institution aid amount
Total	\$3,400	\$2,200	\$3,000	\$4,100	\$3,500	\$1,300	\$800
Attendance status							
Full-time/full-year	5,400	3,700	2,600	4,900	4,800	1,700	1,300
Full-time/part-year	3,100	2,000	2,500	3,400	2,800	1,200	800
Part-time/full-year	3,400	2,100	3,200	4,500	3,500	1,200	700
Part-time/part-year	1,800	1,100	3,600	2,900	2,100	700	500
Housing							
On campus	5,500	3,500	1,500	3,600	4,700	1,100	2,300
Off campus	3,400	2,100	2,900	4,300	3,700	1,300	700
Living with parents	3,300	2,400	3,200	3,500	3,200	1,400	900
Price of attendance							
Less than \$6,000	1,500	1,100	1,500	2,500	1,700	800	400
\$6,000-10,999	3,500	2,400	2,700	3,900	3,400	1,300	900
\$11,000–18,999	5,900	3,500	3,200	5,300	4,900	1,700	1,200
\$19,000 or more	7,300	3,800	4,600	5,500	5,300	1,900	1,300
Gender							
Male	3,300	2,100	3,400	4,000	3,500	1,400	800
Female	3,400	2,300	2,700	4,100	3,500	1,300	800
Race/ethnicity ³							
White	3,500	2,100	3,000	4,200	3,700	1,300	1,000
Black	3,500	2,400	2,800	3,900	3,500	1,200	900
Hispanic	2,900	2,200	3,100	3,700	3,200	1,300	600
Asian	3,200	2,500	2,800	3,700	3,400	1,500	700
American Indian	3,500	2,400	‡	4,300	3,600	1,300	‡
Pacific Islander	2,900	2,000	‡	‡	3,200	‡	900
Two or more races	3,900	2,400	3,200	4,400	3,700	1,400	700
Other	2,900	1,700	‡	‡	3,000	‡	‡
Age as of 12/31/07							
18 years or younger	3,600	2,700	2,900	3,400	3,300	1,500	1,100
19–23 years	3,500	2,500	3,000	3,800	3,400	1,300	1,000
24–29 years	3,700	2,000	3,000	4,400	3,800	1,200	600
30–39 years 40 years or older	3,200 2,800	1,900 1,700	2,700 3,000	4,500 4,500	3,600 3,700	1,200 1,200	700 600
	2,000	.,,,,,	2,000	1,000	5,700	1,200	230
Dependency status	2.500	2 500	2 000	2 700	2 400	1 100	4 000
Dependent	3,500	2,500	3,000	3,700	3,400	1,400	1,000
Independent	3,300	2,000 1,800	2,900	4,400	3,600	1,200	600
Unmarried, no dependents	3,400	1,800	2,800	4,400 4,400	3,600	1,300 1,200	600 700
Married, no dependents Single parent	2,900 3,600	2,300	3,500 2,700	4,400	3,900 3,600	1,200	600
Married parent	3,100	1,800	2,700	4,200	3,600	1,200	700

Table 3.7-B. Average amount of financial aid received by undergraduates who received a specific type of aid in public 2-year institutions, by type and source of aid and selected student characteristics: 2007–08—Continued

Student characteristics	Any aid amount	Any grant amount	Any work- study amount	Any Ioan amount ¹	Any federal aid amount ²	Any state aid amount	Any institution aid amount
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$3,900	\$3,400	\$2,500	\$3,100	\$3,600	\$1,500	\$900
\$20,000–39,999	3,800	2,900	2,500	3,500	3,400	1,400	800
\$40,000–59,999	3,200	1,900	2,800	3,800	2,900	1,400	1,000
\$60,000-79,999	3,200	1,600	3,600	4,000	3,400	1,300	1,500
\$80,000–99,999	3,300	1,700	4,700	3,900	3,700	1,500	1,400
\$100,000 or more	3,100	1,800	3,800	3,700	3,400	1,300	1,300
Independent	,	,	•	,	•	•	•
Less than \$10,000	4,000	2,600	2,400	4,400	3,900	1,300	600
\$10,000–19,999	3,800	2,200	2,700	4,000	3,400	1,200	600
\$20,000–29,999	3,600	2,100	3,100	4,600	3,700	1,200	600
\$30,000-49,999	2,800	1,600	3,200	4,500	3,200	1,100	600
\$50,000 or more	2,300	1,200	3,900	4,700	4,300	1,200	900
Income group ⁴							
Lowest 25 percent	3,900	2,900	2,500	3,900	3,700	1,400	700
Middle 50 percent	3,300	1,900	3,200	4,100	3,300	1,300	900
Highest 25 percent	2,500	1,300	3,800	4,300	4,000	1,200	1,000
Undergraduate class level							
First year	3,100	2,100	2,900	3,600	3,200	1,300	800
Second year	3,900	2,400	3,000	4,500	3,900	1,400	900
Third year	3,900	2,400	‡	4,700	4,300	1,400	600
Aid status							
No aid	†	†	†	†	†	†	†
Received aid	3,400	2,200	3,000	4,100	3,500	1,300	800
Grant status							
No grant	4,100	†	3,800	4,000	3,800	1,800	2,500
Received grant	3,300	2,200	2,500	4,100	3,500	1,300	800

Table 3.7-B. Average amount of financial aid received by undergraduates who received a specific type of aid in public 2-year institutions, by type and source of aid and selected student characteristics: 2007–08—Continued

Student characteristics	Any aid amount	Any grant amount	Any work- study amount	Any Ioan amount ¹	Any federal aid amount ²	Any state aid amount	Any institution aid amount
Loan status ¹							
No loan	\$2,400	\$2,100	\$3,100	†	\$2,600	\$1,300	\$800
Received loan	5,900	2,600	2,500	4,100	5,000	1,300	900

[†] Not applicable.

NOTE: Average aid amounts are calculated only for students receiving a specific type of aid. Those not receiving a specific type of aid (i.e., zero values) are not included in the average for that aid. Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. This table excludes students attending more than one institution. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[‡] Reporting standards not met.

¹ Loans only include loans to students. Excludes federal Parent PLUS loans.

² Includes all federal grants, loans, college work-study awards, and federal Parent PLUS loans. Excludes veterans benefits and education tax credit and tax deduction benefits.

³ Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

⁴ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

Table S3.7-B. Standard errors for table 3.7-B: Average amount of financial aid received by undergraduates who received a specific type of aid in public 2-year institutions, by type and source of aid and selected student characteristics: 2007–08

	Any aid	Any grant	Any work- study	Any Ioan	Any federal aid	Any state aid	Any institution aid
Student characteristics	amount	amount	amount	amount	amount	amount	amount
Total	\$100	\$60	\$120	\$120	\$160	\$30	\$30
Attendance status							
Full-time/full-year	180	100	120	130	190	40	60
Full-time/part-year	140	90	310	110	100	60	120
Part-time/full-year	130	50	260	170	190	30	30
Part-time/part-year	50	30	260	60	60	30	40
Housing							
On campus	330	240	120	310	280	100	350
Off campus	110	50	160	160	200	40	30
Living with parents	90	80	160	90	100	30	50
Price of attendance							
Less than \$6,000	30	20	180	50	40	40	20
\$6,000–10,999	80	60	190	100	110	30	40
\$11,000–18,999	230	100	140	150	250	50	60
\$19,000 or more	300	250	230	270	240	130	250
Gender							
Male	80	50	170	110	150	40	40
Female	120	70	130	140	170	30	30
Race/ethnicity							
White	80	40	120	120	170	30	40
Black	170	100	220	150	200	50	90
Hispanic	130	90	330	180	140	50	40
Asian	140	120	310	350	170	100	60
American Indian	290	170	†	400	320	140	†
Pacific Islander	290	240	†	†	350	†	230
Two or more races	260	170	440	430	210	100	100
Other	500	250	†	†	370	†	†
Age as of 12/31/07							
18 years or younger	130	110	210	140	130	60	70
19–23 years	100	80	150	90	110	30	50
24–29 years	140	60	210	190	210	50	30
30–39 years	130	60	290	200	230	60	50
40 years or older	100	50	300	210	270	60	60
Dependency status							
Dependent	90	80	140	70	100	30	50
Independent	110	50	170	180	210	40	30
Unmarried, no dependents	120	70	240	170	190	50	30
Married, no dependents	110	80	390	220	190	100	60
Single parent	170	70	210	230	230	50	30
Married parent	130	60	340	230	250	60	80

Table S3.7-B. Standard errors for table 3.7-B: Average amount of financial aid received by undergraduates who received a specific type of aid in public 2-year institutions, by type and source of aid and selected student characteristics: 2007–08—Continued

	Any aid	Any grant	Any work- study	Any Ioan	Any federal aid	Any state aid	Any institution aid
Student characteristics	amount	amount	amount	amount	amount	amount	amount
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$180	\$150	\$230	\$150	\$140	\$60	\$90
\$20,000-39,999	150	120	260	140	130	40	60
\$40,000-59,999	100	70	260	130	110	60	70
\$60,000-79,999	110	100	330	170	130	70	190
\$80,000-99,999	160	120	370	200	160	210	190
\$100,000 or more	140	130	420	200	190	70	120
Independent							
Less than \$10,000	230	120	180	220	250	60	40
\$10,000-19,999	170	90	330	190	180	50	30
\$20,000-29,999	170	80	290	210	220	60	50
\$30,000-49,999	110	40	330	250	240	60	40
\$50,000 or more	80	50	540	200	250	100	130
Income group							
Lowest 25 percent	180	120	140	160	180	40	40
Middle 50 percent	100	40	140	130	150	30	40
Highest 25 percent	80	50	320	150	180	70	100
Undergraduate class level							
First year	80	60	190	100	130	40	40
Second year	140	70	130	130	200	40	40
Third year	260	160	†	320	280	180	70
Aid status							
No aid	†	†	†	†	†	†	†
Received aid	100	60	120	120	160	30	30
Grant status							
No grant	90	†	190	90	120	200	360
Received grant	110	60	140	170	170	30	30
Loan status							
No loan	40	40	150	†	70	30	30
Received loan	210	130	140	120	260	50	50

[†] Not applicable.

Table 3.8-A. Percentage of undergraduates in private nonprofit 4-year institutions receiving financial aid, by type and source of aid and selected student characteristics: 2007–08

			Any	Any	Any	Any	Any
Student characteristics	Any aid	Any grants	work- study	Any Ioans ¹	federal aid ²	state aid	institution aid
Ottadent enaracteristics	aid	grants	Study	Ioans	aiu	aiu	alu
Total	84.7	73.6	23.2	58.9	62.5	25.3	53.1
Attendance status							
Full-time/full-year	89.4	81.2	31.5	65.0	69.6	31.3	67.5
Full-time/part-year	79.2	64.7	15.5	53.5	54.6	17.4	43.1
Part-time/full-year	84.2	69.4	9.0	57.3	62.3	21.3	27.3
Part-time/part-year	67.5	48.7	3.3	35.6	35.3	7.4	14.8
Housing							
On campus	86.1	78.0	36.9	62.8	65.3	25.8	72.2
Off campus	82.1	67.0	10.6	54.7	57.5	21.0	32.0
Living with parents	88.4	80.5	15.4	59.1	69.9	38.9	54.8
Price of attendance							
Less than \$6,000	49.4	37.5	‡	12.2	9.2	2.6!	4.9
\$6,000-10,999	73.5	54.1	2.7	36.3	41.3	8.9	12.7
\$11,000–18,999	84.2	68.6	7.8	54.8	64.1	22.4	26.8
\$19,000 or more	88.6	79.7	30.9	66.0	68.5	29.7	67.5
Gender							
Male	82.0	70.6	21.8	54.7	57.2	22.4	51.5
Female	86.8	75.9	24.3	62.2	66.6	27.5	54.3
Race/ethnicity ³							
White	83.6	72.7	23.1	57.6	58.9	23.8	57.0
Black	89.0	73.4	20.3	71.4	73.2	29.3	39.4
Hispanic	92.0	82.6	21.6	60.0	78.7	32.2	42.9
Asian	73.7	64.5	29.9	44.0	47.9	19.1	53.0
American Indian	78.6	69.3	16.2 !	51.5	54.6	16.1 !	47.1
Pacific Islander	77.8	71.3	27.4!	50.6	55.7	17.9 !	56.9
Two or more races	88.3	80.8	36.8	69.9	71.6	31.0	60.8
Other	69.2	58.5	16.6 !	44.3	45.2	21.4 !	27.5
Age as of 12/31/07							
18 years or younger	89.1	82.2	33.6	62.9	68.5	29.6	73.8
19–23 years	85.3	76.1	30.2	60.5	63.9	26.7	66.0
24–29 years	82.7	65.7	9.2	60.4	64.0	21.0	26.3
30–39 years	84.3	66.0	3.9	56.8	60.1	22.1	16.1
40 years or older	76.6	63.2	2.2	42.0	43.4	18.5	13.6
Dependency status							
Dependent	85.9	77.2	31.3	61.1	64.5	27.3	68.6
Independent	82.2	66.4	7.2	54.7	58.4	21.4	22.4
Unmarried, no dependents	80.8	65.2	12.2	59.5	61.7	23.2	28.2
Married, no dependents	80.8	60.1	6.6	44.6	47.5	13.8	23.9
Single parent	87.2	77.5	6.1	63.1	70.5	30.0	21.8
Married parent	80.3	61.9	2.7	47.3	49.9	15.9	15.7

Table 3.8-A. Percentage of undergraduates in private nonprofit 4-year institutions receiving financial aid, by type and source of aid and selected student characteristics: 2007–08

—Continued

	Any	Any	Any work-	Any	Any federal	Any state	Any institution
Student characteristics	aid	grants	study	loans ¹	aid ²	aid	aid
Dependency and income in 2006							
Dependent							
Less than \$20,000	95.0	91.0	39.6	65.1	87.2	42.7	64.0
\$20,000–39,999	96.0	91.6	44.5	77.3	89.3	50.0	74.0
\$40,000–59,999	95.3	87.0	42.0	72.7	75.9	39.3	77.3
\$60,000-79,999	90.4	81.0	35.8	68.8	67.3	31.0	76.3
\$80,000-99,999	88.5	80.4	33.1	64.0	63.8	24.4	76.2
\$100,000 or more	75.4	64.3	19.9	47.7	47.7	12.8	59.6
Independent							
Less than \$10,000	85.7	78.7	22.1	64.7	73.2	28.6	38.6
\$10,000-19,999	85.1	75.0	6.6	64.1	76.2	31.3	27.7
\$20,000-29,999	82.6	65.5	4.1	62.0	69.5	26.1	24.6
\$30,000-49,999	85.6	68.3	3.2	59.1	59.9	23.0	17.4
\$50,000 or more	75.8	53.0	1.0	36.8	33.4	8.3	10.9
Income group ⁴							
Lowest 25 percent	91.8	86.8	33.3	69.0	82.8	40.5	57.1
Middle 50 percent	88.8	78.0	26.1	66.1	68.5	29.4	58.2
Highest 25 percent	75.1	60.3	13.6	43.7	42.8	11.1	44.0
Undergraduate class level							
First year	84.8	75.4	23.4	57.5	62.7	25.9	56.3
Second year	87.8	77.2	28.1	61.6	66.3	26.6	58.5
Third year	87.2	75.3	22.2	64.9	67.9	28.1	53.4
Fourth year or more	81.5	69.5	21.2	54.8	56.7	22.3	47.5
Aid status							
No aid	†	†	†	†	†	†	†
Received aid	100.0	86.9	27.4	69.6	73.8	29.9	62.7
Grant status							
No grant	41.9	†	6.7	35.3	31.0	0.8 !	2.0
Received grant	100.0	100.0	29.2	67.4	73.7	34.1	71.4

Table 3.8-A. Percentage of undergraduates in private nonprofit 4-year institutions receiving financial aid, by type and source of aid and selected student characteristics: 2007–08

—Continued

Student characteristics	Any aid	Any grants	Any work- study	Any Ioans ¹	Any federal aid ²	Any state aid	Any institution aid
Loan status ¹							
No loan	62.7	58.4	10.4	†	17.6	11.9	38.0
Received loan	100.0	84.2	32.2	100.0	93.7	34.6	63.6

[†] Not applicable.

NOTE: Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. This table excludes students attending more than one institution. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico

[!] Interpret data with caution (estimates are unstable).

[‡] Reporting standards not met.

¹ Loans only include loans to students. Excludes federal Parent PLUS loans.

² Includes all federal grants, loans, college work-study awards, and federal Parent PLUS loans. Excludes veterans benefits and education tax credit and tax deduction benefits.

³ Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

⁴ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

Table S3.8-A. Standard errors for table 3.8-A: Percentage of undergraduates in private nonprofit 4-year institutions receiving financial aid, by type and source of aid and selected student characteristics: 2007–08

	Any	Any	Any work-	Any	Any federal	Any state	Any institution
Student characteristics	aid	grants	study	loans	aid	aid	aid
Total	0.97	0.93	0.79	0.54	0.52	0.78	1.24
Attendance status							
Full-time/full-year	0.69	0.93	1.03	0.69	0.67	1.08	1.58
Full-time/part-year	2.39	1.77	1.57	1.93	2.00	1.27	2.09
Part-time/full-year	1.29	1.51	0.96	1.74	1.64	1.60	1.52
Part-time/part-year	2.89	3.12	0.58	2.08	1.77	0.96	1.44
Housing							
On campus	1.25	1.28	1.21	0.96	1.04	1.26	1.51
Off campus	1.18	1.25	0.66	0.72	0.88	0.98	1.38
Living with parents	1.11	1.50	1.14	1.71	1.63	1.73	2.23
Price of attendance							
Less than \$6,000	3.46	2.47	†	2.62	1.63	1.29	1.26
\$6,000–10,999	2.41	3.39	0.59	2.09	2.69	1.08	1.68
\$11,000–18,999	2.14	1.76	0.86	1.71	2.27	1.36	1.74
\$19,000 or more	0.73	0.86	1.03	0.59	0.61	1.04	1.55
Gender							
Male	1.34	1.21	1.10	1.22	1.10	1.07	1.68
Female	0.83	0.92	0.85	0.68	0.66	0.83	1.29
Race/ethnicity							
White	1.25	1.15	1.02	0.84	0.79	0.99	1.56
Black	1.13	2.09	1.94	1.52	1.77	1.92	2.03
Hispanic	0.95	1.34	1.63	2.13	1.69	1.59	2.75
Asian	3.14	3.11	2.60	3.12	2.71	2.20	2.97
American Indian	9.18	11.42	7.97	10.78	12.07	7.35	11.79
Pacific Islander	6.33	7.72	8.89	11.61	11.43	6.42	9.40
Two or more races	2.36	3.26	4.61	3.26	3.61	3.48	4.56
Other	10.63	10.12	5.67	9.26	9.36	7.19	7.68
Age as of 12/31/07							
18 years or younger	1.52	1.88	1.82	1.82	1.81	1.43	2.19
19–23 years	1.04	1.12	0.94	0.87	0.84	1.05	1.53
24–29 years	2.04	1.93	1.21	2.15	2.07	1.15	1.59
30-39 years	1.84	2.08	0.80	1.69	2.04	2.39	1.67
40 years or older	2.16	2.18	0.54	1.84	2.33	2.53	1.47
Dependency status							
Dependent	0.97	1.08	0.95	0.73	0.79	1.02	1.50
Independent	1.37	1.32	0.61	0.95	1.09	1.21	1.06
Unmarried, no dependents	1.47	1.62	1.38	1.85	1.83	1.21	1.62
Married, no dependents	1.90	2.38	1.09	2.45	2.31	1.68	2.06
Single parent	1.14	1.67	1.06	2.10	2.07	2.42	1.88
Married parent	3.01	3.30	0.48	2.30	2.64	2.04	1.60

Table S3.8-A. Standard errors for table 3.8-A: Percentage of undergraduates in private nonprofit 4-year institutions receiving financial aid, by type and source of aid and selected student characteristics: 2007–08—Continued

Student characteristics	Any aid	Any grants	Any work- study	Any Ioans	Any federal aid	Any state aid	Any institution aid
Dependency and income in 2006			•				
Dependent T2000							
Less than \$20,000	1.03	1.98	2.85	2.46	1.44	2.58	3.29
\$20,000–39,999	0.72	1.56	2.30	1.68	1.49	1.99	2.43
\$40,000–59,999	0.71	1.73	2.21	2.06	1.95	2.06	2.90
\$60,000–79,999	1.63	1.89	2.16	1.94	1.87	1.79	2.34
\$80,000–99,999	1.73	2.29	2.13	2.35	2.20	1.90	2.40
\$100,000 or more	1.97	2.04	1.06	1.27	1.36	1.43	2.18
Independent	1.01	2.0 .	1.00		1.00	1.10	2.10
Less than \$10,000	2.02	2.34	2.15	2.63	2.29	1.72	2.39
\$10,000–19,999	1.78	1.66	1.20	3.06	2.32	2.13	2.41
\$20,000–29,999	2.19	2.29	0.68	2.31	2.43	2.06	2.09
\$30,000–49,999	2.01	1.84	0.57	2.92	3.16	2.54	1.77
\$50,000 or more	2.26	3.12	0.27	1.88	2.09	1.76	1.40
Income group							
Lowest 25 percent	0.87	1.06	1.55	1.32	1.08	1.21	2.18
Middle 50 percent	0.71	0.93	1.09	0.89	0.83	0.97	1.46
Highest 25 percent	1.87	1.98	0.81	0.90	0.89	1.14	1.70
Undergraduate class level							
First year	1.57	1.78	1.55	1.48	1.53	1.22	2.12
Second year	1.05	1.48	1.31	1.34	1.33	1.42	1.93
Third year	1.27	1.29	1.01	1.38	1.51	1.17	1.50
Fourth year or more	1.24	1.10	0.93	1.06	1.05	0.94	1.39
Aid status							
No aid	†	†	†	†	t	†	†
Received aid	†	0.58	0.87	0.69	0.63	0.84	1.42
Grant status							
No grant	2.29	†	0.94	1.97	1.70	0.24	0.30
Received grant	†	†	0.90	0.76	0.67	0.95	1.35
Loan status							
No loan	2.15	2.04	0.74	ţ	0.94	0.99	1.56
Received loan	†	0.75	1.03	†	0.50	0.98	1.59

[†] Not applicable.

Table 3.8-B. Average amount of financial aid received by undergraduates who received a specific type of aid in private nonprofit 4-year institutions, by type and source of aid and selected student characteristics: 2007–08

Total \$17,400 \$10,20 Attendance status Full-time/full-year 21,100 12,30 Full-time/part-year 11,800 6,50 Part-time/full-year 11,500 5,80 Part-time/part-year 6,300 3,50 Housing 22,000 13,10 Off campus 12,800 6,90 Living with parents 14,900 8,80 Price of attendance 2,600 2,10 Less than \$6,000 2,600 2,10 \$6,000-10,999 5,200 3,10 \$11,000-18,999 8,500 4,30 \$19,000 or more 21,100 12,20 Gender 4 17,300 10,10 Male 17,300 10,10 Female 17,900 10,40 Black 16,000 8,90 Hispanic 14,700 8,40 Asian 18,800 13,60 American Indian 18,500 11,00 Pacific Islander 16,200 8,60 Two or more races 20,900 12,60 <	2,100 1,800 2,200 2,100 2,000 2,200 2,000 1,900 2,100	\$9,100 9,800 7,600 8,400 6,200 9,300 9,000 8,400 3,200 5,100 6,800 9,900 9,200 8,900 8,900	\$8,700 9,600 6,400 7,800 5,000 9,600 8,100 7,400 2,200 4,300 6,200 9,600	\$3,500 3,800 2,200 2,400 1,600 3,900 3,000 3,000 1,500 3,900 3,300	\$9,600 10,600 6,600 6,200 3,600 11,100 7,200 7,400 1,300 2,400 3,100 10,400
Full-time/full-year 21,100 12,30 Full-time/part-year 11,800 6,50 Part-time/full-year 11,500 5,80 Part-time/part-year 6,300 3,50 Housing 22,000 13,10 Off campus 12,800 6,90 Living with parents 14,900 8,80 Price of attendance 2,600 2,10 Less than \$6,000 2,600 2,10 \$6,000-10,999 5,200 3,10 \$11,000-18,999 8,500 4,30 \$19,000 or more 21,100 12,20 Gender 17,300 10,10 Male 17,300 10,10 Female 17,400 10,20 Race/ethnicity³ White 17,900 10,40 Black 16,000 8,90 Hispanic 14,700 8,40 Asian 18,800 13,60 American Indian 18,500 11,00 Pacific Islander 16,200 8,60 Two or more races 20,900 12,60	1,800 2,200 2,100 2,000 2,200 2,000 2,400 1,900 2,100	7,600 8,400 6,200 9,300 9,000 8,400 3,200 5,100 6,800 9,900	6,400 7,800 5,000 9,600 8,100 7,400 2,200 4,300 6,200 9,600	2,200 2,400 1,600 3,900 3,000 3,000 1,000 1,500 3,900	6,600 6,200 3,600 11,100 7,200 7,400 1,300 2,400 3,100 10,400
Full-time/part-year 11,800 6,50 Part-time/full-year 11,500 5,80 Part-time/part-year 6,300 3,50 Housing 22,000 13,10 Off campus 12,800 6,90 Living with parents 14,900 8,80 Price of attendance 2,600 2,10 Less than \$6,000 2,600 2,10 \$6,000-10,999 5,200 3,10 \$11,000-18,999 8,500 4,30 \$19,000 or more 21,100 12,20 Gender Male 17,300 10,10 Female 17,400 10,20 Race/ethnicity³ White 17,900 10,40 Black 16,000 8,90 Hispanic 14,700 8,40 Asian 18,800 13,60 American Indian 18,500 11,00 Pacific Islander 16,200 8,60 Two or more races 20,900 12,60	1,800 2,200 2,100 2,000 2,200 2,000 2,400 1,900 2,100	7,600 8,400 6,200 9,300 9,000 8,400 3,200 5,100 6,800 9,900	6,400 7,800 5,000 9,600 8,100 7,400 2,200 4,300 6,200 9,600	2,200 2,400 1,600 3,900 3,000 3,000 1,000 1,500 3,900	6,600 6,200 3,600 11,100 7,200 7,400 1,300 2,400 3,100 10,400
Full-time/part-year 11,800 6,50 Part-time/full-year 11,500 5,80 Part-time/part-year 6,300 3,50 Housing 22,000 13,10 Off campus 12,800 6,90 Living with parents 14,900 8,80 Price of attendance 2,600 2,10 Less than \$6,000 2,600 2,10 \$6,000-10,999 5,200 3,10 \$11,000-18,999 8,500 4,30 \$19,000 or more 21,100 12,20 Gender Male 17,300 10,10 Female 17,400 10,20 Race/ethnicity³ White 17,900 10,40 Black 16,000 8,90 Hispanic 14,700 8,40 Asian 18,800 13,60 American Indian 18,500 11,00 Pacific Islander 16,200 8,60 Two or more races 20,900 12,60	2,200 2,100 2,000 2,200 2,000 ‡ 2,400 1,900 2,100	9,300 9,000 8,400 3,200 5,100 6,800 9,900	7,800 5,000 9,600 8,100 7,400 2,200 4,300 6,200 9,600	2,400 1,600 3,900 3,000 3,000 \$\pm\$ \$\pm\$ 1,000 1,500 3,900	6,200 3,600 11,100 7,200 7,400 1,300 2,400 3,100 10,400
Part-time/full-year 11,500 5,80 Part-time/part-year 6,300 3,50 Housing 22,000 13,10 Off campus 12,800 6,90 Living with parents 14,900 8,80 Price of attendance 2,600 2,10 Less than \$6,000 2,600 2,10 \$6,000-10,999 5,200 3,10 \$11,000-18,999 8,500 4,30 \$19,000 or more 21,100 12,20 Gender Male 17,300 10,10 Female 17,400 10,20 Race/ethnicity³ White 17,900 10,40 Black 16,000 8,90 Hispanic 14,700 8,40 Asian 18,800 13,60 American Indian 18,500 11,00 Pacific Islander 16,200 8,60 Two or more races 20,900 12,60	2,200 2,100 2,000 2,200 2,000 ‡ 2,400 1,900 2,100	9,300 9,000 8,400 3,200 5,100 6,800 9,900	5,000 9,600 8,100 7,400 2,200 4,300 6,200 9,600 8,800	1,600 3,900 3,000 3,000 ‡ 1,000 1,500 3,900	3,600 11,100 7,200 7,400 1,300 2,400 3,100 10,400
Housing On campus Off campus Living with parents Price of attendance Less than \$6,000 \$6,000-10,999 \$11,000-18,999 \$11,000-18,999 \$19,000 or more Gender Male Female Race/ethnicity³ White Black Hispanic Asian American Indian Pacific Islander Two or more races 12,800 13,10 2,600 2,10 2,600 2,10 3,10 2,600 2,600 2,10 2,600 2,600 2,600 2,600 2,600 2,600 2,600 2,600 2,600 2,600 2,600 2,600 2,600 2,600 2,600 2,600 2,600 2,600 2	2,000 2,200 2,000 ‡ 2,400 1,900 2,100	9,300 9,000 8,400 3,200 5,100 6,800 9,900	9,600 8,100 7,400 2,200 4,300 6,200 9,600	3,900 3,000 3,000 ‡ 1,000 1,500 3,900	11,100 7,200 7,400 1,300 2,400 3,100 10,400
On campus 22,000 13,10 Off campus 12,800 6,90 Living with parents 14,900 8,80 Price of attendance 2,600 2,10 Less than \$6,000 2,600 2,10 \$6,000-10,999 5,200 3,10 \$11,000-18,999 8,500 4,30 \$19,000 or more 21,100 12,20 Gender Male 17,300 10,10 Female 17,400 10,20 Race/ethnicity³ White 17,900 10,40 Black 16,000 8,90 Hispanic 14,700 8,40 Asian 18,800 13,60 American Indian 18,500 11,00 Pacific Islander 16,200 8,60 Two or more races 20,900 12,60	2,200 2,000 ‡ 2,400 1,900 2,100	9,000 8,400 3,200 5,100 6,800 9,900	8,100 7,400 2,200 4,300 6,200 9,600 8,800	3,000 3,000 ‡ 1,000 1,500 3,900	7,200 7,400 1,300 2,400 3,100 10,400
Off campus 12,800 6,90 Living with parents 14,900 8,80 Price of attendance 2,600 2,10 \$6,000-10,999 5,200 3,10 \$11,000-18,999 8,500 4,30 \$19,000 or more 21,100 12,20 Gender 17,300 10,10 Female 17,400 10,20 Race/ethnicity³ White 17,900 10,40 Black 16,000 8,90 Hispanic 14,700 8,40 Asian 18,800 13,60 American Indian 18,500 11,00 Pacific Islander 16,200 8,60 Two or more races 20,900 12,60	2,200 2,000 ‡ 2,400 1,900 2,100	9,000 8,400 3,200 5,100 6,800 9,900	8,100 7,400 2,200 4,300 6,200 9,600 8,800	3,000 3,000 ‡ 1,000 1,500 3,900	7,200 7,400 1,300 2,400 3,100 10,400
Living with parents 14,900 8,80 Price of attendance 2,600 2,10 \$6,000-10,999 5,200 3,10 \$11,000-18,999 8,500 4,30 \$19,000 or more 21,100 12,20 Gender 30 10,10 Male 17,300 10,10 Female 17,400 10,20 Race/ethnicity³ 30 10,40 Black 16,000 8,90 Hispanic 14,700 8,40 Asian 18,800 13,60 American Indian 18,500 11,00 Pacific Islander 16,200 8,60 Two or more races 20,900 12,60	2,000 ‡ 2,400 1,900 2,100	3,200 5,100 6,800 9,900	7,400 2,200 4,300 6,200 9,600 8,800	3,000 ‡ 1,000 1,500 3,900	7,400 1,300 2,400 3,100 10,400
Price of attendance Less than \$6,000 2,600 2,10 \$6,000-10,999 5,200 3,10 \$11,000-18,999 8,500 4,30 \$19,000 or more 21,100 12,20 Gender Male 17,300 10,10 Female 17,400 10,20 Race/ethnicity³ White 17,900 10,40 Black 16,000 8,90 Hispanic 14,700 8,40 Asian 18,800 13,60 American Indian 18,500 11,00 Pacific Islander 16,200 8,60 Two or more races 20,900 12,60	‡ 2,400 1,900 2,100	3,200 5,100 6,800 9,900	2,200 4,300 6,200 9,600	‡ 1,000 1,500 3,900	1,300 2,400 3,100 10,400
Less than \$6,000 2,600 2,10 \$6,000-10,999 5,200 3,10 \$11,000-18,999 8,500 4,30 \$19,000 or more 21,100 12,20 Gender Male 17,300 10,10 Female 17,400 10,20 Race/ethnicity³ White 17,900 10,40 Black 16,000 8,90 Hispanic 14,700 8,40 Asian 18,800 13,60 American Indian 18,500 11,00 Pacific Islander 16,200 8,60 Two or more races 20,900 12,60	2,400 1,900 2,100 2,000	5,100 6,800 9,900 9,200	4,300 6,200 9,600 8,800	1,000 1,500 3,900	2,400 3,100 10,400
\$6,000–10,999	2,400 1,900 2,100 2,000	5,100 6,800 9,900 9,200	4,300 6,200 9,600 8,800	1,000 1,500 3,900	2,400 3,100 10,400
\$11,000–18,999 8,500 4,30 \$19,000 or more 21,100 12,20 Gender Male 17,300 10,10 Female 17,400 10,20 Race/ethnicity³ White 17,900 10,40 Black 16,000 8,90 Hispanic 14,700 8,40 Asian 18,800 13,60 American Indian 18,500 11,00 Pacific Islander 16,200 8,60 Two or more races 20,900 12,60	1,900 2,100 2,000	6,800 9,900 9,200	6,200 9,600 8,800	1,500 3,900	3,100 10,400
\$19,000 or more 21,100 12,20 Gender Male 17,300 10,10 Female 17,400 10,20 Race/ethnicity³ White 17,900 10,40 Black 16,000 8,90 Hispanic 14,700 8,40 Asian 18,800 13,60 American Indian 18,500 11,00 Pacific Islander 16,200 8,60 Two or more races 20,900 12,60	2,100	9,900 9,200	9,600 8,800	3,900	10,400
Gender Male 17,300 10,10 Female 17,400 10,20 Race/ethnicity³ White 17,900 10,40 Black 16,000 8,90 Hispanic 14,700 8,40 Asian 18,800 13,60 American Indian 18,500 11,00 Pacific Islander 16,200 8,60 Two or more races 20,900 12,60	2,000	9,200	8,800		·
Male 17,300 10,10 Female 17,400 10,20 Race/ethnicity³ White 17,900 10,40 Black 16,000 8,90 Hispanic 14,700 8,40 Asian 18,800 13,60 American Indian 18,500 11,00 Pacific Islander 16,200 8,60 Two or more races 20,900 12,60	•	•	-	3 300	0.600
Female 17,400 10,20 Race/ethnicity³ White 17,900 10,40 Black 16,000 8,90 Hispanic 14,700 8,40 Asian 18,800 13,60 American Indian 18,500 11,00 Pacific Islander 16,200 8,60 Two or more races 20,900 12,60	•	•	-	3 300	0.600
Race/ethnicity ³ White 17,900 10,40 Black 16,000 8,90 Hispanic 14,700 8,40 Asian 18,800 13,60 American Indian 18,500 11,00 Pacific Islander 16,200 8,60 Two or more races 20,900 12,60	2,100	8,900	0.600	5,555	9,600
White 17,900 10,40 Black 16,000 8,90 Hispanic 14,700 8,40 Asian 18,800 13,60 American Indian 18,500 11,00 Pacific Islander 16,200 8,60 Two or more races 20,900 12,60			8,600	3,500	9,600
Black 16,000 8,90 Hispanic 14,700 8,40 Asian 18,800 13,60 American Indian 18,500 11,00 Pacific Islander 16,200 8,60 Two or more races 20,900 12,60					
Hispanic 14,700 8,40 Asian 18,800 13,60 American Indian 18,500 11,00 Pacific Islander 16,200 8,60 Two or more races 20,900 12,60	2,000	9,300	8,700	3,500	9,700
Asian 18,800 13,60 American Indian 18,500 11,00 Pacific Islander 16,200 8,60 Two or more races 20,900 12,60	2,100	8,600	9,300	3,400	8,800
American Indian 18,500 11,00 Pacific Islander 16,200 8,60 Two or more races 20,900 12,60	2,100	8,500	7,700	2,900	8,600
Pacific Islander 16,200 8,60 Two or more races 20,900 12,60	2,400	8,400	8,800	4,500	12,000
Two or more races 20,900 12,60	‡	10,700	9,200	‡	10,100
	‡	8,500	8,800	‡	7,500
Other 15,100 9,90	2,100	8,200	10,100	4,100	10,700
	‡	9,200	8,900	‡	‡
Age as of 12/31/07					
18 years or younger 20,700 13,00	•	7,500	9,200	3,800	10,600
19–23 years 20,100 11,80		9,700	9,100	3,700	10,200
24–29 years 11,900 5,60		9,200	7,900	2,800	5,200
30–39 years 9,800 4,80		8,100	7,500	2,400	3,800
40 years or older 8,500 4,50	2,700	7,900	7,300	2,400	3,900
Dependency status			.		
Dependent 20,500 12,20		9,300	9,100	3,700	10,400
Independent 10,800 5,60		8,500	7,800	2,700	5,200
Unmarried, no dependents 13,300 6,70		9,500	8,500	3,100	6,100
Married, no dependents 9,600 5,40		9,000	7,200	2,700	5,800
Single parent 10,700 5,30 Married parent 8,600 4,60	1,800	7,700 7,800	7,900 7,000	2,500 2,400	4,500 3,800

Table 3.8-B. Average amount of financial aid received by undergraduates who received a specific type of aid in private nonprofit 4-year institutions, by type and source of aid and selected student characteristics: 2007–08—Continued

Student characteristics	Any aid amount	Any grant amount	Any work- study amount	Any Ioan amount ¹	Any federal aid amount ²	Any state aid amount	Any institution aid amount
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$19,500	\$13,300	\$2,000	\$7,600	\$9,600	\$4,100	\$9,700
\$20,000–39,999	23,200	14,700	2,100	8,400	10,300	4,300	10,500
\$40,000–59,999	22,400	13,200	2,100	9,900	8,900	4,200	10,900
\$60,000-79,999	20,900	12,100	2,100	9,300	8,300	3,300	10,900
\$80,000-99,999	21,400	12,000	1,900	10,000	8,300	3,100	10,800
\$100,000 or more	18,400	10,300	2,100	9,600	9,100	3,100	9,800
Independent							
Less than \$10,000	14,600	7,800	2,100	8,400	9,200	3,300	6,500
\$10,000-19,999	12,100	5,600	2,100	9,000	7,800	2,600	5,100
\$20,000–29,999	11,200	5,400	2,600	8,400	7,400	2,600	4,700
\$30,000-49,999	9,700	4,300	2,400	8,300	7,300	2,300	3,700
\$50,000 or more	7,800	4,400	‡	8,600	6,600	2,500	4,400
Income group ⁴							
Lowest 25 percent	19,000	11,900	2,100	8,200	9,700	4,000	9,200
Middle 50 percent	18,100	10,200	2,100	9,300	8,300	3,300	10,000
Highest 25 percent	15,000	8,700	2,100	9,400	8,400	3,000	9,400
Undergraduate class level							
First year	17,000	10,600	1,900	7,600	8,200	3,300	9,800
Second year	18,300	10,700	2,100	9,200	8,500	3,600	9,900
Third year	18,500	10,300	2,100	9,900	9,500	3,600	9,700
Fourth year or more	16,100	9,300	2,300	9,500	8,600	3,300	9,200
Aid status							
No aid	†	†	†	†	†	†	†
Received aid	17,400	10,200	2,100	9,100	8,700	3,500	9,600
Grant status							
No grant	11,200	†	2,700	9,900	7,900	‡	2,700
Received grant	18,300	10,200	2,000	8,900	8,800	3,400	9,700

Table 3.8-B. Average amount of financial aid received by undergraduates who received a specific type of aid in private nonprofit 4-year institutions, by type and source of aid and selected student characteristics: 2007–08—Continued

Student characteristics	Any aid amount	Any grant amount	Any work- study amount	Any Ioan amount ¹	Any federal aid amount ²	Any state aid amount	Any institution aid amount
Loan status ¹							
No loan	\$9,300	\$8,900	\$2,400	†	\$4,200	\$3,000	\$9,200
Received loan	20,900	10,800	2,000	9,100	9,300	3,600	9,800

[†] Not applicable.

NOTE: Average aid amounts are calculated only for students receiving a specific type of aid. Those not receiving a specific type of aid (i.e., zero values) are not included in the average for that aid. Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. This table excludes students attending more than one institution. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[‡] Reporting standards not met.

¹ Loans only include loans to students. Excludes federal Parent PLUS loans.

² Includes all federal grants, loans, college work-study awards, and federal Parent PLUS loans. Excludes veterans benefits and education tax credit and tax deduction benefits.

³ Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

⁴ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

Table S3.8-B. Standard errors for table 3.8-B: Average amount of financial aid received by undergraduates who received a specific type of aid in private nonprofit 4-year institutions, by type and source of aid and selected student characteristics: 2007–08

Student characteristics	Any aid amount	Any grant amount	Any work- study amount	Any Ioan amount	Any federal aid amount	Any state aid amount	Any institution aid amount
Total	\$260	\$170	\$40	\$130	\$140	\$100	\$140
Attendance status							
Full-time/full-year	310	200	40	160	150	110	160
Full-time/part-year	660	420	100	440	250	130	480
Part-time/full-year	350	220	120	190	270	160	310
Part-time/part-year	250	210	300	220	210	160	330
Housing							
On campus	310	220	50	210	200	120	180
Off campus	260	190	70	140	160	130	260
Living with parents	440	320	100	240	190	180	280
Price of attendance							
Less than \$6,000	120	160	†	190	250	†	180
\$6,000-10,999	190	140	350	160	150	130	260
\$11,000–18,999	160	100	220	150	130	130	150
\$19,000 or more	270	190	30	150	150	90	160
Gender							
Male	330	200	40	220	200	140	190
Female	340	210	50	130	150	100	170
Race/ethnicity							
White	320	180	50	160	170	90	160
Black	430	350	100	180	220	190	460
Hispanic	700	390	90	450	280	230	320
Asian	780	670	190	370	300	290	610
American Indian	2,740	1,680	†	2,010	1,960	†	1,010
Pacific Islander	2,150	1,180	†	1,620	700	†	660
Two or more races	820	640	130	680	510	300	680
Other	2,030	1,990	†	1,000	840	†	†
Age as of 12/31/07							
18 years or younger	430	370	100	300	320	160	300
19–23 years	300	190	40	190	180	110	160
24–29 years	350	220	150	190	240	190	270
30–39 years	290	180	280	220	220	210	350
40 years or older	280	220	410	250	250	380	410
Dependency status							
Dependent	290	190	40	170	160	110	150
Independent	220	140	120	140	160	160	220
Unmarried, no dependents	420	310	180	230	260	200	330
Married, no dependents	370	370	290	330	240	300	520
Single parent Married parent	370 270	200 180	250 260	260 220	240 230	250 210	360 320

Table S3.8-B. Standard errors for table 3.8-B: Average amount of financial aid received by undergraduates who received a specific type of aid in private nonprofit 4-year institutions, by type and source of aid and selected student characteristics: 2007–08

—Continued

	Any aid	Any grant	Any work- study	Any Ioan	Any federal aid	Any state aid	Any institution aid
Student characteristics	amount	amount	amount	amount	amount	amount	amount
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$650	\$500	\$120	\$360	\$310	\$220	\$390
\$20,000–39,999	620	490	130	280	300	180	300
\$40,000–59,999	530	420	60	610	350	170	310
\$60,000–79,999	550	410	70	300	310	160	390
\$80,000-99,999	560	410	80	410	290	250	300
\$100,000 or more	390	220	60	250	320	180	200
Independent							
Less than \$10,000	550	340	160	300	290	260	360
\$10,000-19,999	530	330	190	300	260	210	470
\$20,000-29,999	400	300	400	320	220	200	510
\$30,000-49,999	250	180	380	220	210	200	420
\$50,000 or more	310	230	†	330	220	370	460
Income group							
Lowest 25 percent	490	360	60	220	200	170	200
Middle 50 percent	270	200	50	180	160	90	210
Highest 25 percent	370	210	60	200	270	160	210
Undergraduate class level							
First year	430	320	50	260	250	150	250
Second year	430	300	80	300	240	170	250
Third year	330	230	60	190	200	130	260
Fourth year or more	330	200	60	170	150	80	210
Aid status							
No aid	†	†	†	†	†	†	†
Received aid	260	170	40	130	140	100	140
Grant status							
No grant	320	†	210	290	310	†	260
Received grant	280	170	30	140	150	100	140
Loan status							
No loan	220	220	100	†	200	160	210
Received loan	300	190	40	130	140	90	170

[†] Not applicable.

Table 3.9-A. Percentage of undergraduates in for-profit institutions receiving financial aid, by type and source of aid and selected student characteristics: 2007–08

	Any	Any	Any work-	Any	Any federal	Any state	Any institution
Student characteristics	aid	grants	study	loans ¹	aid ²	aid	aid
Total	96.3	70.4	2.0	91.6	93.0	8.1	8.6
Attendance status							
Full-time/full-year	96.7	71.9	2.0	91.6	93.1	9.5	8.2
Full-time/part-year	95.1	68.9	2.0	90.0	91.5	6.7	9.5
Part-time/full-year	98.3	74.0	2.4	94.4	96.5	10.0	10.3
Part-time/part-year	96.5	67.5	1.8 !	92.2	93.1	6.1	6.2
Housing							
On campus	93.5	60.2	‡	93.3	91.6	12.1	21.8 !
Off campus	96.7	72.6	1.9	92.5	93.6	7.0	8.6
Living with parents	95.1	62.8	2.4	87.7	90.7	12.0	8.0
Price of attendance							
Less than \$6,000	76.2	54.6	#	54.9	56.4	3.8 !	5.3 !
\$6,000-10,999	94.3	71.4	1.4 !	85.7	90.1	8.1	‡
\$11,000-18,999	97.4	73.7	2.3	91.2	95.1	10.0	9.3
\$19,000 or more	97.3	69.0	2.2	95.2	94.3	7.1	9.7
Gender							
Male	96.5	63.5	2.4	91.3	92.8	8.0	8.7
Female	96.3	73.5	1.9	91.7	93.1	8.1	8.5
Race/ethnicity ³							
White	95.8	65.8	2.0	92.3	91.8	7.0	8.1
Black	98.1	79.0	2.1	95.6	95.9	7.3	9.4
Hispanic	95.4	73.6	1.8 !	84.1	92.1	11.9	8.2
Asian	94.9	52.5	‡	92.0	92.1	8.7	9.8 !
American Indian	98.3	79.7	‡	96.7	97.0	‡	13.1 !
Pacific Islander	98.5	75.0	‡	98.1	95.8	‡	‡
Two or more races	96.4	64.6	‡	94.0	93.4	3.7	11.8 !
Other	96.1	73.8	#	94.6	92.9	‡	‡
Age as of 12/31/07							
18 years or younger	95.3	71.3	5.4 !	86.6	91.8	16.2	11.1
19–23 years	95.8	66.3	2.7	90.8	92.4	8.9	8.4
24–29 years	96.9	75.8	1.6 !	92.6	94.1	7.5	9.0
30–39 years	96.8	72.0	1.1 !	92.9	93.7	6.2	7.8
40 years or older	96.1	65.9	1.4 !	90.9	91.2	7.2	8.4 !
Dependency status							
Dependent	94.8	58.1	3.3	88.6	90.7	10.7	9.8
Independent	96.8	74.3	1.7	92.5	93.7	7.3	8.2
Unmarried, no dependents	96.5	64.0	2.0	92.7	93.3	7.3	8.8
Married, no dependents	94.9	43.1	4.5 !	90.9	88.7	2.7	8.9 !
Single parent	97.5	90.5	1.3	93.0	95.6	9.2	6.7
Married parent	96.7	68.3	‡	91.8	92.5	5.3	9.8 !

Table 3.9-A. Percentage of undergraduates in for-profit institutions receiving financial aid, by type and source of aid and selected student characteristics: 2007–08—Continued

	Any	Any	Any work-	Any	Any federal	Any state	Any institution
Student characteristics	aid	grants	study	loans ¹	aid ²	aid	aid
Dependency and income in 2006							
Dependent							
Less than \$20,000	98.2	92.8	2.1 !	84.1	96.5	14.5	8.4
\$20,000-39,999	96.6	76.8	3.0 !	91.1	94.0	12.4	9.9
\$40,000-59,999	96.2	53.9	‡	93.1	91.3	14.1	12.8 !
\$60,000-79,999	92.7	14.3 !	‡	91.1	87.5	4.4 !	8.6 !
\$80,000-99,999	93.8	16.6 !	8.6 !	91.6	88.0	‡	11.0 !
\$100,000 or more	82.7	8.6	3.5 !	79.8	72.9	2.1 !	8.1
Independent							
Less than \$10,000	96.8	91.2	1.8	90.6	94.5	10.3	7.7
\$10,000–19,999	98.1	87.2	2.0	94.3	96.4	8.3	6.9
\$20,000-29,999	97.4	66.4	2.5 !	95.0	94.9	6.3	8.4
\$30,000-49,999	97.1	70.2	1.2 !	93.7	93.7	5.3	11.1 !
\$50,000 or more	93.7	30.1	‡	88.6	85.7	2.8 !	6.6 !
Income group ⁴							
Lowest 25 percent	97.0	89.4	2.2	89.7	94.8	11.2	7.8
Middle 50 percent	96.9	67.0	2.1	93.9	93.9	7.1	9.6
Highest 25 percent	91.9	27.5	1.1 !	87.4	83.8	2.8	6.8 !
Undergraduate class level							
First year	95.6	71.6	1.6	89.5	91.7	7.0	7.5
Second year	98.1	70.4	3.2	95.9	96.3	11.4	9.7
Third year	98.7	68.2	2.6 !	96.1	96.0	6.2	15.8
Fourth year or more	98.3	62.0	1.5 !	95.8	95.4	10.9	7.3
Aid status							
No aid	†	†	†	†	†	†	†
Received aid	100.0	73.1	2.1	95.0	96.5	8.4	8.9
Grant status							
No grant	87.6	†	1.7	86.7	80.5	8.0	‡
Received grant	100.0	100.0	2.2	93.6	98.2	11.1	11.6

Table 3.9-A. Percentage of undergraduates in for-profit institutions receiving financial aid, by type and source of aid and selected student characteristics: 2007–08—Continued

Student characteristics	Any aid	Any grants	Any work- study	Any Ioans ¹	Any federal aid ²	Any state aid	Any institution aid
Loan status ¹							
No loan	56.8	53.5	0.6 !	†	41.7	13.0	7.3 !
Received loan	100.0	72.0	2.2	100.0	97.7	7.6	8.7

[†] Not applicable.

NOTE: Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. This table excludes students attending more than one institution. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[#] Rounds to zero.

[!] Interpret data with caution (estimates are unstable).

[‡] Reporting standards not met.

¹ Loans only include loans to students. Excludes federal Parent PLUS loans.

² Includes all federal grants, loans, college work-study awards, and federal Parent PLUS loans. Excludes veterans benefits and education tax credit and tax deduction benefits.

³ Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

⁴ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

Table S3.9-A. Standard errors for table 3.9-A: Percentage of undergraduates in for-profit institutions receiving financial aid, by type and source of aid and selected student characteristics: 2007–08

			Any		Any	Any	Any
	Any	Any	work-	Any	federal	state	institution
Student characteristics	aid	grants	study	loans	aid	aid	aid
Total	0.25	0.91	0.29	0.26	0.38	0.64	1.70
Attendance status							
Full-time/full-year	0.33	1.38	0.35	0.60	0.68	1.16	2.01
Full-time/part-year	0.54	1.80	0.41	0.98	0.94	0.80	2.76
Part-time/full-year	0.45	2.41	0.53	1.47	0.83	1.10	1.93
Part-time/part-year	0.63	2.47	0.76	1.10	1.14	1.11	1.45
Housing							
On campus	2.77	7.69	†	2.78	2.87	2.89	7.28
Off campus	0.33	0.82	0.33	0.34	0.48	0.68	1.93
Living with parents	0.60	2.07	0.52	0.91	0.87	1.25	1.34
Price of attendance							
Less than \$6,000	3.75	5.15	†	5.15	5.41	1.58	2.44
\$6,000-10,999	0.61	2.47	0.55	1.16	0.93	1.23	2.19
\$11,000–18,999	0.34	1.23	0.45	0.75	0.53	0.94	1.50
\$19,000 or more	0.30	1.27	0.35	0.48	0.53	0.96	2.26
Gender							
Male	0.38	2.40	0.47	0.84	0.72	0.84	1.94
Female	0.36	0.79	0.34	0.55	0.58	0.76	1.73
Race/ethnicity							
White	0.44	1.21	0.37	0.79	0.78	0.66	1.83
Black	0.32	1.84	0.44	0.58	0.53	0.92	2.40
Hispanic	0.49	2.63	0.56	1.53	0.92	1.68	1.67
Asian	1.15	5.56	†	1.75	1.62	2.15	3.78
American Indian	1.57	10.02	†	2.71	2.88	†	5.60
Pacific Islander	1.11	13.13	†	1.22	1.99	†	†
Two or more races	1.32	5.20	†	1.83	2.14	1.00	5.88
Other	3.70	10.12	†	4.33	3.11	†	†
Age as of 12/31/07							
18 years or younger	0.86	3.71	1.91	1.12	1.05	2.35	1.93
19–23 years	0.34	1.69	0.45	0.45	0.52	0.72	1.39
24–29 years	0.37	1.14	0.57	0.51	0.46	1.07	2.41
30-39 years	0.37	1.63	0.43	0.52	0.57	0.80	2.08
40 years or older	0.65	2.50	0.53	1.15	1.20	1.23	2.95
Dependency status							
Dependent	0.45	1.98	0.54	0.68	0.66	1.19	1.78
Independent	0.26	0.74	0.30	0.31	0.40	0.71	1.89
Unmarried, no dependents	0.38	1.54	0.60	0.58	0.54	0.78	1.64
Married, no dependents	1.02	2.74	1.66	1.22	1.55	0.60	3.09
Single parent	0.33	1.37	0.28	0.54	0.43	1.17	1.41
Married parent	0.51	2.76	†	0.71	0.88	0.87	4.61

Table S3.9-A. Standard errors for table 3.9-A: Percentage of undergraduates in for-profit institutions receiving financial aid, by type and source of aid and selected student characteristics: 2007–08—Continued

Charles to be a contacting	Any	Any	Any work-	Any	Any federal	Any state	Any institution
Student characteristics	aid	grants	study	loans	aid	aid	aid
Dependency and income in 2006							
Dependent							
Less than \$20,000	0.44	1.46	0.73	1.80	0.64	2.04	1.64
\$20,000-39,999	0.71	2.62	1.09	1.13	0.96	1.78	2.73
\$40,000-59,999	0.99	3.78	†	1.12	1.47	2.20	3.86
\$60,000-79,999	1.62	4.52	†	1.73	2.45	1.87	3.61
\$80,000-99,999	2.26	5.09	3.05	2.58	3.29	†	5.02
\$100,000 or more	3.18	1.97	1.25	3.56	4.26	0.71	1.97
Independent							
Less than \$10,000	0.39	0.80	0.48	0.76	0.67	1.48	1.60
\$10,000-19,999	0.32	1.16	0.55	0.55	0.44	1.15	1.68
\$20,000-29,999	0.46	2.49	0.91	0.66	0.71	0.81	1.81
\$30,000-49,999	0.54	2.12	0.49	0.71	0.77	0.83	3.91
\$50,000 or more	0.98	2.88	†	1.52	1.80	0.87	2.35
Income group							
Lowest 25 percent	0.29	0.66	0.43	0.61	0.50	1.10	1.29
Middle 50 percent	0.31	1.38	0.40	0.39	0.44	0.61	2.19
Highest 25 percent	1.11	2.56	0.49	1.58	1.95	0.72	2.10
Undergraduate class level							
First year	0.33	1.10	0.29	0.42	0.55	0.64	1.56
Second year	0.30	2.14	0.67	0.48	0.41	1.49	2.03
Third year	0.33	2.63	1.19	0.72	0.55	0.80	4.46
Fourth year or more	0.51	3.33	0.51	0.79	0.78	2.33	1.83
Aid status							
No aid	t	†	†	†	†	†	†
Received aid	†	0.86	0.29	0.22	0.23	0.66	1.76
Grant status							
No grant	0.73	†	0.46	0.75	0.95	0.20	†
Received grant	†	†	0.33	0.30	0.19	0.90	2.15
Loan status							
No loan	2.34	2.53	0.21	†	3.21	1.79	2.57
Received loan	†	0.87	0.31	†	0.19	0.65	1.86

[†] Not applicable.

Table 3.9-B. Average amount of financial aid received by undergraduates who received a specific type of aid in for-profit institutions, by type and source of aid and selected student characteristics: 2007–08

Student characteristics	Any aid amount	Any grant amount	Any work- study amount	Any Ioan amount ¹	Any federal aid amount ²	Any state aid amount	Any institution aid amount
Total	\$10,800	\$3,200	\$3,500	\$8,100	\$7,500	\$2,600	\$2,100
Attendance status							
Full-time/full-year	13,100	4,000	3,600	9,600	8,900	3,000	2,200
Full-time/part-year	9,900	2,900	3,800	7,400	7,000	2,700	2,500
Part-time/full-year	11,400	3,100	2,800	8,800	7,900	2,400	1,800
Part-time/part-year	7,200	1,700	3,200	5,800	4,900	1,600	1,100
Housing							
On campus	22,500	4,500	‡	15,200	11,500	3,200	2,000 !
Off campus	10,600	3,100	3,400	8,100	7,400	2,600	2,100
Living with parents	11,100	3,400	3,600	7,800	7,900	2,700	2,000
Price of attendance							
Less than \$6,000	3,300	1,500	‡	3,000	2,600	‡	‡
\$6,000-10,999	5,800	1,800	‡	4,800	4,600	1,300	900 !
\$11,000–18,999	9,400	2,900	3,400	7,200	6,900	2,300	1,900
\$19,000 or more	13,400	3,800	3,800	9,800	8,800	3,400	2,400
Gender							
Male	11,800	3,400	3,800	8,800	7,800	2,600	2,700
Female	10,400	3,100	3,300	7,800	7,400	2,600	1,900
Race/ethnicity ³							
White	10,900	3,000	3,200	8,400	7,500	3,100	1,600
Black	10,600	3,000	3,500	7,900	7,300	2,300	1,800
Hispanic	10,600	3,500	4,000	7,800	7,600	2,200	3,200
Asian	11,700	3,900	‡	8,600	7,500	3,700 !	2,700
American Indian	9,800	4,200	‡	5,800	6,700	‡	‡
Pacific Islander	11,900	2,600	‡	8,300	8,800	‡	‡
Two or more races	11,400	2,800	‡	9,100	8,000	2,800	‡
Other	9,200	2,900	‡	6,800	7,000	‡	‡
Age as of 12/31/07							
18 years or younger	12,100	3,700	2,900	7,800	8,900	2,100	1,300
19–23 years	11,100	3,200	4,000	7,900	7,900	2,600	2,600
24–29 years	10,900	3,100	2,800	8,400	7,400	2,700	2,000
30–39 years 40 years or older	10,400 10,300	3,100 3,400	4,000 ‡	8,200 8,100	7,100 6,900	2,600 3,000	1,900 1,800
•	10,300	0,400	+	3,100	0,000	3,000	1,000
Dependent	12 100	3,400	3 600	g 400	8 400	2 700	2 200
Dependent Independent	12,100	3,400 3,100	3,600	8,400 8,100	8,400 7,200	2,700	2,300
Unmarried, no dependents	10,400 10,800	3,100	3,400 3,500	8,700	7,200 7,200	2,600 2,700	2,000 2,300
Married, no dependents	10,800	3,100	3,500	9,200	7,200 7,100	3,600	2,300
Single parent	10,000	3,100	3,300	7,300	7,100	2,400	1,800
Married parent	10,600	3,100	3,300 ‡	8,300	7,400	2,400	2,000 !

Table 3.9-B. Average amount of financial aid received by undergraduates who received a specific type of aid in for-profit institutions, by type and source of aid and selected student characteristics: 2007–08—Continued

Student characteristics	Any aid amount	Any grant amount	Any work- study amount	Any Ioan amount ¹	Any federal aid amount ²	Any state aid amount	Any institution aid amount
Dependency and income in 2006							
Dependent Dependent							
Less than \$20,000	\$10,200	\$3,700	\$3,000	\$6,800	\$7,700	\$2,300	\$1,500
\$20,000–39,999	12,200	3,500	2,500	8,100	8,700	2,900	1,800
\$20,000 <u></u> —39,999 \$40,000 <u></u> —59,999	12,200	2,400		8,700	8,100	2,500	2,900
\$60,000 - 79,999	12,400	2,400	‡ +	8,700	7,900	2,600	2,900 !
\$80,000-79,999	13,700	2,000 !	‡ +	9,500	9,400		3,000 !
\$100,000 or more	•	2,100 ! 2,600	‡	9,500	9,400	‡ 2,800	4,000 !
Independent	15,700	2,600	‡	11,200	9,800	2,800	4,000 !
•	10.200	2 200	2 900	7,400	7 600	2 400	1 600
Less than \$10,000	10,200	3,200	2,800		7,600	2,400	1,600
\$10,000–19,999 \$20,000,000,000	10,200	2,900	3,200	7,600	7,300	2,300	2,500
\$20,000-29,999	10,600	3,400	3,800	8,100	7,400	2,800	2,000
\$30,000–49,999 \$50,000 or more	10,400	2,700	‡	8,400	7,100	3,100	2,200
\$50,000 or more	11,000	4,300	‡	9,600	6,200	3,700 !	2,200 !
Income group ⁴							
Lowest 25 percent	10,500	3,300	2,600	7,400	7,800	2,400	1,600
Middle 50 percent	10,800	2,900	4,000	8,200	7,500	2,800	2,400
Highest 25 percent	11,800	4,100	‡	9,900	6,700	3,600 !	2,100
Undergraduate class level							
First year	9,600	2,900	3,200	7,200	7,000	2,400	2,000
Second year	12,500	3,600	3,900	9,000	8,200	2,800	2,300
Third year	13,900	3,600	2,900	10,900	9,100	3,100	1,700 !
Fourth year or more	14,400	4,200	3,800	11,300	8,500	3,000	3,200
Aid status							
No aid	†	†	†	†	†	†	†
Received aid	10,800	3,200	3,500	8,100	7,500	2,600	2,100
Grant status							
No grant	10,500	†	4,400	9,200	6,900	7,400	5,300
Received grant	10,900	3,200	3,200	7,700	7,700	2,500	1,900

Table 3.9-B. Average amount of financial aid received by undergraduates who received a specific type of aid in for-profit institutions, by type and source of aid and selected student characteristics: 2007–08—Continued

Student characteristics	Any aid amount	Any grant amount	Any work- study amount	Any Ioan amount ¹	Any federal aid amount ²	Any state aid amount	Any institution aid amount
Loan status ¹							
No loan	\$4,000	\$3,500	\$3,000	†	\$2,900	\$2,400	‡
Received loan	11,200	3,100	3,500	8,100	7,700	2,700	2,100

[†] Not applicable.

NOTE: Average aid amounts are calculated only for students receiving a specific type of aid. Those not receiving a specific type of aid (i.e., zero values) are not included in the average for that aid. Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. This table excludes students attending more than one institution. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[!] Interpret data with caution (estimates are unstable).

[‡] Reporting standards not met.

¹ Loans only include loans to students. Excludes federal Parent PLUS loans.

² Includes all federal grants, loans, college work-study awards, and federal Parent PLUS loans. Excludes veterans benefits and education tax credit and tax deduction benefits.

³ Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

⁴ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

Table S3.9-B. Standard errors for table 3.9-B: Average amount of financial aid received by undergraduates who received a specific type of aid in for-profit institutions, by type and source of aid and selected student characteristics: 2007–08

Student characteristics	Any aid amount	Any grant amount	Any work- study amount	Any Ioan amount	Any federal aid amount	Any state aid amount	Any institution aid amount
Total	\$170	\$50	\$290	\$130	\$110	\$170	\$390
Attendance status							
Full-time/full-year	250	90	430	190	150	230	630
Full-time/part-year	350	110	490	270	270	300	560
Part-time/full-year	210	120	490	170	170	210	370
Part-time/part-year	250	70	740	230	200	270	190
Housing							
On campus	1,520	470	†	1,200	2,320	700	750
Off campus	180	60	350	150	100	210	480
Living with parents	290	90	450	210	240	210	320
Price of attendance							
Less than \$6,000	180	150	†	280	200	†	†
\$6,000-10,999	130	60	†	140	120	220	300
\$11,000-18,999	220	80	480	220	160	200	220
\$19,000 or more	280	90	360	210	170	240	600
Gender							
Male	330	110	370	230	230	270	470
Female	190	60	330	150	100	180	380
Race/ethnicity							
White	240	70	410	190	180	190	260
Black	260	90	570	250	140	240	380
Hispanic	260	120	560	200	200	300	620
Asian	890	470	†	690	690	1,120	640
American Indian	2,110	910	†	1,260	1,690	†	†
Pacific Islander	2,020	610	†	1,630	930	†	†
Two or more races	750	240	†	720	330	490	†
Other	960	660	†	840	770	†	†
Age as of 12/31/07							
18 years or younger	580	140	640	440	450	250	320
19–23 years	280	70	370	220	180	210	480
24-29 years	290	100	500	260	190	310	530
30-39 years	300	100	620	270	140	360	410
40 years or older	340	170	†	270	280	270	520
Dependency status							
Dependent	370	120	410	310	250	220	390
Independent	170	70	340	150	100	210	420
Unmarried, no dependents	250	130	610	230	200	290	450
Married, no dependents	770	250	†	700	400	830	650
Single parent	220	100	410	150	200	270	450
Married parent	300	150	†	270	190	250	630

Table S3.9-B. Standard errors for table 3.9-B: Average amount of financial aid received by undergraduates who received a specific type of aid in for-profit institutions, by type and source of aid and selected student characteristics: 2007–08—Continued

Student characteristics	Any aid amount	Any grant amount	Any work- study amount	Any Ioan amount	Any federal aid amount	Any state aid amount	Any institution aid amount
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$390	\$150	\$460	\$310	\$290	\$270	\$320
\$20,000-39,999	470	180	590	350	430	370	450
\$40,000-59,999	670	180	†	540	390	270	610
\$60,000-79,999	1,120	720	†	1,000	570	460	810
\$80,000-99,999	910	720	†	890	940	†	920
\$100,000 or more	1,030	670	†	1,150	820	630	1,530
Independent	•		•	,			,
Less than \$10,000	270	90	180	240	160	300	330
\$10,000-19,999	270	130	650	200	130	260	590
\$20,000-29,999	300	180	900	260	250	290	460
\$30,000-49,999	270	100	†	240	190	270	650
\$50,000 or more	390	430	†	390	240	1,440	680
Income group							
Lowest 25 percent	250	70	210	180	170	210	310
Middle 50 percent	170	80	430	150	150	180	490
Highest 25 percent	380	390	†	360	240	1,140	500
Undergraduate class level							
First year	180	70	380	150	160	200	370
Second year	300	120	470	240	190	240	560
Third year	450	210	640	380	350	330	320
Fourth year or more	620	290	640	540	380	880	790
Aid status							
No aid	†	†	†	†	†	†	†
Received aid	170	50	290	130	110	170	390
Grant status							
No grant	280	†	450	240	200	1,040	1,320
Received grant	190	50	290	140	100	170	310
Loan status							
No loan	180	140	880	†	150	380	†
Received loan	180	60	290	130	110	200	400

[†] Not applicable.

Table 4.1-A. Average out-of-pocket net price (price of attendance minus total aid) of postsecondary education among undergraduates, by type of institution and selected student characteristics: 2007–08

		Public	: 4-year		vate it 4-year		
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions ¹
Total	\$5,400	\$7,000	\$9,400	\$11,200	\$16,100	\$10,200	\$8,000
Attendance status							
Full-time/full-year	9,100	8,600	11,000	13,900	19,300	16,000	12,000
Full-time/part-year	5,200	5,900	7,500	9,000	12,500	7,200	6,900
Part-time/full-year	6,300	6,900	7,800	8,600	11,000	8,100	6,900
Part-time/part-year	3,200	3,800		4,600	5,700	4,900	3,600
Housing							
On campus	5,100	7,500	10,300	13,700	19,400	9,400	12,400
Off campus	5,300	6,900	9,100	9,100	13,100	10,500	7,500
Living with parents	5,600	6,900	8,500	10,400	12,200	9,100	6,700
Price of attendance							
Less than \$6,000	3,300	3,400	3,500	3,100	3,100	2,500	3,300
\$6,000-10,999	6,200	5,600	5,600	4,700	4,800	3,000	5,800
\$11,000–18,999	9,500	7,900	8,500	7,400	8,400	5,800	8,300
\$19,000 or more	15,900	11,600	13,600	14,200	19,100	15,400	15,300
Gender							
Male	5,500	7,300	9,800	11,300	17,200	10,400	8,300
Female	5,400	6,800	9,000	11,100	15,200	10,100	7,900
Race/ethnicity ²							
White	5,400	7,500	10,000	12,000	17,000	10,400	8,500
Black	4,900	5,000	5,800	9,000	10,000	9,700	6,400
Hispanic	5,500	6,400	7,500	8,500	11,200	9,600	7,000
Asian	6,200	9,200	11,500	15,400	23,100	11,100	10,000
American Indian	5,200	5,500	7,300	12,500	‡	15,400	7,100
Pacific Islander	5,400	8,100	8,000	5,600	28,500	11,400	7,800
Two or more races	5,600	6,800	8,500	10,300	15,500	12,200	8,100
Other	5,400	7,800	7,300	7,400	‡	15,200	7,900
Age as of 12/31/07							
18 years or younger	6,500	8,100	10,100	13,600	18,800	8,400	9,500
19–23 years	5,800	7,700	10,300	12,800	18,200	9,800	9,100
24–29 years	5,000	5,700	6,300	8,300	9,800	9,800	6,400
30-39 years	4,900	5,100	6,000	7,900	8,500	11,200	6,500
40 years or older	4,800	5,400	6,300	8,400	7,600	11,400	6,100
Dependency status							
Dependent	6,000	7,900	10,400	13,300	18,600	9,000	9,500
Independent	5,000	5,700		8,200	9,100	10,600	6,500
Unmarried, no dependents	5,000	5,500		8,200	9,700	9,800	6,300
Married, no dependents	5,200	6,800		8,400	8,800	12,400	6,900
Single parent	4,800	5,700		8,100	9,100	10,500	6,600
Married parent	5,000	5,300	6,300	8,200	8,600	11,100	6,400

Table 4.1-A. Average out-of-pocket net price (price of attendance minus total aid) of postsecondary education among undergraduates, by type of institution and selected student characteristics: 2007–08—Continued

		Public	: 4-year		vate it 4-year		
		Non-		Non-			All
	Public	doctorate-	Doctorate-	doctorate-	Doctorate-		insti-
Student characteristics	2-year	granting	granting	granting	granting	For-profit	tutions ¹
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$4,800	\$4,800	\$6,100	\$9,900	\$11,100	\$8,300	\$6,100
\$20,000-39,999	5,000	5,400	6,400	7,400	10,800	7,900	6,100
\$40,000-59,999	6,100	7,200	8,100	8,900	12,800	9,300	7,500
\$60,000-79,999	6,500	8,100	10,100	12,300	15,600	9,600	9,000
\$80,000-99,999	6,900	9,100	11,300	13,600	18,200	10,100	10,500
\$100,000 or more	7,100	10,600	13,400	18,500	24,300	12,300	13,800
Independent							
Less than \$10,000	5,100	6,000	6,000	8,100	10,400	9,200	6,500
\$10,000-19,999	4,900	5,100	6,300	7,400	9,300	9,600	6,300
\$20,000-29,999	4,800	5,700	6,600	7,900	8,800	10,400	6,400
\$30,000-49,999	5,000	5,400	6,700	8,500	8,100	11,600	6,600
\$50,000 or more	5,000	6,200	7,500	8,600	8,900	14,000	6,600
Income group ³							
Lowest 25 percent	4,900	5,300	6,000	8,400	10,500	8,900	6,300
Middle 50 percent	5,600	7,100	9,100	10,000	14,100	10,300	7,700
Highest 25 percent	5,600	8,900		15,000	21,100	13,800	10,500
Aid status							
No aid	5,900	9,900	15,100	17,800	29,300	17,000	9,100
Received aid	4,900	5,800	7,100	10,300	13,200	10,000	7,500
Grant status							
No grant	5,800	8,700	12,100	15,100	24,200	12,000	8,900
Received grant	4,800	5,500	6,900	10,000	12,800	9,500	7,300
Loan status ⁴							
No loan	5,700	8,700	12,700	15,700	23,200	13,300	8,600
Received loan	3,800	4,900	5,700	8,300	10,700	9,900	7,100

[‡] Reporting standards not met.

NOTE: In calculating the out-of-pocket net price, all financial aid received, including Parent PLUS loans, are subtracted from the price of attendance. Federal education tax benefits are not included in the calculation. This table excludes students attending more than one institution. Averages include students who received no aid. Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

¹ All institutions also include public less-than-2-year and private nonprofit less-than-4-year institutions in addition to those listed separately in this table.

² Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

³ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁴ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S4.1-A. Standard errors for table 4.1-A: Average out-of-pocket net price (price of attendance minus total aid) of postsecondary education among undergraduates, by type of institution and selected student characteristics: 2007–08

		Public	4-year	Priv nonprofi			
	Public		Doctorate-	Non- doctorate-			All insti-
Student characteristics	2-year	granting	granting	granting	granting	For-profit	tutions
Total	\$50	\$150	\$150	\$250	\$370	\$440	\$80
Attendance status							
Full-time/full-year	120	180	160	380	440	410	120
Full-time/part-year	120	370	320	420	1,470	520	230
Part-time/full-year	60	150	220	370	480	520	60
Part-time/part-year	40	100	160	270	280	550	50
Housing							
On campus	250	240	240	510	550	1,470	200
Off campus	60	140	140	300	330	460	80
Living with parents	70	240	180	500	500	490	80
Price of attendance							
Less than \$6,000	20	60	80	170	190	180	20
\$6,000-10,999	50	120	80	200	200	230	50
\$11,000-18,999	110	100	130	240	230	180	80
\$19,000 or more	390	470	270	330	430	390	180
Gender							
Male	60	200	200	400	540	590	110
Female	60	140	140	370	380	460	80
Race/ethnicity							
White	60	170	170	340	410	620	80
Black	90	290	190	680	610	550	130
Hispanic	80	200	200	490	640	470	130
Asian	150	400	440	1,300	1,400	1,080	260
American Indian	310	450	640	2,600	, †	2,630	550
Pacific Islander	300	920	900	990	3,900	1,340	510
Two or more races	180	460	460	1,590	1,430	1,180	290
Other	400	1,100	1,050	880	†	2,980	700
Age as of 12/31/07							
18 years or younger	110	330	230	920	790	570	170
19–23 years	70	170	180	380	460	400	100
24–29 years	80	170	200	580	440	650	130
30–39 years	80	160	270	420	530	590	110
40 years or older	70	220	440	780	480	700	140
Dependency status							
Dependent	70	190	170	390	430	440	100
Independent	60	120	160	390	300	480	90
Unmarried, no dependents	80	160	180	540	460	620	90
Married, no dependents	120	270	340	670	510	940	130
Single parent	80	210	310	630	650	530	160
Married parent	90	170	320	480	420	560	120

Table S4.1-A. Standard errors for table 4.1-A: Average out-of-pocket net price (price of attendance minus total aid) of postsecondary education among undergraduates, by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year	Priv nonprof			
		Non-		Non-			All
	Public	doctorate-	Doctorate-	doctorate-	Doctorate-		insti-
Student characteristics	2-year	granting	granting	granting	granting	For-profit	tutions
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$110	\$230	\$230	\$1,210	\$640	\$550	\$130
\$20,000-39,999	100	190	170	470	750	570	100
\$40,000-59,999	100	220	240	560	670	870	110
\$60,000-79,999	130	190	250	870	1,200	1,040	160
\$80,000-99,999	150	270	230	960	1,030	1,150	200
\$100,000 or more	160	270	220	660	640	950	170
Independent							
Less than \$10,000	90	230	190	690	760	670	170
\$10,000-19,999	90	170	320	540	700	460	120
\$20,000-29,999	80	200	310	480	690	820	170
\$30,000-49,999	80	170	300	640	530	530	120
\$50,000 or more	90	220	340	590	330	700	120
Income group							
Lowest 25 percent	70	170	130	490	490	520	110
Middle 50 percent	50	140	180	300	450	460	70
Highest 25 percent	90	250	200	510	680	660	120
Aid status							
No aid	70	240	190	1,150	850	630	80
Received aid	70	120	90	230	280	450	100
Grant status							
No grant	60	210	220	760	740	490	70
Received grant	70	100	100	250	290	460	100
Loan status							
No loan	50	190	210	480	570	580	70
Received loan	110	170	80	240	330	460	130

[†] Not applicable.

Table 4.1-B. Average net price (price of attendance minus all grants) among undergraduates, by type of institution and selected student characteristics: 2007–08

					vate		
		Public	4-year	nonprof	it 4-year	•	
		Non-		Non-			All
Student characteristics	Public 2-year	doctorate- granting	Doctorate- granting	doctorate- granting	Doctorate- granting	For-profit	insti- tutions ¹
Total	\$6,200	\$10,400		\$18,100	\$23,800	\$18,400	\$11,500
Attendance status							
Full-time/full-year	10,600	13,300	16,100	22,500	28,400	25,800	17,600
Full-time/part-year	6,000	8,600		14,100	,	14,600	10,400
Part-time/full-year	7,100	9,700	•	14,200		16,900	9,200
Part-time/part-year	3,600	5,000		6,900	8,500	10,600	4,700
Housing							
On campus	6,700	12,600	15,200	22,500	28,100	27,700	18,900
Off campus	6,200	10,200	-	14,600	-	18,500	10,700
Living with parents	6,100	8,800		15,600		17,500	8,700
Price of attendance							
Less than \$6,000	3,500	3,800	4,100	3,500	3,800	4,200	3,500
\$6,000–10,999	7,000	7,200		6,900	6,700	7,200	7,100
\$11,000–18,999	11,700	12,200		11,800	12,300	12,800	12,400
\$19,000 or more	19,800	18,800		23,100	28,200	25,800	23,500
Gender							
Male	6,200	10,500	13,900	18,100	24,600	19,600	11,600
Female	6,100	10,300	13,400	18,100	23,000	17,900	11,400
Race/ethnicity ²							
White	6,300	11,200	14,200	19,200	24,600	18,900	12,000
Black	5,700	9,400	11,500	15,900	18,900	17,800	10,500
Hispanic	6,000	8,200	11,600	13,900	19,300	17,200	9,900
Asian	6,600	10,800	14,600	20,300	28,200	20,200	12,300
American Indian	6,100	7,700	11,200	19,300	‡	21,600	9,500
Pacific Islander	5,900	10,700	13,600	11,900	35,300	21,200	10,900
Two or more races	6,600	10,100	12,900	18,200	24,000	21,500	11,900
Other	6,100	9,100	12,400	12,000	‡	22,000	10,700
Age as of 12/31/07							
18 years or younger	7,100	11,600	14,300	21,400	26,700	17,300	13,100
19–23 years	6,500	11,400	14,600	20,900	26,500	18,300	13,000
24–29 years	6,000	8,900	10,900	14,100	16,700	18,000	9,900
30–39 years	5,700	8,000	10,400	12,600	14,200	19,100	9,500
40 years or older	5,400	7,500	8,800	12,000	11,300	19,000	8,100
Dependency status							
Dependent	6,700	11,600		21,400		18,600	13,300
Independent	5,800	8,600		13,100	14,800	18,400	9,400
Unmarried, no dependents	5,900	8,700	10,700	13,900	17,000	18,200	9,500
Married, no dependents	5,900	9,000	11,400	13,200	12,900	21,300	9,300
Single parent	5,600	8,600		13,300		17,400	9,600
Married parent	5,700	8,000	10,000	12,200	13,000	19,200	9,000

Table 4.1-B. Average net price (price of attendance minus all grants) among undergraduates, by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year		vate it 4-year		
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions ¹
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$5,300	\$7,700	\$10,400	\$15,900	\$18,300	\$14,800	\$9,000
\$20,000-39,999	5,700	8,900	10,900	16,200	19,600	17,000	9,800
\$40,000-59,999	6,800	11,000	13,000	18,000	23,500	19,700	11,400
\$60,000-79,999	7,300	12,400	15,200	21,100	24,900	20,800	13,200
\$80,000-99,999	7,700	13,300	16,000	22,500	27,800	22,600	15,000
\$100,000 or more	7,700	14,000	17,000	26,200	31,400	25,100	17,800
Independent							
Less than \$10,000	6,000	9,300	10,800	14,400	16,800	16,200	10,000
\$10,000-19,999	5,900	8,700	11,100	12,900	16,600	17,100	9,700
\$20,000-29,999	5,900	8,500	11,200	13,200	15,400	18,400	9,600
\$30,000-49,999	5,700	8,300	10,300	13,600	13,900	19,900	9,400
\$50,000 or more	5,500	7,800	10,100	12,000	12,700	23,000	8,400
Income group ³							
Lowest 25 percent	5,700	8,600	10,600	15,200	18,000	16,100	9,600
Middle 50 percent	6,400	10,800	13,800	17,600	23,100	18,900	11,400
Highest 25 percent	6,100	11,600	15,800	20,800	27,200	23,500	13,500
Aid status							
No aid	5,900	9,900	15,100	17,800	29,300	17,000	9,100
Received aid	6,500	10,600	13,100	18,100	22,500	18,500	12,700
Grant status							
No grant	6,300	11,500	15,700	19,600	29,000	21,200	11,100
Received grant	5,900	9,400	11,800	17,600	21,600	17,200	11,800
Loan status ⁴							
No loan	5,900	9,000	13,100	16,400	23,900	13,800	8,900
Received loan	8,200	12,200	14,200	19,200	23,700	18,800	15,700

[‡] Reporting standards not met.

NOTE: Federal education tax benefits are not included in this table. This table excludes students attending more than one institution. Averages include students who received no aid. Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

¹ All institutions also include public less-than-2-year and private nonprofit less-than-4-year institutions in addition to those listed separately in this table.

² Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

³ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁴ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S4.1-B. Standard errors for table 4.1-B: Average net price (price of attendance minus all grants) among undergraduates, by type of institution and selected student characteristics: 2007–08

		Public	: 4-year	Priv nonprof	/ate it 4-year		
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions
Total	\$60	\$160	\$100	\$290	\$330	\$460	\$70
Attendance status							
Full-time/full-year	110	160	100	410	380	390	110
Full-time/part-year	80	440	290	490	1,890	510	250
Part-time/full-year	60	180	150	380	480	610	70
Part-time/part-year	60	130	130	260	290	590	80
Housing							
On campus	340	210	170	510	460	1,190	180
Off campus	60	170	100	340	400	490	90
Living with parents	70	270	170	550	790	550	90
Price of attendance							
Less than \$6,000	20	60	60	190	130	140	20
\$6,000–10,999	40	50	70	150	170	190	30
\$11,000–18,999	60	80	90	180	150	170	50
\$19,000 or more	330	310	130	320	350	320	150
Gender							
Male	60	190	130	420	480	720	100
Female	60	160	110	410	360	460	80
Race/ethnicity							
White	80	180	110	340	350	590	80
Black	90	290	180	710	700	720	190
Hispanic	80	240	190	690	1,370	650	220
Asian	150	370	380	1,520	1,170	1,210	270
American Indian	340	490	730	2,140	†	2,810	810
Pacific Islander	270	1,040	830	2,720	4,040	2,490	620
Two or more races	200	530	420	1,290	1,230	1,090	350
Other	440	1,070	910	1,460	†	2,550	810
Age as of 12/31/07							
18 years or younger	120	320	240	770	700	740	220
19–23 years	70	180	120	420	400	510	80
24–29 years	80	180	160	620	570	560	130
30-39 years	90	180	260	530	710	690	140
40 years or older	80	260	420	920	500	730	180
Dependency status							
Dependent	70	190	120	430	370	580	90
Independent	60	130	130	400	390	500	100
Unmarried, no dependents	100	180	150	570	640	700	110
Married, no dependents	140	270	310	880	640	1,210	220
Single parent	100	250		620	770	570	180
Married parent	90	220	340	570	620	630	150

Table S4.1-B. Standard errors for table 4.1-B: Average net price (price of attendance minus all grants) among undergraduates, by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year		vate iit 4-year		
		Non-		Non-			All
	Public	doctorate-	Doctorate-	doctorate-	Doctorate-		insti-
Student characteristics	2-year	granting	granting	granting	granting	For-profit	tutions
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$110	\$230	\$250	\$1,170	\$850	\$630	\$170
\$20,000-39,999	100	240	170	800	970	800	160
\$40,000-59,999	130	270	180	470	730	1,010	140
\$60,000-79,999	160	280	200	800	880	1,460	170
\$80,000-99,999	170	230	200	840	740	860	240
\$100,000 or more	190	260	170	510	550	1,120	160
Independent							
Less than \$10,000	110	220	170	710	860	760	210
\$10,000–19,999	90	220	240	650	910	520	130
\$20,000-29,999	90	230	280	510	820	870	220
\$30,000-49,999	90	190	270	800	610	580	160
\$50,000 or more	120	250	350	700	560	720	160
Income group							
Lowest 25 percent	80	170	120	540	620	610	140
Middle 50 percent	70	170	120	360	340	480	70
Highest 25 percent	120	260	160	470	670	650	130
Aid status							
No aid	70	240	190	1,150	850	630	80
Received aid	70	150	80	300	300	470	150
Grant status							
No grant	90	230	140	750	620	440	100
Received grant	70	120	90	290	320	530	150
Loan status							
No loan	60	190	200	480		620	60
Received loan	120	150	100	320	320	490	150

[†] Not applicable.

Table 4.1-C. Average net price after federal grants among undergraduates, by type of institution and selected student characteristics: 2007–08

		Public	4-year	Priv nonprof			
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions ¹
Total	\$6,500	\$11,700	\$15,800	\$24,200	\$30,800	\$19,000	\$13,200
	4 -,	***,	4 10,000	* = :,===	400,000	*:0,000	¥ : 0,=00
Attendance status	11 200	15 100	10.000	24 200	27 200	26 600	24 400
Full time/part year	11,300 6,300	15,400 9,400	18,900 12,100	31,300 17,200	37,300 22,100	26,600 15,100	21,100 11,500
Full-time/part-year Part-time/full-year	7,500	10,500	13,200	16,900	20,500	17,600	10,000
Part-time/part-year	3,700	5,500	7,100	8,200	10,200	10,900	5,000
Housing							
On campus	7,700	14,800	18,300	32,200	37,100	28,500	24,500
Off campus	6,500	11,400	15,300	17,600	24,200	19,200	11,900
Living with parents	6,500	10,000	13,100	20,700	25,700	18,000	9,700
Price of attendance							
Less than \$6,000	3,600	4,100	4,400	4,200	4,600	4,400	3,700
\$6,000–10,999	7,500	8,000	8,600	8,000	8,200	7,400	7,700
\$11,000–18,999	12,400	13,900	14,700	13,600	14,100	13,300	13,800
\$19,000 or more	20,600	21,900	23,300	31,800	36,900	26,600	28,500
Gender							
Male	6,600	11,900	16,100	24,000	31,400	20,400	13,300
Female	6,500	11,600	15,600	24,400	30,300	18,400	13,200
Race/ethnicity ²							
White	6,600	12,500	16,200	26,000	31,700	19,500	13,900
Black	6,100	11,200	14,200	20,000	25,300	18,300	11,900
Hispanic	6,400	9,200	14,200	17,400	26,100	17,800	11,300
Asian	6,900	12,300	17,200	28,200	36,100	20,900	14,400
American Indian	6,600	9,000	13,700	26,600	‡	23,100	10,900
Pacific Islander	6,300	12,200	16,200	17,900	39,800	21,200	12,200
Two or more races	7,000	11,900	16,100	26,600	33,000	21,900	14,100
Other	6,400	9,800	14,300	15,300	‡	22,500	11,900
Age as of 12/31/07			.=				
18 years or younger	7,600	13,700	17,600	30,700	36,700	17,800	16,200
19–23 years	6,900	12,900	17,000	29,100	34,400	18,700	15,300
24–29 years	6,300	9,700	12,000	16,200	19,500	18,600	10,600
30–39 years 40 years or older	6,100 5,700	8,800 8,400	11,600 10,100	14,500 14,100	17,000 14,000	19,800 20,000	10,200 8,800
•	3,700	0,400	10,100	17,100	1 4,000	25,000	3,000
Dependency status Dependent	7,100	13,300	17,200	30,100	35,400	19,100	16,000
Independent	6,100	9,400	12,000	15,500	17,900	19,000	10,000
Unmarried, no dependents	6,200	9,500	11,900	16,700	20,700	18,900	10,400
Married, no dependents	6,200	9,900	12,900	15,500	16,200	22,000	10,400
Single parent	6,000	9,400	12,100	15,500	17,200	17,900	10,200
Married parent	6,100	8,900	11,200	14,200	15,500	20,000	9,800

Table 4.1-C. Average net price after federal grants among undergraduates, by type of institution and selected student characteristics: 2007–08—Continued

		Dublic	4	Priv			_
		Public	4-year	nonprof	it 4-year		
		Non-		Non-			All insti-
Student characteristics	Public	doctorate-		doctorate-		Ear profit	tutions ¹
Student Characteristics	2-year	granting	granting	granting	granting	For-profit	tutions
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$5,900	\$9,800	\$14,200	\$23,300	\$27,600	\$15,400	\$11,400
\$20,000-39,999	6,200	11,300	14,900	26,100	30,300	17,500	12,600
\$40,000-59,999	7,200	12,900	16,200	28,300	34,300	20,300	14,200
\$60,000-79,999	7,700	14,000	17,600	30,800	34,800	21,100	15,800
\$80,000-99,999	8,000	14,800	18,000	31,800	37,600	22,900	17,900
\$100,000 or more	8,000	15,300	18,700	33,400	37,500	25,300	20,400
Independent							
Less than \$10,000	6,400	10,400	12,300	17,500	21,000	16,600	10,900
\$10,000-19,999	6,200	9,500	12,300	15,200	19,500	17,600	10,400
\$20,000-29,999	6,200	9,200	12,300	15,200	18,500	19,000	10,300
\$30,000–49,999	6,100	8,900	11,500	15,600	16,600	20,700	10,100
\$50,000 or more	5,800	8,700	11,100	14,200	15,400	24,300	9,200
Income group ³							
Lowest 25 percent	6,200	10,300	13,700	21,500	26,000	16,600	11,400
Middle 50 percent	6,800	12,100	15,900	24,200	31,300	19,400	13,200
Highest 25 percent	6,400	12,600	17,400	26,000	32,500	24,600	15,300
Aid status							
No aid	5,900	9,900	15,100	17,800	29,300	17,000	9,100
Received aid	7,200	12,500	16,100	25,100	31,100	19,100	15,400
Grant status							
No grant	6,300	11,500	15,700	19,600	29,000	21,200	11,100
Received grant	6,800	12,000	15,900	25,600	31,500	18,100	15,200
Loan status⁴							
No loan	6,200	10,200	15,300	20,900	28,900	14,500	10,100
Received loan	8,700	13,800	16,500	26,300	32,300	19,400	18,400

[‡] Reporting standards not met.

NOTE: Federal education tax benefits are not included in this table. This table excludes students attending more than one institution. Averages include students who received no aid. Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

¹ All institutions also include public less-than-2-year and private nonprofit less-than-4-year institutions in addition to those listed separately in this table

² Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

³ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁴ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S4.1-C. Standard errors for table 4.1-C: Average net price after federal grants among undergraduates, by type of institution and selected student characteristics: 2007–08

		Public	: 4-year	Priv nonprof			
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions
Total	\$60	\$180	\$90	\$360	\$390	\$470	\$70
Attendance status							
Full-time/full-year	100	180	90	440	440	370	100
Full-time/part-year	70	440	310	510	2,390	530	290
Part-time/full-year	60	200	150	450	550	600	80
Part-time/part-year	70	160	120	260	300	590	90
Housing							
On campus	380	240	140	500	410	1,180	220
Off campus	70	220	100	380	500	500	90
Living with parents	80	290	160	680	1,230	550	100
Price of attendance							
Less than \$6,000	20	50	40	140	80	140	20
\$6,000–10,999	30	40	50	140	100	180	20
\$11,000–18,999	60	80	50	170	250	160	40
\$19,000 or more	290	250	100	400	310	300	130
Gender							
Male	70	200	120	510	540	730	100
Female	60	190	110	530	400	460	90
Race/ethnicity							
White	90	180	100	440	340	600	100
Black	90	370	200	790	740	730	190
Hispanic	80	280	230	900	2,180	660	250
Asian	160	370	360	1,630	1,040	1,210	300
American Indian	350	590	820	3,450	†	2,730	870
Pacific Islander	270	1,140	1,000	3,290	3,380	2,490	650
Two or more races	200	600	380	1,950	1,280	1,110	410
Other	430	1,260	910	2,260	†	2,330	880
Age as of 12/31/07							
18 years or younger	120	350	210	710	720	740	230
19–23 years	70	200	100	450	440	520	90
24–29 years	80	190	160	650	640	570	130
30-39 years	100	200	270	610	830	700	150
40 years or older	80	270	450	1,030	560	740	190
Dependency status							
Dependent	80	220	100	450	440	590	90
Independent	60	140	130	470	410	510	110
Unmarried, no dependents	100	190	140	590	710	710	120
Married, no dependents	140	260	310	880	750	1,210	220
Single parent	90	280	290	730	850	570	190
Married parent	90	240	350	670	660	650	160

Table S4.1-C. Standard errors for table 4.1-C: Average net price after federal grants among undergraduates, by type of institution and selected student characteristics: 2007–08

—Continued

		Public	4-year	Priv nonprof			All insti- tutions
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting		
Dependency and income in 2006	-					•	
Dependent							
Less than \$20,000	\$120	\$260	\$220	\$1,170	\$1,530	\$640	\$210
\$20,000–39,999	100	260	170	870	1,180	820	180
\$40,000–59,999	140	320	180	680	640	1,040	170
\$60,000-79,999	170	300	180	810	750	1,510	270
\$80,000-99,999	180	230	190	870	620	870	310
\$100,000 or more	200	300	130	540	480	1,100	180
Independent							
Less than \$10,000	110	260	160	710	1,110	760	210
\$10,000–19,999	100	250	240	720	1,110	570	140
\$20,000–29,999	90	250	300	630	910	860	230
\$30,000-49,999	90	200	310	830	630	590	170
\$50,000 or more	120	270	380	780	590	700	190
Income group							
Lowest 25 percent	80	190	120	590	1,140	620	150
Middle 50 percent	80	200	120	470	400	490	90
Highest 25 percent	120	290	150	560	590	630	140
Aid status							
No aid	70	240	190	1,150	850	630	80
Received aid	80	190	90	400	440	480	180
Grant status							
No grant	90	230	140	750	620	440	100
Received grant	80	180	100	420	500	540	190
Loan status							
No loan	70	210	160	510	530	660	60
Received loan	130	180	110	440	420	490	160

[†] Not applicable.

Table 4.1-D. Average net tuition (tuition and fees minus all grants) among undergraduates, by type of institution and selected student characteristics: 2007–08

		Public	4-year		vate it 4-year		
			4 your		it 4 your		AII
	Public	Non-	Doctorate-	Non- doctorate-	Doctorate-		insti-
Student characteristics	2-year	granting	granting	granting	granting	For-profit	tutions ¹
Total	\$700	\$2,500	\$4,100	\$8,500	\$13,200	\$8,000	\$3,800
Attendance status							
Full-time/full-year	1,400	3,300	4,900	11,200	16,400	9,200	6,400
Full-time/part-year	800	2,200	3,500	6,100	8,900	7,900	3,900
Part-time/full-year	800	2,200	3,400	5,500	7,800	8,800	2,200
Part-time/part-year	400	1,000	1,800	2,500	3,700	5,000	1,100
Housing							
On campus	900	3,500	5,000	11,900	16,200	15,500	8,600
Off campus	600	2,200	3,800	5,800	10,100	7,700	3,100
Living with parents	900	2,300	3,700	7,000	10,600	8,800	2,500
Price of attendance							
Less than \$6,000	400	800	1,100	1,000	1,300	1,500	500
\$6,000-10,999	800	1,600	2,000	2,100	2,500	3,000	1,300
\$11,000-18,999	1,300	2,900	3,300	3,800	4,500	6,200	3,100
\$19,000 or more	2,200	5,400	7,000	11,600	16,200	11,000	10,600
Gender							
Male	800	2,700	4,400	8,700	14,000	9,000	3,900
Female	700	2,400	3,900	8,400	12,600	7,600	3,700
Race/ethnicity ²							
White	800	3,000	4,500	9,500	14,000	8,300	4,200
Black	500	1,600	2,600	6,500	8,900	7,500	3,000
Hispanic	500	1,300	2,600	4,700	9,300	7,600	2,700
Asian	900	3,100	5,200	11,300	16,900	8,400	4,500
American Indian	500	1,100	2,700	8,800	‡	11,700	2,700
Pacific Islander	600	1,600	3,500	4,500 !	22,200	9,900	3,600
Two or more races	600	2,100	3,500	8,000	13,000	9,100	3,600
Other	700	1,900	3,100	4,400	‡	9,200	3,300
Age as of 12/31/07							
18 years or younger	900	2,900	4,400	10,700	14,900	9,400	4,700
19–23 years	900	3,000	4,700	10,800	15,300	9,000	4,700
24–29 years	600	1,700	2,700	5,700	8,100	7,800	2,800
30-39 years	500	1,300	2,300	4,000	6,100	7,200	2,400
40 years or older	500	1,300	1,800	3,600	4,500	6,900	1,800
Dependency status							
Dependent	1,000	3,100	4,700	11,100	15,600	10,200	5,000
Independent	500	1,600	2,500	4,600	6,600	7,300	2,400
Unmarried, no dependents	600	1,700	2,700	5,800	8,400	8,200	2,700
Married, no dependents	700	1,900	2,800	4,900	5,000	8,200	2,300
Single parent	400	1,300	2,200	4,000	6,200	6,700	2,400
Married parent	500	1,400	2,200	3,800	5,400	7,300	2,200

Table 4.1-D. Average net tuition (tuition and fees minus all grants) among undergraduates, by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year		vate fit 4-year		All insti- tutions ¹
Student characteristics	Public 0 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$500	\$900	\$2,000	\$5,800	\$8,200	\$7,400	\$2,300
\$20,000-39,999	600	1,500	2,300	6,500	9,300	9,200	2,700
\$40,000–59,999	1,000	2,700	3,500	8,400	12,600	11,500	3,700
\$60,000-79,999	1,200	3,700	4,900	11,200	13,700	12,300	4,800
\$80,000-99,999	1,300	4,100	5,500	11,900	15,900	12,600	5,900
\$100,000 or more	1,300	4,600	6,300	15,200	19,500	13,300	8,000
Independent							
Less than \$10,000	500	1,500	2,300	5,000	7,700	6,900	2,700
\$10,000-19,999	500	1,400	2,700	4,700	8,100	7,200	2,600
\$20,000–29,999	500	1,500	2,900	4,800	6,900	7,800	2,600
\$30,000-49,999	500	1,600	2,400	4,700	5,900	7,600	2,400
\$50,000 or more	600	1,700	2,600	4,100	5,500	7,800	2,000
Income group ³							
Lowest 25 percent	500	1,300	2,100	5,600	8,200	7,300	2,600
Middle 50 percent	800	2,700	4,200	8,200	12,500	8,400	3,700
Highest 25 percent	800	3,500	5,700	11,000	16,400	8,800	5,300
Aid status							
No aid	1,000	3,300	6,300	10,700	19,600	7,600	3,100
Received aid	400	2,200	3,300	8,200	11,800	8,000	4,100
Grant status							
No grant	1,000	3,800	6,300	11,200	19,000	10,100	4,100
Received grant	200	1,300	2,300	7,700	10,800	7,100	3,500
Loan status ⁴							
No loan	700	2,200	4,500	8,300	14,300	5,400	2,600
Received loan	900	2,900	3,700	8,600	12,400	8,300	5,700

[!] Interpret data with caution (estimates are unstable).

NOTE: Federal education tax benefits are not included in this table. This table excludes students attending more than one institution. If the grant amount is greater than tuition and fees, the net tuition was set to zero. Averages include students who received no aid. Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[‡] Reporting standards not met.

¹ All institutions also include public less-than-2-year and private nonprofit less-than-4-year institutions in addition to those listed separately in this table.

² Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

³ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁴ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S4.1-D. Standard errors for table 4.1-D: Average net tuition (tuition and fees minus all grants) among undergraduates, by type of institution and selected student characteristics: 2007–08

		Public	4-year		vate fit 4-year		
Student characteristics	Public 2-year	Non-	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions
Total	\$10	\$110	\$80	\$330	\$280	\$490	\$60
Attendance status							
	30	130	80	470	340	530	90
Full-time/full-year	40	180				610	
Full-time/part-year			220	320	1,150		210
Part-time/full-year Part-time/part-year	10 10	130 70	130 70	320 200	530 210	610 520	50 40
,	10	70	70	200	210	520	40
Housing							
On campus	100	120	130	440	420	1,000	150
Off campus	10	110	80	300	330	530	80
Living with parents	20	180	110	480	800	510	60
Price of attendance							
Less than \$6,000	10	50	40	80	100	140	10
\$6,000-10,999	20	70	40	180	160	250	20
\$11,000–18,999	40	100	70	240	380	240	50
\$19,000 or more	180	250	140	440	300	700	190
Gender							
Male	20	130	100	410	410	530	70
Female	10	110	90	390	300	520	80
Race/ethnicity							
White	20	130	80	360	310	660	70
Black	30	180	100	660	580	700	160
Hispanic	20	90	90	590	1,360	370	130
Asian	50	210	250	1,430	1,090	650	190
American Indian	70	160	340	1,890	†	2,370	570
Pacific Islander	90	310	550	1,610	3,360	1,580	480
Two or more races	40	200	240	1,090	1,070	990	230
Other	120	470	710	620	†	2,510	410
Age as of 12/31/07							
18 years or younger	30	210	150	650	590	590	150
19–23 years	20	130	90	380	330	460	60
24–29 years	20	80	90	530	520	660	130
30–39 years	20	100	90	250	500	520	100
40 years or older	10	120	120	310	420	790	140
Dependency status							
Dependent	20	140	90	380	320	530	70
Independent	10	80	60	270	330	560	100
Unmarried, no dependents	20	90	90	480	470	610	90
Married, no dependents	30	120	150	390	440	910	130
Single parent	20	100	120	430	610	590	180
Married parent	10	90	110	240	490	650	110
Marriou paroni	10	30	110	240	430	000	110

Table S4.1-D. Standard errors for table 4.1-D: Average net tuition (tuition and fees minus all grants) among undergraduates, by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year		vate it 4-year		All insti- tutions
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$40	\$120	\$170	\$910	\$770	\$530	\$110
\$20,000-39,999	20	150	100	630	800	700	120
\$40,000–59,999	30	170	130	490	670	820	100
\$60,000-79,999	40	160	140	700	810	1,290	130
\$80,000-99,999	50	140	140	780	650	910	180
\$100,000 or more	40	150	120	490	500	850	130
Independent							
Less than \$10,000	30	110	100	660	690	700	170
\$10,000-19,999	20	80	150	500	670	410	90
\$20,000-29,999	20	100	180	430	640	920	180
\$30,000-49,999	20	100	90	390	570	570	120
\$50,000 or more	20	100	100	270	410	500	80
Income group							
Lowest 25 percent	20	110	80	500	570	550	100
Middle 50 percent	10	120	90	350	320	520	70
Highest 25 percent	20	140	120	490	560	440	100
Aid status							
No aid	20	170	120	900	690	390	50
Received aid	10	90	60	330	260	500	100
Grant status							
No grant	20	170	100	660	520	480	60
Received grant	10	60	50	290	260	530	90
Loan status							
No loan	10	130	130	450	480	300	50
Received loan	30	100	60	360	290	520	140

[†] Not applicable.

Table 4.1-E. Average out-of-pocket net price (price of attendance minus total aid) of postsecondary education among dependent undergraduates, by dependent student family income in 2006 and selected institutional and student characteristics: 2007–08

		Dependen	t student far	nily income	in 2006		
Institutional and student characteristics	Less than \$20,000	\$20,000- 39,999	\$40,000– 59,999	\$60,000- 79,999	\$80,000- 99,999	\$100,000 or more	All dependent students
Total	\$6,100	\$6,100	\$7,500	\$9,000	\$10,500	\$13,800	\$9,500
Institution type							
Public							
Less-than-2-year	8,600 !	9,800	8,100	9,900	9,400	‡	9,100
2-year	4,800	5,000	6,100	6,500	6,900	7,100	6,000
4-year	5,600	6,000	7,800	9,400	10,700	12,700	9,600
Non-doctorate-granting	4,800	5,400	7,200	8,100	9,100	10,600	7,900
Doctorate-granting	6,100	6,400	8,100	10,100	11,300	13,400	10,400
Private nonprofit							
Less-than-4-year	7,700	7,500	8,200	11,500	13,200	12,200	9,500
4-year	10,400	9,000	10,700	14,000	16,100	21,900	16,100
Non-doctorate-granting	9,900	7,400	8,900	12,300	13,600	18,500	13,300
Doctorate-granting	11,100	10,800	12,800	15,600	18,200	24,300	18,600
For-profit	8,300	7,900	9,300	9,600	10,100	12,300	9,000
Attendance status							
Full-time/full-year	7,400	7,500	9,600	11,500	13,100	16,900	12,100
Full-time/part-year	5,500	5,500	6,200	7,500	8,300	10,100	7,500
Part-time/full-year	5,600	5,600	6,800	7,500	7,600	9,500	7,200
Part-time/part-year	3,100	3,100	3,400	3,700	3,900	4,900	3,700
Housing							
On campus	7,000	6,900	8,800	11,200	13,600	17,700	12,800
Off campus	6,600	6,600	8,100	9,600	10,600	13,400	9,800
Living with parents	5,400	5,500	6,500	7,400	8,000	8,900	6,900
Price of attendance							
Less than \$6,000	2,900	3,200	3,400	3,400	3,500	3,500	3,300
\$6,000-10,999	4,900	5,000	6,400	6,800	7,000	7,200	6,200
\$11,000–18,999	6,200	6,400	8,100	9,500	10,200	11,500	8,900
\$19,000 or more	10,800	9,600	11,500	13,900	15,500	20,300	15,500
Gender							
Male	6,400	6,400	7,500	9,200	10,600	13,700	9,700
Female	5,900	5,900	7,500	8,900	10,500	13,900	9,200

Table 4.1-E. Average out-of-pocket net price (price of attendance minus total aid) of postsecondary education among dependent undergraduates, by dependent student family income in 2006 and selected institutional and student characteristics: 2007–08—Continued

		Dependen	t student far	nily income	in 2006		
Institutional and student characteristics	Less than \$20,000	\$20,000- 39,999	\$40,000- 59,999	\$60,000- 79,999	\$80,000- 99,999	\$100,000 or more	All dependent students
Race/ethnicity ¹							
White	\$6,400	\$6,400	\$7,600	\$9,100	\$10,500	\$14,000	\$10,200
Black	5,300	4,900	6,300	7,300	9,000	10,300	6,400
Hispanic	5,800	6,000	6,900	7,500	9,000	11,300	7,300
Asian	7,400	8,000	10,100	13,800	15,100	17,600	11,800
American Indian	5,000	6,300	6,900	7,900	10,000	11,100	7,800
Pacific Islander	5,300	6,700	7,400	7,200	14,600	10,700	8,400
Two or more races	7,800	6,400	8,100	8,600	9,200	13,600	9,200
Other	‡	6,400	‡	‡	‡	‡	8,400
Age as of 12/31/07							
18 years or younger	5,700	6,200	7,600	9,300	11,100	14,300	9,600
19–23 years	6,200	6,100	7,500	9,000	10,400	13,700	9,400
24–29 years	‡	‡	‡	‡	‡	‡	‡
30-39 years	‡	‡	‡	‡	‡	‡	‡
40 years or older	‡	‡	‡	‡	‡	‡	‡
Income group ²							
Lowest 25 percent	6,100	6,000	†	†	†	†	6,000
Middle 50 percent	†	7,000	7,500	9,000	10,500	11,000	8,900
Highest 25 percent	†	†	†	†	†	14,200	14,200
Aid status							
No aid	8,400	7,700	8,300	9,900	11,800	15,500	11,500
Received aid	5,600	5,700	7,100	8,500	9,800	12,500	8,400
Grant status							
No grant	8,200	7,600	8,000	9,100	10,600	14,200	10,700
Received grant	5,500	5,600	7,100	8,900	10,400	13,200	8,300
Loan status ³							
No loan	6,800	6,800	8,600	10,200	12,100	15,500	10,900
Received loan	5,100	5,200	6,000	7,500	8,300	10,400	7,300

[†] Not applicable.

NOTE: Federal education tax benefits are not included in this table. Students age 24 or older are independent. This table excludes students attending more than one institution. Averages include students who received no aid. Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[!] Interpret data with caution (estimates are unstable).

[‡] Reporting standards not met.

¹ Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

² Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

³ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S4.1-E. Standard errors for table 4.1-E: Average out-of-pocket net price (price of attendance minus total aid) of postsecondary education among dependent undergraduates, by dependent student family income in 2006 and selected institutional and student characteristics: 2007–08

		Dependen	t student far	nily income	in 2006		
Institutional and student characteristics	Less than \$20,000	\$20,000- 39,999	\$40,000- 59,999	\$60,000- 79,999	\$80,000- 99,999	\$100,000 or more	All dependent students
Total	\$130	\$100	\$110	\$160	\$200	\$170	\$100
Institution type Public							
Less-than-2-year	3,930	2,850	1,690	1,420	1,520	†	2,190
2-year	110	100	100	130	150	160	70
4-year	170	130	190	190	200	200	160
Non-doctorate-granting	230	190	220	190	270	270	190
Doctorate-granting	230	170	240	250	230	220	170
Private nonprofit							
Less-than-4-year	1,900	1,110	1,160	1,390	1,110	1,530	980
4-year	710	450	400	790	690	460	290
Non-doctorate-granting	1,210	470	560	870	960	660	390
Doctorate-granting	640	750	670	1,200	1,030	640	430
For-profit	550	570	870	1,040	1,150	950	440
Attendance status							
Full-time/full-year	220	140	180	260	260	230	120
Full-time/part-year	290	280	230	320	510	500	290
Part-time/full-year	180	170	220	160	180	220	110
Part-time/part-year	120	100	110	150	160	200	70
Housing							
On campus	530	300	270	480	520	340	200
Off campus	180	140	200	210	210	190	110
Living with parents	120	120	120	130	160	170	80
Price of attendance							
Less than \$6,000	70	60	50	60	80	60	30
\$6,000–10,999	100	90	90	110	110	100	50
\$11,000–18,999	140	130	160	140	140	130	100
\$19,000 or more	500	270	310	480	410	330	200
Gender							
Male	210	140	160	250	220	240	130
Female	160	130	130	190	330	220	110

Table S4.1-E. Standard errors for table 4.1-E: Average out-of-pocket net price (price of attendance minus total aid) of postsecondary education among dependent undergraduates, by dependent student family income in 2006 and selected institutional and student characteristics: 2007–08—Continued

		Dependen	t student far	nily income	in 2006		
Institutional and student characteristics	Less than \$20,000	\$20,000– 39,999	\$40,000– 59,999	\$60,000– 79,999	\$80,000– 99,999	\$100,000 or more	All dependent students
Race/ethnicity							
White	\$190	\$140	\$120	\$170	\$220	\$170	\$100
Black	350	190	280	440	610	670	190
Hispanic	200	200	250	250	390	470	170
Asian	400	390	550	1,790	1,240	740	380
American Indian	890	930	920	830	1,950	1,530	550
Pacific Islander	810	1,090	1,030	1,010	3,850	2,920	820
Two or more races	1,630	430	1,160	640	900	760	380
Other	†	1,260	†	†	†	†	680
Age as of 12/31/07							
18 years or younger	200	230	220	280	630	390	170
19–23 years	160	120	110	180	190	190	100
24–29 years	†	†	†	†	†	†	†
30–39 years	†	†	†	†	†	†	†
40 years or older	†	†	†	†	†	†	†
Income group							
Lowest 25 percent	130	110	†	†	†	†	100
Middle 50 percent	†	240	110	160	200	350	90
Highest 25 percent	†	†	†	†	†	190	190
Aid status							
No aid	330	230	180	330	370	260	130
Received aid	140	100	130	160	250	170	100
Grant status							
No grant	350	230	170	230	290	240	110
Received grant	130	100	140	220	290	210	100
Loan status							
No loan	180	130	140	240	320	180	110
Received loan	270	160	160	160	200	230	100

[†] Not applicable.

Table 4.1-F. Average out-of-pocket net price (price of attendance minus total aid) of postsecondary education among independent undergraduates, by independent student income in 2006 and selected institutional and student characteristics: 2007–08

		Independent s	student incom	e in 2006		
Institutional and student characteristics	Less than \$10,000	\$10,000– 19,999	\$20,000- 29,999	\$30,000– 49,999	\$50,000 or more	All indepen- dent students
Total	\$6,500	\$6,300	\$6,400	\$6,600	\$6,600	\$6,500
Institution type						
Public						
Less-than-2-year	6,900	6,300	6,400	6,300	6,200	6,500
2-year	5,100	4,900	4,800	5,000	5,000	5,000
4-year	6,000	5,700	6,100	6,100	6,900	6,200
Non-doctorate-granting	6,000	5,100	5,700	5,400	6,200	5,700
Doctorate-granting	6,000	6,300	6,600	6,700	7,500	6,600
Private nonprofit						
Less-than-4-year	7,900	7,900	7,100	8,700	9,900	8,300
4-year	9,000	8,000	8,200	8,400	8,700	8,500
Non-doctorate-granting	8,100	7,400	7,900	8,500	8,600	8,200
Doctorate-granting	10,400	9,300	8,800	8,100	8,900	9,100
For-profit	9,200	9,600	10,400	11,600	14,000	10,600
Attendance status						
Full-time/full-year	9,900	11,100	11,900	13,800	16,400	11,900
Full-time/part-year	5,800	5,500	6,300	6,400	8,000	6,200
Part-time/full-year	6,200	6,300	6,300	6,700	7,700	6,700
Part-time/part-year	3,600	3,300	3,500	3,700	3,700	3,600
Housing						
On campus	7,200	6,200	6,300	6,800	6,700	6,800
Off campus	6,500	6,300	6,600	6,600	6,600	6,500
Living with parents	6,300	5,900	5,100	6,100	6,300	6,000
Price of attendance						
Less than \$6,000	3,200	3,200	3,200	3,300	3,300	3,300
\$6,000-10,999	4,900	4,800	5,200	5,800	6,300	5,400
\$11,000–18,999	6,500	6,800	6,900	7,600	9,100	7,200
\$19,000 or more	13,200	13,800	14,500	15,300	17,300	14,700
Gender						
Male	6,400	6,300	6,100	6,300	6,400	6,300
Female	6,600	6,200	6,600	6,700	6,700	6,600
Race/ethnicity ¹						
White	6,300	6,100	6,300	6,400	6,300	6,300
Black	6,200	6,100	6,500	6,700	7,100	6,500
Hispanic	6,700	6,800	6,200	7,000	6,700	6,700
Asian	8,800	6,700	7,000	7,200	7,700	7,700
American Indian	5,600	5,200	6,400	6,200	9,800	6,700
Pacific Islander	7,900	8,200	5,900	5,800	5,800	6,900
Two or more races	6,400	5,900	6,600	7,800	7,264	6,800
Other	9,500 !	6,600 !	‡	‡	‡	7,400

Table 4.1-F. Average out-of-pocket net price (price of attendance minus total aid) of postsecondary education among independent undergraduates, by independent student income in 2006 and selected institutional and student characteristics: 2007–08—Continued

		Independent	student incom	e in 2006		
Institutional and student characteristics	Less than \$10,000	\$10,000– 19,999	\$20,000– 29,999	\$30,000– 49,999	\$50,000 or more	All indepen- dent students
Age as of 12/31/07						
18 years or younger	\$6,900	\$5,200	‡	‡	‡	\$6,800
19–23 years	7,100	7,300	6,900	8,000	7,700	7,300
24-29 years	6,500	6,100	6,200	6,600	7,000	6,400
30-39 years	6,300	6,200	6,700	6,500	6,600	6,500
40 years or older	5,700	5,500	6,100	6,200	6,300	6,100
Independent students						
Unmarried, no dependents	6,200	6,100	6,200	6,700	6,300	6,300
Married, no dependents	6,700	7,100	7,400	7,200	6,500	6,900
Single parent	7,000	6,300	6,400	6,500	6,900	6,600
Married parent	6,000	6,100	6,100	6,300	6,600	6,400
Income group ²						
Lowest 25 percent	6,500	6,200	†	†	†	6,500
Middle 50 percent	†	6,300	6,400	6,600	†	6,400
Highest 25 percent	†	†	†	7,000	6,600	6,600
Aid status						
No aid	8,100	6,600	6,500	6,100	6,100	6,600
Received aid	6,000	6,100	6,300	6,900	7,000	6,400
Grant status						
No grant	7,800	6,300	6,500	6,500	6,700	6,800
Received grant	5,900	6,200	6,300	6,600	6,300	6,200
Loan status ³						
No loan	7,200	6,400	6,100	6,000	5,900	6,300
Received loan	5,700	6,100	6,800	7,800	9,100	6,900

[†] Not applicable.

NOTE: Federal education tax benefits are not included in this table. Students age 24 or older are independent. This table excludes students attending more than one institution. Averages include students who received no aid. Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[!] Interpret data with caution (estimates are unstable).

[‡] Reporting standards not met.

¹ Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

² Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

³ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S4.1-F. Standard errors for table 4.1-F: Average out-of-pocket net price (price of attendance minus total aid) of postsecondary education among independent undergraduates, by independent student income in 2006 and selected institutional and student characteristics: 2007–08

		Independent	student incom	e in 2006		
Institutional and student characteristics	Less than \$10,000	\$10,000– 19,999	\$20,000- 29,999	\$30,000– 49,999	\$50,000 or more	All indepen- dent students
Total	\$170	\$120	\$170	\$120	\$120	\$90
Institution type Public						
Less-than-2-year	1,490	780	690	960	740	750
2-year	90	90	80	80	90	60
4-year	140	200	200	200	220	110
Non-doctorate-granting	230	170	200	170	220	120
Doctorate-granting	190	320	310	300	340	160
Private nonprofit						
Less-than-4-year	900	780	760	910	1,140	640
4-year	540	430	430	500	390	280
Non-doctorate-granting	690	540	480	640	590	390
Doctorate-granting	760	700	690	530	330	300
For-profit	670	460	820	530	700	480
Attendance status						
Full-time/full-year	240	280	440	340	410	200
Full-time/part-year	350	220	500	210	370	230
Part-time/full-year	110	180	130	120	100	70
Part-time/part-year	130	60	100	80	100	60
Housing						
On campus	650	480	880	750	730	390
Off campus	180	140	190	120	130	100
Living with parents	240	280	220	510	600	130
Price of attendance						
Less than \$6,000	80	60	40	40	30	20
\$6,000–10,999	90	110	100	80	80	60
\$11,000–18,999	130	140	170	170	190	80
\$19,000 or more	400	380	540	350	430	260
Gender						
Male	180	200	170	150	160	100
Female	220	140	220	150	170	110
Race/ethnicity						
White	240	140	190	150	140	110
Black	250	250	400	270	350	160
Hispanic	220	270	240	260	250	130
Asian	420	390	570	390	630	240
American Indian	780	660	1,380	1,510	2,590	770
Pacific Islander	760	1,470	1,190	700	1,210	470
Two or more races	660	500	1,370	810	730	380
Other	3,070	2,180	†	†	†	1,150

Table S4.1-F. Standard errors for table 4.1-F: Average out-of-pocket net price (price of attendance minus total aid) of postsecondary education among independent undergraduates, by independent student income in 2006 and selected institutional and student characteristics: 2007–08—Continued

		Independent	student income	e in 2006		
Institutional and student characteristics	Less than \$10,000	\$10,000– 19,999	\$20,000- 29,999	\$30,000– 49,999	\$50,000 or more	All indepen- dent students
Age as of 12/31/07						
18 years or younger	\$490	\$810	†	†	†	\$430
19–23 years	210	280	380	390	550	160
24–29 years	210	160	200	210	210	130
30-39 years	310	260	310	180	170	110
40 years or older	220	210	390	240	160	140
Independent students						
Unmarried, no dependents	150	140	190	210	340	90
Married, no dependents	410	390	500	210	190	130
Single parent	270	190	310	210	400	160
Married parent	350	330	280	200	140	120
Income group						
Lowest 25 percent	170	270	†	†	†	160
Middle 50 percent	†	130	170	120	†	110
Highest 25 percent	†	†	†	710	120	120
Aid status						
No aid	160	150	130	110	120	60
Received aid	200	160	240	180	190	140
Grant status						
No grant	170	130	200	130	140	70
Received grant	210	160	200	180	220	140
Loan status						
No loan	150	100	110	80	120	50
Received loan	320	200	410	290	340	220

[†] Not applicable.

Table 4.1-G. Average out-of-pocket net price (price of attendance minus total aid) of postsecondary education among full-time/full-year undergraduates, by type of institution and selected student characteristics: 2007–08

		Public	4-year		/ate it 4-year		
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions ¹
Total	\$9,100	\$8,600	\$11,000	\$13,900	\$19,300	\$16,000	\$12,000
Housing							
On campus	5,400	8,000	10,900	14,500	20,200	10,000	13,300
Off campus	9,700	9,200	11,400	13,500	19,100	16,800	12,300
Living with parents	8,800	8,600	10,000	11,900	14,100	13,400	9,800
Price of attendance							
Less than \$6,000	‡	‡	‡	‡	‡	‡	‡
\$6,000–10,999	6,600	6,000	6,600	‡	‡	4,900	6,500
\$11,000–18,999	10,000	8,100	8,800	7,800	9,000	6,700	8,900
\$19,000 or more	16,100	11,500	13,600	14,300	19,700	17,000	15,600
Gender							
Male	9,400	8,900	11,500	14,300	20,400	16,000	12,400
Female	8,800	8,400	10,600	13,600	18,300	16,000	11,800
Race/ethnicity ²							
White	9,200	9,100	11,800	14,500	20,100	16,300	12,600
Black	7,600	5,600	6,500	12,600	12,500	16,400	9,500
Hispanic	8,800	8,300	8,500	10,700	12,900	14,400	10,200
Asian	11,100	10,800	13,300	17,800	26,300	17,500	15,000
American Indian	7,400	7,100	9,300	‡	‡	15,700	9,700
Pacific Islander	9,600	‡	8,400	8,300	‡	13,900	12,200
Two or more races	8,900	8,000	9,500	12,100	17,400	18,000	11,400
Other	‡	‡	8,200	‡	‡	‡	12,500
Age as of 12/31/07							
18 years or younger	8,900	8,700	10,600	14,400	19,400	10,800	11,600
19–23 years	9,100	8,900	11,400	14,000	19,700	15,200	12,200
24–29 years	9,000	7,700	8,200	12,400	15,100	16,000	11,100
30-39 years	9,200	6,100	8,500	13,000	15,000	17,600	12,900
40 years or older	9,300	7,900	9,600	14,800	15,200	17,300	13,200
Dependency status							
Dependent	9,000	8,800	11,300	14,200	19,800	13,100	12,100
Independent	9,100	7,900	8,900	12,600	15,000	16,900	11,900
Unmarried, no dependents	9,200	7,300	8,000	11,100	14,600	16,600	10,700
Married, no dependents	10,800	9,400	11,000	12,800	14,800	19,400	13,200
Single parent	8,500	8,000	9,000	12,700	15,200	16,500	12,200
Married parent	9,300	7,600	9,600	14,600	15,600	16,900	12,900

Table 4.1-G. Average out-of-pocket net price (price of attendance minus total aid) of postsecondary education among full-time/full-year undergraduates, by type of institution and selected student characteristics: 2007–08—Continued

		Public	: 4-year		vate fit 4-year		All insti- tutions ¹
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate-	For-profit	
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$6,600	\$5,000	\$6,300	\$10,700	\$11,300	\$11,300	\$7,400
\$20,000–39,999	7,100	5,600	6,900	7,300	11,200	12,100	7,500
\$40,000–59,999	9,200	8,200	8,900	9,800	13,400	14,300	9,600
\$60,000–79,999	10,200	9,200	10,900	13,200	16,500	14,500	11,500
\$80,000–99,999	10,500	9,800	12,700	14,800	18,900	13,900	13,100
\$100,000 or more	11,100	12,100	14,700	20,100	26,100	15,100	16,900
Independent							
Less than \$10,000	8,300	7,600	7,400	10,000	13,500	14,900	9,900
\$10,000–19,999	8,700	6,600	8,900	11,500	14,100	16,200	11,100
\$20,000-29,999	9,000	7,900	9,400	11,600	14,300	16,300	11,900
\$30,000-49,999	9,900	8,000	10,500	14,600	16,100	18,000	13,800
\$50,000 or more	11,600	12,600	13,300	17,500	19,900	20,100	16,400
Income group ³							
Lowest 25 percent	7,300	5,900	6,600	9,300	11,500	14,000	8,200
Middle 50 percent	9,600	8,900	10,900	12,400	16,600	16,400	11,600
Highest 25 percent	11,300	12,200	14,800	20,300	26,200	19,300	17,200
Aid status							
No aid	12,300	16,300	20,500	35,500	41,200	28,900	19,600
Received aid	7,400	6,900	8,200	12,400	15,500	15,600	10,200
Grant status							
No grant	11,500	12,400	15,700	28,100	34,400	19,700	16,400
Received grant	7,200	6,400	7,800	11,600	14,600	14,600	9,700
Loan status ⁴							
No loan	10,100	12,200	16,000	23,200	30,600	20,500	15,500
Received loan	5,600	5,600	6,400	9,700	12,100	15,600	9,000

[‡] Reporting standards not met.

NOTE: Federal education tax benefits are not included in this table. Averages include students who received no aid. This table excludes students attending more than one institution. Full-time/full-year students represent 38.5 percent of all undergraduates attending only one institution. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico. SOURCE: U.S. Department of Education, National Center for Education Statistics, 2007–08 National Postsecondary Student Aid Study (NPSAS:08).

¹ All institutions also include public less-than-2-year and private nonprofit less-than-4-year institutions in addition to those listed separately in this table.

² Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

³ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁴ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S4.1-G. Standard errors for table 4.1-G: Average out-of-pocket net price (price of attendance minus total aid) of postsecondary education among full-time/full-year undergraduates, by type of institution and selected student characteristics: 2007–08

		Public	4-year	Priv nonprof	/ate it 4-year		
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate-	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions
Total	\$120	\$180	\$160	\$380	\$440	\$410	\$120
Housing							
On campus	410	270	220	580	610	1,380	210
Off campus	190	200	190	420	550	420	130
Living with parents	120	240	240	590	600	550	130
Price of attendance							
Less than \$6,000	†	†	†	†	†	†	†
\$6,000-10,999	80	220		†	†	320	80
\$11,000–18,999	130	110	130	230	550	310	70
\$19,000 or more	460	500	240	400	460	380	180
Gender							
Male	140	240	210	600	660	660	170
Female	140	180	170	480	500	460	120
Race/ethnicity							
White	130	190	170	490	500	590	130
Black	280	430	270	1,400	870	440	250
Hispanic	240	200	250	870	800	680	220
Asian	540	520	510	1,740	1,650	1,680	460
American Indian	620	930	820	†	†	2,730	790
Pacific Islander	770	†	1,460	2,330	†	1,950	1,280
Two or more races	340	630	550	2,060	1,900	1,700	490
Other	†	†	1,500	†	†	†	1,600
Age as of 12/31/07							
18 years or younger	150	370	240	1,010	790	700	210
19–23 years	140	180	170	460	530	520	140
24–29 years	300	360	400	970	990	560	240
30-39 years	380	540	670	840	1,300	630	370
40 years or older	410	710	880	870	1,780	770	400
Dependency status							
Dependent	130	200	150	460	480	600	120
Independent	240	250	310	610	740	410	200
Unmarried, no dependents	310	320	400	1,020	980	550	230
Married, no dependents	500	610	590	1,120	1,240	750	380
Single parent	260	440		1,090	1,530	530	300
Married parent	450	570	630	990	1,410	590	350

Table S4.1-G. Standard errors for table 4.1-G: Average out-of-pocket net price (price of attendance minus total aid) of postsecondary education among full-time/full-year undergraduates, by type of institution and selected student characteristics: 2007–08—Continued

		Public	Private 4-year nonprofit 4-ye				
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$210	\$250	\$270	\$1,500	\$680	\$870	\$220
\$20,000-39,999	200	210	200	550	740	830	140
\$40,000–59,999	210	270	300	660	840	1,380	180
\$60,000-79,999	260	240	250	1,010	1,400	1,090	260
\$80,000-99,999	250	260	250	1,200	1,180	1,580	260
\$100,000 or more	220	240	200	780	760	1,340	230
Independent							
Less than \$10,000	320	340	340	1,110	1,110	720	240
\$10,000–19,999	330	350	670	1,100	1,450	560	280
\$20,000-29,999	340	490	690	870	1,820	970	440
\$30,000-49,999	420	490	700	1,160	1,450	480	340
\$50,000 or more	530	870	690	890	1,500	700	410
Income group							
Lowest 25 percent	170	190	170	760	500	610	130
Middle 50 percent	150	180	210	450	630	430	140
Highest 25 percent	240	210	200	750	760	670	240
Aid status							
No aid	170	230	180	1,370	910	730	210
Received aid	170	130	100	330	350	410	100
Grant status							
No grant	190	210	210	1,290	900	440	200
Received grant	190	120	100	330	350	460	100
Loan status							
No loan	140	220	210	780	710	980	180
Received loan	230	190	110	320	400	420	130

[†] Not applicable.

Table 4.1-H. Average net tuition (tuition and fees minus all grants) among full-time/full-year undergraduates, by type of institution and selected student characteristics: 2007–08

		Public	4-year	Priv nonprof			
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions ¹
Total	\$1,400	\$3,300	\$4,900	\$11,200	\$16,400	\$9,200	\$6,400
Housing							
On campus	800	3,800	5,200	12,700	17,100	16,900	9,300
Off campus	1,300	3,300	4,900	9,300	16,100	8,700	5,700
Living with parents	1,400	2,800	4,200	8,200	12,700	10,400	3,900
Price of attendance							
Less than \$6,000	‡	‡	‡	‡	‡	‡	‡
\$6,000-10,999	1,100	1,300	2,200	‡	‡	‡	1,200
\$11,000–18,999	1,400	2,900	3,200	1,800	2,000 !	4,800	2,600
\$19,000 or more	2,600	5,500	6,900	11,800	17,000	9,700	10,500
Gender							
Male	1,600	3,600	5,200	11,500	17,200	10,500	6,800
Female	1,200	3,100	4,600	11,000	15,700	8,600	6,100
Race/ethnicity ²							
White	1,600	3,900	5,400	12,100	17,200	9,400	7,000
Black	700	2,000	3,100	9,600	11,900	8,700	4,800
Hispanic	800	1,700	3,000	6,100	11,300	8,600	4,300
Asian	2,000	3,700	6,000	13,700	20,000	10,700	7,700
American Indian	800 !	1,500	3,500	‡	‡	14,400 !	4,600
Pacific Islander	1,300	‡	4,100	8,100	‡	11,200 !	6,900
Two or more races	900	2,400	3,900	9,800	14,700	10,500	5,800
Other	‡	‡	3,500 !	‡	‡	‡	5,400
Age as of 12/31/07							
18 years or younger	1,200	3,100	4,400	11,400	15,800	10,800	6,000
19–23 years	1,500	3,600	5,200	12,000	16,900	11,300	6,900
24–29 years	1,200	2,400	3,800	8,900	14,100	9,200	5,400
30–39 years	1,100	1,700	3,200	7,100	11,900	7,500	5,100
40 years or older	1,100	2,300	2,700	6,300	11,600	6,700	4,500
Dependency status							
Dependent	1,500	3,600	5,100	12,100	16,900	13,200	6,900
Independent	1,100	2,200	3,500	7,700	12,400	8,000	5,000
Unmarried, no dependents	1,400	2,300	3,700	8,700	14,100	9,700	5,300
Married, no dependents	2,100	2,700	4,000	8,400	10,100	8,600	5,400
Single parent	500	1,800	3,000	6,700	11,400	7,200	4,400
Married parent	1,000	2,200	3,200	7,200	11,600	7,500	4,900

Table 4.1-H. Average net tuition (tuition and fees minus all grants) among full-time/full-year undergraduates, by type of institution and selected student characteristics: 2007–08

—Continued

		Public	4-year	Priv nonprof			
		Non-	. ,	Non-			All
	Public	doctorate-	Doctorate-	doctorate-	Doctorate-		insti-
Student characteristics	2-year	granting	granting	granting	granting	For-profit	tutions ¹
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$600	\$700	\$1,900	\$6,200	\$8,600	\$8,700	\$2,800
\$20,000-39,999	700	1,500	2,400	6,700	9,800	12,200	3,600
\$40,000-59,999	1,600	3,200	3,800	9,100	13,700	14,600	5,200
\$60,000-79,999	2,000	4,400	5,200	11,900	15,100	16,300	6,600
\$80,000-99,999	2,100	4,500	6,100	13,100	17,100	16,800	7,800
\$100,000 or more	2,100	5,400	6,900	16,800	21,200	17,200	10,400
Independent							
Less than \$10,000	800	1,900	2,700	6,700	10,900	7,700	4,000
\$10,000-19,999	800	1,900	3,800	7,600	12,500	8,400	4,800
\$20,000-29,999	1,000	2,400	4,300	8,100	11,000	8,400	5,300
\$30,000-49,999	1,200	2,800	3,600	8,600	14,400	7,800	5,600
\$50,000 or more	2,400	3,900	5,300	8,600	16,300	8,000	6,700
Income group ³							
Lowest 25 percent	700	1,300	2,200	6,400	9,400	8,600	3,400
Middle 50 percent	1,600	3,800	5,000	10,600	15,300	9,500	6,200
Highest 25 percent	2,200	5,300	7,000	16,000	21,100	9,400	10,100
Aid status							
No aid	2,400	5,800	8,700	23,600	28,400	10,200	8,400
Received aid	800	2,800	3,700	10,300	14,300	9,200	5,900
Grant status							
No grant	2,500	6,100	8,400	22,100	27,700	12,000	9,000
Received grant	400	1,700	2,500	9,400	12,900	8,100	5,100

Table 4.1-H. Average net tuition (tuition and fees minus all grants) among full-time/full-year undergraduates, by type of institution and selected student characteristics: 2007–08

—Continued

Student characteristics		Public	4-year	Priva nonprofit			
	Public 2-year			Non- doctorate- Doctorate- granting granting		For-profit	All insti- tutions ¹
Loan status ⁴							
No loan	1,300	3,200	5,700	13,000	19,600	6,300	5,800
Received loan	1,400	3,400	4,200	10,300	14,400	9,500	7,000

[!] Interpret data with caution (estimates are unstable).

NOTE: Federal education tax benefits are not included in this table. If the grant amount is greater than tuition and fees, the net tuition was set to zero. Averages include students who received no aid. This table excludes students attending more than one institution. Full-time/full-year students represent 38.5 percent of all undergraduates attending only one institution. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[‡] Reporting standards not met.

¹ All institutions also include public less-than-2-year and private nonprofit less-than-4-year institutions in addition to those listed separately in this

² Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

³ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁴ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S4.1-H. Standard errors for table 4.1-H: Average net tuition (tuition and fees minus all grants) among full-time/full-year undergraduates, by type of institution and selected student characteristics: 2007–08

				Priv			
		Public 4-year Non-		nonprof	it 4-year		
				Non-			All
Student characteristics	Public 2-year	doctorate- granting	Doctorate- granting	doctorate- granting	Doctorate- granting	For-profit	insti- tutions
Total	\$30	\$130		\$470	\$340	\$530	\$90
	,,,,	****	7	****	****	7	***
Housing	100	4.40	4.40	400	450	000	400
On campus	120	140		460	450	680	180
Off campus	40	160		680	500	610	120
Living with parents	40	210	140	620	1,070	540	100
Price of attendance							
Less than \$6,000	†	†	†	†	†	†	†
\$6,000–10,999	30	150	190	†	†	†	40
\$11,000–18,999	50	80	70	220	690	390	50
\$19,000 or more	240	280	120	490	330	600	170
Gender							
Male	40	150	100	610	470	730	130
Female	40	130	110	490	380	540	110
Race/ethnicity							
White	40	150	80	500	380	730	110
Black	90	290	160	1,110	840	660	220
Hispanic	50	130	120	970	1,700	560	200
Asian	190	280	320	1,720	1,230	1,260	390
American Indian	260	230	530	†	†	4,350	1,020
Pacific Islander	340	†	860	1,980	†	3,390	1,160
Two or more races	140	240	310	1,310	1,430	1,860	450
Other	†	†	1,220	†	†	†	930
Age as of 12/31/07							
18 years or younger	50	240	160	700	600	660	180
19–23 years	40	130	90	420	380	650	100
24–29 years	80	170	210	910	900	800	230
30–39 years	110	290	260	1,060	1,340	660	270
40 years or older	110	340	340	1,200	1,710	810	350
Dependency status							
Dependent	40	140	90	410	350	650	100
Independent	40	110	160	790	790	620	190
Unmarried, no dependents	100	150	220	920	760	740	200
Married, no dependents	200	280	300	890	1,310	740	230
Single parent	70	200	280	1,140	1,610	730	310
Married parent	70	220	270	1,260	1,490	740	310

Table S4.1-H. Standard errors for table 4.1-H: Average net tuition (tuition and fees minus all grants) among full-time/full-year undergraduates, by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year	Priv nonprofi			
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$80	\$100	\$200	\$1,090	\$840	\$570	\$170
\$20,000-39,999	50	140	120	660	840	1,000	170
\$40,000-59,999	80	210	160	600	710	1,340	170
\$60,000-79,999	80	210	150	790	950	1,180	200
\$80,000-99,999	110	160	160	950	760	1,140	240
\$100,000 or more	70	180	120	540	590	1,170	190
Independent							
Less than \$10,000	90	170	200	1,020	1,040	750	200
\$10,000-19,999	80	180	360	1,010	1,220	630	200
\$20,000-29,999	110	200	430	1,070	1,560	1,040	370
\$30,000-49,999	120	280	250	1,630	1,460	860	400
\$50,000 or more	160	250	420	1,850	1,140	520	310
Income group							
Lowest 25 percent	40	100	110	640	620	570	120
Middle 50 percent	50	150	100	490	420	660	110
Highest 25 percent	80	160	130	730	590	470	190
Aid status							
No aid	60	200	190	1,300	810	790	200
Received aid	40	110	60	440	310	540	100
Grant status							
No grant	50	190	120	990	580	560	150
Received grant	30	80	60	380	300	610	100
Loan status							
No loan	50	160	140	780	610	480	130
Received loan	100	130	80	460	330	560	130

[†] Not applicable.

Table 4.2. Average federal expected family contribution (EFC) among undergraduates, by type of institution and selected student characteristics: 2007–08

		Public	4-year	Priv nonprof			
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions ¹
Total	\$8,300	\$9,800	\$14,000	\$13,000	\$16,600	\$4,200	\$10,200
Attendance status Full-time/full-year Full-time/part-year Part-time/full-year Part-time/part-year	7,900	10,900	15,000	14,200	18,300	4,700	12,400
	7,700	9,200	14,700	12,900	15,900	3,900	9,100
	8,200	8,700	11,100	9,800	11,800	3,900	8,700
	8,600	8,400	11,400	11,300	11,100	3,800	8,700
Housing ² On campus Off campus Living with parents	10,200	12,800	16,600	17,500	20,300	7,200	16,800
	7,600	8,800	13,200	9,600	13,300	3,800	8,700
	9,200	9,500	12,100	10,100	11,700	5,300	9,400
Price of attendance ² Less than \$6,000 \$6,000–10,999 \$11,000–18,999 \$19,000 or more	8,600	8,800	11,300	12,700	9,900	4,000	8,800
	8,200	8,300	12,800	9,900	10,100	2,400	8,500
	7,200	10,300	13,100	7,800	11,300	3,200	9,600
	9,300	12,300	16,500	15,000	18,400	5,300	13,800
Gender Male Female	9,100 7,600	10,600 9,100	14,300 13,800	14,000 12,300	18,100 15,400	5,300 3,700	11,200 9,400
Race/ethnicity ³ White Black Hispanic Asian American Indian Pacific Islander Two or more races Other	9,700 4,900 6,500 6,800 6,200 7,400 8,600 9,600 !	11,800 5,300 7,000 7,500 7,100 6,300 7,300 11,200	16,200 6,600 9,500 11,700 9,800 11,200 12,300 7,700 !	15,900 6,500 5,500 12,400 12,200 ! 8,700 8,500 5,600 !	14,500 13,200	5,200 2,600 3,100 6,300 ‡ 4,300 3,300!	12,400 5,100 6,600 9,100 7,200 8,200 9,300 8,400
Age as of 12/31/07 18 years or younger 19–23 years 24–29 years 30–39 years 40 years or older	10,000	12,200	17,200	18,200	19,500	4,900	13,400
	9,100	11,700	16,100	15,600	19,300	4,700	12,600
	4,900	3,600	4,200	5,000	4,100	2,700	4,300
	7,900	7,700	7,700	8,000	8,200	4,200	7,200
	9,400	9,500	11,000	11,300	11,300	5,800	9,300
Dependency status Dependent Independent Unmarried, no dependents Married, no dependents Single parent Married parent	10,500	12,600	17,000	17,000	20,100	6,900	14,000
	6,600	5,500	5,700	7,200	6,700	3,300	5,900
	4,100	3,500	3,500	4,600	4,300	4,000	4,000
	12,100	8,700	11,200	11,800	10,400	8,700	11,100
	3,100	2,600	2,200	2,700	3,900	900	2,500
	10,400	9,600	10,400	11,100	10,600	5,000	9,600

Table 4.2. Average federal expected family contribution (EFC) among undergraduates, by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year	Priv nonprofi			
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions ¹
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$100	‡	\$600 !	\$600 !	\$1,700 !	\$300 !	\$400
\$20,000–39,999	2,300	1,900	2,300	2,000	2,000	1,700	2,200
\$40,000–59,999	6,200	5,900	5,800	5,700	5,600	5,100	5,900
\$60,000-79,999	11,100	10,600	11,200	11,000	10,700	10,600	11,000
\$80,000-99,999	16,500	16,200	15,900	16,400	16,300	16,300	16,200
\$100,000 or more	29,300	29,500	33,100	34,700	35,500	28,800	32,300
Independent							
Less than \$10,000	100	‡	200	200 !	100	100	200
\$10,000–19,999	1,500	1,800	1,900	1,300	1,600	1,000	1,500
\$20,000-29,999	3,000	3,300	3,800	2,800	2,300	2,400	3,000
\$30,000-49,999	5,000	5,200	6,300	4,700	5,900	4,500	5,100
\$50,000 or more	17,100	16,500	18,300	18,700	16,100	14,400	17,000
Income group ⁴							
Lowest 25 percent	700	800	1,000	800	1,100	400	800
Middle 50 percent	6,500	8,000	9,500	7,900	9,500	3,900	7,300
Highest 25 percent	20,800	25,100	31,800	29,400	31,900	16,800	25,900
Aid status							
No aid	10,500	13,300	19,500	19,100	24,700	8,900	13,100
Received aid	5,700	8,300	11,900	12,100	14,800	4,000	8,700
Grant status							
No grant	10,500	13,400	18,300	17,200	23,100	10,200	13,300
Received grant	4,900	6,500	10,200	11,700	13,900	1,600	7,300
Loan status ⁵							
No loan	8,500	11,000	17,400	16,600	21,100	5,700	11,300
Received loan	6,400	8,300	10,300	10,800	13,100	4,000	8,400

[!] Interpret data with caution (estimates are unstable).

NOTE: Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[‡] Reporting standards not met.

¹ All institutions also include those attending public less-than-2-year, private nonprofit less-than-4-year, and multiple institutions in addition to those listed separately in this table.

² Excludes those attending more than one institution.

³ Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

⁴ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁵ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S4.2. Standard errors for table 4.2: Average federal expected family contribution (EFC) among undergraduates, by type of institution and selected student characteristics: 2007–08

		Public	: 4-year		vate it 4-year		
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions
Total	\$160	\$230	\$150	\$420	\$380	\$160	\$80
Attendance status							
Full-time/full-year	190	350	190	650	460	240	130
Full-time/part-year	440	520	460	750	940	310	300
Part-time/full-year	170	340	300	620	750	290	110
Part-time/part-year	240	430	390	640	630	360	180
Housing							
On campus	1,430	470	310	760	440	1,290	250
Off campus	140	230	170	280	580	160	90
Living with parents	220	360	350	670	1,010	280	170
Price of attendance							
Less than \$6,000	250	490	510	1,240	1,030	680	210
\$6,000-10,999	170	270	380	570	760	270	140
\$11,000-18,999	230	310	200	750	1,180	150	150
\$19,000 or more	790	620	280	630	380	270	190
Gender							
Male	190	330	220	500	620	350	120
Female	160	200	200	580	420	150	90
Race/ethnicity							
White	150	210	180	590	400	190	100
Black	230	410	290	430	810	170	150
Hispanic	230	420	450	620	1,430	280	130
Asian	300	760	610	1,720	940	1,200	260
American Indian	1,150	1,000	1,430	4,860	†	†	770
Pacific Islander	750	1,050	1,680	1,920	2,580	†	640
Two or more races	800	850	900	1,370	1,400	660	380
Other	3,140	5,290	2,320	2,550	†	1,320	1,470
Age as of 12/31/07							
18 years or younger	300	600	520	2,190	740	780	270
19–23 years	240	300	160	470	420	220	100
24–29 years	170	220	280	430	340	130	90
30–39 years	280	380	390	520	620	300	150
40 years or older	280	680	870	640	800	430	200
Dependency status							
Dependent	220	300	170	590	400	310	110
Independent	140	200	270	310	470	130	80
Unmarried, no dependents	110	170	260	360	310	200	70
Married, no dependents	400	500	580	830	790	680	230
Single parent	300	380	220	370	1,090	140	130
Married parent	320	550	620	530	850	360	190

Table S4.2. Standard errors for table 4.2: Average federal expected family contribution (EFC) among undergraduates, by type of institution and selected student characteristics: 2007–08

—Continued

		Public	4-year		vate fit 4-year		
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$20	†	\$190	\$190	\$560	\$170	\$50
\$20,000-39,999	80	120	110	200	180	120	50
\$40,000-59,999	110	220	120	240	250	270	80
\$60,000-79,999	150	150	160	420	300	410	70
\$80,000-99,999	170	260	160	380	390	610	90
\$100,000 or more	480	420	310	1,200	570	900	230
Independent							
Less than \$10,000	20	†	60	70	20	10	20
\$10,000-19,999	50	90	80	120	270	60	30
\$20,000-29,999	110	170	230	190	330	240	70
\$30,000-49,999	120	240	310	290	350	290	80
\$50,000 or more	310	500	610	580	660	600	200
Income group							
Lowest 25 percent	30	110	70	110	180	50	20
Middle 50 percent	110	170	100	230	240	190	60
Highest 25 percent	340	430	380	1,000	500	630	190
Aid status							
No aid	200	420	330	2,220	910	830	200
Received aid	140	210	140	370	400	150	80
Grant status							
No grant	170	310	260	1,280	650	280	150
Received grant	150	200	160	410	440	100	80
Loan status							
No loan	180	380	250	910	630	600	130
Received loan	200	190	190	330	350	150	80

[†] Not applicable.

Table 4.3-A. Percentage of undergraduates who have financial need, by type of institution and selected student characteristics: 2007–08

		Dublic	4	Priv			
		Public	4-year	nonprof	it 4-year		
		Non-		Non-			All insti-
Student characteristics	Public 2-year	doctorate- granting	Doctorate- granting	doctorate- granting	granting	For-profit	tutions ¹
Total	59.1	70.8		82.4	82.9	97.5	69.7
Attendance status							
Full-time/full-year	74.6	77.0	72.1	89.8	87.8	98.7	79.7
Full-time/part-year	64.0	71.4		74.9	74.7	96.6	73.5
Part-time/full-year	65.0	72.5	73.1	83.3	83.8	98.1	70.3
Part-time/part-year	46.2	53.7		57.6	61.1	96.4	51.9
Housing							
On campus	62.8	70.4	66.9	84.6	84.3	96.9	74.9
Off campus	62.1	72.2		79.4	80.0	98.1	71.5
Living with parents	54.0	68.1	66.5	85.8	86.6	95.2	62.5
Price of attendance							
Less than \$6,000	45.8	46.3	42.5	36.7	46.6	78.2	46.0
\$6,000–10,999	66.5	68.8		65.4	61.2	95.6	67.6
\$11,000–18,999	81.5	76.6		84.8	76.2	98.0	78.6
\$19,000 or more	93.7	86.7		89.1	87.5	98.7	87.4
	00	00		33	00		• • • • • • • • • • • • • • • • • • • •
Gender Male	E4.2	60.0	60.0	70.0	90.6	07.6	66.1
Female	54.3 62.8	68.2 72.9	68.9 68.8	79.2 84.7	80.6 84.7	97.6 97.5	72.4
	02.0	12.9	00.0	04.7	04.7	91.5	12.4
Race/ethnicity ²							
White	53.2	65.7		79.6	79.9	96.4	64.5
Black	72.5	84.6	84.6	87.4	89.4	99.3	82.2
Hispanic	66.6	74.8	78.5	90.1	90.6	97.7	76.9
Asian	63.9	78.7		84.8	86.7	97.8	73.2
American Indian	71.1	76.2		98.6	‡	99.1	77.8
Pacific Islander	59.2	87.7		69.3	99.0	96.5	71.2
Two or more races	61.5	76.9	76.2	91.5	86.5	98.5	74.0 73.7
Other	59.6	77.9	85.5	88.0	‡	97.4	13.1
Age as of 12/31/07							
18 years or younger	57.0	70.8		83.3	85.6	95.2	67.4
19–23 years	56.3	67.8	65.6	84.5	82.8	96.3	67.5
24–29 years	72.2	86.5	88.9	89.0	91.6	99.1	82.6
30–39 years	59.2	67.6		78.6	79.0	98.4	70.5
40 years or older	51.7	60.4	63.0	65.5	68.7	96.4	60.2
Dependency status							
Dependent	51.2	66.1	63.6	83.6	82.7	94.0	64.4
Independent	65.0	78.0		80.5	83.2	98.6	75.5
Unmarried, no dependents	73.0	84.9		90.5	92.7	99.0	82.6
Married, no dependents	43.1	65.4		67.5	68.9	94.6	57.2
Single parent	81.6	88.7		91.3	86.6	99.8	88.0
Married parent	49.1	63.4	65.8	67.9	75.3	97.5	61.4

Table 4.3-A. Percentage of undergraduates who have financial need, by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year		vate it 4-year		
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions ¹
Dependency and income in 2006		<u> </u>	<u> </u>	<u> </u>	<u> </u>		
Dependent							
Less than \$20,000	99.8	99.4	99.6	99.5	98.5	99.9	99.6
\$20,000–39,999	87.2	97.9	98.6	98.4		100.0	93.8
\$40,000–59,999	57.9	85.8	93.5	99.5		97.5	77.8
\$60,000-79,999	30.4	71.4	83.9	94.7	97.3	94.8	64.1
\$80,000–99,999	12.6	47.8	60.8	90.1	94.0	89.3	50.8
\$100,000 or more	4.6	22.1	24.3	59.9	63.4	60.1	29.9
Independent							
Less than \$10,000	99.8	99.5	99.9	99.9	100.0	100.0	99.8
\$10,000-19,999	90.7	94.9	98.9	99.9	100.0	100.0	94.8
\$20,000-29,999	75.1	87.7	91.5	93.7	99.6	99.9	84.5
\$30,000-49,999	68.2	77.6	82.2	89.9	88.7	98.5	77.3
\$50,000 or more	17.9	31.3	39.0	44.6	53.6	91.5	31.1
Income group ³							
Lowest 25 percent	96.5	98.9	99.6	99.3	99.6	100.0	98.2
Middle 50 percent	59.8	75.7	80.7	94.1	96.2	98.4	74.4
Highest 25 percent	14.5	24.5	25.0	52.6	59.9	86.1	30.1
Aid status							
No aid	43.8	48.6	49.4	54.1	63.9	75.4	46.7
Received aid	75.9	80.2	76.4	86.5	87.1	98.4	81.9
Grant status							
No grant	46.5	56.3	57.0	66.2	69.7	92.9	54.4
Received grant	78.2	83.8	79.2	87.4	88.3	99.4	84.0
Loan status ⁴							
No loan	56.1	60.5	55.9	66.9	71.4	85.9	58.4
Received loan	79.0	84.2	82.9	92.2	91.7	98.6	88.3

[‡] Reporting standards not met.

NOTE: This table excludes students attending more than one institution. Financial need is defined as the total price of attendance minus the expected family contribution (EFC). Those with a negative or zero balance after subtraction are considered to have zero need. Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

¹ All institutions also include public less-than-2-year and private nonprofit less-than-4-year institutions in addition to those listed separately in this table.

² Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

³ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁴ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S4.3-A. Standard errors for table 4.3-A: Percentage of undergraduates who have financial need, by type of institution and selected student characteristics: 2007–08

		Public	4-year		/ate it 4-year		
		Non-	7	Non-	,		All
	Public	doctorate-	Doctorate-		Doctorate-		insti-
Student characteristics	2-year	granting	granting	granting	granting	For-profit	tutions
Total	1.12	0.65	0.38	0.77	0.71	0.19	0.60
Attendance status							
Full-time/full-year	0.72	1.03	0.50	0.73	0.76	0.20	0.27
Full-time/part-year	2.23	1.92	1.18	2.68	2.44	0.34	0.90
Part-time/full-year	0.69	1.36	0.78	1.51	1.65	0.35	0.55
Part-time/part-year	1.76	2.04		2.65	2.96	0.61	1.42
Housing							
On campus	4.19	1.32	0.78	1.12	0.96	1.38	0.58
Off campus	1.02	0.85	0.50	1.20	1.38	0.20	0.60
Living with parents	1.46	0.99	0.92	1.72	2.14	0.45	0.88
Price of attendance							
Less than \$6,000	1.47	2.52	2.04	4.09	5.42	3.56	1.29
\$6,000–10,999	0.86	1.06	1.19	2.68	3.08	0.63	0.62
\$11,000–18,999	0.76	0.93	0.60	1.91	2.76	0.24	0.50
\$19,000 or more	1.32	1.38	0.64	0.75	0.67	0.18	0.30
Gender							
Male	1.20	0.91	0.63	1.11	1.06	0.29	0.69
Female	1.11	0.70	0.51	0.91	0.90	0.25	0.57
Race/ethnicity							
White	1.02	0.75	0.42	0.91	0.90	0.33	0.57
Black	1.60	1.38	0.90	1.96	1.50	0.17	1.03
Hispanic	1.29	1.66	1.24	1.70	1.84	0.35	0.54
Asian	2.13	2.48	1.84	4.05	1.76	0.86	1.20
American Indian	5.63	5.00	5.13	2.12	†	1.38	2.95
Pacific Islander	4.36	4.20	5.28	10.65	2.31	2.28	2.71
Two or more races	2.24	3.42	2.60	2.75	3.99	0.96	1.32
Other	7.94	10.27	6.75	10.21	†	1.95	4.00
Age as of 12/31/07							
18 years or younger	1.37	1.87	1.50	2.50	1.38	0.93	0.80
19–23 years	1.41	0.72	0.53	0.84	0.78	0.34	0.57
24–29 years	1.02	1.52	1.18	1.28	1.52	0.16	0.62
30–39 years	1.72	1.97	1.36	2.34	2.94	0.30	1.13
40 years or older	1.41	3.06	3.06	4.21	3.53	0.56	1.21
Dependency status							
Dependent	1.36	0.75		0.89	0.78	0.45	0.55
Independent	1.04	1.10		1.44	1.59	0.17	0.70
Unmarried, no dependents	1.07	1.60		1.38	1.13	0.22	0.68
Married, no dependents	1.49	2.38		3.45	3.51	0.94	1.19
Single parent	2.05	1.55		1.48	4.64	0.10	1.18
Married parent	1.41	2.06	2.24	2.52	3.06	0.41	0.89

Table S4.3-A. Standard errors for table 4.3-A: Percentage of undergraduates who have financial need, by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year	Priv nonprof			
		Non-		Non-			All
Student characteristics	Public 2-year	doctorate- granting	Doctorate- granting	doctorate- granting	Doctorate- granting	For-profit	insti- tutions
Dependency and income in 2006							
Dependent							
Less than \$20,000	0.14	0.54	0.28	0.35	0.75	0.13	0.11
\$20,000-39,999	0.93	0.98	0.31	0.80	†	†	0.43
\$40,000-59,999	2.34	2.13	0.67	0.55	0.41	0.74	1.15
\$60,000-79,999	1.81	1.74	1.00	1.56	0.64	1.47	0.92
\$80,000-99,999	1.08	1.94	1.25	2.47	1.08	2.71	0.97
\$100,000 or more	0.62	1.38	0.79	2.02	1.60	3.82	0.53
Independent							
Less than \$10,000	0.11	0.44	0.09	0.10	†	†	0.06
\$10,000-19,999	1.13	1.22	0.43	0.13	†	#	0.60
\$20,000-29,999	1.67	1.61	1.49	1.95	0.58	0.03	0.98
\$30,000-49,999	1.24	1.85	2.11	1.88	2.81	0.36	0.89
\$50,000 or more	0.88	2.09	2.01	3.67	2.97	1.04	0.98
Income group							
Lowest 25 percent	0.29	0.45	0.10	0.38	0.20	0.02	0.14
Middle 50 percent	1.33	0.82	0.54	0.69	0.45	0.19	0.71
Highest 25 percent	0.70	1.15	0.82	1.74	1.51	1.23	0.54
Aid status							
No aid	0.72	1.31	0.98	3.48	2.15	3.01	0.51
Received aid	0.70	0.63	0.42	0.65	0.84	0.12	0.20
Grant status							
No grant	0.72	0.91	0.69	2.59	1.57	0.50	0.52
Received grant	0.75	0.72	0.48	0.74	0.89	0.09	0.25
Loan status							
No loan	1.22	1.16	0.59	1.69	1.21	1.72	0.89
Received loan	0.68	0.65	0.55	0.51	0.61	0.11	0.25

[†] Not applicable.

[#] Rounds to zero.

Table 4.3-B. Average financial need (price of attendance minus expected family contribution) among undergraduates with financial need, by type of institution and selected student characteristics: 2007–08

		Public	: 4-year	Priv nonprof			
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions ¹
Total	\$6,000	\$9,400	\$11,400	\$18,600	\$22,000	\$17,300	\$11,400
Attendance status							
Full-time/full-year	10,000	11,700	13,300	22,800	25,600	24,500	16,600
Full-time/part-year	5,500	7,400	8,400	12,200	15,800	13,700	9,800
Part-time/full-year	6,200	8,100		13,300	14,800	15,900	8,400
Part-time/part-year	3,100	4,700	5,400	6,500	7,600	8,900	4,400
Housing							
On campus	6,900	10,600	12,700	22,300	24,900	24,400	17,600
Off campus	6,200	9,600	11,300	15,300	18,400	17,500	10,800
Living with parents	5,600	8,100	9,500	16,800	20,900	15,900	8,800
Price of attendance							
Less than \$6,000	3,100	3,400	3,600	3,500	3,700	4,700	3,200
\$6,000–10,999	6,200	6,700	6,400	6,500	6,500	7,100	6,400
\$11,000–18,999	10,200	10,500	10,600	11,400	11,000	12,300	10,800
\$19,000 or more	13,900	14,600	15,200	22,600	25,100	23,700	20,800
Gender							
Male	5,800	9,200	11,300	18,200	21,900	17,300	11,200
Female	6,100	9,600	11,500	18,900	22,000	17,300	11,600
Race/ethnicity ²							
White	5,700	8,900	10,600	18,500	21,100	16,800	11,100
Black	6,400	11,100	13,300	18,600	22,300	17,700	12,100
Hispanic	6,100	9,000	12,000	17,000	22,200	17,400	11,200
Asian	6,600	11,300	14,000	22,700	26,700	16,500	12,700
American Indian	6,500	8,500	10,500	15,800	‡	18,300	9,900
Pacific Islander	5,600	8,900	12,000	17,100	26,400	19,300	11,100
Two or more races	6,300	10,700	12,100	23,100	26,500	19,400	12,900
Other	5,100	‡	11,800	14,700	‡	21,800	11,500
Age as of 12/31/07							
18 years or younger	6,300	10,300	-	21,400	24,600	16,400	12,800
19–23 years	6,100	9,600	11,500	20,700	23,600	16,900	12,300
24–29 years	6,000	9,400	11,400	15,900	18,800	17,800	10,800
30–39 years	6,000	8,500		13,400	14,900	17,600	10,400
40 years or older	5,400	7,800	8,900	12,700	11,300	16,600	8,700
Dependency status							
Dependent	5,900	9,600	-	20,800	23,800	15,600	12,400
Independent	6,000	9,300		15,100	17,000	17,800	10,500
Unmarried, no dependents	5,700	9,300		16,000	19,600	16,400	10,100
Married, no dependents	5,000	8,200		13,100	14,300	15,500	9,200
Single parent	6,700	10,600		17,000	18,400	19,300	11,800
Married parent	5,800	8,100	10,100	12,900	12,400	17,200	9,900

Table 4.3-B. Average financial need (price of attendance minus expected family contribution) among undergraduates with financial need, by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year	Priv nonprof	/ate it 4-year		
		Non-		Non-			All
	Public	doctorate-	Doctorate-	doctorate-	Doctorate-		insti-
Student characteristics	2-year	granting	granting	granting	granting	For-profit	tutions ¹
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$7,400	\$12,900	\$17,200	\$26,600	\$30,900	\$17,900	\$13,700
\$20,000-39,999	6,000	12,000	15,500	27,700	31,700	18,000	13,400
\$40,000–59,999	5,000	9,600	12,100	23,700	30,000	16,400	12,300
\$60,000-79,999	4,200	6,900	8,800	21,300	25,000	11,400	11,200
\$80,000-99,999	4,300	5,700	7,700	18,100	23,300	8,500	12,000
\$100,000 or more	3,800	5,200	6,100	13,100	16,000	8,000	10,600
Independent							
Less than \$10,000	7,400	12,300	14,200	20,200	23,200	19,000	12,600
\$10,000-19,999	6,100	9,600	11,800	15,600	19,600	18,600	10,800
\$20,000–29,999	6,200	8,200	10,500	14,700	17,500	18,300	10,400
\$30,000-49,999	4,800	7,100	8,600	13,400	13,100	17,700	8,900
\$50,000 or more	3,600	4,600	6,600	9,700	9,800	12,200	7,300
Income group ³							
Lowest 25 percent	7,000	12,300	15,400	24,200	28,400	18,700	13,000
Middle 50 percent	5,400	8,100	10,000	18,500	23,400	17,300	10,900
Highest 25 percent	3,700	4,800	6,200	11,700	14,400	12,000	9,000
Aid status							
No aid	4,700	6,900	9,600	13,300	17,500	15,000	6,900
Received aid	6,800	10,100	11,900	19,100	22,700	17,300	12,800
Grant status							
No grant	4,700	6,600	8,600	12,900	16,900	13,000	7,600
Received grant	7,100	11,100	13,200	19,900	23,600	19,000	13,700
Loan status ⁴							
No loan	5,700	8,300	10,500	15,900	18,700	13,800	8,300
Received loan	7,200	10,500	12,100	19,800	24,000	17,500	14,900

[‡] Reporting standards not met.

NOTE: This table excludes students attending more than one institution and students who have no need. Financial need is defined as the total price of attendance minus the expected family contribution (EFC). Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

¹ All institutions also include public less-than-2-year and private nonprofit less-than-4-year institutions in addition to those listed separately in this table.

² Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

³ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁴ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S4.3-B. Standard errors for table 4.3-B: Average financial need (price of attendance minus expected family contribution) among undergraduates with financial need, by type of institution and selected student characteristics: 2007–08

			_		ate		
		Public	4-year	nonprof	it 4-year		
	5 :	Non-	.	Non-			All
Student characteristics	Public 2-year	granting	Doctorate- granting	granting	Doctorate- granting	For-profit	insti- tutions
Total	\$60	\$100	\$90	\$310	\$290	\$480	\$100
Attendance status							
Full-time/full-year	170	130	110	310	360	400	110
Full-time/part-year	110	340		550	1,830	550	250
Part-time/full-year	70	130	140	420	560	650	100
Part-time/part-year	50	150	140	300	300	440	60
Housing							
On campus	360	280	140	420	380	2,310	190
Off campus	70	150	110	330	380	520	130
Living with parents	80	190	180	500	850	530	110
Price of attendance							
Less than \$6,000	40	90		200	310	120	30
\$6,000–10,999	40	80	100	150	200	280	40
\$11,000–18,999	140	120		180	620	170	40
\$19,000 or more	750	260	150	320	290	410	170
Gender							
Male	70	140	110	390	380	650	110
Female	80	130	130	410	360	480	110
Race/ethnicity							
White	80	120		360	300	660	110
Black	110	380		640	820	690	190
Hispanic	120	190		580	1,380	490	260
Asian	180	380		1,260	660	1,220	240
American Indian	410	660		4,430	†	3,750	900
Pacific Islander	530	800		2,320	2,180	2,080	710
Two or more races	250	630		2,410	1,360	1,010	480
Other	660	†	1,180	2,090	†	3,470	1,240
Age as of 12/31/07							
18 years or younger	160	290	310	790	680	660	190
19–23 years	80	120	100	380	350	530	100
24–29 years	80	210		680	570	560	160
30–39 years	120	290		650	1,120	740	220
40 years or older	110	250	360	820	580	800	220
Dependency status	00	400	440	400	000	500	400
Dependent	90	130		400	360	580	100
Independent	70	140		400	450	520	150
Unmarried, no dependents	80	260		710	820	780 1.250	150
Married, no dependents	150	290		690	1,020	1,250	280
Single parent Married parent	110 140	380 280		770 510	740 780	560 620	200
manicu parent	140	200	340	510	700	020	250

Table S4.3-B. Standard errors for table 4.3-B: Average financial need (price of attendance minus expected family contribution) among undergraduates with financial need, by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year	Priv nonprof			
		Non-		Non-			All
Student characteristics	Public 2-year	doctorate- granting	Doctorate- granting	doctorate- granting	Doctorate- granting	For-profit	insti- tutions
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$140	\$250	\$200	\$1,030	\$1,570	\$670	\$220
\$20,000–39,999	130	280	180	870	1,080	840	210
\$40,000–59,999	110	250	220	670	640	990	200
\$60,000–79,999	170	220	190	780	630	1,680	270
\$80,000-99,999	390	180	210	790	610	1,260	300
\$100,000 or more	290	260	170	560	390	700	190
Independent							
Less than \$10,000	120	280	180	720	1,130	770	220
\$10,000-19,999	110	220	280	730	1,140	560	160
\$20,000–29,999	120	250	320	680	1,010	960	260
\$30,000-49,999	100	190	360	1,050	690	710	220
\$50,000 or more	160	250	360	570	710	710	220
Income group							
Lowest 25 percent	100	180	130	610	1,110	630	160
Middle 50 percent	70	120	120	380	340	580	100
Highest 25 percent	160	190	180	430	340	640	170
Aid status							
No aid	60	230	250	800	830	630	90
Received aid	150	120	100	310	330	490	170
Grant status							
No grant	60	150	140	620	590	470	90
Received grant	160	140	100	320	370	550	200
Loan status							
No loan	60	110	140	440	470	450	70
Received loan	310	160	110	340	350	510	170

[†] Not applicable.

Table 4.4-A. Percentage of undergraduates who have remaining need after receiving financial aid (price of attendance minus the expected family contribution and any financial aid), by type of institution and selected student characteristics: 2007–08

		Public	: 4-year		/ate it 4-year		
			, you		-		All
	Public	Non- doctorate-	Doctorate-	Non- doctorate-	Doctorate-	For-profit	insti- tutions ¹
Student characteristics	2-year	granting	granting	granting	granting		
Total	50.1	46.5	41.4	49.1	50.5	71.9	50.2
Attendance status							
Full-time/full-year	64.2	48.4	43.5	53.8	54.8	86.3	55.0
Full-time/part-year	53.3	46.1	36.0	41.4	44.8	65.2	51.1
Part-time/full-year	55.1	50.0	43.3	53.0	47.6	67.0	53.5
Part-time/part-year	39.1	38.8	33.9	33.5	32.8	57.2	39.4
Housing							
On campus	42.7	35.6	36.2	42.2	49.7	57.8	40.6
Off campus	52.6	49.2	43.5	53.4	49.3	74.0	53.4
Living with parents	46.1	49.6	43.7	57.3	57.8	64.5	48.6
Price of attendance							
Less than \$6,000	41.7	40.3	34.9	29.4	33.8	62.0	41.3
\$6,000–10,999	55.4	50.5	39.1	41.0	39.1	59.5	52.4
\$11,000–18,999	64.2	46.5	41.5	57.2	46.3	63.4	51.7
\$19,000 or more	57.4	47.0	43.7	50.2	52.9	81.3	55.7
Gender							
Male	45.9	44.7	41.8	45.8	49.2	67.8	46.8
Female	53.4	47.9	41.0	51.4	51.5	73.8	52.8
Race/ethnicity ²							
White	43.8	39.0	37.8	42.6	46.6	67.4	43.9
Black	62.7	54.0	46.6	59.3	52.1	77.2	61.2
Hispanic	59.3	60.3	49.6	68.5	60.2	75.7	61.4
Asian	58.6	69.1	55.3	64.4	65.2	67.2	60.2
American Indian	59.2	55.7	44.1	54.6	‡	74.7	58.6
Pacific Islander	50.9	63.6	50.6	38.1	79.2	77.6	55.3
Two or more races	50.9	51.6	44.1	54.0	51.0	75.8	52.5
Other	51.8	65.1	49.4	65.6	‡	80.1	57.6
Age as of 12/31/07							
18 years or younger	48.1	45.8	36.7	44.7	51.6	68.6	46.4
19–23 years	47.7	42.2	38.1	45.6	50.2	69.2	45.8
24–29 years	60.7	63.3	60.6	64.3	63.2	73.6	63.5
30–39 years	50.3	44.2	43.1	54.6	45.7	75.3	53.9
40 years or older	44.6	43.7		43.8		70.8	47.5
Dependency status							
Dependent	42.4	39.9	35.8	43.4	49.0	58.7	41.7
Independent	55.9	56.6		57.4		76.0	59.5
Unmarried, no dependents	62.4	61.4		64.0	60.6	68.2	63.2
Married, no dependents	36.2	45.6		40.1	40.5	59.5	41.8
Single parent	73.4	70.4		74.1	67.8	86.1	75.8
Married parent	39.6	41.3		45.5	42.2	73.0	45.4

Table 4.4-A. Percentage of undergraduates who have remaining need after receiving financial aid (price of attendance minus the expected family contribution and any financial aid), by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year	Priv nonprof			
		Non-		Non-			All
	Public	doctorate-	Doctorate-	doctorate-	Doctorate-		insti-
Student characteristics	2-year	granting	granting	granting	granting	For-profit	tutions ¹
Dependency and income in 2006							
Dependent							
Less than \$20,000	93.8	83.3	75.9	81.7	80.9	84.9	86.4
\$20,000–39,999	75.4	74.0	68.0	68.0	72.5	72.2	72.7
\$40,000-59,999	45.8	52.8	55.4	56.4	65.3	59.1	51.8
\$60,000–79,999	19.7	32.2	39.9	46.0	52.8	37.3	32.6
\$80,000-99,999	7.0	17.2	27.5	36.0	50.2	18.9	22.7
\$100,000 or more	0.9	5.4	9.2	19.6	32.3	10.4	11.4
Independent							
Less than \$10,000	91.4	80.9	74.9	84.0	83.9	85.2	86.0
\$10,000–19,999	78.4	69.4	69.1	78.2	74.4	83.4	77.2
\$20,000-29,999	62.1	62.4	60.3	68.0	69.2	75.8	65.2
\$30,000-49,999	58.8	52.7	51.9	62.8	50.4	73.9	60.1
\$50,000 or more	12.4	16.3	20.3	20.1	20.8	45.9	17.3
Income group ³							
Lowest 25 percent	87.8	78.9	72.9	78.7	78.3	83.4	82.4
Middle 50 percent	49.0	45.3	45.3	55.1	57.9	71.2	51.2
Highest 25 percent	9.4	9.3	10.1	19.7	28.9	40.3	14.2
Aid status							
No aid	43.8	48.6	49.4	54.1	63.9	75.4	46.7
Received aid	57.0	45.6	38.3	48.3	47.5	71.8	52.0
Grant status							
No grant	41.4	38.9	38.5	47.1	55.3	51.3	42.1
Received grant	63.4	53.3	43.9	49.7	48.5	80.6	57.7
Loan status ⁴							
No loan	52.1	53.7	47.0	56.2	59.3	83.5	52.6
Received loan	36.9	37.1	35.2	44.5	43.6	70.9	46.2

[‡] Reporting standards not met.

NOTE: Federal education tax benefits are not included in this table. This table excludes students attending more than one institution. Remaining need is defined as the total price of attendance minus the expected family contribution (EFC) minus all types of financial aid. The calculation of the percentage with remaining need includes students who had no need and those who received no aid. Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

¹ All institutions also include public less-than-2-year and private nonprofit less-than-4-year institutions in addition to those listed separately in this table.

² Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

³ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁴ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S4.4-A. Standard errors for table 4.4-A: Percentage of undergraduates who have remaining need after receiving financial aid (price of attendance minus the expected family contribution and any financial aid), by type of institution and selected student characteristics: 2007–08

		Public	4-year	Priv nonprof			
Student characteristics	Public 2-year	Non-	Doctorate- granting	Non- doctorate- granting	-	For-profit	All insti- tutions
Total	0.77	0.77	0.55	1.26	1.10	1.68	0.38
Attendance status							
Full-time/full-year	0.84	1.11	0.58	1.55	1.40	1.20	0.39
Full-time/part-year	1.87	2.60	1.35	2.64	2.89	2.55	0.97
Part-time/full-year	0.86	1.50	1.44	2.44	2.22	2.71	0.59
Part-time/part-year	0.95	2.10	1.45	2.92	2.45	2.97	0.70
Housing							
On campus	4.20	1.15	0.80	1.60	1.52	6.62	0.58
Off campus	0.78	1.02	0.66	1.61	1.46	1.75	0.43
Living with parents	1.06	1.24	1.09	2.51	3.35	2.06	0.65
Price of attendance							
Less than \$6,000	1.01	2.46	1.99	4.77	4.47	4.64	0.89
\$6,000–10,999	0.81	1.08	1.17	2.80	3.20	3.53	0.56
\$11,000–18,999	0.97	1.17	0.78	2.02	4.85	1.84	0.53
\$19,000 or more	4.96	1.89	0.91	1.57	1.18	1.44	0.77
Gender							
Male	0.83	1.00	0.71	1.79	1.66	2.12	0.42
Female	0.84	0.91	0.70	1.29	1.16	1.90	0.45
Race/ethnicity							
White	0.66	0.84	0.55	1.63	1.05	2.46	0.37
Black	1.52	1.85	1.49	2.99	2.36	2.22	0.87
Hispanic	1.24	2.13	1.55	3.17	5.27	1.70	0.74
Asian	2.04	2.72	2.12	5.20	2.90	3.28	1.16
American Indian	5.49	7.37	7.28	13.61	†	11.47	3.44
Pacific Islander	4.62	7.55	7.11	6.32	8.15	13.90	2.83
Two or more races	2.45	3.89	2.99	6.72	4.29	3.70	1.56
Other	7.38	9.00	9.41	13.68	†	9.44	4.18
Age as of 12/31/07							
18 years or younger	1.28	2.43	1.36	2.71	2.52	3.72	0.89
19–23 years	1.00	0.85	0.72	1.37	1.24	1.61	0.36
24–29 years	0.94	2.13	1.08	2.59	2.40	2.92	0.70
30–39 years	1.34	1.83	2.33	3.39	3.67	2.55	0.74
40 years or older	1.40	2.86	3.14	4.10	3.00	3.23	1.06
Dependency status							
Dependent	0.90	0.79	0.61	1.33	1.19	1.73	0.37
Independent	0.83	1.39	0.91	1.97	1.91	1.91	0.50
Unmarried, no dependents	1.03	2.14		2.75	2.33	3.24	0.76
Married, no dependents	1.47	2.33	2.30	2.99	3.21	5.30	1.03
Single parent	1.99	2.09		2.68	4.08	1.34	1.03
Married parent	1.29	2.34	1.81	3.32	3.36	2.42	0.90

Table S4.4-A. Standard errors for table 4.4-A: Percentage of undergraduates who have remaining need after receiving financial aid (price of attendance minus the expected family contribution and any financial aid), by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year		vate it 4-year		All insti- tutions
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate-	For-profit	
Dependency and income in 2006							
Dependent							
Less than \$20,000	0.67	1.66	1.52	2.92	2.63	1.70	0.58
\$20,000-39,999	1.16	2.14	1.27	2.87	2.76	2.99	0.77
\$40,000–59,999	1.75	2.23	1.46	3.47	2.89	4.00	0.93
\$60,000-79,999	1.10	1.56	1.40	2.79	2.58	7.07	0.71
\$80,000-99,999	0.97	1.60	1.26	4.19	3.42	4.96	0.86
\$100,000 or more	0.21	0.71	0.54	1.49	1.67	2.06	0.39
Independent							
Less than \$10,000	0.47	1.29	1.28	2.75	2.49	1.90	0.50
\$10,000–19,999	1.07	2.43	1.99	2.66	3.09	1.85	0.71
\$20,000-29,999	1.52	2.25	2.59	2.61	3.80	3.69	1.08
\$30,000-49,999	1.19	2.15	2.63	3.50	3.05	3.24	0.91
\$50,000 or more	0.63	1.65	1.81	3.50	2.10	3.59	0.59
Income group							
Lowest 25 percent	0.53	1.17	0.82	1.74	_	1.42	0.41
Middle 50 percent	0.88	0.98	0.88	1.53	1.35	2.43	0.51
Highest 25 percent	0.47	0.72	0.59	1.55	1.53	2.90	0.33
Aid status							
No aid	0.72	1.31	0.98	3.48	2.15	3.01	0.51
Received aid	0.83	0.84	0.53	1.42	1.22	1.73	0.42
Grant status							
No grant	0.70	1.01	0.77	2.16	1.62	2.72	0.48
Received grant	1.02	0.93	0.66	1.51	1.33	1.55	0.42
Loan status							
No loan	0.99	1.22	0.62	1.84		1.75	0.67
Received loan	1.04	0.92	0.69	1.57	1.16	1.82	0.59

[†] Not applicable.

Table 4.4-B. Average amount of remaining financial need (price of attendance minus the expected family contribution and any financial aid), among undergraduates with financial need after receiving financial aid, by type of institution and selected student characteristics: 2007–08

		Public	4-year	Priv nonprof			
Charles to be a sector of the	Public	Non- doctorate-	Doctorate-	Non- doctorate-	Doctorate-	Fan madit	All insti-
Student characteristics	2-year	granting	granting	granting	granting	For-profit	tutions ¹
Total	\$4,500	\$5,500	\$7,000	\$9,400	\$12,200	\$10,500	\$6,800
Attendance status							
Full-time/full-year	6,800	6,300	7,700	10,800	13,600	14,300	9,300
Full-time/part-year	4,300	4,900	5,900	7,500	10,300	7,800	6,200
Part-time/full-year	4,800	5,700	6,400	7,700	9,000	8,600	5,600
Part-time/part-year	2,600	3,100	4,200	4,800	5,400	5,400	3,200
Housing							
On campus	3,700	5,200	7,400	10,300	13,500	10,100	9,300
Off campus	4,700	5,700	7,000	8,800	11,200	10,700	6,800
Living with parents	4,200	5,100	6,100	9,000	10,200	9,700	5,500
Price of attendance							
Less than \$6,000	2,600	2,800	3,100	3,100	3,400	2,600	2,700
\$6,000-10,999	4,700	4,500	4,600	4,600	4,300	3,600	4,600
\$11,000–18,999	7,500	5,900	6,100	6,700	7,800	6,300	6,600
\$19,000 or more	13,800	9,300	9,900	11,200	13,700	14,200	12,400
Gender							
Male	4,500	5,600	7,400	9,500	12,700	10,400	6,800
Female	4,500	5,400	6,600	9,300	11,800	10,500	6,700
Race/ethnicity ²							
White	4,300	5,300	6,900	9,600	12,000	10,700	6,700
Black	4,500	5,000	6,200	9,300		10,200	6,600
Hispanic	4,600	5,500	6,200	8,200	10,000	10,200	6,500
Asian	5,500	7,900	9,600	11,500	16,900	9,800	8,400
American Indian	4,800	4,200	5,900	‡	‡	12,100	6,100
Pacific Islander	4,500	5,400	8,000	‡		11,000	7,200
Two or more races	4,700	6,000	6,300	10,400		12,000	7,200
Other	3,700	‡	6,700	‡	‡	16,400	7,500
Age as of 12/31/07							
18 years or younger	4,500	5,400	6,100	8,900	12,700	8,700	6,500
19–23 years	4,600	5,600	7,300	10,100	12,900	10,300	7,100
24–29 years	4,600	5,800	6,900	9,400	11,400	10,700	6,800
30–39 years	4,500	4,700	6,600	8,000	9,600	10,900	6,700
40 years or older	4,300	4,400	5,400	8,300	7,700	10,400	5,800
Dependency status							
Dependent	4,400	5,300	6,900	9,900	12,800	9,300	6,900
Independent	4,600	5,700	7,100	8,800	10,600	10,800	6,700
Unmarried, no dependents	4,700	5,600	6,900	9,000		10,100	6,500
Married, no dependents	4,200	6,200	7,600	8,200	10,100	11,100	6,500
Single parent	4,800	6,200	7,700	9,100		11,300	7,200
Married parent	4,300	4,800	6,200	8,300	8,700	10,300	6,400

Table 4.4-B. Average amount of remaining financial need (price of attendance minus the expected family contribution and any financial aid), among undergraduates with financial need after receiving financial aid, by type of institution and selected student characteristics: 2007–08

—Continued

		Public	4-year		/ate it 4-year			
		Non-		Non-	, ,		All	
	Public		Doctorate-		Doctorate-		insti-	
Student characteristics	2-year	granting	granting	granting	granting	For-profit	tutions ¹	
Dependency and income in 2006								
Dependent								
Less than \$20,000	\$5,000	\$5,600	\$7,700	\$11,700	\$12,700	\$9,500	\$6,800	
\$20,000-39,999	4,100	5,300	6,900	8,900	12,700	9,300	6,100	
\$40,000-59,999	4,100	5,500	7,200	8,800	13,100	9,500	6,500	
\$60,000-79,999	3,900	4,800	6,600	11,200	14,400	8,000	7,300	
\$80,000-99,999	3,800	4,800	6,600	10,100	13,300	7,000	8,300	
\$100,000 or more	2,900	4,900	5,600	9,400	11,800	9,300	8,900	
Independent								
Less than \$10,000	5,400	7,200	7,800	9,500	12,300	10,800	7,500	
\$10,000–19,999	4,500	5,400	6,900	8,100	10,800	10,500	6,500	
\$20,000-29,999	4,800	5,400	6,900	8,800	10,400	11,500	6,800	
\$30,000-49,999	3,800	4,200	6,200	8,500	8,300	11,200	5,900	
\$50,000 or more	3,200	3,800	5,600	8,600	8,900	9,100	5,600	
Income group ³								
Lowest 25 percent	4,900	6,100	7,400	10,100	12,500	10,300	6,900	
Middle 50 percent	4,200	5,100	6,900	9,000	12,600	10,800	6,600	
Highest 25 percent	3,200	3,900	5,700	9,200	11,100	9,400	7,000	
Aid status								
No aid	4,700	6,900	9,600	13,300	17,500	15,000	6,900	
Received aid	4,400	4,900	5,600	8,700	10,600	10,300	6,700	
Grant status								
No grant	4,600	6,400	8,700	13,100	16,200	11,100	7,000	
Received grant	4,400	4,900	5,600	8,300	10,300	10,300	6,600	
Loan status ⁴								
No loan	4,600	6,400	8,500	11,800	15,100	11,700	6,500	
Received loan	3,400	3,800	4,700	7,500	9,200	10,400	7,200	

[‡] Reporting standards not met.

NOTE: Federal education tax benefits are not included in this table. This table excludes students attending more than one institution. Remaining need is defined as the total price of attendance minus the expected family contribution (EFC) minus all types of financial aid. These averages are shown only for those who had need remaining after receiving financial aid, if any. Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

¹ All institutions also include public less-than-2-year and private not-for-profit less-than-4-year institutions in addition to those listed separately in this table.

² Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

³ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁴ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S4.4-B. Standard errors for table 4.4-B: Average amount of remaining financial need (price of attendance minus the expected family contribution and any financial aid), among undergraduates with financial need after receiving financial aid, by type of institution and selected student characteristics: 2007–08

		Dublic	4	Priv			
		Public	4-year	nonprof	it 4-year		
	5	Non-		Non-			All
Student characteristics	Public 2-year	doctorate- granting	Doctorate- granting	granting	Doctorate- granting	For-profit	insti- tutions
Total	\$60	\$90	\$130	\$260	\$350	\$380	\$90
	φου	φθΟ	φ130	φ200	φυυ	φοου	φθΟ
Attendance status							
Full-time/full-year	130	130	160	360	390	350	120
Full-time/part-year	110	320	250	460	1,480	580	220
Part-time/full-year	70	130	160	520	570	500	80
Part-time/part-year	30	90	200	380	350	490	50
Housing							
On campus	330	220	210	600	530	1,890	230
Off campus	60	90	170	340	390	410	110
Living with parents	70	150	160	540	460	370	90
Price of attendance							
Less than \$6,000	30	100	90	200	240	160	20
\$6,000–10,999	50	80	120	170	320	300	40
\$11,000–18,999	110	100	110	190	540	140	60
\$19,000 or more	550	340	220	340	390	350	180
Gender							
Male	70	130	200	420	490	470	120
Female	60	110	120	340	380	400	100
Race/ethnicity							
White	70	130	160	410	370	520	110
Black	100	240	210	940	710	550	160
Hispanic	90	160	200	350	650	400	150
Asian	200	410	470	1,040	1,380	830	260
American Indian	350	520	740	†	†	1,870	550
Pacific Islander	450	840	910	÷	· †	1,720	540
Two or more races	220	450	490	2,330	2,500	1,340	410
Other	500	†	1,096	†	†	3,330	1,080
Age as of 12/31/07							
18 years or younger	130	250	260	650	750	460	150
19–23 years	80	130	140	470	410	380	100
24–29 years	90	160	280	640	630	500	150
30–39 years	100	240	330	470	880	530	170
40 years or older	90	190	400	760	560	550	160
Dependency status							
Dependent	80	120	130	480	380	360	100
Independent	60	100	200	360	510	400	120
Unmarried, no dependents	80	220	230	670	660	600	110
Married, no dependents	150	320	520	710	930	830	190
Single parent	100	270	280	690	770	490	170
Married parent	110	220	360	430	730	470	180

Table S4.4-B. Standard errors for table 4.4-B: Average amount of remaining financial need (price of attendance minus the expected family contribution and any financial aid), among undergraduates with financial need after receiving financial aid, by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year		Private nonprofit 4-year			
		Non-		Non-			All	
	Public	doctorate-		doctorate-			insti- tutions	
Student characteristics	2-year	granting	granting	granting	granting	For-profit		
Dependency and income in 2006								
Dependent								
Less than \$20,000	\$110	\$220	\$260	\$1,260	\$750	\$570	\$150	
\$20,000–39,999	110	150	210	590	670	560	110	
\$40,000–59,999	120	220	260	580	790	850	150	
\$60,000–79,999	240	230	240	1,370	1,310	730	310	
\$80,000–99,999	600	290	270	1,690	970	1,220	400	
\$100,000 or more	720	630	360	710	490	1,950	350	
Independent								
Less than \$10,000	100	260	220	690	820	640	180	
\$10,000–19,999	100	190	380	610	880	430	130	
\$20,000-29,999	110	220	420	580	800	730	210	
\$30,000-49,999	90	190	380	900	740	450	160	
\$50,000 or more	170	270	570	610	1,140	630	200	
Income group								
Lowest 25 percent	70	160	140	540	550	500	110	
Middle 50 percent	70	100	170	360	480	390	100	
Highest 25 percent	170	290	350	620	500	640	250	
Aid status								
No aid	60	230	250	800	830	630	90	
Received aid	70	100	70	250	280	390	120	
Grant status								
No grant	60	170	250	790	680	450	90	
Received grant	70	100	80	240	260	420	120	
Loan status								
No loan	60	120	190	470	510	460	90	
Received loan	120	140	90	290	370	400	180	

[†] Not applicable.

Table 4.5-A. Percentage of undergraduates with financial need who received any financial aid, by type of institution and selected student characteristics: 2007–08

		Public	: 4-year		/ate it 4-year		All
		Non-		Non-	, , , , , , , , , , , , , , , , , , ,		
	Public		Doctorate-		Doctorate-		insti-
Student characteristics	2-year	granting	granting	granting	granting	For-profit	tutions ¹
Total	61.2	79.5	79.9	91.7	85.9	97.2	76.8
Attendance status							
Full-time/full-year	75.6	87.7	83.6	94.7	87.4	97.0	86.1
Full-time/part-year	65.9	79.2	78.2	90.9	80.2	96.6	80.9
Part-time/full-year	63.3	74.0	75.9	90.3	87.0	98.7	71.5
Part-time/part-year	46.7	58.3	62.8	74.5	79.4	97.5	57.7
Housing							
On campus	79.9	87.3	83.3	94.1	85.2	95.3	86.8
Off campus	61.8	78.4	78.4	89.1	85.0	97.4	76.1
Living with parents	59.5	75.4	79.1	92.1	91.6	96.6	71.3
Price of attendance							
Less than \$6,000	45.3	46.5	40.3	45.3	49.9	83.6	45.7
\$6,000–10,999	67.5	76.1	71.1	85.9	78.9	96.0	72.4
\$11,000–18,999	75.9	86.3	84.8	92.0	86.2	98.0	85.2
\$19,000 or more	80.1	86.9	81.6	94.2	87.0	97.5	89.4
Gender							
Male	56.2	76.5	76.8	91.0	84.0	97.2	73.3
Female	64.6	81.9	82.7	92.2	87.4	97.1	79.2
Race/ethnicity ²							
White	59.0	80.3	77.7	91.4	84.4	96.9	75.5
Black	69.5	86.7	89.8	92.5	94.2	98.4	83.6
Hispanic	64.4	76.3	86.2	96.0	92.7	96.4	79.2
Asian	46.5	61.0	69.6	80.6	76.6	95.6	62.7
American Indian	68.5	77.3	84.6	85.4	‡	98.6	78.1
Pacific Islander	57.4	79.7	71.7	85.7	75.4	98.4	71.1
Two or more races	64.5	79.8	86.5	94.7	89.2	96.6	80.1
Other	59.3	‡	79.2	67.2	‡	96.0	73.9
Age as of 12/31/07							
18 years or younger	67.6	89.8	89.6	94.5	90.3	97.2	83.2
19–23 years	62.2	81.9	80.7	93.9	84.9	97.0	78.4
24–29 years	58.7	69.8	72.8	86.0	82.6	97.2	72.5
30–39 years	63.6	78.2	80.2	89.3	90.4	97.4	77.8
40 years or older	55.4	76.1	68.3	87.1	85.6	97.1	69.3
Dependency status							
Dependent	64.0	84.7	82.8	94.3	85.9	96.8	80.3
Independent	59.5	72.8	73.6	87.6	85.8	97.3	73.5
Unmarried, no dependents	51.4	68.9		82.9	82.7	97.0	67.5
Married, no dependents	48.1	70.3	73.5	87.0	86.7	96.1	68.1
Single parent	65.7	77.2		92.6	85.3	97.6	77.9
Married parent	67.1	76.5	79.8	88.9	91.4	97.4	79.2

Table 4.5-A. Percentage of undergraduates with financial need who received any financial aid, by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year	Private nonprofit 4-year			
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions ¹
Dependency and income in 2006							
Dependent							
Less than \$20,000	67.2	88.4	85.9	94.8	95.1	98.2	80.3
\$20,000-39,999	65.6	89.0	90.5	98.4	93.5	96.6	81.4
\$40,000-59,999	59.3	81.3	83.0	97.3	93.6	97.3	78.1
\$60,000-79,999	56.2	80.6	81.8	94.5	88.7	94.4	79.8
\$80,000-99,999	64.8	80.4	75.9	96.3	86.0	95.2	81.7
\$100,000 or more	87.3	86.5	76.3	88.2	76.1	96.6	80.8
Independent							
Less than \$10,000	60.5	73.7	73.5	87.6	82.8	96.8	73.6
\$10,000-19,999	60.8	74.2	75.0	86.9	81.4	98.1	73.9
\$20,000-29,999	63.3	73.3	75.4	85.6	81.9	97.5	75.0
\$30,000-49,999	55.9	70.8	72.2	87.6	91.6	97.8	70.8
\$50,000 or more	54.2	69.9	70.7	90.2	90.6	96.1	74.8
Income group ³							
Lowest 25 percent	63.9	82.4	83.3	93.0	90.1	97.0	77.6
Middle 50 percent	59.4	77.7	78.8	92.2	87.9	97.5	75.9
Highest 25 percent	57.5	78.9	75.2	88.4	78.9	96.2	78.0
Aid status							
No aid	†	†	†	†	†	†	†
Received aid	100.0	100.0	100.0	100.0		100.0	100.0
Grant status							
No grant	18.4	45.9	48.2	56.5	42.8	90.0	38.3
Received grant	100.0	100.0	100.0	100.0	100.0	100.0	100.0
Loan status ⁴							
No loan	52.9	57.7	52.6	73.6	62.4	62.1	55.3
Received loan	100.0	100.0	100.0	100.0	100.0	100.0	100.0

[†] Not applicable.

NOTE: Federal education tax benefits are not included in this table. This table excludes students attending more than one institution and students who have no need. Financial need is defined as the total price of attendance minus the expected family contribution (EFC). Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[‡] Reporting standards not met.

¹ All institutions also include public less-than-2-year and private nonprofit less-than-4-year institutions in addition to those listed separately in this table

² Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

³ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁴ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S4.5-A. Standard errors for table 4.5-A: Percentage of undergraduates with financial need who received any financial aid, by type of institution and selected student characteristics: 2007–08

		Public	4-year	Private nonprofit 4-year		_	
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions
Total	2.48	0.79	0.82	0.63		0.19	1.16
Attendance status							
Full-time/full-year	1.28	0.64	0.86	0.68	1.05	0.32	0.56
Full-time/part-year	3.26	1.93		1.44	2.59	0.51	1.22
Part-time/full-year	2.03	1.57		1.78	1.55	0.32	1.59
Part-time/part-year	4.13	2.14	1.89	2.18	2.51	0.46	2.66
Housing							
On campus	4.79	1.18	0.90	0.95	1.46	1.93	0.78
Off campus	2.68	0.86	0.98	0.94	1.21	0.25	1.34
Living with parents	2.31	1.64	1.52	1.45	1.30	0.56	1.28
Price of attendance							
Less than \$6,000	3.85	2.91	2.84	5.36	6.96	3.34	3.33
\$6,000–10,999	1.95	1.69	1.94	3.15	2.92	0.66	1.39
\$11,000–18,999	1.56	0.70	1.04	1.66	2.40	0.29	0.83
\$19,000 or more	3.24	1.31	0.93	0.69	1.06	0.28	0.55
Gender							
Male	2.91	1.16	1.14	1.02	1.45	0.36	1.36
Female	2.21	0.89	0.67	0.70	0.94	0.25	1.02
Race/ethnicity							
White	2.84	0.84	0.86	0.82	1.22	0.33	1.29
Black	2.41	2.09	1.02	1.53	1.06	0.30	1.27
Hispanic	2.24	1.69	1.42	1.13	1.45	0.46	0.83
Asian	3.40	4.66		3.73	4.33	1.08	1.99
American Indian	5.04	6.98	5.70	8.04	†	1.28	2.82
Pacific Islander	5.76	9.63		4.83	14.09	1.16	3.60
Two or more races	3.22	3.34		1.97	3.11	1.26	1.53
Other	9.63	†	7.85	15.89	†	3.84	4.93
Age as of 12/31/07							
18 years or younger	2.44	1.40	1.11	1.97	1.81	0.86	0.96
19–23 years	2.34	0.88	0.82	0.59	1.20	0.32	0.93
24–29 years	3.12	1.92		2.12	2.84	0.33	1.81
30–39 years	3.44 2.37	2.20 2.33	1.84 4.46	2.16 2.38	1.78 2.52	0.32 0.51	1.81 1.56
40 years or older	2.37	2.33	4.40	2.30	2.52	0.51	1.50
Dependency status	2.20	0.04	0.05	0.70	4.44	0.07	0.04
Dependent	2.38	0.84		0.73		0.37	0.91
Independent Unmarried, no dependents	2.64 3.14	1.14 1.89		1.19 1.89	1.28 1.97	0.23 0.31	1.50
Married, no dependents Married, no dependents	2.74	2.85		2.36	2.49	0.80	1.79 1.95
Single parent	2.74	2.00	1.96	1.14		0.80	1.79
Married parent	1.99	2.98		2.32		0.33	0.95

Table S4.5-A. Standard errors for table 4.5-A: Percentage of undergraduates with financial need who received any financial aid, by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year	Priv nonprof			
		Non-		Non-			All
	Public	doctorate-	Doctorate-	doctorate-	Doctorate-		insti-
Student characteristics	2-year	granting	granting	granting	granting	For-profit	tutions
Dependency and income in 2006							
Dependent							
Less than \$20,000	2.24	1.90	1.42	1.58	1.38	0.44	1.29
\$20,000-39,999	2.53	1.58	0.90	0.54	1.39	0.71	1.16
\$40,000-59,999	3.05	1.46	1.63	1.12	0.96	0.78	1.26
\$60,000-79,999	3.91	1.64	1.32	1.35	2.74	1.36	1.14
\$80,000-99,999	5.41	2.03	1.34	0.78	2.93	1.93	0.98
\$100,000 or more	4.12	2.14	1.61	2.24	2.27	1.44	1.06
Independent							
Less than \$10,000	2.76	1.69	1.38	2.10	2.87	0.39	1.46
\$10,000-19,999	3.11	1.79	2.36	2.03	4.14	0.32	1.74
\$20,000-29,999	3.14	2.22	2.69	2.47	3.62	0.46	2.03
\$30,000-49,999	2.71	2.65	2.78	2.48	1.95	0.45	1.62
\$50,000 or more	2.51	3.47	2.89	2.56	1.96	0.70	1.29
Income group							
Lowest 25 percent	2.34	1.44	0.73	0.88	1.42	0.29	1.23
Middle 50 percent	2.78	0.94	1.11	0.70	1.20	0.27	1.26
Highest 25 percent	2.49	1.99	1.42	1.80	2.10	0.58	0.89
Aid status							
No aid	†	†	†	†	†	†	†
Received aid	†	†	†	†	†	†	†
Grant status							
No grant	1.80	1.39	1.59	3.00	2.53	0.66	1.41
Received grant	†	†	†	†	†	†	†
Loan status							
No loan	2.92	1.75	1.57	1.88	2.40	2.33	2.22
Received loan	†	†	†	†	†	†	†

[†] Not applicable.

Table 4.5-B. Average amount of financial aid received by undergraduates who had financial need, by type of institution and selected student characteristics: 2007–08

		Public	: 4-year	Priv nonprof		For-profit	
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting		All insti- tutions ¹
Total	\$3,800	\$8,700	\$11,000	\$17,000	\$20,000	\$10,900	\$10,000
Attendance status							
Full-time/full-year	5,800	10,400	12,400	20,400	23,100	13,100	13,600
Full-time/part-year	3,300	6,400	8,100	11,400	14,300	9,900	8,000
Part-time/full-year	3,700	6,900	9,400	11,300	13,000	11,400	6,700
Part-time/part-year	2,100	4,800	6,000	6,800	7,700	7,200	4,300
Housing							
On campus	6,200	11,300	13,100	22,300	23,800	22,600	17,600
Off campus	3,900	8,400	10,700	12,300	15,900	10,700	8,600
Living with parents	3,600	6,600	8,300	13,800	17,300	11,200	7,200
Price of attendance							
Less than \$6,000	1,600	2,300	2,900	2,700	3,100	3,300	1,800
\$6,000–10,999	3,700	5,000	5,600	5,200	5,600	5,800	4,400
\$11,000–18,999	6,100	9,500	10,000	8,700	8,600	9,500	8,800
\$19,000 or more	7,400	13,700	14,600	20,700	22,800	13,400	17,100
Gender							
Male	3,900	8,700	11,000	16,800	20,100	11,800	10,300
Female	3,800	8,600	11,100	17,100	20,000	10,400	9,800
Race/ethnicity ²							
White	4,100	9,000	10,700	18,200	20,400	11,000	10,600
Black	3,800	10,000	12,400	14,800	18,900	10,600	9,400
Hispanic	3,200	6,500	10,800	12,200	18,700	10,700	8,400
Asian	3,400	7,700	11,500	19,300	20,200	11,600	10,400
American Indian	3,900	7,700	10,100	‡	‡	9,800	7,700
Pacific Islander	3,200	6,800	12,800	19,200	16,100	12,100	9,100
Two or more races	4,100	9,100	11,600	20,200	23,100	11,500	11,000
Other	3,500	‡	12,200	‡	‡	9,200	8,600
Age as of 12/31/07							
18 years or younger	4,000	9,100		21,300	22,400	12,200	11,900
19–23 years	3,800	9,100		19,900	22,200	11,200	11,400
24–29 years	4,000	8,000	10,000	11,500	14,300	10,900	8,300
30-39 years	3,800	7,800	9,700	10,000	11,900	10,400	7,700
40 years or older	3,400	6,800	8,600	9,600	10,000	10,400	6,800
Dependency status							
Dependent	3,900	9,300		20,500	22,600	12,300	12,100
Independent	3,800	7,700	9,900	11,100	12,900	10,400	7,800
Unmarried, no dependents	3,800	8,200		12,700	15,600	10,800	8,500
Married, no dependents	3,700	6,500		11,300	11,700	10,800	8,000
Single parent	3,700	7,500		10,600		10,000	7,300
Married parent	3,800	7,600	9,400	9,700	9,800	10,700	7,500

Table 4.5-B. Average amount of financial aid received by undergraduates who had financial need, by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year	Private nonprofit 4-year			
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions ¹
Dependency and income in 2006	•					•	
Dependent							
Less than \$20,000	\$3,900	\$9,400	\$13,300	\$18,000	\$21,700	\$10,200	\$9,700
\$20,000–39,999	4,000	9,300	12,200	22,400	24,400	12,200	10,900
\$40,000–59,999	3,600	8,900	11,100	21,000	24,200	12,300	11,500
\$60,000-79,999	3,900	9,100	10,300	20,300	22,200	12,700	12,400
\$80,000-99,999	4,700	9,600	11,100	20,400	23,700	14,600	15,100
\$100,000 or more	4,700	9,900	11,500	20,300	20,700	18,500	16,200
Independent							
Less than \$10,000	4,000	8,800	11,400	13,900	15,600	10,200	8,400
\$10,000-19,999	3,900	8,000	9,700	11,000	14,600	10,200	7,700
\$20,000-29,999	3,900	6,700	9,200	10,500	13,200	10,600	7,500
\$30,000-49,999	3,200	7,000	8,100	9,700	10,800	10,500	7,000
\$50,000 or more	3,700	6,500	8,600	10,000	10,100	11,200	8,400
Income group ³							
Lowest 25 percent	4,000	9,100	12,300	17,600	20,800	10,500	9,400
Middle 50 percent	3,700	8,300	10,300	16,900	20,700	10,900	9,900
Highest 25 percent	3,900	8,500	10,800	16,600	18,000	12,100	12,500
Aid status							
No aid	†	†	†	†	†	t	†
Received aid	3,800	8,700	11,000	17,000	20,000	10,900	10,000
Grant status							
No grant	4,400	7,800	9,400	9,600	13,100	10,600	8,600
Received grant	3,700	8,900	11,600	18,000	21,000	11,000	10,300
Loan status ⁴							
No loan	2,800	5,000	7,200	9,400	11,200	4,000	4,800
Received loan	6,400	10,600	12,500	19,600	23,300	11,200	13,100

[†] Not applicable.

NOTE: Federal education tax benefits are not included in this table. This table excludes students attending more than one institution, those who did not have any financial need, and those who did not receive any financial aid. Financial need is defined as the total price of attendance minus the expected family contribution (EFC). Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico.

[‡] Reporting standards not met.

¹ All institutions also include public less-than-2-year and private nonprofit less-than-4-year institutions in addition to those listed separately in this table

² Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

³ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁴ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S4.5-B. Standard errors for table 4.5-B: Average amount of financial aid received by undergraduates who had financial need, by type of institution and selected student characteristics: 2007–08

		Public	: 4-year	Priv nonprof			
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions
Total	\$140	\$130	\$100	\$390	\$420	\$170	\$150
Attendance status							
Full-time/full-year	190	150	120	460	510	250	110
Full-time/part-year	150	320		440	1,420	350	250
Part-time/full-year	160	160	200	520	580	220	110
Part-time/part-year	60	290	210	410	350	250	130
Housing							
On campus	310	260	150	470	460	1,520	180
Off campus	160	180	130	310	440	180	150
Living with parents	120	150	180	520	1,390	280	150
Price of attendance							
Less than \$6,000	40	170	150	260	360	200	40
\$6,000-10,999	90	130	100	250	230	130	70
\$11,000–18,999	250	110	90	180	360	220	90
\$19,000 or more	310	340	220	450	370	280	160
Gender							
Male	120	160	120	490	600	330	180
Female	160	160	140	480	440	190	160
Race/ethnicity							
White	140	140	120	510	410	240	140
Black	200	360	200	650	650	270	190
Hispanic	150	260	270	750	2,220	270	290
Asian	170	370	360	1,320	790	880	310
American Indian	390	780	810	†	†	2,130	780
Pacific Islander	320	1,160		3,970	1,260	2,100	650
Two or more races	280	760		960	1,520	780	330
Other	710	†	1,010	†	†	990	800
Age as of 12/31/07							
18 years or younger	160	350		660	640	600	290
19–23 years	130	160		450	490	280	130
24–29 years	180	190	180	550	560	290	190
30–39 years	210	280		310	700	300	190
40 years or older	160	270	390	410	490	350	160
Dependency status							
Dependent	120	170		430	480	380	140
Independent	160	150		300	360	170	150
Unmarried, no dependents	160	190		650	640	250	170
Married, no dependents	180	300		580	810	790	270
Single parent	190	330		420	530	220	170
Married parent	210	290	410	330	510	310	200

Table S4.5-B. Standard errors for table 4.5-B: Average amount of financial aid received by undergraduates who had financial need, by type of institution and selected student characteristics: 2007–08—Continued

		Public	4-year	Private nonprofit 4-year			
Student characteristics	Public 2-year	Non- doctorate- granting	Doctorate- granting	Non- doctorate- granting	Doctorate- granting	For-profit	All insti- tutions
Dependency and income in 2006							
Dependent							
Less than \$20,000	\$180	\$300	\$290	\$640	\$1,710	\$390	\$240
\$20,000-39,999	160	290	180	930	1,030	470	260
\$40,000-59,999	120	280	220	710	880	680	250
\$60,000-79,999	170	310	240	790	740	1,160	260
\$80,000-99,999	340	380	330	870	840	970	310
\$100,000 or more	340	510	310	620	700	1,320	300
Independent							
Less than \$10,000	240	290	240	800	990	270	220
\$10,000-19,999	180	250	270	650	790	270	200
\$20,000-29,999	200	360	320	470	850	300	160
\$30,000-49,999	150	240	320	340	400	270	160
\$50,000 or more	200	460	550	530	540	430	200
Income group							
Lowest 25 percent	180	190	140	630	1,150	250	200
Middle 50 percent	130	140	130	400	450	170	140
Highest 25 percent	180	420	310	710	570	420	220
Aid status							
No aid	†	†	†	†	†	†	†
Received aid	140	130	100	390	420	170	150
Grant status							
No grant	110	180	160	460	510	300	150
Received grant	160	150	110	430	470	190	170
Loan status							
No loan	70	130		360	510	180	100
Received loan	260	140	110	450	430	180	100

[†] Not applicable.

Table 5.1-A. Percentage distribution of undergraduates, by dependency status and selected institutional and student characteristics: 2007–08

			ent			
		Unmarried,	-			
		no	Married, no	Single	Married	
Institutional and student characteristics	Dependent	dependents	dependents	parent	parent	
Total	53.0	15.7	5.9	13.4	12.0	
Institution type						
Public						
Less-than-2-year	29.3	18.8	8.5	23.4	20.1	
2-year	42.8	17.7	7.4	16.4	15.7	
4-year	69.1	13.6	4.4	6.4	6.6	
Non-doctorate-granting	60.7	15.0	5.4	9.5	9.5	
Doctorate-granting	73.7	12.8	3.8	4.7	5.0	
Private nonprofit						
Less-than-4-year	31.5	23.4	5.2	23.5	16.4	
4-year	66.4	11.1	4.8	8.1	9.7	
Non-doctorate-granting	59.6	12.6	5.6	9.7	12.4	
Doctorate-granting	73.9	9.3	3.9	6.2	6.7	
For-profit	23.8	20.7	6.0	31.5	17.9	
More than one institution	58.2	15.2	5.8	10.7	10.1	
Attendance status						
Full-time/full-year	75.8	8.5	2.9	7.8	5.0	
Full-time/part-year	51.9	16.9	5.1	15.7	10.4	
Part-time/full-year	39.7	19.8	7.6	16.1	16.8	
Part-time/part-year	29.0	22.8	9.8	18.4	20.0	
Housing ¹						
On campus	93.8	3.7	0.7	1.2	0.6	
Off campus	29.2	21.3	9.4	20.1	20.0	
Living with parents	80.6	10.4	1.3	6.3	1.6	
Price of attendance ¹						
Less than \$6,000	35.6	20.8	9.2	16.4	18.0	
\$6,000–10,999	44.5	18.1	6.8	15.7	14.9	
\$11,000–18,999	61.2	14.0	4.2	12.0	8.6	
\$19,000 or more	69.2	9.8	3.5	10.1	7.2	
Gender						
Male	58.0	18.0	5.6	8.2	10.2	
Female	49.1	14.0	6.2	17.2	13.5	

Table 5.1-A. Percentage distribution of undergraduates, by dependency status and selected institutional and student characteristics: 2007–08—Continued

			Independ	lent	
		Unmarried,	-		
		no	Married, no	Single	Married
Institutional and student characteristics	Dependent	dependents	dependents	parent	parent
Race/ethnicity ²					
White	56.4	15.0	6.5	9.7	12.4
Black	39.0	17.0	4.3	28.1	11.6
Hispanic	50.5	16.3	5.0	16.2	12.0
Asian	56.9	17.6	6.2	8.9	10.4
American Indian	39.5	17.8	5.1	20.5	17.0
Pacific Islander	55.9	17.2	3.8	14.3	8.8
Two or more races	53.9	16.5	5.9	13.6	10.2
Other	48.5	17.3	7.1	14.5	12.6
Age as of 12/31/07					
18 years or younger	96.2	1.2	0.4	1.9	0.3
19–23 years	87.2	2.6	2.7	5.8	1.7
24–29 years	†	52.2	10.3	21.7	15.8
30–39 years	†	22.8	10.0	31.3	35.9
40 years or older	†	23.3	14.6	24.6	37.5
Income group ³					
Lowest 25 percent	52.9	24.1	2.1	18.3	2.5
Middle 50 percent	53.0	17.1	5.1	15.1	9.8
Highest 25 percent	52.9	4.1	11.6	4.7	26.6
Aid status					
No aid	51.6	17.1	7.5	10.8	13.0
Received aid	53.6	15.0	5.1	14.7	11.6
Grant status					
No grant	53.2	17.2	7.5	9.3	12.7
Received grant	52.7	14.3	4.4	17.1	11.4
Loan status ⁴					
No loan	51.5	15.7	7.0	12.6	13.2
Received loan	55.3	15.7	4.3	14.6	10.2

[†] Not applicable. All students age 24 or older are considered to be independent.

NOTE: Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico. Detail may not sum to totals because of rounding. SOURCE: U.S. Department of Education, National Center for Education Statistics, 2007–08 National Postsecondary Student Aid Study (NPSAS:08).

¹ Excludes students attending more than one institution.

² Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

³ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁴ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S5.1-A. Standard errors for table 5.1-A: Percentage distribution of undergraduates, by dependency status and selected institutional and student characteristics: 2007–08

			Independ	ent	
		Unmarried,	·		
		no	Married, no	Single	Married
Institutional and student characteristics	Dependent	dependents	dependents	parent	parent
Total	0.31	0.20	0.12	0.21	0.23
Institution type					
Public					
Less-than-2-year	5.33	1.63	1.01	3.11	2.99
2-year	0.49	0.35	0.22	0.40	0.48
4-year	0.70	0.32	0.17	0.19	0.36
Non-doctorate-granting	0.85	0.50	0.26	0.43	0.65
Doctorate-granting	0.84	0.40	0.19	0.20	0.41
Private nonprofit					
Less-than-4-year	3.26	2.44	0.49	2.60	1.69
4-year	0.94	0.42	0.27	0.50	0.48
Non-doctorate-granting	1.74	0.70	0.46	0.84	0.88
Doctorate-granting	0.89	0.48	0.29	0.53	0.52
For-profit	1.25	1.05	0.64	1.56	0.91
More than one institution	0.70	0.55	0.35	0.48	0.38
Attendance status					
Full-time/full-year	0.42	0.16	0.12	0.27	0.19
Full-time/part-year	1.47	0.49	0.36	1.13	0.48
Part-time/full-year	0.47	0.38	0.21	0.52	0.41
Part-time/part-year	0.59	0.53	0.35	0.43	0.57
Housing					
On campus	0.29	0.23	0.08	0.12	0.09
Off campus	0.36	0.35	0.20	0.32	0.39
Living with parents	0.44	0.29	0.11	0.29	0.12
Price of attendance					
Less than \$6,000	0.61	0.59	0.35	0.43	0.57
\$6,000-10,999	0.65	0.43	0.32	0.42	0.46
\$11,000–18,999	0.88	0.39	0.20	0.30	0.38
\$19,000 or more	0.84	0.29	0.23	0.47	0.35
Gender					
Male	0.38	0.32	0.18	0.37	0.30
Female	0.38	0.26	0.15	0.28	0.24

Table S5.1-A. Standard errors for table 5.1-A: Percentage distribution of undergraduates, by dependency status and selected institutional and student characteristics: 2007–08

—Continued

			Independent					
		Unmarried,						
		no	Married, no	Single	Married			
Institutional and student characteristics	Dependent	dependents	dependents	parent	parent			
Race/ethnicity								
White	0.35	0.23	0.16	0.30	0.34			
Black	0.90	0.60	0.34	0.91	0.55			
Hispanic	0.64	0.52	0.23	0.55	0.47			
Asian	0.94	0.65	0.52	0.67	0.66			
American Indian	2.40	1.61	1.11	2.41	1.91			
Pacific Islander	3.07	2.57	1.05	2.98	1.42			
Two or more races	1.54	1.29	0.67	1.08	0.82			
Other	3.81	2.71	1.83	2.95	2.57			
Age as of 12/31/07								
18 years or younger	0.32	0.17	0.08	0.20	0.07			
19–23 years	0.24	0.09	0.11	0.18	0.08			
24–29 years	†	0.60	0.32	0.52	0.42			
30–39 years	†	0.65	0.45	0.71	0.89			
40 years or older	†	0.71	0.55	0.80	0.90			
Income group								
Lowest 25 percent	0.46	0.38	0.15	0.39	0.14			
Middle 50 percent	0.39	0.29	0.16	0.29	0.26			
Highest 25 percent	0.64	0.21	0.31	0.25	0.62			
Aid status								
No aid	0.51	0.39	0.28	0.55	0.46			
Received aid	0.43	0.25	0.14	0.28	0.22			
Grant status								
No grant	0.43	0.30	0.27	0.43	0.35			
Received grant	0.40	0.26	0.13	0.29	0.26			
Loan status								
No loan	0.38	0.27	0.16	0.25	0.35			
Received loan	0.43	0.32	0.23	0.37	0.25			

[†] Not applicable.

Table 5.1-B. Percentage distribution of undergraduates, by attendance status and selected institutional and student characteristics: 2007–08

	Full-t	ime	Part-time		
Institutional and student characteristics	Full-year	Part-year	Full-year	Part-year	
Total	39.3	14.0	22.1	24.6	
Institution type					
Public					
Less-than-2-year	30.6	34.7	12.9 !	21.8	
2-year	17.5	11.9	30.9	39.7	
4-year	57.3	12.4	15.4	14.9	
Non-doctorate-granting	49.3	11.8	18.6	20.3	
Doctorate-granting	61.6	12.7	13.7	12.0	
Private nonprofit					
Less-than-4-year	25.3	33.1	19.4	22.1	
4-year	62.3	14.3	11.3	12.1	
Non-doctorate-granting	58.5	15.1	12.8	13.6	
Doctorate-granting	66.4	13.4	9.6	10.5	
For-profit	37.0	33.4	13.2	16.4	
More than one institution	48.8	6.6	30.5	14.2	
Housing ¹					
On campus	78.7	12.1	5.7	3.6	
Off campus	29.0	14.8	24.3	31.8	
Living with parents	35.9	16.0	23.9	24.2	
Price of attendance ¹					
Less than \$6,000	#	10.7	11.8	77.5	
\$6,000–10,999	15.3	24.4	44.9	15.4	
\$11,000–18,999	58.5	13.9	21.5	6.1	
\$19,000 or more	81.4	9.7	7.1	1.8	
Gender					
Male	39.8	14.7	20.6	24.9	
Female	38.9	13.5	23.2	24.4	
Race/ethnicity ²					
White	41.7	13.3	20.8	24.1	
Black	32.1	17.0	22.8	28.1	
Hispanic	34.7	14.7	26.7	23.9	
Asian	42.9	12.2	21.0	23.9	
American Indian	31.4	17.9	23.3	27.4	
Pacific Islander	32.4	13.5	27.3	26.8	
Two or more races	43.2	14.3	21.0	21.5	
Other	29.7	16.2	32.3	21.7	
Age as of 12/31/07					
18 years or younger	61.7	14.7	13.4	10.2	
19–23 years	51.8	14.3	17.8	16.1	
24–29 years	22.0	16.9	27.5	33.5	
30–39 years	18.1	12.4	29.5	40.0	
40 years or older	12.3	9.2	32.8	45.6	

Table 5.1-B. Percentage distribution of undergraduates, by attendance status and selected institutional and student characteristics: 2007–08—Continued

	Full-t	time	Part-time		
Institutional and student characteristics	Full-year	Part-year	Full-year	Part-year	
Dependency status					
Dependent	56.2	13.8	16.5	13.5	
Independent	20.2	14.4	28.3	37.2	
Unmarried, no dependents	21.4	15.1	27.8	35.7	
Married, no dependents	18.9	12.1	28.3	40.6	
Single parent	22.9	16.5	26.7	34.0	
Married parent	16.4	12.1	30.7	40.8	
Dependency and income in 2006					
Dependent					
Less than \$20,000	51.1	15.1	19.0	14.8	
\$20,000-39,999	50.8	13.4	20.2	15.7	
\$40,000-59,999	50.7	14.3	18.1	16.9	
\$60,000-79,999	54.7	13.4	16.8	15.1	
\$80,000-99,999	60.4	13.5	14.8	11.2	
\$100,000 or more	63.9	13.5	13.0	9.7	
Independent					
Less than \$10,000	31.0	20.6	21.9	26.5	
\$10,000–19,999	23.6	17.1	26.8	32.4	
\$20,000–29,999	19.9	15.9	29.7	34.5	
\$30,000–49,999	16.5	10.9	31.9	40.6	
\$50,000 or more	10.6	8.3	31.6	49.5	
Income group ³					
Lowest 25 percent	41.3	17.1	20.9	20.6	
Middle 50 percent	38.3	13.9	22.8	24.9	
Highest 25 percent	39.1	11.1	21.7	28.1	
•	39.1	11.1	21.7	20.1	
Aid status					
No aid	23.4	11.8	24.9	39.9	
Received aid	47.6	15.2	20.6	16.6	
Grant status					
No grant	29.0	13.6	23.5	34.0	
Received grant	48.9	14.5	20.8	15.8	
Loan status ⁴					
No loan	30.1	12.3	25.1	32.5	
Received loan	54.0	16.8	17.2	12.1	

[#] Rounds to zero.

NOTE: Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico. Detail may not sum to totals because of rounding. SOURCE: U.S. Department of Education, National Center for Education Statistics, 2007–08 National Postsecondary Student Aid Study (NPSAS:08).

[!] Interpret data with caution (estimates are unstable).

¹ Excludes students attending more than one institution.

² Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

³ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁴ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S5.1-B. Standard errors for table 5.1-B: Percentage distribution of undergraduates, by attendance status and selected institutional and student characteristics: 2007–08

	Full-t	ime	Part-time		
Institutional and student characteristics	Full-year	Part-year	Full-year	Part-year	
Total	0.28	0.28	0.33	0.35	
Institution type					
Public					
Less-than-2-year	3.54	7.77	4.07	4.40	
2-year	0.32	0.57	0.41	0.71	
4-year	0.51	0.28	0.35	0.31	
Non-doctorate-granting	0.83	0.51	0.58	0.51	
Doctorate-granting	0.60	0.35	0.44	0.39	
Private nonprofit					
Less-than-4-year	2.36	5.89	3.28	3.67	
4-year	0.89	0.74	0.45	0.53	
Non-doctorate-granting	1.16	0.73	0.73	0.84	
Doctorate-granting	1.56	1.37	0.48	0.50	
For-profit	1.71	1.27	0.95	1.27	
More than one institution	2.29	0.65	0.79	2.35	
Housing					
On campus	0.61	0.55	0.24	0.25	
Off campus	0.35	0.62	0.32	0.55	
Living with parents	0.49	0.44	0.43	0.58	
Price of attendance					
Less than \$6,000	†	0.61	0.53	0.86	
\$6,000–10,999	0.51	0.79	0.76	0.58	
\$11,000–18,999	0.68	0.43	0.40	0.34	
\$19,000 or more	0.72	0.59	0.35	0.16	
Gender					
Male	0.37	0.31	0.36	0.42	
Female	0.42	0.43	0.37	0.41	
Race/ethnicity					
White	0.33	0.30	0.27	0.42	
Black	0.77	0.93	0.60	0.72	
Hispanic	0.72	0.54	0.73	0.57	
Asian	1.00	0.74	1.21	1.27	
American Indian	2.14	2.43	2.22	3.16	
Pacific Islander	2.35	2.40	2.88	2.70	
Two or more races	1.62	0.98	1.46	1.47	
Other	3.79	2.98	4.21	3.57	
Age as of 12/31/07					
18 years or younger	0.76	0.70	0.68	0.57	
19–23 years	0.35	0.27	0.34	0.34	
24–29 years	0.47	0.59	0.58	0.70	
30–39 years	0.68	0.85	0.60	0.99	
40 years or older	0.59	0.79	0.80	0.90	

Table S5.1-B. Standard errors for table 5.1-B: Percentage distribution of undergraduates, by attendance status and selected institutional and student characteristics: 2007–08—Continued

	Full-t	ime	Part-time		
Institutional and student characteristics	Full-year	Part-year	Full-year	Part-year	
Dependency status					
Dependent	0.36	0.31	0.34	0.32	
Independent	0.41	0.62	0.41	0.59	
Unmarried, no dependents	0.41	0.45	0.64	0.74	
Married, no dependents	0.69	0.90	0.85	1.24	
Single parent	0.75	1.41	0.70	0.87	
Married parent	0.65	0.62	0.70	0.87	
Dependency and income in 2006					
Dependent					
Less than \$20,000	1.17	0.63	0.70	0.80	
\$20,000-39,999	0.90	0.56	0.66	0.61	
\$40,000-59,999	0.93	0.64	0.64	0.90	
\$60,000–79,999	0.86	0.61	0.66	0.91	
\$80,000–99,999	0.90	0.67	0.63	0.68	
\$100,000 or more	0.67	0.62	0.41	0.37	
Independent					
Less than \$10,000	1.02	0.99	0.60	0.61	
\$10,000–19,999	0.74	0.81	0.72	0.87	
\$20,000–29,999	0.67	0.71	0.75	0.95	
\$30,000–49,999	0.65	0.65	0.71	0.85	
\$50,000 or more	0.53	0.72	0.73	1.24	
Income group					
Lowest 25 percent	0.80	0.58	0.40	0.40	
Middle 50 percent	0.29	0.31	0.38	0.49	
Highest 25 percent	0.52	0.34	0.47	0.60	
Aid status					
No aid	0.57	0.29	0.53	0.93	
Received aid	0.92	0.36	0.35	0.51	
Grant status					
No grant	0.45	0.28	0.37	0.76	
Received grant	0.98	0.39	0.41	0.52	
Loan status					
No loan	0.32	0.32	0.44	0.70	
Received loan	0.88	0.41	0.32	0.55	

[†] Not applicable.

Table 5.1-C. Percentage distribution of undergraduates, by student housing and selected institutional and student characteristics: 2007–08

Institutional and student characteristics	On campus	Off campus	Living with parents
Total	15.4	58.6	26.0
Institution type			
Public			
Less-than-2-year	‡	72.7	23.7
2-year	1.0	62.0	37.0
4-year	26.3	54.8	18.9
Non-doctorate-granting	20.9	53.9	25.2
Doctorate-granting	29.3	55.3	15.4
Private nonprofit			
Less-than-4-year	‡	69.5	30.4
4-year	46.0	42.6	11.4
Non-doctorate-granting	42.7	45.9	11.3
Doctorate-granting	49.6	39.0	11.4
For-profit	0.8	78.4	20.8
Attendance status			
Full-time/full-year	31.5	44.3	24.3
Full-time/part-year	12.6	59.1	28.2
Part-time/full-year	4.1	66.8	29.1
Part-time/part-year	2.1	73.2	24.7
Price of attendance			
Less than \$6,000	1.5	64.4	34.1
\$6,000–10,999	4.2	62.3	33.5
\$11,000–18,999	18.0	58.2	23.7
\$19,000 or more	38.9	49.2	11.9
Gender			
Male	16.7	55.0	28.3
Female	14.3	61.5	24.2
Race/ethnicity ¹			
White	17.5	58.6	23.9
Black	13.5	65.3	21.2
Hispanic	7.5	55.4	37.1
Asian	17.4	50.7	31.9
American Indian	9.3	65.7	25.0
Pacific Islander	9.9	54.1	36.0
Two or more races	17.2	57.0	25.8
Other	9.7	63.1	27.2
Age as of 12/31/07			
18 years or younger	37.9	18.7	43.3
19–23 years	22.3	41.4	36.3
24–29 years	2.4	81.3	16.3
30–39 years	0.9	94.3	4.8
40 years or older	0.5	96.3	3.3

Table 5.1-C. Percentage distribution of undergraduates, by student housing and selected institutional and student characteristics: 2007–08—Continued

Institutional and student characteristics	On campus	Off campus	Living with parents
Dependency status			
Dependent	27.5	32.6	39.8
Independent	2.0	87.4	10.6
Unmarried, no dependents	3.6	79.3	17.1
Married, no dependents	1.9	92.6	5.5
Single parent	1.3	86.7	12.0
Married parent	0.7	96.0	3.3
Dependency and income in 2006			
Dependent			
Less than \$20,000	19.7	31.8	48.5
\$20,000-39,999	21.0	30.7	48.3
\$40,000-59,999	22.4	32.3	45.3
\$60,000–79,999	25.9	30.3	43.8
\$80,000–99,999	29.4	34.1	36.4
\$100,000 or more	37.8	35.0	27.2
Independent			
Less than \$10,000	4.3	78.8	16.8
\$10,000–19,999	2.5	82.7	14.9
\$20,000–29,999	1.3	87.6	11.1
\$30,000–49,999	1.0	90.8	8.2
\$50,000 or more	0.7	95.9	3.4
Income group ²			
Lowest 25 percent	12.7	53.7	33.6
Middle 50 percent	14.3	58.4	27.3
Highest 25 percent	20.5	64.3	15.2
Aid status			
No aid	8.8	58.3	32.9
Received aid	18.9	58.8	22.3
Grant status			
No grant	10.6	59.2	30.2
Received grant	19.8	58.2	22.0
Loan status ³			
No loan	10.6	58.3	31.1
Received loan	23.2	59.2	17.6

[‡] Reporting standards not met.

NOTE: This table excludes students who attended more than one institution. Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico. Detail may not sum to totals because of rounding.

¹ Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

² Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

³ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S5.1-C. Standard errors for table 5.1-C: Percentage distribution of undergraduates, by student housing and selected institutional and student characteristics: 2007–08

Institutional and student characteristics	On campus	Off campus	Living with parents
Total	0.28	0.42	0.31
Institution type			
Public			
Less-than-2-year	†	6.63	2.65
2-year	0.11	0.52	0.51
4-year	0.67	0.71	0.40
Non-doctorate-granting	0.79	0.78	0.75
Doctorate-granting	0.86	0.96	0.47
Private nonprofit			
Less-than-4-year	†	3.93	3.96
4-year	1.05	0.93	0.50
Non-doctorate-granting	1.62	1.54	0.61
Doctorate-granting	1.45	1.13	0.94
For-profit	0.21	1.15	1.20
Attendance status			
Full-time/full-year	0.49	0.47	0.37
Full-time/part-year	0.71	1.36	0.99
Part-time/full-year	0.18	0.51	0.50
Part-time/part-year	0.15	0.59	0.54
Price of attendance			
Less than \$6,000	0.13	0.67	0.68
\$6,000–10,999	0.21	0.63	0.58
\$11,000–18,999	0.58	0.78	0.48
\$19,000 or more	0.75	0.80	0.37
Gender			
Male	0.36	0.46	0.39
Female	0.32	0.49	0.37
Race/ethnicity			
White	0.30	0.50	0.40
Black	0.61	0.80	0.64
Hispanic	0.40	0.71	0.75
Asian	0.97	1.13	1.25
American Indian	1.92	2.64	2.73
Pacific Islander	1.98	3.34	2.93
Two or more races	1.33	1.55	1.46
Other	2.26	3.50	3.26
Age as of 12/31/07			
18 years or younger	0.91	0.62	0.87
19–23 years	0.37	0.48	0.45
24–29 years	0.17	0.49	0.45
30–39 years	0.11	0.35	0.34
40 years or older	0.09	0.31	0.29

Table S5.1-C. Standard errors for table 5.1-C: Percentage distribution of undergraduates, by student housing and selected institutional and student characteristics: 2007–08—Continued

Institutional and student characteristics	On campus	Off campus	Living with parents
Dependency status			
Dependent	0.42	0.44	0.46
Independent	0.10	0.27	0.26
Unmarried, no dependents	0.25	0.56	0.50
Married, no dependents	0.22	0.49	0.46
Single parent	0.14	0.52	0.51
Married parent	0.10	0.27	0.25
Dependency and income in 2006			
Dependent			
Less than \$20,000	0.89	0.88	1.11
\$20,000-39,999	0.70	0.76	0.89
\$40,000-59,999	0.63	0.73	0.84
\$60,000–79,999	0.92	0.83	1.05
\$80,000–99,999	0.96	0.87	0.87
\$100,000 or more	0.75	0.82	0.62
Independent			
Less than \$10,000	0.29	0.63	0.62
\$10,000–19,999	0.25	0.67	0.64
\$20,000–29,999	0.19	0.64	0.63
\$30,000–49,999	0.14	0.49	0.48
\$50,000 or more	0.10	0.33	0.31
Income group			
Lowest 25 percent	0.38	0.53	0.55
Middle 50 percent	0.26	0.47	0.43
Highest 25 percent	0.58	0.80	0.43
Aid status			
No aid	0.35	0.51	0.51
Received aid	0.46	0.48	0.31
Grant status			
No grant	0.30	0.58	0.47
Received grant	0.50	0.44	0.38
Loan status			
No loan	0.31	0.43	0.39
Received loan	0.41	0.59	0.39

[†] Not applicable.

Table 5.2-A. Percentage distribution of undergraduates, by race/ethnicity and selected institutional and student characteristics: 2007–08

Institutional and student characteristics	White	Black	Hispanic	Asian	American Indian	Pacific Islander	Two or more races	Other
Total	61.8	14.0	14.1	5.9	0.8	0.7	2.4	0.3
Institution type								
Public								
Less-than-2-year	56.1	14.2	19.9	4.4	1.6 !	1.8 !	1.8	‡
2-year	60.2	14.4	14.8	6.1	1.0	0.9	2.3	0.3
4-year	66.5	11.5	11.9	6.2	8.0	0.5	2.3	0.3
Non-doctorate-granting	61.9	13.1	16.1	4.8	1.2	0.4	2.0	0.3
Doctorate-granting	69.0	10.6	9.6	6.9	0.6	0.5	2.5	0.2
Private nonprofit								
Less-than-4-year	29.8 !	12.5 !	33.5	8.7 !	8.6 !	‡	‡	‡
4-year	67.7	11.7	11.6	5.5	0.3	0.6	2.2	0.3
Non-doctorate-granting	67.7	13.1	12.1	3.7	0.4	0.7 !	2.1	0.3
Doctorate-granting	67.8	10.2	11.0	7.5	0.2	0.5 !	2.4	0.3
For-profit	46.6	24.8	20.8	3.2	1.0 !	0.7	2.5	0.3 !
More than one institution	61.2	12.2	14.4	7.7	0.7	8.0	2.8	0.2
Attendance status								
Full-time/full-year	65.6	11.4	12.5	6.4	0.7	0.6	2.6	0.2
Full-time/part-year	58.6	16.9	14.9	5.1	1.1	0.7	2.4	0.3
Part-time/full-year	58.3	14.5	17.1	5.6	0.9	0.9	2.2	0.4
Part-time/part-year	60.6	15.9	13.7	5.7	0.9	0.8	2.1	0.3
Housing ¹								
On campus	70.4	12.4	6.9	6.5	0.5	0.5	2.6	0.2
Off campus	61.8	15.7	13.3	5.0	1.0	0.7	2.3	0.3
Living with parents	56.8	11.5	20.2	7.1	0.8	1.0	2.3	0.3
Price of attendance ¹								
Less than \$6,000	60.8	14.2	15.4	5.7	1.0	0.8	1.9	0.3
\$6,000–10,999	58.7	15.0	16.3	5.4	1.0	0.9	2.4	0.4
\$11,000–18,999	62.1	14.3	14.2	5.3	0.8	0.5	2.5	0.3
	-	_						
\$19,000 or more	65.7	13.0	10.4	6.7	0.7	0.7	2.6	0.2
Gender								
Male	64.2	12.0	13.3	6.4	0.9	0.7	2.2	0.4
Female	59.9	15.5	14.8	5.6	0.8	0.7	2.5	0.2
Age as of 12/31/07								
18 years or younger	61.0	11.7	16.3	6.3	0.7	0.6	3.0	0.3
19–23 years	65.5	10.9	13.6	6.0	0.7	0.8	2.3	0.2
24–29 years	56.3	15.6	16.9	6.8	1.0	0.8	2.4	0.3
30–39 years	54.2	21.9	14.5	5.1	1.1	0.6	2.2	0.4
40 years or older	63.0	18.7	9.5	4.4	1.3	0.6	2.2	0.3

Table 5.2-A. Percentage distribution of undergraduates, by race/ethnicity and selected institutional and student characteristics: 2007–08—Continued

Institutional and student					American	Pacific	Two or more	
characteristics	White	Black	Hispanic	Asian	Indian	Islander	races	Other
Dependency status								
Dependent	65.8	10.3	13.5	6.3	0.6	0.8	2.4	0.3
Independent	57.2	18.1	14.9	5.4	1.1	0.7	2.3	0.3
Unmarried, no dependents	59.0	15.1	14.7	6.6	1.0	0.8	2.5	0.3
Married, no dependents	68.1	10.0	11.9	6.1	0.7	0.5	2.3	0.3
Single parent	44.6	29.5	17.2	3.9	1.3	0.8	2.4	0.3
Married parent	63.3	13.5	14.1	5.1	1.2	0.5	2.0	0.3
Dependency and income in 2006 Dependent								
Less than \$20,000	39.1	21.7	24.3	9.6	0.8	1.2	2.9	0.3
\$20,000-39,999	47.8	18.2	21.4	7.7	0.8	0.8	2.9	0.4
\$40,000-59,999	62.2	11.4	15.4	6.7	0.7	0.7	2.4	0.3
\$60,000-79,999	72.6	7.0	10.8	5.6	0.6	0.9	2.2	0.3 !
\$80,000-99,999	76.7	5.6	8.6	5.3	0.6	0.8	2.1	‡
\$100,000 or more	81.2	4.0	6.7	4.8	0.4	0.5	2.1	0.2
Independent								
Less than \$10,000	48.9	21.6	16.9	6.9	1.6	0.7	3.0	0.3
\$10,000–19,999	54.7	19.5	16.4	5.0	0.7	1.0	2.3	0.4
\$20,000-29,999	53.5	20.3	17.1	4.6	1.0	0.7	2.5	0.3
\$30,000-49,999	58.3	18.6	14.6	4.7	1.1	0.6	2.0	0.2 !
\$50,000 or more	68.4	12.0	10.7	5.4	0.9	0.5	1.9	0.3
Income group ²								
Lowest 25 percent	46.1	20.8	20.1	7.7	1.2	0.8	3.0	0.4
Middle 50 percent	63.2	13.6	13.8	5.3	0.8	0.8	2.3	0.3
Highest 25 percent	75.1	7.8	8.6	5.2	0.7	0.5	2.0	0.3
Aid status								
No aid	65.6	9.7	12.8	8.0	0.7	8.0	2.2	0.3
Received aid	59.8	16.2	14.9	4.8	0.9	0.7	2.5	0.3
Grant status								
No grant	66.2	10.6	12.3	7.0	0.7	8.0	2.3	0.3
Received grant	57.6	17.1	15.9	4.9	1.0	0.7	2.5	0.3
Loan status ³								
No loan	62.0	11.5	15.1	7.2	0.9	0.8	2.3	0.3
Received loan	61.3	18.0	12.6	3.9	0.8	0.6	2.5	0.3

[!] Interpret data with caution (estimates are unstable).

NOTE: Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico. Detail may not sum to totals because of rounding.

[‡] Reporting standards not met.

¹ Excludes students attending more than one institution.

² Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

³ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S5.2-A. Standard errors for table 5.2-A: Percentage distribution of undergraduates, by race/ethnicity and selected institutional and student characteristics: 2007–08

Institutional and student characteristics	White	Black	Hispanic	Asian	American Indian	Pacific Islander	Two or more races	Other
Total	0.52	0.28	0.44	0.18	0.09	0.06	0.12	0.02
Institution type								
Public								
Less-than-2-year	5.66	4.19	3.71	1.07	0.71	0.58	0.45	†
2-year	0.86	0.53	0.78	0.32	0.19	0.09	0.19	0.05
4-year	0.63	0.33	0.59	0.26	0.08	0.06	0.10	0.03
Non-doctorate-granting	1.14	0.85	1.25	0.34	0.23	0.07	0.18	0.05
Doctorate-granting	0.77	0.33	0.56	0.35	0.06	0.08	0.13	0.04
Private nonprofit								
Less-than-4-year	9.54	4.29	8.94	3.60	2.64	†	†	†
4-year	0.82	0.62	0.73	0.40	0.06	0.16	0.24	0.05
Non-doctorate-granting	1.40	0.99	1.07	0.42	0.10	0.26	0.34	0.07
Doctorate-granting	1.35	0.75	1.35	0.60	0.05	0.16	0.26	0.07
For-profit	1.55	1.78	1.09	0.47	0.34	0.19	0.29	0.13
More than one institution	1.07	0.45	0.67	0.56	0.11	0.15	0.26	0.05
Attendance status								
Full-time/full-year	0.39	0.27	0.35	0.23	0.08	0.05	0.13	0.03
Full-time/part-year	0.91	0.87	0.49	0.33	0.15	0.14	0.19	0.07
Part-time/full-year	1.00	0.42	0.74	0.26	0.15	0.12	0.20	0.06
Part-time/part-year	0.78	0.48	0.74	0.37	0.15	0.10	0.23	0.05
Housing								
On campus	0.61	0.50	0.31	0.43	0.10	0.10	0.20	0.05
Off campus	0.53	0.40	0.40	0.23	0.10	0.07	0.13	0.04
Living with parents	0.90	0.40	0.83	0.38	0.16	0.11	0.19	0.04
Price of attendance								
Less than \$6,000	1.07	0.60	0.96	0.38	0.16	0.09	0.23	0.05
\$6,000–10,999	0.66	0.46	0.61	0.29	0.17	0.11	0.18	0.06
\$11,000–18,999	0.71	0.53	0.49	0.23	0.08	0.07	0.12	0.04
\$19,000 or more	0.71	0.33	0.49	0.23	0.00	0.07	0.12	0.04
Gender								
Male	0.54	0.30	0.50	0.22	0.08	0.08	0.16	0.04
Female	0.54	0.35	0.30	0.22	0.08	0.06	0.10	0.04
Age as of 12/31/07								
18 years or younger	0.78	0.46	0.72	0.45	0.14	0.10	0.25	0.08
19–23 years	0.54	0.40	0.45	0.43	0.08	0.07	0.23	0.04
24–29 years	0.76	0.23	0.61	0.28	0.17	0.07	0.13	0.05
30–39 years	0.70	0.31	0.63	0.40	0.17	0.09	0.20	0.03
40 years or older	0.98	0.82	0.57	0.40	0.13	0.12	0.23	0.06

Table S5.2-A. Standard errors for table 5.2-A: Percentage distribution of undergraduates, by race/ethnicity and selected institutional and student characteristics: 2007–08—Continued

Institutional and student characteristics	White	Black	Hispanic	Asian	American Indian	Pacific Islander	Two or more races	Other
Dependency status								
Dependent	0.45	0.25	0.46	0.23	0.07	0.07	0.12	0.03
Independent	0.71	0.46	0.49	0.19	0.13	0.07	0.17	0.04
Unmarried, no dependents	0.88	0.58	0.53	0.24	0.15	0.14	0.26	0.06
Married, no dependents	0.98	0.77	0.62	0.51	0.17	0.13	0.32	0.10
Single parent	1.46	1.15	0.84	0.34	0.23	0.17	0.25	0.07
Married parent	0.77	0.73	0.69	0.33	0.17	0.09	0.19	0.07
Dependency and income in 2006 Dependent								
Less than \$20,000	1.15	0.79	1.37	0.55	0.23	0.19	0.43	0.07
\$20,000-39,999	0.87	0.61	0.77	0.36	0.17	0.14	0.25	0.08
\$40,000-59,999	0.83	0.54	0.81	0.48	0.10	0.14	0.26	0.09
\$60,000-79,999	0.66	0.42	0.56	0.36	0.09	0.15	0.21	0.09
\$80,000-99,999	0.77	0.39	0.45	0.53	0.14	0.19	0.25	†
\$100,000 or more	0.42	0.21	0.32	0.23	0.09	0.07	0.17	0.05
Independent								
Less than \$10,000	1.10	0.73	0.62	0.37	0.31	0.13	0.23	0.10
\$10,000-19,999	1.13	0.76	0.79	0.39	0.12	0.19	0.26	0.11
\$20,000-29,999	1.14	0.92	0.85	0.45	0.22	0.16	0.30	0.09
\$30,000-49,999	0.82	0.70	0.65	0.39	0.17	0.13	0.29	0.06
\$50,000 or more	0.82	0.60	0.70	0.38	0.17	0.11	0.24	0.07
Income group								
Lowest 25 percent	0.83	0.50	0.68	0.26	0.20	0.09	0.17	0.05
Middle 50 percent	0.52	0.35	0.41	0.25	0.07	0.08	0.15	0.03
Highest 25 percent	0.47	0.31	0.43	0.23	0.10	0.07	0.16	0.04
Aid status								
No aid	0.74	0.41	0.75	0.34	0.08	0.10	0.24	0.04
Received aid	0.47	0.34	0.34	0.13	0.11	0.06	0.10	0.03
Grant status								
No grant	0.54	0.31	0.46	0.27	0.08	0.08	0.20	0.04
Received grant	0.57	0.37	0.46	0.14	0.12	0.07	0.09	0.03
Loan status								
No loan	0.79	0.36	0.75	0.24	0.11	0.07	0.16	0.03
Received loan	0.49	0.41	0.27	0.16	0.11	0.07	0.12	0.04

[†] Not applicable.

Table 5.2-B. Percentage distribution of undergraduates, by gender, citizenship, and selected institutional and student characteristics: 2007–08

				Citizenship	
	0-	nder			Foreign/
Institutional and student characteristics	Male	Female	U.S. citizen	Resident alien	international student
Total	43.1	56.9	94.1	4.5	1.4
Institution type					
Public					
Less-than-2-year	37.2	62.8	94.6	5.4!	#
2-year	43.7	56.3	93.3	5.6	1.0
4-year	46.6	53.4	94.6	3.6	1.8
Non-doctorate-granting	45.0	55.0	93.1	5.1	1.8
Doctorate-granting	47.5	52.5	95.4	2.8	1.8
Private nonprofit					
Less-than-4-year	39.4	60.6	84.8	12.9	‡
4-year	44.0	56.0	95.0	2.4	2.7
Non-doctorate-granting	42.5	57.5	96.0	1.9	2.1
Doctorate-granting	45.6	54.4	93.8	2.9	3.3
For-profit	30.8	69.2	95.3	4.6	0.1 !
More than one institution	39.5	60.5	94.2	4.7	1.1
Attendance status					
Full-time/full-year	43.6	56.4	94.6	3.6	1.8
Full-time/part-year	45.2	54.8	93.6	4.1	2.2
Part-time/full-year	40.2	59.8	93.1	6.1	0.8
Part-time/part-year	43.5	56.5	94.7	4.6	0.7
Housing ¹					
On campus	47.2	52.8	95.2	2.0	2.9
Off campus	40.7	59.3	94.1	4.7	1.2
Living with parents	47.3	52.7	93.6	5.4	1.0
	47.5	32.1	93.0	5.4	1.0
Price of attendance ¹					
Less than \$6,000	45.0	55.0	94.6	4.9	0.5
\$6,000–10,999	42.6	57.4	93.3	5.6	1.1
\$11,000–18,999	42.7	57.3	94.5	4.2	1.3
\$19,000 or more	43.1	56.9	94.0	3.2	2.8
Race/ethnicity ²					
White	44.8	55.2	98.0	1.4	0.6
Black	36.9	63.1	93.0	5.8	1.2
Hispanic	40.4	59.6	88.0	10.6	1.4
Asian	46.4	53.6	70.9	18.5	10.6
American Indian	45.1	54.9	97.5	1.9 !	‡
Pacific Islander	44.0	56.0	83.6	14.5	1.9 !
Two or more races	40.4	59.6	96.6	2.6	0.9
Other	52.5	47.5	90.1	6.2	3.7 !
Age as of 12/31/07					
18 years or younger	43.6	56.4	95.7	3.2	1.1
19–23 years	45.8	54.2	94.6	3.8	1.7
24–29 years	44.6	55.4	92.2	6.0	1.8
30–39 years	37.5	62.5	92.6	6.6	0.8
40 years or older	33.8	66.2	95.5	4.1	0.4 !

Table 5.2-B. Percentage distribution of undergraduates, by gender, citizenship, and selected institutional and student characteristics: 2007–08—Continued

				Citizenship	
				-	Foreign/
		ender		5	international
Institutional and student characteristics	Male	Female	U.S. citizen	Resident alien	student
Dependency status					
Dependent	47.2	52.8	94.9	3.5	1.6
Independent	38.4	61.6	93.3	5.6	1.1
Unmarried, no dependents	49.5	50.5	93.3	5.0	1.7
Married, no dependents	40.3	59.7	91.1	7.7	1.2
Single parent	26.5	73.5	94.6	4.5	0.9
Married parent	36.3	63.7	93.1	6.4	0.6
Dependency and income in 2006					
Dependent					
Less than \$20,000	42.1	57.9	90.0	7.9	2.2
\$20,000-39,999	44.3	55.7	92.4	6.3	1.3
\$40,000-59,999	46.0	54.0	94.6	3.6	1.7
\$60,000-79,999	48.6	51.4	96.2	2.1	1.7
\$80,000-99,999	48.2	51.8	96.5	2.2	1.3
\$100,000 or more	50.5	49.5	97.0	1.4	1.6
Independent					
Less than \$10,000	39.3	60.7	92.1	5.3	2.6
\$10,000-19,999	37.7	62.3	92.6	6.5	0.9
\$20,000-29,999	34.6	65.4	93.9	5.4	0.7
\$30,000-49,999	38.3	61.7	93.8	5.7	0.5
\$50,000 or more	40.7	59.3	94.2	5.0	0.7
Income group ³					
Lowest 25 percent	41.2	58.8	91.7	6.3	2.0
Middle 50 percent	42.6	57.4	94.6	4.2	1.2
Highest 25 percent	45.9	54.1	95.6	3.2	1.2
Aid status					
No aid	48.4	51.6	92.6	4.6	2.8
Received aid	40.3	59.7	94.9	4.4	0.7
Grant status					
No grant	47.5	52.5	93.7	4.1	2.1
Received grant	38.9	61.1	94.5	4.8	0.7
Loan status ⁴					
No loan	45.7	54.3	92.6	5.3	2.1
Received loan	38.9	61.1	96.6	3.2	0.3

[#] Rounds to zero.

NOTE: Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico. Detail may not sum to totals because of rounding. SOURCE: U.S. Department of Education, National Center for Education Statistics, 2007–08 National Postsecondary Student Aid Study (NPSAS:08).

[!] Interpret data with caution (estimates are unstable).

[‡] Reporting standards not met.

¹ Excludes students attending more than one institution.

² Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

³ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁴ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S5.2-B. Standard errors for table 5.2-B: Percentage distribution of undergraduates, by gender, citizenship, and selected institutional and student characteristics: 2007–08

				Citizenship	
	_				Foreign/
Institutional and student characteristics	Male	ender Female	U.S. citizen	Resident alien	international student
Total	0.60	0.60	0.14	0.14	0.07
Institution type					
Public					
Less-than-2-year	3.90	3.90	1.62	1.62	†
2-year	0.65	0.65	0.26	0.24	0.08
4-year	0.44	0.44	0.24	0.20	0.17
Non-doctorate-granting	0.76	0.76	0.46	0.49	0.22
Doctorate-granting	0.47	0.47	0.30	0.18	0.19
Private nonprofit	0	0	0.00	00	00
Less-than-4-year	6.05	6.05	3.97	3.72	†
4-year	1.01	1.01	0.31	0.21	0.28
Non-doctorate-granting	1.48	1.48	0.37	0.22	0.32
Doctorate-granting	1.05	1.05	0.54	0.39	0.45
For-profit	1.84	1.84	0.41	0.40	0.04
More than one institution	0.61	0.61	0.46	0.36	0.18
	0.01	0.01	0.10	0.00	0.10
Attendance status	0.46	0.46	0.17	0.12	0.12
Full-time/full-year	1.38	1.38	0.35	0.12	0.12
Full-time/part-year Part-time/full-year	0.68	0.68	0.35	0.33	0.28
•		0.00			
Part-time/part-year	0.77	0.77	0.28	0.25	0.10
Housing					
On campus	0.75	0.75	0.35	0.20	0.28
Off campus	0.64	0.64	0.18	0.16	80.0
Living with parents	0.67	0.67	0.30	0.28	0.10
Price of attendance					
Less than \$6,000	0.74	0.74	0.30	0.28	0.08
\$6,000–10,999	0.64	0.64	0.30	0.28	0.12
\$11,000–18,999	0.81	0.81	0.22	0.19	0.13
\$19,000 or more	0.72	0.72	0.30	0.18	0.22
Race/ethnicity White	0.62	0.62	0.00	0.08	0.05
	0.62	0.62	0.08		0.05
Black	0.98	0.98	0.31	0.30	0.14
Hispanic	0.71	0.71 0.99	0.49 0.95	0.44 0.84	0.16 0.65
Asian	0.99				
American Indian	3.02	3.02	0.82	0.69	†
Pacific Islander	2.99	2.99	2.26	2.20	0.92
Two or more races	1.34	1.34	0.55	0.45	0.23
Other	3.90	3.90	2.17	1.72	1.33
Age as of 12/31/07	_				
18 years or younger	0.77	0.77	0.33	0.31	0.14
19–23 years	0.63	0.63	0.17	0.14	0.11
24–29 years	0.98	0.98	0.29	0.30	0.16
30–39 years	0.92	0.92	0.33	0.32	0.12
40 years or older	0.76	0.76	0.32	0.28	0.20

Table S5.2-B. Standard errors for table 5.2-B: Percentage distribution of undergraduates, by gender, citizenship, and selected institutional and student characteristics: 2007–08

—Continued

				Citizenship	
				•	Foreign/
		ender			international
Institutional and student characteristics	Male	Female	U.S. citizen	Resident alien	student
Dependency status					
Dependent	0.62	0.62	0.16	0.14	0.10
Independent	0.65	0.65	0.20	0.19	0.08
Unmarried, no dependents	0.68	0.68	0.29	0.23	0.17
Married, no dependents	1.14	1.14	0.57	0.56	0.19
Single parent	1.33	1.33	0.29	0.25	0.16
Married parent	0.68	0.68	0.40	0.40	0.09
Dependency and income in 2006					
Dependent					
Less than \$20,000	0.78	0.78	0.55	0.49	0.28
\$20,000–39,999	0.70	0.70	0.36	0.35	0.16
\$40,000-59,999	1.07	1.07	0.40	0.33	0.20
\$60,000–79,999	1.19	1.19	0.35	0.23	0.27
\$80,000-99,999	1.10	1.10	0.36	0.33	0.18
\$100,000 or more	0.72	0.72	0.24	0.14	0.18
Independent					
Less than \$10,000	0.69	0.69	0.41	0.36	0.23
\$10,000–19,999	0.81	0.81	0.45	0.43	0.13
\$20,000-29,999	1.06	1.06	0.43	0.45	0.14
\$30,000-49,999	1.21	1.21	0.35	0.35	0.09
\$50,000 or more	0.95	0.95	0.40	0.37	0.18
Income group					
Lowest 25 percent	0.46	0.46	0.27	0.26	0.15
Middle 50 percent	0.82	0.82	0.17	0.16	0.08
Highest 25 percent	0.67	0.67	0.23	0.20	0.11
Aid status					
No aid	0.75	0.75	0.25	0.22	0.16
Received aid	0.48	0.48	0.16	0.14	0.05
Grant status					
No grant	0.76	0.76	0.19	0.17	0.13
Received grant	0.44	0.44	0.18	0.16	0.06
	0.11	0.17	5.10	0.10	0.00
Loan status	0.70	0.70	0.00	0.04	0.40
No loan	0.70	0.70	0.22	0.21	0.12
Received loan	0.59	0.59	0.14	0.14	0.03

[†] Not applicable.

Table 5.3. Percentage distribution of dependent undergraduates, by parents' income and selected institutional and student characteristics: 2007–08

	Less than	\$20,000-	\$40,000-	\$60,000-	\$80,000-	\$100,000 or
Institutional and student characteristics	\$20,000	39,999	59,999	79,999	99,999	more
Total	11.4	16.7	16.7	14.9	12.6	27.7
Institution type						
Public						
Less-than-2-year	16.5	32.2	15.7	18.5	10.8	6.3 !
2-year	14.3	20.4	20.8	15.7	11.3	17.4
4-year	9.2	14.3	14.8	15.1	14.0	32.7
Non-doctorate-granting	11.9	16.4	16.8	17.1	12.9	25.0
Doctorate-granting	8.0	13.4	13.9	14.2	14.5	36.1
Private nonprofit						
Less-than-4-year	27.2	19.1	15.1	14.9	8.3	15.4
4-year	7.9	12.2	13.4	13.0	14.1	39.3
Non-doctorate-granting	9.4	13.7	15.0	13.4	13.6	34.9
Doctorate-granting	6.5	10.9	12.1	12.6	14.6	43.3
For-profit	24.2	29.5	17.1	11.9	6.9	10.4
More than one institution	10.2	15.7	15.9	15.3	11.4	31.4
Attendance status						
Full-time/full-year	10.4	15.1	15.1	14.5	13.6	31.5
Full-time/part-year	12.5	16.2	17.3	14.5	12.4	27.1
Part-time/full-year	13.1	20.4	18.3	15.1	11.3	21.7
Part-time/part-year	12.5	19.5	21.0	16.6	10.5	19.9
Housing ¹						
On campus	8.2	12.9	13.7	14.0	13.6	37.6
Off campus	11.2	15.8	16.6	13.7	13.3	29.3
Living with parents	14.0	20.4	19.1	16.3	11.6	18.6
Price of attendance ¹						
Less than \$6,000	14.2	20.3	20.6	16.3	10.7	17.9
\$6,000–10,999	14.1	20.0	19.7	13.9	11.2	21.0
\$11,000–18,999	12.2	17.0	16.5	16.0	13.0	25.4
\$19,000 or more	7.7	12.6	13.0	13.4	14.6	38.7
Gender						
Male	10.2	15.7	16.3	15.3	12.9	29.6
Female	12.5	17.6	17.1	14.4	12.4	26.0
Race/ethnicity ²						
White	6.8	12.2	15.8	16.4	14.7	34.2
Black	24.0	29.6	18.5	10.1	6.9	10.9
Hispanic	20.5	26.6	19.1	11.9	8.1	13.8
Asian	17.3	20.2	17.7	13.1	10.6	21.1
American Indian	15.3	20.8	19.7	13.1	12.4	18.7
Pacific Islander	17.4	17.3	16.2	16.8	13.8	18.5
Two or more races	13.9	19.9	16.8	13.8	11.1	24.4
Other	12.9	25.6	21.7	15.5	6.0 !	18.3

Table 5.3. Percentage distribution of dependent undergraduates, by parents' income and selected institutional and student characteristics: 2007–08—Continued

Institutional and student characteristics	Less than \$20,000	\$20,000– 39,999	\$40,000– 59,999	\$60,000– 79,999	\$80,000– 99,999	\$100,000 or more
Age as of 12/31/07						
18 years or younger	11.5	17.3	16.9	14.8	12.5	27.0
19–23 years	11.4	16.6	16.7	14.9	12.6	27.9
Income group ³						
Lowest 25 percent	44.7	55.3	†	†	†	†
Middle 50 percent	†	5.2	33.4	29.7	25.2	6.4
Highest 25 percent	†	†	†	†	†	100.0
Aid status						
No aid	6.6	10.7	16.5	16.3	13.9	36.0
Received aid	13.8	19.7	16.8	14.1	12.0	23.5
Grant status						
No grant	5.5	9.7	16.1	17.8	15.2	35.7
Received grant	17.0	23.3	17.3	12.0	10.2	20.2
Loan status ⁴						
No loan	11.5	15.5	15.9	14.2	12.2	30.7
Received loan	11.3	18.5	17.9	15.8	13.3	23.2

[†] Not applicable.

NOTE: All undergraduate students were considered to be financially dependent on their parents for federal financial aid purposes in 2007–08 unless they were age 24 or older on 12/31/07 or if they met any of the following criteria: were married; had legal dependents; were veterans of the U.S. armed forces or on active duty; were orphans or wards of the court; or could document that they were receiving no parental support and were determined to be independent by a financial aid officer using professional judgment. Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico. Detail may not sum to totals because of rounding.

[!] Interpret data with caution (estimates are unstable).

¹ Excludes students attending more than one institution.

² Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

³ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁴ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S5.3. Standard errors for table 5.3: Percentage distribution of dependent undergraduates, by parents' income and selected institutional and student characteristics: 2007–08

Institutional and student characteristics	Less than \$20,000	\$20,000– 39,999	\$40,000– 59,999	\$60,000– 79,999	\$80,000 <u>–</u> 99,999	\$100,000 or more
Total	0.20	0.20	0.23	0.24	0.20	0.31
Institution type						
Public						
Less-than-2-year	2.65	3.74	1.69	2.36	2.57	1.99
2-year	0.55	0.55	0.46	0.54	0.40	0.56
4-year	0.22	0.35	0.33	0.34	0.29	0.46
Non-doctorate-granting	0.56	0.64	0.62	0.61	0.50	0.89
Doctorate-granting	0.27	0.38	0.35	0.37	0.36	0.49
Private nonprofit						
Less-than-4-year	4.67	2.31	3.31	1.34	1.09	4.45
4-year	0.38	0.41	0.51	0.56	0.72	0.85
Non-doctorate-granting	0.86	0.68	0.84	0.79	1.05	1.32
Doctorate-granting	0.50	0.59	0.74	0.76	0.85	1.20
For-profit	1.52	1.39	1.42	1.32	0.86	0.83
More than one institution	0.55	0.69	0.52	0.66	0.51	0.73
Attendance status						
Full-time/full-year	0.20	0.24	0.33	0.28	0.27	0.37
Full-time/part-year	0.69	0.71	0.73	0.65	0.57	1.07
Part-time/full-year	0.45	0.51	0.59	0.54	0.46	0.58
Part-time/part-year	0.90	0.84	0.90	0.94	0.56	0.72
Housing						
On campus	0.34	0.38	0.42	0.51	0.53	0.81
Off campus	0.38	0.41	0.40	0.35	0.39	0.47
Living with parents	0.46	0.43	0.42	0.53	0.31	0.51
Price of attendance						
Less than \$6,000	0.92	0.94	0.80	0.92	0.53	0.79
\$6,000–10,999	0.47	0.56	0.61	0.51	0.47	0.63
\$11,000–18,999	0.32	0.43	0.45	0.42	0.34	0.50
\$19,000 or more	0.29	0.33	0.36	0.36	0.49	0.58
Gender						
Male	0.33	0.32	0.36	0.40	0.29	0.42
Female	0.23	0.28	0.31	0.27	0.27	0.37
Race/ethnicity						
White	0.18	0.22	0.25	0.27	0.26	0.37
Black	0.77	0.79	0.75	0.58	0.43	0.48
Hispanic	0.85	0.65	0.69	0.56	0.49	0.58
Asian	0.84	0.93	0.97	0.77	0.94	0.99
American Indian	3.26	3.56	2.84	2.31	2.84	2.91
Pacific Islander	2.80	2.51	2.41	2.65	2.76	2.48
Two or more races	1.74	1.50	1.39	1.31	1.34	1.53
Other	2.87	4.75	4.63	4.47	2.85	4.34

Table S5.3. Standard errors for table 5.3: Percentage distribution of dependent undergraduates, by parents' income and selected institutional and student characteristics: 2007–08

—Continued

Institutional and student characteristics	Less than \$20,000	\$20,000– 39,999	\$40,000– 59,999	\$60,000- 79,999	\$80,000 <u>–</u> 99,999	\$100,000 or more
Age as of 12/31/07						
18 years or younger	0.51	0.52	0.70	0.52	0.53	0.77
19–23 years	0.20	0.25	0.23	0.25	0.21	0.30
Income group						
Lowest 25 percent	0.51	0.51	†	†	†	†
Middle 50 percent	†	0.19	0.43	0.42	0.36	0.19
Highest 25 percent	†	†	†	†	†	†
Aid status						
No aid	0.29	0.37	0.45	0.44	0.39	0.55
Received aid	0.21	0.23	0.31	0.24	0.26	0.31
Grant status						
No grant	0.24	0.29	0.33	0.39	0.31	0.49
Received grant	0.26	0.28	0.33	0.25	0.27	0.35
Loan status						
No loan	0.37	0.37	0.28	0.33	0.25	0.46
Received loan	0.27	0.36	0.36	0.31	0.28	0.40

[†] Not applicable.

Table 5.4. Percentage distribution of independent undergraduates, by students' income and selected institutional and student characteristics: 2007–08

	Less than	\$10,000-	\$20,000-	\$30,000-	\$50,000
Institutional and student characteristics	\$10,000	19,999	29,999	49,999	or more
Total	23.0	17.4	15.3	20.3	24.0
Institution type					
Public					
Less-than-2-year	25.7	18.4	18.1	21.7	16.1
2-year	19.3	16.0	15.3	21.6	27.8
4-year	27.7	19.4	14.6	17.7	20.7
Non-doctorate-granting	24.9	18.8	16.1	18.7	21.5
Doctorate-granting	30.0	19.8	13.3	16.9	20.0
Private nonprofit					
Less-than-4-year	31.1	16.6	14.8	20.7	16.7
4-year	21.4	14.0	14.1	20.3	30.2
Non-doctorate-granting	19.9	14.7	14.9	20.9	29.6
Doctorate-granting	23.8	12.8	12.8	19.2	31.3
For-profit	29.5	21.3	16.9	19.6	12.6
More than one institution	23.9	18.2	14.4	19.9	23.6
Attendance status					
Full-time/full-year	35.4	20.4	15.0	16.6	12.6
Full-time/part-year	33.0	20.8	16.9	15.5	13.8
Part-time/full-year	17.8	16.5	16.0	22.9	26.8
Part-time/part-year	16.4	15.2	14.2	22.2	32.0
Housing ¹					
On campus	49.9	21.6	9.8	9.8	9.0
Off campus	20.7	16.4	15.4	21.2	26.4
Living with parents	36.4	24.2	16.0	15.8	7.6
Price of attendance ¹					
Less than \$6,000	16.5	14.8	14.2	22.0	32.5
\$6,000-10,999	21.9	18.0	16.0	20.2	23.9
\$11,000–18,999	30.3	20.1	16.0	18.8	14.8
\$19,000 or more	29.1	18.1	15.6	19.1	18.1
Gender					
Male	23.6	17.1	13.7	20.2	25.4
Female	22.7	17.6	16.2	20.4	23.1
Race/ethnicity ²					
White	19.7	16.6	14.3	20.7	28.7
Black	27.5	18.8	17.1	20.8	15.8
Hispanic	26.1	19.2	17.5	19.9	17.3
Asian	29.3	16.1	13.0	17.8	23.8
American Indian	34.6	11.0	14.1	19.9	20.4
Pacific Islander	24.1	25.0	15.5	19.2	16.1
Two or more races	29.5	17.4	16.3	17.5	19.3
Other	24.9	23.1	15.9	12.9	23.3

Table 5.4. Percentage distribution of independent undergraduates, by students' income and selected institutional and student characteristics: 2007–08—Continued

	Less than	\$10,000-	\$20,000-	\$30,000-	\$50,000
Institutional and student characteristics	\$10,000	19,999	29,999	49,999	or more
Age as of 12/31/07					
18 years or younger	75.2	12.7	4.7	6.8	‡
19–23 years	43.6	22.8	14.5	13.0	6.1
24–29 years	26.9	22.8	18.3	19.4	12.5
30–39 years	14.8	13.3	14.6	23.8	33.5
40 years or older	12.1	10.2	11.9	22.5	43.3
Independent students					
Unmarried, no dependents	35.9	24.3	17.8	15.9	6.1
Married, no dependents	8.0	10.0	11.1	24.0	46.9
Single parent	31.3	22.6	19.3	18.4	8.4
Married parent	4.5	6.4	9.6	26.3	53.3
Income group ³					
Lowest 25 percent	90.3	9.7	†	†	†
Middle 50 percent	†	29.9	30.5	39.6	†
Highest 25 percent	†	†	†	2.1	97.9
Aid status					
No aid	17.0	14.0	13.5	21.3	34.2
Received aid	26.4	19.2	16.2	19.8	18.4
Grant status					
No grant	15.7	13.7	15.4	21.5	33.8
Received grant	29.8	20.8	15.1	19.3	15.0
Loan status ⁴					
No loan	20.0	14.9	14.3	20.9	29.9
Received loan	28.4	21.7	16.9	19.4	13.7

[†] Not applicable.

NOTE: Students were considered to be financially independent of their parents for federal financial aid purposes in 2007–08 if they were age 24 or older on 12/31/07 or if they met any of the following criteria: were married; had legal dependents; were veterans of the U.S. armed forces or on active duty; were orphans or wards of the court; or were enrolled in a graduate or first-professional degree program (beyond the bachelor's degree) in 2007–08. All other undergraduate students under 24 were considered to be dependent unless they could document that they were receiving no parental support and were determined to be independent by a financial aid officer using professional judgment. Independent student income includes the student's income as well as any income from the student's spouse. Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico. Detail may not sum to totals because of rounding.

[‡] Reporting standards not met.

¹ Excludes students attending more than one institution.

² Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

³ Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

⁴ Loans only include loans to students. Excludes federal Parent PLUS loans.

Table S5.4. Standard errors for table 5.4: Percentage distribution of independent undergraduates, by students' income and selected institutional and student characteristics: 2007–08

Institutional and student characteristics	Less than \$10,000	\$10,000– 19,999	\$20,000– 29,999	\$30,000– 49,999	\$50,000 or more
institutional and student characteristics	\$10,000	19,999	29,999	49,999	or more
Total	0.29	0.28	0.25	0.34	0.37
Institution type					
Public					
Less-than-2-year	3.06	1.53	1.54	1.59	1.87
2-year	0.48	0.38	0.36	0.47	0.59
4-year	0.66	0.54	0.45	0.53	0.96
Non-doctorate-granting	0.69	0.77	0.78	0.76	1.06
Doctorate-granting	1.08	0.71	0.61	0.73	1.45
Private nonprofit					
Less-than-4-year	3.08	1.69	1.42	2.84	2.81
4-year	0.95	0.87	0.60	0.81	1.13
Non-doctorate-granting	1.34	1.24	0.80	1.17	1.58
Doctorate-granting	2.04	1.04	1.04	1.06	2.20
For-profit	1.06	0.91	0.90	0.93	0.67
More than one institution	0.96	0.81	0.64	0.80	1.25
Attendance status					
Full-time/full-year	0.75	0.56	0.65	0.57	0.59
Full-time/part-year	0.83	0.67	0.79	0.62	0.71
Part-time/full-year	0.41	0.42	0.41	0.55	0.51
Part-time/part-year	0.48	0.49	0.43	0.53	0.77
Housing					
On campus	2.05	1.73	1.49	1.30	1.17
Off campus	0.35	0.32	0.29	0.37	0.43
Living with parents	1.26	0.97	0.84	0.94	0.74
Price of attendance					
Less than \$6,000	0.69	0.52	0.49	0.58	0.83
\$6,000-10,999	0.63	0.59	0.47	0.54	0.67
\$11,000-18,999	0.96	0.57	0.45	0.60	0.74
\$19,000 or more	1.02	0.76	0.94	0.80	0.86
Gender					
Male	0.56	0.37	0.38	0.58	0.60
Female	0.36	0.36	0.31	0.38	0.39
Race/ethnicity					
White	0.37	0.32	0.31	0.40	0.46
Black	0.73	0.65	0.70	0.72	0.81
Hispanic	0.66	0.70	0.70	0.79	0.79
Asian	1.88	1.17	1.05	1.39	1.35
American Indian	3.74	2.04	2.13	2.41	3.69
Pacific Islander	3.98	3.83	3.33	3.47	3.45
Two or more races	1.94	1.62	1.57	1.59	1.92
Other	5.85	5.51	4.08	3.63	4.33

Table S5.4. Standard errors for table 5.4: Percentage distribution of independent undergraduates, by students' income and selected institutional and student characteristics: 2007–08

—Continued

	Less than	\$10,000-	\$20,000-	\$30,000-	\$50,000
Institutional and student characteristics	\$10,000	19,999	29,999	49,999	or more
Age as of 12/31/07					
18 years or younger	3.21	2.55	1.33	1.86	†
19–23 years	0.86	0.76	0.69	0.63	0.45
24–29 years	0.44	0.45	0.44	0.45	0.38
30–39 years	0.51	0.51	0.42	0.60	0.65
40 years or older	0.54	0.52	0.48	0.75	0.88
Independent students					
Unmarried, no dependents	0.54	0.50	0.44	0.48	0.34
Married, no dependents	0.62	0.54	0.72	0.79	0.99
Single parent	0.66	0.66	0.60	0.96	0.44
Married parent	0.27	0.33	0.37	0.70	0.76
Income group					
Lowest 25 percent	0.31	0.31	†	†	†
Middle 50 percent	†	0.49	0.45	0.59	†
Highest 25 percent	†	†	†	0.21	0.21
Aid status					
No aid	0.42	0.43	0.51	0.55	1.00
Received aid	0.37	0.35	0.35	0.37	0.40
Grant status					
No grant	0.35	0.35	0.41	0.45	0.69
Received grant	0.44	0.42	0.40	0.46	0.42
Loan status					
No loan	0.44	0.32	0.28	0.44	0.54
Received loan	0.64	0.49	0.45	0.47	0.41

[†] Not applicable.

Table 5.5. Percentage distribution of undergraduates, by type of institution and selected institutional and student characteristics: 2007–08

Institutional and student characteristics			Public	Public 4-vear		Private rivate nonprofit 4-year			More
	Public less- than-	Public	Non- doctorate-		nonprofit less- than-	Non-	Doctorate-	For-	than one insti-
	2-year	2-year	granting	granting	4-year	granting	granting	profit	tution
Total	0.4	40.0	10.3	19.0	0.4	6.8	6.2	9.0	8.0
Attendance status									
Full-time/full-year	0.3	17.8	12.9	29.7	0.3	10.2	10.4	8.5	9.9
Full-time/part-year	1.1	33.9	8.7	17.1	1.0	7.4	5.9	21.4	3.7 !
Part-time/full-year	0.3 !	56.0	8.7	11.8	0.4	4.0	2.7	5.4	11.0
Part-time/part-year	0.4	64.5	8.5	9.2	0.4	3.8	2.6	6.0	4.6 !
Housing									
On campus	‡	2.8	15.2	39.2	#	20.6	21.6	0.5	†
Off campus	0.6	45.9	10.3	19.4	0.5	5.8	4.4	13.1	†
Living with parents	0.4	62.0	10.9	12.2	0.5	3.2	2.9	7.8	†
Price of attendance									
Less than \$6,000	0.6	82.3	7.5	5.9	0.2	1.8	0.7	0.9	†
\$6,000-10,999	0.5	63.6	11.2	12.3	0.4	3.6	2.1	6.3	†
\$11,000-18,999	0.5	23.8	19.6	35.9	0.6	4.9	3.4	11.4	†
\$19,000 or more	0.2 !	2.6	5.8	27.9	0.5	20.3	21.5	21.1	†
Gender									
Male	0.4	40.6	10.8	20.9	0.4	6.8	6.5	6.4	7.3
Female	0.5	39.5	9.9	17.5	0.4	6.9	5.9	10.9	8.5
Race/ethnicity ¹									
White	0.4	39.0	10.3	21.2	0.2 !	7.5	6.8	6.8	7.9
Black	0.4	41.3	9.7	14.4	0.4 !		4.5	15.9	6.9
Hispanic	0.6	41.7	11.7	12.9	1.0	5.8	4.8	13.2	8.1
Asian	0.3	41.1	8.4	22.2	0.6 !		7.9	4.9	10.3 !
American Indian	0.8 !	45.2	15.1	13.2	4.2 !		1.6	10.6 !	6.3
Pacific Islander	1.1 !	49.2	6.1	12.6	‡	6.3 !	4.5	8.5	8.8 !
Two or more races	0.3	39.6	8.9	19.7	‡	6.0	6.2	9.7	9.4 !
Other	‡	43.5	9.2	16.3	‡	7.3	6.2	10.4 !	6.6
Age as of 12/31/07									
18 years or younger	0.2	37.6	11.5	21.5	0.2 !	8.1	9.4	4.7	6.8
19–23 years	0.3	32.9	11.4	25.4	0.3	7.1	7.7	5.8	9.1
24–29 years	0.5	43.3	10.1	14.3	0.7	5.4	3.3	14.6	7.7
30–39 years	0.7	49.8	8.0	8.0	0.7	7.1	3.4	15.9	6.5
40 years or older	0.9	58.4	6.7	6.7	0.6	6.3	3.9	10.6	5.9 !
Dependency status									
Dependent	0.2	32.3	11.8	26.4	0.2	7.7	8.6	4.0	8.7
Independent	0.6	48.6	8.6	10.6	0.6	5.9	3.4	14.6	7.1
Unmarried, no dependents	0.5	44.9	9.8	15.5	0.6	5.5	3.7	11.9	7.7
Married, no dependents	0.6	50.0	9.4	12.2	0.4	6.5	4.1	9.1	7.8 !
Single parent	0.7	49.1	7.3	6.6	0.7	5.0	2.9	21.2	6.4
Married parent	0.7	52.2	8.1	7.9	0.6	7.0	3.4	13.4	6.7

Table 5.5. Percentage distribution of undergraduates, by type of institution and selected institutional and student characteristics: 2007–08—Continued

Institutional and student characteristics	Public		Public 4-year		Private nonprofit	Private nonprofit 4-year			More than
	less- than- 2-year	Public 2-year	Non- doctorate- granting	Doctorate- granting	less- than- 4-year	Non- doctorate- granting	Doctorate- granting	For- profit	one insti- tution
Dependency and income in 200	6								
Dependent									
Less than \$20,000	0.3	40.5	12.3	18.5	0.6	6.4	4.9	8.6	7.9
\$20,000–39,999	0.5	39.4	11.5	21.1	0.3	6.3	5.6	7.1	8.2
\$40,000-59,999	0.2	40.2	11.8	21.9	0.2	6.9	6.2	4.2	8.3
\$60,000–79,999	0.3	34.2	13.6	25.2	0.2	7.0	7.3	3.3	9.0 !
\$80,000–99,999	0.2	29.0	12.1	30.2	0.2	8.3	10.0	2.2	7.9
\$100,000 or more	0.1 !	20.3	10.6	34.3	0.1 !	9.7	13.4	1.5	9.9
Independent									
Less than \$10,000	0.7	40.8	9.3	13.8	0.8	5.1	3.5	18.6	7.4
\$10,000–19,999	0.7	44.7	9.3	12.1	0.6	5.0	2.5	17.8	7.4
\$20,000–29,999	8.0	48.8	9.1	9.3	0.6	5.7	2.9	16.2	6.7
\$30,000–49,999	0.7	51.7	7.9	8.8	0.6	6.0	3.2	14.1	6.9
\$50,000 or more	0.4	56.3	7.7	8.9	0.4	7.2	4.5	7.7	7.0 !
Income group ²									
Lowest 25 percent	0.6	40.4	10.8	16.9	0.6	5.6	4.4	13.0	7.7
Middle 50 percent	0.5	41.4	10.6	18.3	0.4	6.6	5.3	9.2	7.8
Highest 25 percent	0.2	36.7	9.2	22.4	0.3	8.6	9.7	4.4	8.5
Aid status									
No aid	0.6	60.9	8.9	15.5	0.2	2.5	3.3	1.0	7.2 !
Received aid	0.4	29.0	11.0	20.8	0.5	9.1	7.7	13.2	8.4
Grant status									
No grant	0.5	50.0	10.1	18.4	0.3	3.4	3.7	5.5	8.0 !
Received grant	0.4	30.6	10.5	19.5	0.5	10.1	8.4	12.3	7.9
Loan status ³									
No loan	0.6	56.4	9.5	16.1	0.4	4.3	4.4	1.2	7.2 !
Received loan	0.2	13.7	11.6	23.5	0.5	10.9	9.0	21.4	9.3

[†] Not applicable.

NOTE: Students who were enrolled part time for any amount of time during the 9-month academic year, including those who were enrolled full time for part of the year, are included in the part-time/full-year or part-time/part-year categories. Estimates include students enrolled in Title IV eligible postsecondary institutions in the 50 states, the District of Columbia, and Puerto Rico. Detail may not sum to totals because of rounding. SOURCE: U.S. Department of Education, National Center for Education Statistics, 2007–08 National Postsecondary Student Aid Study (NPSAS:08).

[#] Rounds to zero.

[!] Interpret data with caution (estimates are unstable).

[‡] Reporting standards not met.

¹ Black includes African American, Hispanic includes Latino, American Indian includes Alaska Native, Pacific Islander includes Native Hawaiian, and Other includes respondents having origins in a race not listed.

² Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent. Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year.

³ Loans only include loans to students. Excludes federal Parent PLUS.

Table S5.5. Standard errors for table 5.5: Percentage distribution of undergraduates, by type of institution and selected institutional and student characteristics: 2007–08

	Public		Public 4-year		Private nonprofit	Private nonprofit 4-year			More than
Institutional and student characteristics	less- than- 2-year	Public 2-year		Doctorate- granting	less- octorate- than-	Non- doctorate- granting	Doctorate- granting	For- profit	one insti- tution
Total	0.01	1.02	0.23	0.56	0.01	0.11	0.11	0.12	2.13
Attendance status Full-time/full-year Full-time/part-year Part-time/full-year	0.04 0.21 0.08	0.46 0.90 1.88	0.38 0.36 0.32	0.85 0.80 0.54	0.03 0.18 0.06	0.26 0.31 0.20	0.26 0.58 0.15	0.42 0.80 0.31	2.13 1.36 2.54
Part-time/part-year	0.08	1.87	0.24	0.33	0.06	0.24	0.13	0.44	2.25
Housing On campus Off campus Living with parents	† 0.05 0.05	0.30 0.38 0.51	0.51 0.15 0.33	0.91 0.22 0.35	† 0.03 0.07	0.75 0.20 0.20	0.58 0.13 0.25	0.14 0.26 0.37	† † †
Price of attendance Less than \$6,000 \$6,000–10,999 \$11,000–18,999 \$19,000 or more	0.11 0.06 0.07 0.09	0.47 0.49 0.53 0.25	0.46 0.26 0.41 0.34	0.23 0.36 0.63 0.55	0.03 0.06 0.06 0.10	0.23 0.29 0.30 0.42	0.08 0.14 0.30 0.46	0.11 0.36 0.68 0.76	† † †
Gender Male Female	0.04 0.04	1.04 1.05	0.25 0.26	0.49 0.63	0.06 0.04	0.24 0.17	0.17 0.13	0.36 0.27	2.05 2.19
Race/ethnicity White Black Hispanic Asian American Indian Pacific Islander Two or more races Other	0.04 0.13 0.12 0.08 0.37 0.36 0.07	1.12 1.54 1.08 2.02 5.04 4.39 1.59 3.99	0.27 0.74 0.63 0.65 2.67 0.94 0.79 1.94	0.57 0.57 0.88 1.09 1.99 1.87 1.79 2.44	0.06 0.12 0.25 0.26 1.35 †	0.19 0.50 0.42 0.48 0.83 2.28 0.79 1.57	0.14 0.33 0.61 0.61 0.42 1.31 0.57	0.23 0.96 1.05 0.70 3.40 2.35 1.06 3.55	2.05 1.74 2.28 3.62 1.84 3.56 2.84 1.94
Age as of 12/31/07 18 years or younger 19–23 years 24–29 years 30–39 years 40 years or older	0.04 0.04 0.04 0.11 0.14	0.82 0.99 1.32 1.68 1.66	0.62 0.27 0.33 0.28 0.30	1.12 0.83 0.34 0.49 0.56	0.06 0.03 0.07 0.08 0.07	0.42 0.20 0.28 0.40 0.66	0.47 0.18 0.19 0.20 0.33	0.32 0.23 0.49 0.56 0.66	2.00 2.35 2.02 1.81 1.93

Table S5.5. Standard errors for table 5.5: Percentage distribution of undergraduates, by type of institution and selected institutional and student characteristics: 2007–08—Continued

	Public		Public	4-year	Private nonprofit	Private nonprofit 4-year			More than one
	less-		Non-		less-	Non-			
Institutional and student	than-	Public	doctorate-	Doctorate-	than-	doctorate-	Doctorate-	For-	insti-
characteristics	2-year	2-year	granting	granting	4-year	granting	granting	profit	tution
Dependency status									
Dependent	0.04	0.83	0.35	0.88	0.03	0.23	0.20	0.23	2.32
Independent	0.06	1.38	0.18	0.34	0.02	0.27	0.11	0.24	1.93
Unmarried, no dependents	0.05	1.53	0.37	0.42	0.06	0.32	0.22	0.54	2.18
Married, no dependents	0.08	1.96	0.51	0.50	0.03	0.57	0.30	0.84	2.37
Single parent	0.11	1.83	0.35	0.34	0.07	0.40	0.23	0.87	1.64
Married parent	0.11	1.08	0.40	0.63	0.06	0.52	0.22	0.82	1.82
Dependency and income in 2000 Dependent	6								
Less than \$20,000	0.09	1.00	0.46	1.14	0.09	0.51	0.41	0.53	1.79
\$20,000-39,999	0.12	0.77	0.47	0.72	0.05	0.38	0.35	0.54	1.91
\$40,000-59,999	0.04	1.21	0.49	0.82	0.06	0.49	0.34	0.36	2.16
\$60,000-79,999	0.04	1.54	0.60	1.03	0.04	0.44	0.40	0.45	2.76
\$80,000-99,999	0.05	1.30	0.59	1.00	0.02	0.59	0.55	0.33	2.33
\$100,000 or more	0.02	0.91	0.55	1.20	0.05	0.47	0.51	0.14	2.78
Independent									
Less than \$10,000	0.09	0.98	0.35	0.51	0.08	0.33	0.34	0.69	1.78
\$10,000-19,999	0.06	1.41	0.40	0.48	0.05	0.37	0.22	0.62	1.81
\$20,000-29,999	0.10	1.47	0.45	0.52	0.05	0.34	0.26	0.76	1.77
\$30,000-49,999	0.09	1.74	0.33	0.45	0.09	0.49	0.18	0.61	2.07
\$50,000 or more	0.07	2.05	0.42	0.77	0.08	0.56	0.36	0.44	2.28
Income group									
Lowest 25 percent	0.07	0.62	0.30	0.61	0.04	0.22	0.30	0.37	1.74
Middle 50 percent	0.03	1.24	0.21	0.52	0.02	0.15	0.12	0.20	2.17
Highest 25 percent	0.03	1.25	0.39	0.72	0.03	0.32	0.26	0.21	2.51
Aid status									
No aid	0.04	2.25	0.20	0.41	0.04	0.18	0.18	0.06	2.85
Received aid	0.02	0.25	0.37	0.76	0.02	0.23	0.21	0.37	1.76
Grant status									
No grant	0.02	1.93	0.21	0.39	0.02	0.15	0.15	0.14	2.68
Received grant	0.02	0.30	0.30	0.78	0.03	0.26	0.25	0.28	1.65
Loan status	-							-	
No loan	0.02	1.49	0.17	0.58	0.01	0.14	0.12	0.04	2.39
Received loan	0.02	0.17	0.17	0.56	0.01	0.14	0.12	0.04	1.73

[†] Not applicable.

Glossary

All of the variables that were used in these tables are described in this glossary. The variables were taken directly from the 2007-08 National Postsecondary Student Aid Study (NPSAS:08) Data Analysis System (DAS). The DAS is a web-based application that allows users to create tables with estimates and standard errors from the NPSAS:08 data and other surveys (see the Data Analysis System section for a more detailed description), and it can be accessed at http://www.nces.ed.gov/das/. In the glossary, the items are listed in alphabetical order by the variable label. The name of each variable appears to the right of the variable label.

Age as of 12/31/07 AGE

18 years or younger

19-23 years

24-29 years

30-39 years

40 years or older

Any campus-based aid CAMPAMT

Total amount of federal aid received during 2007–08 that is campus-based. Includes the Supplemental Educational Opportunity (SEOG) grant (SEOGAMT), federal college work-study (TFEDWRK), and Perkins loans (PERKAMT), all of which are awarded on the basis of need. Federal campus-based aid is available only at institutions that participate in the campus-based aid programs and is administered directly by the financial aid office at each participating school. The amount of aid received by students depends on the availability of funds at their institution. Schools set their own deadlines for awarding campus-based funds.

Attendance status ATTNSTAT

Number of months enrolled full time or part time at all institutions attended during the 2007–08 academic year. In some cases, only full-time, full-year students who attended one institution were included to allow for comparisons in tuition, price of attendance, and financial aid. Those attending full time generally have higher overall expenses than do all students. They also generally qualify for federal aid and other assistance not available to many part-time students. Full-year is defined as enrollment for 9 or more months during the NPSAS year. Months did not have to be contiguous or at the same institution, and students did not have to be enrolled for a full month in order to be considered enrolled for that month. Full-time is usually defined as 12 or more credit hours. The categories are as follows:

Full-time/full-year Enrolled full time for 9 or more months.

Full-time/part-year Enrolled full time, but for less than 9 months.

Part-time/full-year Enrolled for 9 or more months, but less than

9 months were full time.

Part-time/part-year Enrolled for less than 9 months, and these

months were not all full time.

G1

Citizenship CITIZEN2

U.S. citizen Student was a U.S. citizen or U.S. national in

2007-08.

Resident alien Student was a permanent or temporary U.S.

resident eligible for federal financial aid in

2007-08.

Foreign/international

student

Student was not a U.S. citizen and was not eligible for financial aid (includes those hold-

ing student or exchange visitor visas).

Dependency and income in 2006

INCOME

For dependent undergraduates, this is the total income of the student's parents. For independent undergraduates, this is the total income of the student (and spouse if married). Prior calendar year income is reported in the federal financial aid application and is used in determining the expected family contribution (EFC) in federal need analysis. That is, 2006 income was used to determine financial aid eligibility for the 2007–08 academic year.

Dependent students

Less than \$20,000

\$20,000-39,999

\$40,000-59,999

\$60,000-79,999

\$80,000-99,999

\$100,000 or more

Independent students

Less than \$10,000

\$10,000-19,999

\$20,000-29,999

\$30,000-49,999

\$50,000 or more

Dependency status DEPEND

Student's dependency status for federal financial aid purposes during the 2007–08 academic year. Students were considered to be financially independent of their parents for federal financial aid purposes in 2007–08 if they were age 24 or older on 12/31/07 or if they met any of the following criteria: were married; had dependents; were veterans of the U.S. military or on active duty; were orphans or wards of the court; or were enrolled in a graduate or first-professional degree program in 2007–08. All other students under 24 were considered to be dependent unless they could document that they were receiving no parental support and were determined to be independent by a financial aid officer using professional judgment.

Dependent students' parent income

DEPINC

For dependent undergraduates, this is the total income of the student's parents in 2006. Prior calendar year income is reported in the federal financial aid application and is used in determining the expected family contribution (EFC) in federal need analysis. That is, 2006 income was used to determine financial aid eligibility for the 2007–08 academic year.

Less than \$20,000 \$20,000–39,999 \$40,000–59,999 \$60,000–79,999 \$80,000–99,999 \$100,000 or more

Federal aid total TFEDAID

Total amount of federal financial aid received in 2007–08. This includes federal grants, federal student loans, federal work-study, and federal Parent PLUS loans. It does not include federal tax benefits, federal veterans benefits, or Department of Defense programs.

Federal expected family contribution

EFC

The federal expected family contribution (EFC) used in federal need analysis is an estimate of the amount that students and parents should be able to contribute toward educational expenses. A zero EFC means that the student and family cannot afford to contribute any amount. Students with a zero EFC are eligible for a maximum Pell grant (\$4,310 in 2007–08). The EFC is calculated by taking into account a family's income, assets, family size, number of children in college, and other factors affecting the ability to pay for a postsecondary education. The income of parents is included in calculating the EFC of dependent students, but not for independent students. EFC values were taken from federal sources for Pell grant recipients and students with federal financial aid application records or from institutional records. If the EFC was not available from any of these sources, it was imputed by regression using dependency, family size, income, and number of children in college. EFC is generally imputed for students who did not apply for or receive any federal aid.

Federal grants TFEDGRT

Total amount of federal grants received by a student during the 2007–08 academic year. These are Pell grants, Supplemental Educational Opportunity Grants (SEOG), Academic Competitiveness Grants (ACG), Science and Mathematics Access to Retain Talent Grants (SMART), Robert Byrd Scholarships. and several smaller federal grant programs received during the 2007–08 academic year. This variable does not include federal tax benefits, federal veterans benefits, or benefits from Department of Defense programs.

G3

Federal loans TFEDLN

Total amount of federal student loans received by the student during 2007–08. Primarily Stafford loans (subsidized and unsubsidized), Perkins loans, and a small percentage of federal loans through the Public Health Service. Does not include federal Parent PLUS loans. The major sources of loans to students are the Federal Stafford loan programs (see STAFFAMT).

Federal Parent PLUS loans

PLUSAMT

Total amount of federal Parent PLUS loans received in 2007–08. Federal Parent PLUS loans are available to the parents of dependent undergraduates in addition to any federal Stafford loans for which students are eligible. Parent PLUS loans are not based on need; however parents must meet certain creditworthiness criteria in order to take out a loan. There is no fixed limit to the amount of a Parent PLUS loan, but the Parent PLUS loan may not exceed the total price of attendance minus any other financial aid.

Federal Pell grant PELLAMT

Total amount of federal Pell grants received at all institutions attended during the 2007–08 academic year. Pell grants are need-based grants awarded to undergraduates who have not yet received a bachelor's degree and for students in teaching certificate programs. They are intended as a financial base, to which other financial aid awards can be added. The amount of a Pell grant depends on the expected family contribution (EFC), the price of attendance, and attendance status (full-time or part-time, full-year or part-year). In the 2007–08 academic year the maximum Pell grant was \$4,310.

Federal work-study TFEDWRK

Total amount of federal work-study aid received by a student in 2007–08. The federal work-study program subsidizes students' wages and is available to students with financial need. Includes institutional matching funds.

Financial need (price of attendance minus EFC)

SNEED1

The student's total need for need-based financial aid as determined by federal need analysis. Equal to the total price of attendance (BUDGETAJ) minus the federal expected family contribution (EFC). The fundamental rule in federal financial aid need analysis is that the student's need is determined by subtracting the EFC from the price of attendance (student budget). If the student's need is zero or negative, the student is not eligible for any need-based federal aid. When the EFC is greater than the price of attendance (and therefore the student has no need), the result is a negative number that has been set to zero. The averages in the tables include only students with positive need values. A student with no federal need may still be eligible for federal non-need-based aid (primarily unsubsidized Stafford loans or Parent PLUS loans). Similarly, a student without federal need may also receive any non-federal grants, loans, or any other aid as long as no federal need-based aid is accepted. Calculated only for students who attended one institution in 2007–08. See the NOTE under NETCST1 for an explanation of the difference between remaining need and net price variable calculations.

Gender GENDER

Male Female

Housing LOCALRES

Students' residence while enrolled

On campus Institution-owned living quarters for stu-

dents. These are typically on-campus or offcampus dormitories, residence halls, or other

facilities.

Off campus Student lived off campus in non-institution-

owned housing but not with her or his par-

ents or relatives.

Living with parents/ Student lived at home with parents or other

other relatives relatives.

Income group PCTALL

Indicates income percentiles for all students. Calculated separately for dependent and independent students and then combined into this variable. Each ranking thus compares the student only with other students of the same dependency status. Uses parents' total income if the student is dependent and the student's own income if the student is independent (see DEPINC and INDEPINC). Total income in 2006 is used because this was the income reported on the federal financial aid application and used for federal need analysis for the 2007–08 academic year. The categories are as follows:

Lowest 25 percent 25 percent or lower
Middle 50 percent 26 percent to 75 percent
Highest 25 percent 76 percent to 100 percent

Independent student categories

DEPEND5B

Indicates an independent student's marital status and whether or not the student had dependents.

Unmarried, no dependents Student was single or separated and had no

dependents (includes those who were wi-

dowed or divorced).

Married, no dependents Student was married and had no dependents

(a spouse is not considered a dependent).

Single parent Student was single or separated and had de-

pendents (includes those who were widowed

or divorced).

Independent student categories—continued

DEPEND5B

Married parent

Student was married and had dependents (a spouse is not considered a dependent).

Independent students' income

INDEPINC

For independent students, this is the total income of the student (and spouse if married) in 2006. Prior calendar year income is reported in the financial aid application and is used in determining the expected family contribution (EFC) in federal need analysis. That is, 2006 income was used to determine financial aid eligibility for the 2007–08 academic year.

Less than \$10,000 \$10,000–19,999 \$20,000–29,999 \$30,000–49,999 \$50,000 or more

Institution type AIDSECT

Control and level of the NPSAS sample institution attended by the student in 2007–08, based on the classification in the 2007 Integrated Postsecondary Education Data System (IPEDS) Institutional Characteristics (IC) file. Control concerns the source of revenue and control of operations (public, private nonprofit, for-profit), and level concerns the highest degree or award offered by the institution in any program. Doctorate-granting institutions award a doctoral or first-professional degree in one or more programs: 4-year non-doctorate-granting institutions award at least a bachelor's degree; 2-year institutions award at least an associate's degree; and less-than-2-year institutions award certificates or other credentials in vocational programs. It should be noted that some community colleges have begun offering bachelor's degrees in a few programs and have been reclassified as 4-year institutions in IPEDS.

The nine categories of types of institutions were used as the sampling frame for the national NPSAS sample. Sampled students determined to have attended more than one institution during the NPSAS study year have been reclassified into a separate category ("More than one institution") because all or part of their financial aid may not have been received at the NPSAS sample institution.

Public
Less-than-2-year
2-year
4-year
Non-doctorate-granting
Doctorate-granting
Private nonprofit
Less-than-4-year
4-year
Non-doctorate-granting
Doctorate-granting

Institution type—continued

AIDSECT

For-profit (less-than-2-year, 2-year, and 4-year) More than one institution

Institutional aid total INSTAMT

Total amount of institutional aid received during 2007–08. This includes all types of institutional grants and scholarships from institutional funds, tuition waivers, institutional loans, and institutional work-study (outside of the federal work-study program).

Institutional grants INGRTAMT

Total amount of grant aid received during the 2007–08 academic year that was funded by the postsecondary institution attended. This includes all need- and merit-based grants, scholarships, and tuition waivers from institutional funds.

Institutional loans INLNAMT

Total amount of all institutional loans (from funds provided by the educational institution) received during the 2007–08 academic year.

Institutional merit-only grants

INSMERIT

Institutional grants and scholarships received during the 2007–08 academic year that were based only on merit rather than need, including all athletic scholarships. Students who receive need-based grants may also receive merit scholarships that are not based on need.

Institutional work-study

INSTWRK

Total amount of all institution-sponsored work-study aid (outside of the federal work-study program) received during the 2007–08 academic year. Includes any institutional amounts that were classified as work-study or campus employment.

Net price after federal grants

NETCST2

The net price of attendance after federal grants received in 2007–08. Equal to the attendance-adjusted student budget (BUDGETAJ) minus all federal grants (TFEDGRT). Federal grants include Pell grants, Supplemental Educational Opportunity Grants (SEOG), Academic Competitiveness Grants (ACG), Science and Mathematics Access to Retain Talent Grants (SMART), and a small number of Robert Byrd Scholarships. They do not include federal veterans benefits or education tax benefits. All students are included in the calculation of the average, whether or not they received any federal grants. For students who did not receive any federal grants, this amount is the same as the price of attendance. Calculated only for students who attended one institution during 2007–08. See the NOTE under NETCST1 for more information.

G7

Net price (price of attendance minus all grants)

NETCST3

The net price of attendance after all grants for the 2007–08 academic year. Equal to the total price of attendance (BUDGETAJ) minus total grant aid from federal, state, institutional, or other sources (TOTGRT). Grants include tuition waivers and employer tuition reimbursements. They do not include federal veterans benefits or military education benefits. All students are included in the calculation of the average, whether or not they received any grant aid. For students who did not receive any grants, this amount is the same as the price of attendance. Calculated only for students who attended one institution during 2007–08. See the NOTE under NETCST1 for more information.

Net tuition (tuition and fees minus all grants)

NETCST9

The net tuition and fees after all grants for 2007–08. Equal to tuition and fees (TUITION2) minus total grant aid from federal, state, institutional, or other sources (TOTGRT). Grants include tuition waivers and employer tuition reimbursements. They do not include federal veterans benefits or military education benefits. All students are included in the calculation of the average, whether or not they received any grant aid. For students who did not receive any grants, this amount is the same as the tuition and fees. Grants help cover the entire price of attendance (not just tuition), so the grant amount may be greater than tuition alone. Negative values (if TOTGRT was greater than TUITION2) were set to zero. The average net tuition shown in the tables includes these zero values. Calculated only for students who attended one institution during 2007–08. See the NOTE under NETCST1 for more information.

Other type of aid OTHTYPE

Indicates the total amount of aid received during 2007-08 that was not classified as grants, student loans, or work study. Includes Parent PLUS loans, vocational rehabilitation and job training grants, veterans benefits, and benefits from Department of Defense programs. It does not include federal education tax benefits.

Out-of-pocket net price (price of attendance minus total aid)

NETCST1

The net price of attendance after all financial aid received in 2007–08. Equal to the total price of attendance (BUDGETAJ) minus total aid (TOTAID). It represents the estimated "out-of-pocket" expense to students remaining after all financial aid, including loans, is received. All students are included in the calculation of the average, whether or not they received any financial aid. For students who did not receive any financial aid, this amount is the same as the price of attendance. Calculated only for students who attended one institution during 2007–08.

NOTE: Net price of attendance variables are calculated by subtracting financial aid (total or only some types, such as grants) from the price of attendance (the student budget). The net price variables are never less than zero because a basic principle of financial aid is that total aid may not exceed the student budget. Need and remaining need variables subtract the EFC as well as financial aid from the student budget. Because the EFC may be greater than the student budget, need may be negative. Even if the EFC is less than the student budget (so that

NETCST1

Out-of-pocket net price (price of attendance minus total aid)—continued

need is positive), financial aid may include non-need-based aid (such as unsubsidized Stafford loans), so the remaining need (student budget minus EFC minus aid) may be negative. In the Data Analysis System (DAS), the negative values for need and remaining need are set to zero.

Perkins loans PERKAMT

Total amount of federal Perkins loans received at all institutions attended during the 2007–08 academic year. The Perkins loan is a campus-based, low-interest loan administered by each institution for undergraduate and graduate students who show exceptional financial need. Priority is given to Pell grant recipients. The total annual Perkins loan cannot exceed \$4,000 for undergraduate students and \$6,000 for graduate level students.

Price of attendance BUDGETAJ

Total educational expenses of the student during the 2007–08 academic year, adjusted for attendance status. The total price of attendance includes the tuition and fees as well as all other expenses related to enrollment: books and supplies, room and board (or housing and meal allowances for off-campus students), transportation, and other personal living expenses. The total price of attendance is based on the student budgets reported by the institutions. Student budgets are prepared by institutional financial aid offices to estimate the typical educational expenses of various types of students (e.g., dependent, independent, on-campus, off-campus, living with parents) and are used as part of the federal need analysis that determines the student's need for financial aid. For students who attended one institution only.

Race/ethnicity RACE

Student's race/ethnicity with Hispanic or Latino origin as a separate category. Based on the 2000 census race categories, but the race categories exclude Hispanic origin unless specified.

American Indian/Alaska

Native

A person having origins in any of the original peoples of North America and who maintains cultural identification through tribal affilia-

tion or community recognition.

Asian A person having origins in any of the peoples

of the Far East, Southeast Asia, or the Indian subcontinent. This includes people from China, Japan, Korea, the Philippine Islands, India,

and Vietnam.

Black/African American A person having origins in any of the Black

racial groups of Africa.

Race/ethnicity—continued RACE

Hispanic/Latino A person of Mexican, Puerto Rican, Cuban,

Central or South American, or other Spanish

culture or origin.

Pacific Islander/Native A person having origins in the Pacific Islands

Hawaiian including Hawaii and Samoa.

White A person having origins in any of the original

peoples of Europe, North Africa, or the Mid-

dle East.

Other A person reporting having origins in a race

not listed above.

Two or more races A person reporting origins in more than one

race.

Remaining need after receiving financial aid (price of attendance minus EFC minus total aid)

SNEED2

The remaining need after all financial aid (need-based and non-need-based) received in 2007–08. Equal to the total price of attendance (BUDGETAJ) minus the federal expected family contribution (EFC) and minus total financial aid (TOTAID). Negative values have been set to zero. The averages in the tables include only students with positive remaining need values after financial aid was subtracted from need (SNEED1). Calculated only for students who attended one institution in 2007–08. See the NOTE under NETCST1 for an explanation of the difference between remaining need and net price variable calculations.

Stafford subsidized STAFSUB

Total amount of federal subsidized Stafford loans received during the 2007–08 academic year. Subsidized Stafford loans are awarded on the basis of financial need. If a student qualifies for a subsidized loan, the federal government pays the interest on the loan until the student begins repayment, and during authorized periods of deferment thereafter. See STAFFAMT for more information about Stafford loans.

Stafford unsubsidized STAFUNSB

Total amount of federal unsubsidized Stafford loans received during the 2007–08 academic year. Unsubsidized Stafford loans are not awarded on the basis of need. Students who qualify for an unsubsidized loan are charged interest from the time the loan is disbursed until it is paid in full. Students can choose to pay the interest or allow it to accumulate. If the student allows the interest to accumulate, it will be capitalized (i.e., added to the principal amount of the loan). See STAFFAMT for more information about Stafford loans.

State aid total STATEAMT

Total amount of state-funded financial aid received by the student during the 2007–08 academic year. This includes state grants, state loans, state-sponsored work-study, and vocational rehabilitation and job training grants, including any federal Workforce Investment Act (WIA) funds.

State grants STGTAMT

Total amount of state-funded grants, scholarships, and fellowships received during the 2007–08 academic year. These include the federal matching funds to states through the Leveraging Education Assistance Partnership (LEAP) program. State grants include both need- and merit-based grant programs.

State loans STLNAMT

Total amount of state loans received during 2007–08. These are student loans that are funded by state government agencies.

State merit-only grants

STMERIT

Total amount of state grants and scholarships received during the 2007–08 academic year that were based only on merit, such as high school grades or admission test scores. Does not include state grants based on both merit and need or special purpose grants that are neither need-based nor merit-based, such as waivers for National Guard or state employees.

State work-study STWKAMT

Total amount of state work-study aid received during the 2007–08 academic year.

Supplemental Educational Opportunity Grant (SEOG)

SEOGAMT

Amount of Supplemental Educational Opportunity Grant (SEOG) received during the 2007–08 academic year. The SEOG is a federal, campus-based grant (administered by each institution) for undergraduates who have not yet received a bachelor's or first-professional degree, and who show exceptional financial need. It is intended to supplement the Pell grant (priority is given to Pell grant recipients). The maximum SEOG was \$4,000 in 2007–08. However, unlike the Pell grant, eligibility does not guarantee an SEOG award, as the funds available to a particular institution may be limited.

Total aid TOTAID

Total amount of financial aid received by a student in 2007–08 from any source except parents, relatives, or friends. It includes any grants, student loans, workstudy, Parent PLUS loans, job training funds, and veterans benefits and benefits from Department of Defense programs. It does not include federal tax benefits.

Total grants TOTGRT

Total amount of all grants and scholarships received by a student in 2007–08. Grants are a type of financial aid that do not require repayment or employment. This variable is equal to the sum of all federal grants, state grants, institutional grants, and grants from employers or private sources. All need-based grants, merit scholarships, tuition waivers, and employer tuition reimbursements are included.

TOTLOAN Total loans

Total amount of all student loans received during the 2007–08 academic year. This variable includes all student loans received through federal, state, institutional, or private programs. A student loan is defined as a type of financial aid that advances funds to a student and is evidenced by a promissory note requiring the recipient to repay the specified amounts under prescribed conditions. It excludes federal Parent PLUS loans and any loans from family or friends.

Total Stafford loan STAFFAMT

Total amount of federal subsidized and unsubsidized Stafford loans received either through the Direct Loan program or Federal Family Education Loan Program (FFELP), at all institutions attended during the 2007-08 academic year. Includes loans borrowed to attend schools other than the NPSAS sample school. Annual loan limits for Stafford loans vary by class level and dependency status.

Federal Stafford loans were available in 2007-08 through two different loan programs. Although the source of funds is different, both programs offer the same types of Stafford loans. These programs also offer Parent PLUS (PLUSAMT) and Graduate PLUS (GPLUSAMT) loans.

- 1. Direct Loan program. Under this program, the federal government makes loans directly to students and parents through postsecondary institutions participating in the Direct loan program. Institutions process the applications and disburse the loans directly to students.
- 2. Federal Family Education Loan Program (FFELP). Under this program, private lenders such as banks, credit unions, and savings & loan associations process the applications and make the loans to students. The federal government guarantees these loans.

There are two types of federal Stafford loans, both available through the two different loan programs: subsidized and unsubsidized Stafford loans.

- Subsidized Stafford Ioan (STAFSUB). These are awarded on the basis of financial need. If a student qualifies for a subsidized loan, the federal government pays the interest on the loan until the student begins repayment, and during authorized periods of deferment thereafter.
- 2. Unsubsidized Stafford Ioan (STAFUNSB). These are not awarded on the basis of need. Students who qualify for an unsubsidized loan are charged interest from the time the loan is disbursed until it is paid in full. Students can choose to pay the interest or allow it to accumulate. If the student allows the interest to accumulate, it will be capitalized (i.e., added to the principal amount of the loan).

TOTWKST Total work-study

Total amount of work-study aid received during the 2007-08 academic year. This variable includes all federal work-study, state work-study, and institutional workstudy aid, including undergraduates who had assistantships, tutoring, dormitory, or advising jobs.

TUITION2 Tuition and fees

Tuition and fees at the sampled institution for students who attended only one institution during the 2007-08 academic year. Students who attended more than one institution were not included. The tuition as reported by institutions was the primary source, edited and adjusted for attendance status if necessary. If tuition data were not available from institutions, they were taken from the most recent IPEDS report and adjusted for attendance status. The tuition and fees categories are as follows:

> Less than \$1,000 \$1,000-2,999 \$3,000-6,999 \$7,000 or more

AIDTYPE Type and source of aid

Aid package by type of aid received during the 2007-08 academic year. For students with any aid, indicates combinations of grants, loans, work-study, and other types of aid.

> Grant only Grant and work-study Grant, loan, and work-study Grant and loan Loan only Loan and work-study Other combinations

Undergraduate class level

UGLVL1

Undergraduate class level during the 2007-08 academic year. This variable gives priority to the institution-reported class level. If not available, the order of priority was student interview, then the federal financial aid application, and then the National Student Loan Data System federal loan records.

> First year Second year Third year Fourth year or more (public 4-year and private nonprofit 4-year institutions only)